THE UNIVERSITY OF HULL

Integrated HIV and maternal/reproductive health service utilization: trajectory for prevention of mother to child transmission of HIV in Zimbabwe

Being a Thesis submitted for the Degree of Doctor of Philosophy in the University of Hull

By

Marufu Gazimbi BA (hons) Leeds Metropolitan University M, P.H, University of Sheffield

February 2018

ABSTRACT

Background: The persistent high mortality related to HIV infection in sub-Saharan Africa has prompted calls for scaling-up access to sexual and reproductive health services including family planning as a trajectory to prevent HIV infection. Thus, HIV prevention programs have been integrated into family planning and reproductive health care services as a means to reach out to men and women who are HIV positive and those who are vulnerable to HIV infection. While the rationale for integration of HIV and reproductive health (RH) services is strong, there is paucity of information on which population groups most utilize these services. Due to the considerable stigma attached to HIV/AIDS, people living with HIV and those who perceive themselves to be at risk of HIV infection may be less likely to use integrated health care services. This thesis aims to inform policy and programs on better integration of HIV testing, maternal health care, and family planning services in order to optimize HIV prevention programs such as prevention of mother to child HIV transmission and condom use with the broader aim to reduce the HIV pandemic.

Objectives: Focusing on individual and community-level predisposing, enabling and perceived need factors (PREP), the specific objectives of the study are to: (i) examine the effects of HIV on reproductive health care services; (ii) identify the determinants of HIV testing, antenatal and delivery care services; (iii) examine contraceptive methods choice among women who know their HIV-sero status; and (iv) establish community-level variation in service utilization.

Data and Methods: The study applied multilevel binary and multinomial logistic regression models to nationally-representative samples of women and men who participated in the 2005/6 and 2010/11 Zimbabwe Demographic and Health Surveys.

Results: Overall, those who were ever tested for HIV and with low HIV stigma were more likely to use maternal health services than their counterparts who had never been tested or with high HIV stigma. These groups were also more likely to use condoms and long-term

contraceptive methods as a means to prevent both unwanted pregnancies and HIV infection. The results from the analysis of HIV testing showed an evidence of improvement in HIV testing uptake between 2005/6 and 2010/11, especially for women. Most individual level socioeconomic and demographic factors associated with HIV testing are largely consistent with patterns in Southern Africa (e.g higher uptake by women and those who are wealthier), but important patterns have also emerged. In particular, results reveal notable gender differences in the determinants of HIV testing: rural residence is associated with lower uptake of HIV testing for women but higher for men; for women, average wealth in a community is a more important factor in enabling HIV testing than household wealth, but the converse is true for men; individual-level, rather than community-level stigma is important for women, while for men, it is community-level stigma that is important.

The analysis of determinants of maternal health care shows that use of antenatal and delivery care services in Zimbabwe are improving and are determined by a wide range of individual-level factors relating to women's economic and demographic status as well as HIV factors relating to stigma, HIV awareness, ever been tested for HIV during pregnancy, knowing someone who died due to HIV, and factors relating to availability and access to health care and media within the community. The individual-level enabling factors that were particularly strong for women included high socio-economic status and not having observed HIV stigmatisation and discrimination. These groups of individuals have an extremely high likelihood of having been ever tested for HIV during pregnancy, or having an early or more than four ANC visits; and have delivered their babies in a health institution with a professional delivery attendant, particularly if they live in richer communities or in communities with low stigma and HIV prevalence.

The analysis of determinants of contraceptive methods choice among women who know their HIV status identified a number of potential pathways of the determinants of this outcome. The analysis revealed that women who know their HIV-positive status were more likely to use condoms and long-term methods than those who know their HIV-negative status. The study also revealed that even though wealth status has no direct effect on the choice of contraceptive methods, it has an indirect impact on the choice of condom versus hormonal methods through intermediate factors such as HIV sero-status.

Conclusions: First, the observed gender disparities in determinants of HIV testing calls for a gender specific response. Couple-oriented HIV counselling and testing services where men accompany their spouses to HIV screening during pregnancy may help increase HIV testing uptake for males and reduce gender disparities. Second, the fact that enabling factors such as socioeconomic status, having been tested for HIV as part of ANC and stigmatization are predictors of maternal health care utilization suggests that being wealthier, having been HIV tested during pregnancy; and having low HIV stigma do translate into expected behavior for pregnant women. Third, knowing own HIV status emerges as a major predictor of condom use and long-term contraceptive methods for women who are HIV positive. These findings have important policy and integrated programme implications for addressing unmet need for HIV and RH services in Zimbabwe.

Acknowledgements

For this thesis to be a success, I received support and encouragement from several people. It is now time to acknowledge those who helped me along the way though too numerous to mention. Firstly, I acknowledge, with immense gratitude and appreciation, the help and support I got from Professor Monica A Magadi and Dr Ruth Butler who guided me all the way in processing my thoughts. Without their support this thesis would not have been successful.

Let me also thank my family, especially my wife Portia and my children Luke and Amanda who always asked- what are you doing Daddy? Everyone's support and encouragement is appreciated.

Acronyms and abbreviations

AIDS	Acquired immune deficiency syndrome
ABC	Abstinence, Be faithful, Use a condom
ANC	Antenatal care
ART	Antiretroviral therapy
BEAM	Basic Education Assistance Module
CI	Confidence interval
CPR	Contraceptive prevalence rate
DHS	Demographic and Health Ssurvey
DIC	Deviance Information Criterion
FP	Family planning
HIV	Human Immunodeficiency Virus
IEC	Information, education, and communication
ICC	Intra-cluster corrélation
ICF	International Calverton, Maryland, USA
IGLS	Iterative generalised least squares

IUD	Intrauterine Device
МСМС	Markov Chain Monte Carlo
MDG	Millennium Development Goal
MQL	Marginalized Quasi Likelihood
МТСТ	Mother-to-child HIV transmission
OR	Odds ratio
PCV	Proportional change in variance
РМТСТ	Prevention of mother-to-child HIV transmission
PNC	Postnatal care
PNC PREP	Postnatal care Predisposing, Enabling and perceived /need factors
PNC PREP PQL	Postnatal care Predisposing, Enabling and perceived /need factors Penalized Quasi Likelihood
PNC PREP PQL RR	Postnatal care Predisposing, Enabling and perceived /need factors Penalized Quasi Likelihood Relative risk
PNC PREP PQL RR SDGS	Postnatal care Predisposing, Enabling and perceived /need factors Penalized Quasi Likelihood Relative risk Sustainable Development Goals
PNC PREP PQL RR SDGS SE	Postnatal care Predisposing, Enabling and perceived /need factors Penalized Quasi Likelihood Relative risk Sustainable Development Goals Standard error
PNC PREP PQL RR SDGS SE SES	Postnatal care Predisposing, Enabling and perceived /need factors Penalized Quasi Likelihood Relative risk Sustainable Development Goals Standard error Socio-economic status

STI	Sexually transmitted infections
TBA	Traditional Birth Attendant
UNAIDS	United Nations Programmes on HIV and AIDS
UNICEF	United Nations Children's Fund
VPC	Variance Partition Coefficient
WHO	World Health Organization
ZDHS	Zimbabwe Demographic and Health Survey

TABLE OF CONTENT

Abstract		iv
Acknowled	lgement	v
Acronyms	and abbreviations	viii
List of Tab	les	Xv
List of Fig	ures	xvi
1 CHAP 1.1 Int	PTER ONE: BACKGROUND AND OBJECTIVES	1
1.2 Ge	neral background	2
1.3 Glo	bal HIV prevention strategies	3
1.3.1	HIV burden and prevention approaches in Zimbabwe	6
1.4 Co	mponents of HIV and SRH integrated services in Zimbabwe	7
1.4.1	Problem Statement	8
1.4.2	Rationale for the study	9
15 Str	ucture of the thesis	12
1.5 50		
2 CHAP	PTER TWO: LITERATURE REVIEW AND CONC	EPTUAL
2 CHAF FRAME	PTER TWO: LITERATURE REVIEW AND CONC WORK	EPTUAL
2 CHAF FRAME 2.1 Int	PTER TWO: LITERATURE REVIEW AND CONC WORK	EPTUAL
2 CHAF FRAME 2.1 Int 2.2 Ev	PTER TWO: LITERATURE REVIEW AND CONC WORK	2 EPTUAL
2 CHAF FRAME 2.1 Int 2.2 Ev 2.2.1 2.2.2	PTER TWO: LITERATURE REVIEW AND CONC WORK roduction olution of HIV Counselling and Testing HIV and maternal health services	EPTUAL 1414141414
2 CHAF FRAME 2.1 Int 2.2 Ev 2.2.1 2.2.2 2.3 Th	PTER TWO: LITERATURE REVIEW AND CONC WORK roduction olution of HIV Counselling and Testing HIV and maternal health services HIV and Family planning	EPTUAL 141414141619 23
2 CHAF FRAME 2.1 Int 2.2 Ev 2.2.1 2.2.2 2.3 Th 2.3 1	PTER TWO: LITERATURE REVIEW AND CONC WORK roduction olution of HIV Counselling and Testing HIV and maternal health services HIV and Family planning eoretical Frameworks The Health Belief Model	EPTUAL 141414161923 23
2 CHAF FRAME 2.1 Int 2.2 Ev 2.2.1 2.2.2 2.3 Th 2.3.1 2.3.2	PTER TWO: LITERATURE REVIEW AND CONC WORK roduction olution of HIV Counselling and Testing HIV and maternal health services HIV and Family planning eoretical Frameworks The Health Belief Model AIDS Risk Reduction Model (ARRM)	EPTUAL 141414161923232323
2 CHAF FRAME 2.1 Int 2.2 Ev 2.2.1 2.2.2 2.3 Th 2.3.1 2.3.2 2.3.3	PTER TWO: LITERATURE REVIEW AND CONC WORK roduction olution of HIV Counselling and Testing olution of HIV Counselling and Testing HIV and maternal health services HIV and Family planning eoretical Frameworks The Health Belief Model AIDS Risk Reduction Model (ARRM) Social Ecological Models of Health Promotion and Structural and E	EPTUAL 1414141619232324 nvironment
2 CHAF FRAME 2.1 Int 2.2 Ev 2.2.1 2.2.2 2.3 Th 2.3.1 2.3.2 2.3.3	PTER TWO: LITERATURE REVIEW AND CONC WORK roduction olution of HIV Counselling and Testing HIV and maternal health services HIV and Family planning eoretical Frameworks The Health Belief Model AIDS Risk Reduction Model (ARRM) Social Ecological Models of Health Promotion and Structural and E	2EPTUAL 14 14 14 14 16 19 23 23 23 24 nvironment 26
2 CHAF FRAME 2.1 Int 2.2 Ev 2.2.1 2.2.2 2.3 Th 2.3.1 2.3.2 2.3.3 2.3.4	PTER TWO: LITERATURE REVIEW AND CONC WORK roduction olution of HIV Counselling and Testing olution of HIV Counselling and Testing HIV and maternal health services HIV and Family planning eoretical Frameworks The Health Belief Model AIDS Risk Reduction Model (ARRM) Social Ecological Models of Health Promotion and Structural and E The Behavioural Model of Health Service Use	2EPTUAL 14 14 14 14 16 19 23 23 23 24 nvironment 26 27
2 CHAF FRAME 2.1 Int 2.2 Ev 2.2.1 2.2.2 2.3 Th 2.3.1 2.3.2 2.3.3 2.3.4 2.3.5	PTER TWO: LITERATURE REVIEW AND CONC WORK roduction olution of HIV Counselling and Testing HIV and maternal health services HIV and Family planning eoretical Frameworks The Health Belief Model AIDS Risk Reduction Model (ARRM) Social Ecological Models of Health Promotion and Structural and E The Behavioural Model of Health Service Use Limitation of the Theoretical Frameworks	25 EPTUAL 14 14 14 16 19 23 23 23 24 nvironment 26 27 29

3 CHA	PTER THREE: DATA AND METHODS	
3.1 St 3.2 St	udy Design	34
3.2 St	ampling Design	35
3.3.1	Ethical consideration	35
3.3.2	Data limitations	36
3.3.3	Missing data	
3.4 In	troduction to multilevel modelling analytical methods	
3.4.1	Multilevel Logistic Model	
3.4.2	Multilevel Multinomial logistic regression Model	
3.5 Fi	xed and Random Effects	43
3.5.1	Inter –class correlations	44
3.5.2	Proportional change in Variance (PCV)	44
3.5.3	Parameter estimation	45
3.6 Si	ummary	45
4 CHA	PTER FOUR: FACTORS ASSOCIATED WITH HIV TH	ESTING
IN ZIM 4.1 In	BABWE	46
4.1.1	Background context	46
4.1.2	Factors associated with HIV testing	47
4.2 C	onceptual framework	
4.3 D	ata and Methods	
4.3.1	The Data	
4.3.2	Outcome variables	
4.3.3	Explanatory Variables	
4.3.4	Individual-level socio-economic and demographic predisposing variab	oles53
4.3.5	Enabling factors	54
4.3.6	HIV and risky sexual behaviour variables	56
4.3.7	Methods of analysis	56
4.4 R	esults	59
4.4.1	Descriptive and bivariate analysis	59
4.4.2	Multivariate multilevel Analyses	64
4.5 D	iscussion, conclusion and recommendation	70

4.5.1	Key findings	70
4.5.2	Study limitations	75
4.5.3	Conclusion and way forward	76
5 CHAI	PTER FIVE: DETERMINANTS OF HIV AND ANC	SERVICES
UTILIS	ATION	77
5.1 Int	roduction	77
5.2 Ba	ckground	77
5.3	Factors associated with HIV and maternal healthcare: evidence from	m available
liter	ature	
5.4 HI	V and maternal healthcare in Zimbabwe	
5.5 Ai	nalytical framework	
5.6 M	ethods and data	
5.6.1	Data source	
5.6.2	Measures of outcome variables	
5.6.3	Key independent variables	
5.6.4	HIV factors	
5.6.5	Methods of analysis	
5.7 Re	esults	
5.7.1	Proportions of mothers who gave birth in the last 5 years precedi	ng the surveys.
5.7.2	Description of Study Population by Background predisposing fac	ctors 95
573	Percentage distribution of women who had HIV test had 4 ANC	visits and ANC
5.7.5	in the first trimester	97
571	Multilevel logistic regression analysis	104
5.7.4	soussion	
5.0 D	scussion	
J.9 Kt		
6 CHAI 6.1 Int	PIER SIX: HIV AND DELIVERY CARE	126
6.1.1	Background	126
612	Rationale for the study of HIV and delivery care in Zimbabwe	120
6.2 M	ethods and data	120
621	Outcome variables	120
622	Evolopotory variables	121
0.2.2	Explanatory variables	131

6.3	Results
6.3.1	General trends in institutional and professional delivery care136
6.3.2	2 Delivery attended by a professional delivery attendant and institutional delivery
6.3.3	Bivariate association between institutional and professional delivery by
	individual level predisposing factors
6.3.4	Bivariate association between institutional and professional delivery by enabling
	factors141
6.3.5	5 Multivariate multilevel analysis
6.4	Discussion
6.5	Conclusion
7 CE	IAPTER SEVEN: KNOWLEDGE OF HIV STATUS AND
CON	TRACEPTIVE METHODS CHOICE AMONG WOMEN IN
ZIIVII 7.1	Introduction
7.2	Background and Objectives
7.3	Data and Methods
7.3.1	Data source
7.3.2	2 Methods of analysis
7.3.3	3 Outcome variables
7.3.4	Key predictor variables
7.3.5	5 Selection of control variables
7.4	Results
7.4.1	Distribution of contraceptive methods choice in Zimbabwe
7.4.2	2 Bivariate association between HIV status, background characteristics and
	contraceptive method choice
7.4.3	B Determinants of contraceptive methods choice among women who knew their
	HIV status in Zimbabwe184
7.5	Discussion and conclusion
8 CE	HAPTER EIGHT: CONCLUSIONS AND RECOMMENDATIONS
••••	
8.1	Introduction
8.2	Summary of key findings

8.2.1	Association between individual-level predisposing factors and HIV and SRH-
	integrated health outcomes198
8.2.2	Association between individual and community-level social and economic
	enabling factors and HIV and SRH-integrated health outcomes203
8.2.3	Association between HIV 'enabling' and 'perceived risk/need'factors and HIV
	and SRH-integrated health outcomes
8.3 Rec	commendations, policy and research implications
8.3.1	Poverty as a potential impediment to utilization of services
8.3.2	The need to target the youth
8.3.3	HIV perceived risk/need factors
8.3.4	Knowledge of HIV status as trajectory to better choice of contraceptive methods
8.3.5	Final remarks
REFERI	ENCE
Appendix (95% of females	x 4.1: Average odds of HIV testing from multilevel logistic regression models confidence intervals are given in square brackets) males and
Appendix (95% cor	6.1: Average odds of HIV testing from multilevel logistic regression models nfidence intervals are given in square brackets) institutional and professional

LIST OF TABLES

Table 3-1: Samples showing ZDHS Hierarchical pooled datasets 35
Table 4-1: Trends in HIV testing coverage and receipt of results, 2005/6-2010/11 ZDHS59
Table 4-2: Distribution of HIV testing by background socio-economic and demographic
characteristics
Table 4-3: Distribution of testing by enabling and need factors
Table 4-4: Average odds of HIV testing from multilevel logistic regression models (95%
confidence intervals are given in square brackets)-females65
Table 4-5: Average odds of HIV testing from multilevel logistic regression models (95%
confidence intervals are given in square brackets)-males
Table 5-1: List of variables related to each set of variables shown in Figure 5.1 above
Table 5-2: Percentages of mothers who gave births in the last 5 years preceding the surveys,
2005-6; 2010-11 ZDHS
Table 5-3: Proportion of last birth in the last 5 years by selected socio-demographic factors 96
Table 5-4: Percentage distribution of HIV and reproductive health care outcomes in the
samples, 2005/6 and 2010/11 ZDHS
Table 5-5: Bivariate analysis of timing of ANC, antenatal HIV testing and frequency of
ANCvisits by predisposing factors, 2005/6 and 2010/11 ZDHS101
Table 5-6: Bivariate analysis of timing of ANC, antenatal HIV testing and frequency of ANC
visits by enabling and need factors, 2005/6 and 2010/11 ZDHS103
Table 5-6 Continued 104
Table 5-7: Average odds ratios of ANC visit in the 1st trimester from multilevel logistic

Table 5-8: Average odds ratios of ANC- HIV testing from multilevel logistic regression models
(95% confidence intervals are given in square brackets)112
Table 5-9: Average odds ratios of at least four ANC visits from multilevel logistic regression
models (95% confidence intervals are given in square brackets)
Table 6-1: Summary description of study variables 134
Table 6-2: Percentage Distribution of institutional and professional care delivery in Zimbabwe,
2005/6 and 2010/11 ZDHS
Table 6-3: Bivariate analysis of institutional and professional delivery by predisposing
individual factors
Table 6-4: Bivariate association between institutional and professional delivery by enabling
and perceived risk/need factors142
Table 6-5: Average odds ratios of institutional delivery from multilevel logistic regression
models (95% confidence intervals are given in square brackets)
Table 6-6: Average odds ratios of professional delivery from multilevel logistic regression
models (95% confidence intervals are given in square brackets)150
Table 7-1: Selected Covariates for Method Choice Estimation
Table 7-2: Bivariate associations between demographic variables and contraceptive method
choice among women aged between 15-49 years of age in Zimbabwe
Table 7-3: Results of the multilevel multinomial logistic regression models to explore
associations between HIV status and contraceptive method choice (adjusted RR and 95% CI
of RR)
Table 8-1: Summary of the effects of individual and community factors of HIV and SRH
outcomes
Table 8-2: Summary implication for health policy in Zimbabwe

LIST OF FIGURES

Figure 1-1: Pathways to achievement of 90-90-90 HIV strategic milestones
Figure 2-1: Socio-ecological model
Figure 2-2: Analytical framework for the study of HIV and SRH service utilisation32
Figure 3-1: Study area and population
Figure 4-1 Conceptual framework for the analysis of determinants of HIV testing51
Figure 5-1: Analytical framework on determinants of HIV and maternal health integrated
service utilization
Figure 5-2 : Percentage distribution of HIV and reproductive health care outcomes in the study
population by survey year
Figure 6-1: Analytical framework for the analysis of determinants of delivery in Zimbabwe
Figure 6-2: Percentages of birth deliveries that were attended by a health professional by
institutional delivery
Figure 6-3: Percentages of health professional birth deliveries by institutional delivery138
Figure 7-1: Percentage distribution of method choice among women who ever used any
contraceptive methods (N=13698)167
Figure 7-2: Distribution of contraceptive methods choice for the 2005-11 ZDHS (N=3757)

1 CHAPTER ONE: BACKGROUND AND OBJECTIVES

1.1 Introduction

This thesis focuses on the subject of HIV and sexual and reproductive health (SRH) integrated services, which has emerged as a strong strategy to enhance efforts to reduce the HIV epidemic. The persistent high mortality related to HIV infection has prompted calls for scaling-up access to family planning and reproductive health care services as a trajectory to prevent HIV infection (Druce et al., 2009; Kennedy et al., 2010). Thus, HIV prevention programs have been integrated into family planning and reproductive health care services as a means to reach out to men and women who are HIV positive and those who are vulnerable to HIV infection (Church et al., 2013; Hope et al., 2014). While the rationale for integration of HIV and sexual and reproductive health (SRH) services seems to be strong, there is a paucity of information on which population groups are most likely to use these services. Due to the considerable stigma attached to HIV/AIDS, people living with HIV and those who perceive themselves to be at risk of HIV infection may be less likely to use integrated health care services (Turan et al., 2008). The efficiency of HIV and SRH integrated services has been improved in recent years and most of the improvements have been achieved by removing socio-economic barriers to SRH services, in particular antenatal and delivery care services. Nonetheless, it is possible to further improve the efficiency by increasing uptake of HIV testing services, better timing of maternal health care and family planning choices. This study seeks to inform policy and programs on better integration of HIV testing, maternal health care, and family planning services in order to optimize HIV prevention programs such as HIV testing, prevention of mother to child HIV transmission and condom use with the broader aim to reduce the HIV pandemic. This chapter includes a background of intervention programs followed as pathways to increase uptake of HIV testing. Included in the chapter is also a statement of the problem of HIV burden in Zimbabwe and specific objectives of the thesis are presented in the last section, leading to an outline of thesis chapters.

1.2 General background

By the end of 2013, the total number of people living with HIV was estimated to be 35 million (WHO, 2015). Quite recently, considerable attention has been paid on HIV and sexual and reproductive health integrated services as a pathway to reducing new HIV infection by 90% between 2015 and 2020 (UNAIDS, 2015). Notwithstanding the fact that risk of HIV transmission remains high, the last two decades have seen a remarkable decline by 40% between 2000 and 2015, from an estimated 3.5 million cases to 2.1 million (GARPR, 2016). The percentage reduction in number of new HIV cases does not only reflect better data available and statistical methodologies (Granich et al., 2015), but also reflects effectiveness of HIV intervention programs in some parts of the world (WHO, 2016). Overall, the annual reduction in the number of new HIV infection was estimated at 4% between 2010 and 2015 (USAID, 2016). The current 4% annual reduction in new HIV infection is slower than the 5.5% annual reduction needed to reach the millennium development goal number six (MDG 6)-to reducing new HIV infection. Although the rate of new HIV infection is decreasing in most parts of the world, it is probable that new HIV infection rates will continue to increase in most of the Sub-Saharan African (SSA) countries due to insufficient uptake of HIV intervention programs (Granich et al., 2015; Musheke et al., 2013). As is well known, the proportion of new HIV infection is highest in Sub-Saharan African (SSA) countries and about 66% of global new HIV infection was attributed to countries in SSA by the end of 2015 (USAID, 2016). The majority of these cases come from 20 SSA countries that contributed 69% of all new HIV infection, with Zimbabwe being among the top ten of the countries with the highest number of new infection (Wanga et al., 2016). While the current global agenda to end AIDS within the Sustainable Development Goals (SDGS) is more likely to be achieved in majority of countries in the developed world, this is less likely in most of the countries in SSA unless major efforts are made to increase uptake of HIV intervention programs (Schneider et al., 2016). While lessons learnt from the successes of the MGD 6 at national and global levels can act as a pathfinder for the 2030 agenda for Sustainable Development, it may be difficult for low to middle income countries to overcome social, economic and gender inequalities to promote inclusive societies (Maleche, 2015). In setting out a new approach to end HIV and AIDS, the UNAIDS 2016-2021 strategy firmly acknowledges the need to anchor the HIV response in Agenda 2030, recognising the interdependence between HIV and SDGS, from ending poverty (SDG 1), to promoting inclusive societies (SDG 16) and strengthening partnerships (SDGS 17) within the countries and at global level (UNAIDS, 2015). It has been argued that the SDGS provide a key opportunity to address the structural drivers that underlie the HIV epidemic. It may be that ending HIV and AIDS is a feminist issue, thus giving women access to sexual and health, education and services and reducing gender inequality and improvement of rights may be integral to a sustainable HIV response (Femnet, 2016)

1.3 Global HIV prevention strategies

To solve the problem of persistent HIV burden, WHO (2015) has set up a 90-90-90 strategic milestones to be achieved by united nation (UN) member states by 2020. The milestones means that by 2020, 90% of all people living with HIV will know their HIV status, 90% of all people with diagnosed HIV infection will receive sustained antiretroviral therapy and 90% of all people receiving antiretroviral therapy will have viral load suppression. To achieve these new ambitious goals, USAID's (2015) strategic plan has called for renewed efforts to increase uptake of sexual and reproductive services (SRH) including family planning services as a trajectory to prevention of new HIV infection around the world (Church et al., 2013;Keogh et al., 2009). Consequently, integration of HIV with SRH services, including family planning, has become a central strategy in an effort to reduce HIV burden. This approach is based on UNAIDS' recognition that reducing HIV infection is closely tied to advancement in addressing maternal health and family planning needs; and in providing comprehensive HIV prevention strategies and SRH care services (Druce et al., 2006; Church et al., 2013; Turan et al., 2012). The HIV and SRH integrated care model links different types of SRH and HIV services by physically locating services in one place or by system referral (Hope et al., 2014). This is important because two or more services can be offered at the same location, during the same opening hours, with the provider actively encouraging clients to consider using other services (Turan et al., 2012). It has been highlighted that integration of HIV services into SRH can increase the pool of people who might not normally be reached through traditional HIV testing clinics e.g. youth, men and commercial sex workers (Turan et al., 2015). There is evidence that individuals who use SRH services often benefit from HIV services and vice versa (Kennedy et al., 2010). For example adding family planning to HIV testing, treatment and care may facilitate the uptake of condom use and long term acting contraceptive methods both of which are crucial for preventing unwanted pregnancies and HIV infection (Mumah et al., 2014). In the context of prevention of mother to child HIV transmission (PMTCT), the benefits of integrating services includes timely identification of HIV positive mothers, early initiation of antiretroviral (ARV) treatment for both mother and newborn, shorter waiting times, lower transport costs, and reduced stigma and discrimination (Hope et al., 2014).

There is evidence from intervention programs pre-2015 that coordinated HIV and SRH integrated programhas helped galvanise support for HIV response (Druce et al., 2006; Hope et al., 2014; Turan et al., 2015), and that programs post-2015 have seen renewed efforts to improve the efficacy of integrated programs (Schneider et al., 2016). In light of the benefits accrued from offering integrated services, most SRH programmes are being revolutionalised and there is a call for a strong momentum at country level to optimize integration of HIV prevention services into SRH care, including family planning services (UNAIDS, 2014). The United Nation (UN) member states are now being urged to build a vibrant coalition between HIV prevention programmes, maternal health care; and family planning constituencies around the goal of ending HIV by 2030, and to promote greater synergies and the strategic integration of HIV counselling, testing and treatment, prevention of mother to child HIV transmission (PMTCT), maternal health and family planning programmes (Rutaremwa and Kabagenyi, 2016).

Despite the fact that integrated services are critical for increasing uptake of HIV intervention programs, maternal health and family planning literature demonstrates that HIV/AIDS factors such as risk perception, stigma and discrimination are some of the major barriers to access to integrated services in many developing countries (Akwara et al., 2003; Church et al., 2013; Matake and Matake, 2016; Meiberg et al., 2008; Murire et al., 2014; Lemnge et al., 2011). It can thus be reasonably assumed that those who are HIV positive and those who perceive themselves to be at risk of HIV transmission are less likely to use integrated services due to fear of stigma and discrimination (Turan et al., 2012). In the literature, HIV positive people tend to experience stigma and discrimination in the community, family and in health facilities (Jurgensen et al., 2012; Stephenson et al., 2013; Turan et al., 2012; Venkatesh et al., 2011; Yeatman, 2009). Nonetheless, it is worthwhile noting that the changing HIV stigma and discrimination environment, impact of HIV treatment and longevity of many people living with HIV- may alter individual's health seeking behaviour (Magadi, 2016). It is likely that individuals who are HIV positive and those who may perceive themselves to be HIV positive

may have reconsidered their decisions on SRH services as a result of improved HIV intervention programs (Adedimeji et al., 2012; Kennedy et al., 2010; Harrison et al., 2010). The past decade has seen a rapid growth of HIV intervention programs that are targeted at reducing HIV-related stigma and discrimination in the communities (Kalichman et al., 2009), in schools and hospitals (Kevin et al., 2014), and in faith-based organisations (Hartwigg and Kissioki, 2009). While we may know intuitively that reduction in HIV related stigma and discrimination are associated with increased uptake of integrated services, there is much to learn about how specific HIV factors including sero-status, HIV awareness and sexual behaviour affect HIV and sexual and reproductive health integrated services (Kalichman and Simbayi, 2003; Mashuba and Hemalata, 2015; Turan et al., 2008). Many hypotheses regarding the impact of HIV on uptake of integrated services appear to be not well grounded due to lack of data and inappropriate statistical methodologies which do not control for HIV factors.

Besides the effects of HIV on uptake of integrated services, several classical measurements of socio-economic and demographic factors have also been linked to service uptake. These factors include age, gender, marital status, media, education, employment status, religion, place of residence and media exposure (Adamu, 2011; Achia and Mageto, 201; Alam et al., 2015; Ahmed et al., 2010; Golis et al., 2013; Hartwig et al., 2006; Jesmin et al., 2013). A key problem with much of the literature in relation to socio-economic determinants of integrated services relates to its lack of consistency in their findings. There is still considerable ambiguity with regards to the effects of socio-economic and demographic factors on HIV and SRH integrated services. One question that needs to be asked is 'by how much do socio-economic and demographic factors contribute to the variation in service utilisation of integrated services?' after HIV factors are controlled for. Experts contend that the evidence relating to the association of socio-economic and geographic factors with utilisation of integrated services is not conclusive, unless the impact of HIV is controlled for (Magadi and Magadi, 2017; Martin-Herz et al., 2006; Mitchel et al., 2010; Sahlu et al., 2014). Unfortunately, most of the studies neglect to explain the role of HIV in either promoting or inhibiting utilisation of integrated services. Taking advantage of the ever increasing unmet need for sexual and reproductive health services as a means to improve HIV intervention programs, the focus should now be placed on not only addressing socio-economic and structural barriers, but also on targeting people who are vulnerable to HIV infection and those who suffer stigmatisation (Lemnge et al., 2011; Murire et al., 2014) so that no one is left behind (Maleche, 2015). This presents a challenge to the renewed international efforts to end the HIV epidemic by 2030. Notwithstanding the shortcomings, the HIV and SRH integrated approach is being applied in SSA and in Zimbabwe as a means to reduce HIV new infection by 90 percent between 2015 and 2020.

1.3.1 HIV burden and prevention approaches in Zimbabwe

By 2015, Zimbabwe sought to half new HIV infection as prescribed by the World Health Organisation's strategic plan (USAIDS, 2016). Although it has been a daunting task, adult HIV prevalence in Zimbabwe has almost halved from its peak of nearly 30% around 1997 to around 13.7% in 2011 (ZDHS, 2010/11). HIV related deaths have also been reduced by over 60% as a result of a successful HIV testing, treatment and support programme (ZNASP 2011-2015). However, despite the government's efforts to halt new infections of HIV, Zimbabwe remains one of the countries in Sub-Saharan African countries still burdened with the HIVepidemic (Granich et al., 2015). The current aim in the post-2015 era is to end the AIDS epidemic by 2030. In line with the USAID's goal, the government of Zimbabwe has set up the 90-90-90 strategic milestones to be achieved by 2020 (ZNASP III, 2015-2018). To achieve these goals, HIV services are being made available and accessible through sexual health clinics, maternal and child health care and family planning services as a strategy to reach out to populations who may not otherwise seek HIV services (Wiegert et al., 2014). An example of this integration is where individuals visiting family planning and sexual health clinics for other services are offered HIV counselling and testing (Demisse, 2009; Hope et al., 2014; Obure et al., 2015). In the case of prevention of mother to child HIV transmission, pregnant women who visit maternal health care are screened for HIV and are given treatment, care and support throughout to delivery (Sibanda et al., 2013). At these clinics, sex education is offered such as on risk reduction strategies including practicing safer sex, provision of males and female condoms for both family planning and prophylactic reasons and provision of effective long-term contraceptives to avoid unwanted pregnancies (McCoy et al., 2014; Mumah et al., 2014). The bi-directional integrated services also includes outreach HIV testing and counselling services and referral for special populations such as people living in remote rural areas, vulnerable groups such as prisoners, sex workers and highly mobile populations such as long distance truck drivers (Zimbabwe National HIV and AIDS Strategic Plan 2011-2015).

1.4 Components of HIV and SRH integrated services in Zimbabwe.

In Zimbabwe, the public health system is the largest provider of health-care services, complemented by mission hospitals and health care delivery by non-governmental organisations (NGOS). About 68% of 1589 primary health facilities are providing HIV services i.e HIV testing, counselling and testing, at the same time they provide SRH services (ZNNPS, 2015-2018). The visual depiction representation in figure 1.1 indicates a simulation of four domains of SRH services that are integrated with HIV intervention programs in Zimbabwe. The multiple tiered interventions, HIV testing, comprehensive maternal health care services and family planning are assumed to be the pathways through which the 90-90-90 strategic milestones may be realised in Zimbabwe by 2020. The first domain relates to HIV testing in the general population. This is achieved through an out reach program which is targeted at special populations such as people living in remote areas and the other hard to reach populations. The outreach HIV program is backed by a referral support system to integrated health centres. In addition to the outreach program, HIV testing and counselling targeting the general populationcan also be initiated and recommended by health care providers to all clients attending health care facilities as a standard component of medical care.

The second domain relates to the antenatal care package putting emphasis on early ANC booking, frequent ANC visits and antenatal HIV testing. Zimbabwe National Guidelines on HIV testing and counselling stipulates that all pregnant women be offered routine HIV testing as part of ANC. The framework requires pregnant women to initiate a first ANC visit in the first trimester, and to have at least four more ANC visits as well as having antenatal care HIV testing. Thirdly, institutional delivery and child birth with a trained health professional is an additional intention in prevention of mother to child HIV transmission. Lastly, HIV prevention such as condom promotion and permanent contraceptive methods are also integrated with family planning services. The guidance draws from WHO standards in health care service delivery (WHO, 2012) with specific reference to strategies on increasing HIV testing and prevention of mother to child HIV transmission. It is hoped that identification of determinants of each of the components of HIV and SRH integrated care may help us identify populations that may be targeted in order to optimise the on-going HIV and SRH integrated program.

Figure 1-1: Pathways to achievement of 90-90-90 HIV strategic milestones



Source: Author's construct

1.4.1 Problem Statement

Zimbabwe is still burdened with an HIV epidemic and the government has an ambitious aim in the post-2015 era to end the AIDS epidemic by 2030. It is a daunting task to increase uptake of HIV and SRH integrated services to meet the government targets of reducing HIV by 90% between 2015 and 2020. Only 36% (men) and 57% (women) aged 15-59 years have ever beeen tested for HIV (ZDHS, 2010/11); suggesting that 64% (men) and 43% (women) in Zimbabwe are not aware of their HIV status. Despite the government efforts in encouraging pregnant women to initiate antenatal care in the first trimester and to have at least four ANC visits as well as to have an HIV test during pregnancy, only 19 % of these women had an ANC visit in the 1st trimester, 67 % had attended the recommended four or more ANC visits and 59 % had received counselling and accepted an HIV test as part of ANC care (ZDHS, 2010/11). In addition, the proportion of pregnant women having institutional and professional delivery falls short of what should be expected in Zimbabwe. By 2010/11 only 67% and 68% of pregnant women had institutional and professional delivery, respectively (ZDHS, 2010/11). While the contraceptive prevalence rate (CPR) is high in Zimbabwe, the use of condoms and long term methods for HIV prevention and unplanned pregnancy is low. According to the (ZNASP 2011-2015) report, the CPR is estimated at 65% and of these 63% are using modern contraception and 2% traditional methods. The contraceptive method commonly used is a pill (48%) and it is also estimated that 12% and 37% of births are unplanned and mistimed, respectively. In the absence of prevention of mother to child HIV transmission services, unplanned and mistimed births are risk factors for mother to child HIV transmission among HIV positive mothers (Admachak et al., 2007; Agadjanian and Hayford, 2009).

1.4.2 Rationale for the study

Despite remarkable gains in improvement of integrated services in Zimbabwe over the last few decades, there is a lack of evidence of a corresponding increase in the uptake of HIV prevention programs through sexual and reproductive health services over the years. Testing the link between the efficacy of integrated services and uptake of HIV intervention programs is difficult. Despite the interest in service integration, methodological problems prevent most studies from explicitly testing whether integrated services increase the uptake of HIV intervention or maternal health and family planning programs (Rutaremwa and Kabagenyi, 2016). In the literature, this problem rests on the many determinants of SRH that also impact on decisions on accessing HIV programs (Car et al., 2012; Church et al., 2013). Many program managers and policy makers have recognised these joint determinants when they noticed that some individuals are more receptive to both SRH and HIV services (Fetene and Feleke, 2010). In statistical terms, 'HIV and SRH integrated service' utilisation is an endogeneity variable (Obure et al., 2016). The authors underline that ignoring this endogeneity produces biased and misleading estimates of the impact of SRH utilisation on HIV service receptiveness. Studying the impact of integrated services on HIV prevention programs requires complex statistical models on determinants of each of the individual components of the HIV and SRH integrated service program. However, our knowledge of determinants of HIV and SRH integrated services is largely based on very limited data. Thus, studying the determinants of HIV testing, maternal health and family planning separately becomes at the fore.

I am aware of population based studies on the individual components of HIV and SRH care services in Zimbabwe. For example, Muchabaiwa et al (2012) examined determinants of antenatal, delivery and postnatal care, Maruva et al (2014) studied knowledge of HIV status and antenatal care and utilisation of maternal health care services, Ngobe and Odimegwu (2014) examined the influence of both individual, household and community determinants of adolescence contraceptive use; and (Mumah et al., 2014) also explored the influence of

knowledge of HIV status and contraceptive choice in Zimbabwe. Others such as (Buzdugan et al., 2015; Chandisarewa et al., 2012; Mucheto et al., 2011) have all studied antenatal care HIV testing, facility-based delivery, prevention of mother to child HIV transmission in localised settings respective.

A key problem with many of the studies on SRH in Zimbabwe relates to methodology. All of them have not taken into account the role of community level factors which may also influence health seeking behaviour (Achia and Mageto, 2015; Adamu, 2011; Aremu et al., 2011; Alam et al., 2015; Magadi and Agwanda, 2010). The shortcomings of their methodologies in Zimbabwe have been clearly recognised in literature (Ngome and Odimegwu, 2014). It has been argued that one of the major problems of investigating the effects of HIV and socioeconomic factors on a wide range of HIV and SRH integrated outcomes in Southern Africa is the lack of well structured multilevel multivariate techniques to determine the relative importance of various HIV and socio-demographic factors that have an influence on access to SRH outcomes (Chitalu et al., 2013; Stephenson et al., 2013). This study is the first to apply multilevel models to investigate the determinants of HIV and SRH outcomes in Zimbabwe. The study shifts from not only explaining the classical individual level factors but also includes the independent effects of community level factors. The rationale for considering community level factors on service utilization has been well recognized in the social sciences field (Aremu et al., 2011; Delprato, 2016; Diez-Roux, 1998; Magadi and Agwanda and Obare, 2007; Utman, 2010; Vu, 2005). Murire et al (2014) have pointed out that community environment governs our behaviour, thus poverty, HIVprevalence, stigma or awareness in the community also must govern people's SRH behaviour. It has been suggested that many contextual or aggregated variables that have been hypothesized to affect health care service utilization represent properties that vary over geographical and social units (Babalola and Fatusi, 2009; O'meara et al., 2013; Jones, 1991)

The ZDHS (2010/11) estimates of HIV testing and SRH utilisation including contraceptive methods use show wide differences between communities in Zimbabwe. There is considerable literature noting that the explanation of this diversity in service utilisation is numerous-but the closest are the indicators of individual and community level socio-economic and demographic factors (Buzdugan et al., 2015; Makate and Makate, 2016; Kabagenyi et al., 2016). Makate and Makate (2016) recognise that Zimbabwe is a country characterised by huge internal

differences-demography, cultural, economic, cultural and environment. All these factors are therefore likely to have a significant contribution to the service utilisation in Zimbabwe.

Whilst literature on access to SRH indicates low uptake of SRH services is as a result of social and economic disadvantages (Ahmed et al., 2010; Goli et al., 2013; Kevany et al., 2012; Obermeyer et al., 2013), studies on HIV suggests that factors such as HIV awareness and stigma may also lead to low uptake of health care services (Chirawu et al., 2013; Murire et al., 2014; Maruva et al., 2014; Mumah et al., 2014; Sambisa, 2008). These perspectives have rarely been considered and discussed together at a national level in Zimbabwe.

It is important that factors associated with HIV and SRH outcomes are well understood in order to devise appropriate strategies to improve uptake of HIV and SRH services. HIV testing, adequate antenatal and delivery care as well as family planning are all useful in addressing HIV new infections. An investigation of factors associated with HIV and SRH utilisation is necessary for an improved understanding of barriers to effective HIV and SRH integrated services for the development of appropriate interventions. Efforts to increase uptake of HIV and SRH are held back by a lack of comprehensive understanding of the breadth and complexity of factors that influence the use of a wide range of HIV and SRH care services (Piot et al., 2015). This thesis hypothesises that the levels of HIV and SRH care service utilization observed in Zimbabwe are not only as a result of socio-economic and demographic characteristics, but also as a result of the impact of HIV. The overall aim of the thesis, therefore, is to improve understanding of factors associated with utilisation of HIV prevention programmes within the integrated health care service system in Zimbabwe.

Focusing on individual and community-level predisposing, enabling and perceived risk/need factors, the specific objectives of the thesis are:

- i. To identify determinants of HIV testing
- ii. examine determinants of maternal care service utilisation i.e antenatal (antenatal HIV testing, timing and frequency of antenatal care visits) and delivery care (institutional and professional delivery) including the independent effects of HIV on these services
- iii. Explore determinants of contraceptive methods choice among women who know their HIV sero-status

- iv. To establish community variation in service utilisation, and
- v. To synthesise and interpret results to give health policy and planning recommendations.

Research questions

To achieve these objectives the following questions were to be answered:

Question 1: To what extent do individual and community factors influence HIV and SRH integrated service utilisation in Zimbabwe?

Question 2: To what extent do HIV factors independently influence utilisation of HIV and SRH care service including contraceptive methods choices?

Question 3: To what extent do HIV and SRH service utilisation vary across the communities in Zimbabwe?

Community and neighbourhood effects are a matter of practical concern for policy makers and program planners. Thus, the findings of the focused studies also addressed the following question:

Question 4: Is there evidence that would call for policy interventions targeted at the community level in addition to those targeting the individuals?

1.5 Structure of the thesis

The thesis is divided into eight chapters. Chapter two presents a literature review and the conceptual framework. The first part of the literature review focuses on determinants of HIV/AIDS testing in Zimbabwe, the impact of HIV/AIDS on maternal health care services utilization and on contraceptive methods choice. Finally, the limitations of the theoretical framework are presented.

The third chapter on methodology includes a description of the data that will be used and statistical methods to be adopted. The first section describes data, including the statistical methods. The selection of variables is based on the literature and conceptual framework

discussed in chapter two. The last part of the chapter gives model specifications and the estimation procedure.

Chapter four presents findings from a multilevel logistic regression model to identify factors influencing HIV testing. The independent effect of HIV, individual and community factors associated with HIV testing are examined. The chapter concludes with a summary of findings and their implication for policy and program intervention.

Chapter five identifies determinants of HIV and antenatal care (ANC) services. These services include HIV testing during pregnancy, four or more ANC visits and ANC in the first trimester. The analysis of these study outcomes uses a two-level multilevel logistic regression analysis. In addition to the reproductive factors relating and the background socio-economic and HIV perceived risk/need factors; the analysis includes factors relating to the availability and accessibility of health care services within communities.

Chapter 6 relates to delivery care. Factors associated withinstitutional and professional delivery areexamined using two-level multilevel logistic regression models, leading to chapter seven.

Chapter 7 examines determinants of contraceptive choice among women who know their HIV sero-status. The response outcome is an unordered categorical variable that distinguished between different types of methods among users, thus multinomial regression models were applied to contraceptive choices of women who are using contraception who knew their HIV status.

Chapter 8 concludes the analysis part of the thesis by bringing together all of the evidence from the study. It presents a summary of the most interesting and important outcomes from the analyses, along with the conclusions and their implication for HIV and SRH health services. The results are discussed with reference to the conceptual framework perspectives outlined in chapter 2, which attempt to explain the pathways to HIV and SRH care service utilization, as well as with reference to previous work undertaken in this field.

2 CHAPTER TWO: LITERATURE REVIEW AND CONCEPTUAL FRAMEWORK

2.1 Introduction

To set out the context of this focused study, chapter two presents a general background of the study of HIV and SRH services. Included in the chapter is the historical overview of HIV testing, and the rationale for integration of HIV with maternal health and family planning services. Various relevant theoretical frameworks are also presented and discussed, and the chapter concludes with a discussion of a general analytical framework that has been adapted to guide the study. Comprehensive literaturerelevant for different chapters are provided in the respective chapters rather than presented here.

2.2 Evolution of HIV Counselling and Testing

HIV testing is considered to be one of the most important components of HIV prevention strategies and is an entry point to treatment and care (Berkman, 2001). The first test for HIV became widely available in developed countries in the mid-1980s, a few years after the virus was discovered to be the cause of AIDS (Bayer, 1991). The initial approach to HIV testing emphasised that it should be client based initiated and to be done voluntarily (Bayer and Fairchild, 2006). In the light of activism which demanded the need for protection of human rights of those infected which was highly stigmatised, counselling was later co-opted as a crucial component of HIV prevention (Bayer, 1991; DeCock, 1998; Heywood, 2005). The chief advantages of post-test counselling is that it provides an opportunity to offer emotional, psychological support and to help clients understand the significance of their test results (whether HIV positive or negative) and to plan for the future (Angotti et al., 2009; Chirawu et al., 2013).

With the introduction of antiretroviral treatment in the mid to late 1990s, policy documents began to emphasize the importance of HIV testing and counselling not only for HIV prevention, but also in linking HIV positive individuals to treatment and care (De Cock, 1998; Bayer and Fairchild, 2006). By the early 2000s, international organisations were examining ways to increase the uptake of HIV counselling and testing services and scale up antiretroviral (ARV) use in SSA (Granich et al., 2015). Debates ensued about 'opt out' and opt in' approaches to testing coming from the human rights health perspective (Bayer and Edington, 2009).

However, in recognition that many opportunities to counselling and to diagnose individuals attending health facilities were being missed in 2007, the WHO and USAIDs published guidelines on HIV counselling and testing to be offered by health care workers within facilities and in the communities (Heywood, 2005). The guidelines recommended opt out as a standard part of medical care. While there has been a dramatic increase in HIV counselling and testing uptake in SSA in recent years, there are considerable regional and community variations in HIV testing. The Demographic and Health Survey (DHS) data from 2004-2010, show that the proportion of women ever tested for HIV ranges from one percent in Chad in 2004 to 76 percent in Rwanda in 2010, and for men ranges from 4 to 69 percent, also in Chad in 2004 and Rwanda in 2010, respectively (Staveteig et al., 2013).

Several facility based HIV intervention programs in SSA countries have reported a substantial increase in HIV testing across SSA (Fetene and Feleke, 2010; Kennedy et al., 2010; Nakanjako et al., 2006,) with much of this being attributed to the testing of pregnant women during antenatal care (Godif et al., 2015; Mitchel et al., 2010). The last few years have also seen a remarkable increase in mobile community-based outreach HIV testing programs (Dugas et al., 2015; Sharma et al., 2017;Underwood et al., 2016; Wheeler et al., 2015). Backed up by an effective referral system, these outreach programs have been found to substantially increase the uptake of HIV testing (Elul et al., 2014; Naik et al., 2015; Sanga et al., 2016). The outreach program helps by reducing logistical costs for individuals who have to travel to a health facility (Muketo et al., 2015).

A number of studies have reported on the general socio-economic advantages that are associated with individuals who reported ever having been tested for HIV (Godif et al., 2015; Musheke. 2013; Moyer and Mustafa, 2013; Venkatesh et al., 2011). Some studies have found the use of HIV testing to be associated with higher risk sexual behaviours (Sambisa, 2008; Rosenberg et al., 2013; Yeatman et al., 2009). Others found HIV factors such as awareness, stigma and risk perception to be major barriers to HIV testing within health facilities and in the community (Martin-Herz et al., 2006; Jurgensen et al., 2012). In spite of the shortcomings of HIV testing programs, there is a general acceptance that increasing uptake of HIV testing may reduce stigma, discourage risky sexual behaviours and may encourage use of other SRH services such as maternal health care and family planning services (Rosenberg et al., 2013; Tedrow et al. 2011). Whilst, HIV testing is broadly directed at specific high-risk populations,

pregnant women have perhaps received the most attention for HIV testing, presumably because mother-to-child transmission is the leading cause of HIV infection among children (Godif et al., 2015). Documents reviewed have shown that mother to child HIV transmission is now a major contributor to new HIV infection in SSA countries (Granich et al., 2015; Heemelaar. 2014). In order to prevent mother to child HIV transmission, antenatal care (ANC) and delivery care became prime locations for HIV screening, treatment and care for pregnant women (WHO 2010). In the last decade, HIV testing, counselling and treatment have been incorporated in the maternal health care system in order to enhance the (PMTCT) program, and the next decade is likely to see a considerable effort to entrench the integration of HIV with other services within the sustainable Development Goals framework.

2.2.1 HIV and maternal health services

In the general population, HIV testing may help by discouraging risky sexual behaviours and individuals can plan for the future, including childbearing of HIV free infants (Wringe et al., 2008), thereby reducing new HIV transmission. Given the importance of HIV testing in prevention of mother to child HIV transmission services, HIV services have also been integrated with maternal health related care (Church et al., 2013; Turan et al., 2015; Washington et al., 2015; Wiegert et al., 2014). While, all pregnant women are encouraged to use maternal health care and to use HIV services to prevent mother to child HIV transmission, there is scant information relating to the sub-groups of pregnant women who are less likely to use HIV and RH integrated services. Whereas there are reports suggesting that integration of HIV and maternal health has been successful in many countries in SSA (Car et al., 2012; Turan et al., 2012), there is evidence that many pregnant women face barriers relating to socioeconomic status and HIV in accessing maternal health care services. It has been noted that, due to social stigma, access to HIV and maternal health services is still a crucial issue among pregnant women (Jayleen et al., 2016; Wiegert et al., 2014). Since most people still perceive HIV/AIDS as a disease of bad character (Musheke et al., 2013), few pregnant women are willing to disclose their HIV status to anyone, even to health providers or to the nearest and dearest family members and friends (Daftary et al., 2007; Denison et al., 2008; McCoy et al., 2014; Mushaba and Hemalata, 2015; Mucheto et al., 2011). The considerable stigma attached to HIV/AIDS is reflected in the importance placed on confidentiality (Angotti et al., 2009). Worries about confidentiality and unwillingness to disclose HIV status to health care providers,

families and friends, mean that people who are HIV positive and those who perceive themselves to be at risk of HIV infection are much less likely to present themselves early for maternal health care services that are integrated with HIV services (Mucheto et al., 2011; Rutaremwa and Kabagenyi, 2016).

It has been a consistent finding that individual psychosocial factors such as knowledge of HIV transmission, perception of risk of HIV infection and how confidential HIV status will be maintained are key issues that influence women in seeking maternal health services in a facility (Asefa et al., 2013; Kevin et al., 2014; Orne-Gliemann et al, 2006; Rochat et al., 2006; Sama et al., 2017). Although a heightened risk perception of HIV infection may provide impetus to access HIV services (Oshi et al., 2007), there is also a disjunction between perceived risk of HIV infection and uptake of maternal health care services that involves HIV testing, as those who perceive themselves as being infected already do not see the value of knowing their HIV status during their pregnancies (Martin-Herz et al., 2006). On the other hand, those who are HIV positive and those who perceive themselves to be at risk of HIV infection and have knowledge of HIV transmission and PMTCT services may seek maternal health care services in order to prevent potential danger of HIV transmission to their infants (Turan et al., 2008). As Wiegert et al (2014) postulated, HIV services are often sought when there is clear risk in HIV infection or decline in an individual's health status which necessitates access to health care.

It has also been argued that interpretations of being at low risk of infection, sometimes because of trusting sexual partners may negatively affect uptake of reproductive health care services (Martin-Herz et al., 2006). That type of behaviour has been observed by Morrill and Noland (2006) who pointed out that proxy testing, for example, adopting the status of a sexual partner, is sometimes used as a marker for one's own HIV status. Kalichman and Simbayi (2003) found that people who felt they were at low risk of infection because they trusted their partners or because HIV was mainly perceived as a problem for sex workers were less likely to use maternal health care services than their counterparts.

While HIV factors may directly affect access to maternal health care services, Turan et al (2008) stated that maternal health care problems may manifest through existing root causes of HIV/AIDs and sexual behaviour, such as poverty, gender inequality, stigma and

discrimination, and marginalisation of vulnerable groups. Given this association, and to be able to identify factors associated with maternal health care (Rutaremwa and Kabagenyi, 2016) have argued for a need to understand how predisposing issues such as age, gender inequality, education, marital status, social and economic status interact with HIV factors like risk perception, risky sexual behaviour and HIV awareness to influence utilisation of HIV and maternal health integrated care services. These interactions may lead to an increased risk perception of HIV infection which may in turn decrease the willingness to use or to delay access to HIV and RH integrated services (Manzi et al., 2014; Tedrow et al., 2011; Turan et al., 2006). Studies have shown that underprivileged groups such as young, unmarried (never married, divorced and widowed) women and those who are poor and uneducated are the most vulnerable groups in protecting themselves against HIV infection as a result of engaging in risky sexual behaviour, often due to limited resources, peer pressure, lack of access to condoms or lack of knowledge of HIV (Asefa and Beyene, 2013; Makate and Makate, 2017; Magadi and Agwanda, 2007; Lepine et al., 2014). All of these behaviours may lead to an increased risk perception of HIV infection, leading to low use of maternal health care services.

There is also evidence of an association between HIV knowledge and utilisation of maternal health care services. It was argued that HIV knowledge may be negatively associated with maternal health care services such as HIV testing if knowledge affects risk perception and associated behaviours (Birhane et al., 2015; Byamugisha et al., 2010; Lepine et al., 2014, Sahlu et al., 2014; Weiser et al., 2006). Firstly, because women who have a good understanding of HIV prevention and transmission methods could be less likely to adopt risky behaviour and thus may perceive less need for using any maternal health care services that are linked to HIV as they are at less risk (Birhane et al., 2015). Secondly, pregnant women who overestimate their likelihood of getting infected through their poor health knowledge may have higher odds of using maternal health care thinking they are more at risk (Asefa and Beyene, 2013).

Although extant literature indicates that health care seeking behaviour is an individual responsibility, there is also a body of knowledge arguing that health seeking behaviour may also depend on the social and geographical contexts in which the individuals live (Bond et al., 2005; Chitalu et al., 2013; Fotso et al., 2009; Mumah, 2014). An example of this behaviour can be found in religious community groups, where issues of HIV are seen as a result of promiscuity and sinful acts (Hartwig et al., 2006). These negative community beliefs and norms

may lead to stigmatization and prejudice among those who want to get tested for HIV (Chirawu et al., 2008). In such circumstances researchers such as, Hendriksen et al (2009) have argued that behaviour is influenced by how a person thinks the community views his or her actions, for example, an HIV positive mother would think 'what will the community think of me if I do not breastfeed my baby?', as I am not supposed to breastfeed due to my HIV status. Within communities, gossip and rumours about mothers who are perceived to be HIV positive was found to be common in qualitative studies in South Africa (Sennott and Majola , 2017), Botswana (Dahl, 2012), Zambia (Hazemba et al., 2016) and elsewhere in SSA countries (Liddel et al., 2004). While HIV/AIDS stigma exists within an individual due to how an individual feels and the fear imagined due perceived risk of HIV infection, based on self assessment (Anyait et al., 2012), stigma can also be exerted from the outside world such as from family, community, and friends(Denison, et al, 2008; Gausset et al., 2011). Murire et al (2014) postulated that negative community beliefs that people living with HIV are wayward, coupled with HIV misconceptions and fears of being seen queuing for HIV services have kept away potential candidates for HIV and maternal health care services.

2.2.2 HIV and Family planning

Traditionally, family planning (FP) support for women has been mainly for prevention of unwanted pregnancy in order to reduce fertility and maternal mortality (UNAIDS, 2009) However, with high rates of unmet family planning needs and unplanned pregnancies reported among HIV positive women, HIV and FP services have been integrated in most countries with higher HIV prevalence rates (Haddad et al., 2015; Keogh et al., 2009). The rationale for integrating these services is to increase access to contraceptives, including condoms and long acting methods such as implants and intrauterine device (IUDS) in order to prevent HIV and unwanted pregnancy (Johnson et al., 2009; Keogh et al., 2009). One important thrust for HIV and Family planning integrated services is to encourage the use of condoms with or without other contraceptive methods for protection against unwanted pregnancy and STIs/HIV (Haddad et al., 2015). Upon visiting HIV and FP integrated services women are informed of the fact that non-barrier contraceptives are not effective against STI/HIV transmission and mother-to-child transmission (Agha, 2012; Bogale et al., 2010). Previous studies have highlighted the importance of condom use and long acting contraceptive methods in preventing HIV infection and unwanted pregnancy (Akinyemi et

al., 2013; Church et al., 2014; Haddad et al., 2015; Rosenberg et al., 2010). Extant literature indicates that there are several barriers that service providers face running HIV and FP integrated services, due to problems associated with condom use and long-acting contraceptive methods (Adamchak et al., 2007; Warren et al., 2012). Although condom and long-acting methods have been well promoted for primary prevention of HIV infection and unwanted pregnancies, some studies have indicated that women's choice to use or not use a particular contraceptive is strongly influenced by perceptions of method side effects (Heys et al., 2009), effectiveness (Morrison et al., 2011), availability (Kurewa et al., 2011); and whether or not it will be used for disease prevention of unwanted pregnancy (Mumah et al., 2014)

Another important barrier for HIV and FP integrated services is that, in Sub-Saharan African countries, the scale-up of HIV and FP services has largely focused on availability of contraceptives, and has not comprehensively addressed women's needs based on their HIV serostatus and fertility decisions (WHO, 2010). It was noted that, focusing on increasing access to contraceptives only, without consideration of specific contraceptive needs of women may lead to increased risk of HIV transmission to women, their sexual partners and children (Church et al., 2014). Although, HIV sero-status and fertility desires for women may be only minor considerations among the variety of factors that impact contraceptive use and choice within HIV and FP integrated services (Kurewa et al., 2011), extant literature indicates that these factors are also important determinants of contraceptive choice (Church et al., 2014; Habte and Namasusu. 2015; Johnson et al., 2009; Magadi. 2016; Mayondi et al., 2016; Mumah et al., 2014; Laher et al., 2009). Some scholars such as Snow et al(2012) and Turan et al., (2008) have claimed that women's knowledge of HIV sero-status and an elevated mortality and morbidity known to be attributable to HIV may cause behavioural change in contraceptive methods choice and fertility intentions (Church et al., 2014; Johnson et al., 2009). For example, if an HIV negative woman sees heightened mortality or morbidity that she can reasonably attribute to HIV, she may worry about her own contraction of the virus, and in that sense the threat is also to herself, sexual partners and of any future birth (Mayondi et al., 2016). She may think she may die before her children are old enough to care for themselves (Kakaire et al., 2010; Kaida et al., 2013); or that her ability to bear children may be impacted if she contracts the HIV virus (Habte and Namasasu, 2015).
Another body of scholarship claims that if a woman's community status relies on her fertility, she may perceive contraction of HIV as a threat to her standing if it affects her ability to have children (Snow et al., 2012; Turan et al., 2009). In South Africa, studies have documented societal expectations in relation to childbearing, specifically, pressures to have children, the need to have boys as heirs, and large families (Mantell et al., 2014; Rochat et al., 2006). Similarly, in Malawi, respondents noted the need to have a child in order to strengthen and maintain their relationships and even those who already have children may be forced to have more when they get a new sexual partner (Bisika, 2008). Literature reviewed indicates that, once the threat such as HIV-related mortality and risk of HIV infection has been perceived, it would need to be translated into change in sexual behaviour, such that women who know their HIV negative status and those who know the risks associated with risky sexual behaviours may be more likely to use condoms for HIV prevention, and other methods for FP (Haddad et al., 2015; Kurewa et al., 2011). Studies have found that in addition to condom use, reversible contraception and pill have been well tolerated among HIV negative women (Wagner and Wanyenze, 2013). Whereas several studies show that HIV negative women are more likely to want to have more children, literature also shows that a substantial proportion of women living with HIV desire to have children (Wekesa and Coast, 2014), and thus are less likely to use long-term contraceptive methods such as implants and sterilization than hormonal methods.

One behavioural model suggests that women respond to their HIV positive status by desiring more children, in order to increase the chance that more will live to adulthood (Wilcher et al., 2009; Yeatman et al., 2009) such that this group of women will choose reversible contraceptive methods because they want to increase their childbearing pace in order to achieve their unmet reproduction goals (McCoy et al., 2014; Mumah et al., 2014). Several studies have shown that after HIV diagnosis, the greater majority of women who had children already expressed a wish to become pregnant again and expressed a stronger wish to use reversible methods, despite the risk of HIV infection and mother to child HIV transmission (Nakayiwa et al., 2006; Wagner and Wanyeze , 2013).

Some studies have also noted that women may not want to have more children when they learn their HIV positive status (Homsy et al., 2009; McCoy et al., 2014), and they have better use of contraception as compared with their HIV negative counters (Johnson et al., 2009). Some

women fear that additional children would accelerate their disease progression (Curtis et al., 2009; Morrison et al., 2011), that they might transmit the virus to their babies (Hoffman et al., 2008), and that they may die and leave their children alone (Muheriwa et al., 2012). Other studies cited ill health as influencing the decision not to have children which may impact on their economic ability to support children (Haddad et al., 2015). However, Elul et al (2009) noted that this may change when they improve after getting into care or starting treatment and when they improve knowledge on HIV prevention methods.

Literature reviewed supports the claim that contraceptive choice among women who know their HIV status can also be altered if they access to information and ARV treatment (Church et al., 2014; Magadi and Magadi, 2016). A change in risk perception may occur through widespread media and education on HIV and its transmission and increased availability of antiretroviral therapy (Bertrand et al., 2006; Magadi, 2016). An increasing number of studies have also found that fertility intentions and contraceptive use are associated with available educational materials on the topic of HIV/AIDS and providers' attitude towards HIV positive women (Duflo et al., 2015; Blackie et al., 2016; Gustav et al., 2014). For example, women who are exposed to correct HIV and reproductive health messages were found to be more likely to use condom and long-term contraceptive methods in Zimbabwe (Mumah et al., 2014) and Kenya (Turan et al., 2008). In Malawi, providers were reported to have ambivalence about supporting childbearing among their clients with HIV and they lack specific knowledge about safer conception strategies, have a low level of efficacy in relation to PMTCT and the risks of pregnancy for HIV-infected women (Kawale et al., 2015).

As the movement towards integration of HIV and FP progresses, HIV sero-status of women must be considered to maximise the integration of HIV and contraceptive provision. Thus, HIV and FP providers should offer comprehensive support to meet the needs of both HIV positive and negative women who are sexually active and wish to prevent both HIV and unwanted pregnancy (Mayondi et al., 2016). However, this balance is not always achieved for various reasons including lack of knowledge of barriers associated with contraceptive choice after HIV diagnosis (Agadjanian et al., 2009). Thus, examining contraceptive method choice among women who know their HIV serostatus in order to prevent HIV and mother to child HIV transmission becomes at the fore in countries with high HIV prevalence rates such as Zimbabwe.

2.3 Theoretical Frameworks

In this section, a number of theoretical frameworks are considered. Due to the fact that most of the frameworks examined in this thesis have been barely considered from the point of view of the effect of HIV, a development of an overarching analytical framework that takes into account both HIV and the context in which people live has been proposed. The thesis recognises that some variables are very specific to parts of the study outcomes; and do not overlap, for example, some of the determinants of HIV testing and determinants of maternal health services are not the same and cannot crossover, or some of the variables are filtered out depending on the study outcome because of their incompleteness. To solve this problem the proposed framework is modified to suit the requirements of each of the respective chapters of the thesis.

Models that are generally used to explain health behavior are those that focus on individual's characteristics as determinants of services utilisation, thereby, reducing their capacity to explain broader determinants of service utilisation. These models include 'Behavioral Model of Health service use and Social influence model' (Iriyana, 2007), Construct Alone and Trans theoretical Models and AIDS Risk reduction Model (ARRM) developed by (Catania, 1990).

2.3.1 The Health Belief Model

The model posits that health behaviour is a function of an individual's socio-demographic characteristics, knowledge and attitude (Iriyana, 2007). According to this model a person must have the following beliefs in order to change behaviour.

- a) Perceived susceptibility to a particular health problem (e.g., am I at risk of HIV)?.
- b) Perceived seriousness of the condition (e.g., how serious is HIV/AIDS, how hard would be my life if I get it)?.
- c) Belief in effectiveness of the new behavior (e.g., HIV testing is effective in reducing mother to child HIV transmission)?.
- d) Cues to action (e.g., witness the death or illness of a close friend or a family member due to HIV)?.
- e) Perceived benefits of prevention action (e.g., if I deliver my child in a healthcare facility, can I avoid transmitting HIV infection to my child)?.

Case study for HBM:

'A woman has become aware that her husband has other girl friends (**perceived threat**). She knows that her auntie became HIV positive as a result of her husband also having multiple sexual partnerships (**perceived severity**). Her health provider tells her that abstaining from sexual intercourse or using condom will decrease the risk of contracting HIV (**perceived benefit**). Abstaining is difficult because they are married and she is worried about using condom (**perceived barrier**). On talking to people she hears a lady talk about her positive experience with condoms. She begins to use condom and finds success with this routine (**self-efficacy**)'.

In this model, promoting action to change behaviour includes changing individual personal beliefs, e.g through HIV awareness campaigns. Individuals will then weigh the benefits against the perceived costs and barriers to change. For a change to occur benefit must outweigh costs (Catania, 1990). The HBM provides insight on why people make health decisions and it creates a process of encouraging change. It is useful in understanding how to design programs such as health education and to design persuasive messages (Albarracin et al., 2005).

2.3.2 AIDS Risk Reduction Model (ARRM)

Closely related to Health behavioural Model is the AIDS Risk reduction Model (ARRM) developed by (Catania, 1990). The ARRM provides a framework for explaining and predicting the behaviour change efforts of individuals particularly in relationship to the sexual transmission of HIV/AIDS. The model incorporates several variables from other behaviour change theories, including the Health Behaviour Model, 'efficacy' theory, emotional influences, social cognitive theory and diffusion of innovation theories to describe the process individuals (or groups) pass through while changing behaviour regarding HIV risk. The model has three stages involved in reducing risk of HIV. The stages, as well as the hypothesised factors that influence the successful completion of each stage are as follows (Catania, 1990):

Stage 1: Recognition and labelling of one's behaviour as high risk

Hypothesised influences

• Knowledge of sexual activities associated with transmission of HIV

- Believing that one is personally susceptible to contraction of HIV
- Believing that having HIV is undesirable
- Social norms and networking

Stage 2: Making a commitment to reduce high-risk sexual contacts and to increase low-risk activities

Hypothesised influences

- Cost and benefits
- Enjoyment (e.g will the changes affect my enjoyment of sex?)
- Response efficacy (e.g, will the changes successfully reduce my risk of HIV infection?)
- Self efficacy
- Knowledge of health utility, as well as social factors (group norms and social support) are believed to influence individual's cost and benefit and self-efficacy beliefs.

Stage 3: This stage is broken down into three phases: 1) information seeking; 2) obtaining remedies; 3) enacting solutions

Hypothesised influences:

- Social networks and problem-solving choices (self-help, informal and formal help)
- Prior experiences with problems and solutions
- Level of self-esteem
- Resource requirements of acquiring help
- Ability to communicate verbally with sexual partner
- Sexual partner's beliefs and behaviours.

The authors of ARRM further identified other internal and external factors that may motivate individual movement across stages. For example, aversive emotional states, for instance, high levels of distress over HIV/AIDS that blunt emotional states, may facilitate or hinder the label of one's behaviours. External motivators including public campaigns, an image of a person dying from AIDs, or informal support groups may also cause people to examine and potentially change their health seeking behaviour or sexual activities.

The key problem with much of the frameworks discussed above is that they narrowly focus on health seeking decision making of the individuals, neglecting the environmental underpinnings of health and illness (Achia and Mageto, 2015; Bond et al., 2005). Based on the premises that individuals have some kind of interaction with their own communities (Stokols, 1996), Social Ecological Models of Health Promotion and Environment (SEM) are also reviewed for this thesis.

2.3.3 Social Ecological Models of Health Promotion and Structural and Environment

The Sociolecological Model (SEM) was developed under the tree of ecological theory where SEM contributes significantly in explaining the phenonmenon of community participation in various field of development (Stokol, 1996). SEM highlights a premise that 'individuals must have some kind of interaction with their own community and to understand and gain perspectives on factors that shape their behaviours, contextual factors must be considered' (Stokols, 1996). The SEM comprises of several immediate environmental contexts including individual, social, institution and policy (Stokols, 1996). The core principle of SEM as postulated by Stokols is that 'multiple factors influence behaviour, environments are multidimensional and complex, human-environment interactions can be described at varying levels of organisation and the interralationships between people and their environment are dynamic'(Stokols, 1996:282). Based on the SEM model, physical and socio-environmental conditions influence health seeking behaviour at multiple levels (i.e. at both the individual and the community level). This perspective echoes Diez-Roux's (1996) argument that environmental settings, physical, social and cultural dimensions can influence health outcomes. Another interesting framework of understanding how multilevel factors influence individual's health behavior is that of Dahlgren and Whitehead (1991) which demonstrates the relative importance of places (context) and people (composition) as predictors of health behaviour. Dahlgren and Whitehead pointed out that individuals are nested within multiple-spheres of influence and added that contextual factors can be conceptualised as operating at a higher level of the hierarchy, and features of individuals operating at a lower level. Their framework draws together characteristics from individuals' makeup to the socio-economic, cultural and physical environment, living and working conditions, community and social influences.

The social ecological model avoids exclusive focus on individual, community or geographical factors as separate analytical factors, but pays attention to the dynamic interaction of both individual and non-individual factors. This implies that factors influencing individuals' health-seeking behavior are located in the community where they live as well as within the social, economic and physical environments that surround them. Likewise some scholars such as Golden et al (2012) believe that the decision for an individual to use health care services, for example HIV testing or maternal health , must not be viewed as something that exclusively resides in the individual, but rather as a reflection of wider interactive situational processes. Generally the social ecological framework shows how joint factors can influence an individual to decide to go for HIV and sexual and reproductive services. Fig 2.1 provides a basic illustration of how factors relate to service utilisation interaction at different levels in a country.







2.3.4 The Behavioural Model of Health Service Use

This model is similar to SEM models in that it integrates contextual factors with individual characteristics for studying health-related behaviours (Golden et al., 2012). However, the main ideology behind the behavioural model of health service utilization is that the use of health care services is a function of environmental, predisposing, enabling or impeding and need factors

(Andersen, 1995). In this view, Andersen offered four sets of individual and contextual characteristics which influence individual's use of health care services. The importance of environmental factors such as regional infrastructure (i.e., hospitals, schools, roads etc); socioeconomic development (i,e., employment creation and poverty) is underscored in literature (Achia et al., 2015; Aremu et al., 2011; Angles et al., 2005; Bond et al., 2005; Magadi and Desta, 2011). Gourly et al (2016) have pointed out that community environment governs our behaviour, thus in the context of HIV and SRH services utilisation in Zimbabwe, HIV prevalence, stigma or awareness in the community also must govern people's SRH behaviour. Extended SEM has also suggested that contextual or environmental factors that may affect health care service utilization may also vary geographically (Babalola and Fatusi, 2009).

The second domain of Andersen's framework relates to predisposing factors. While there may be necessary pre-existing conditions in the area or community where people live that may either facilitate or impede health services utilisation, individual predisposing factors may also play an important role in health seeking behaviour. An individual's access to and use of health services is considered to be a function of socio-cultural characteristics of individuals that exist prior to their illness (Kabagenyi et al., 2016). In the literature social structure including (education, occupation, ethnicity, social networks, social interactions, and culture), health beliefs (i.e attitudes, values, and knowledge that people have concerning and towards the health care system); and demographic factors such as age, marital status, and gender are pre-existing conditions that are associated with health care service utilization (Andersen, 1995;Titaley et al., 2010).

The enabling factors in Andersen's framework represent resources that provide individuals with the means to use health care services. These resources must be available for use at any time they are needed, and can be found at the family and community levels. According to Aremu et al (2011) when the pre-existing conditions of an individual are good, the logistical aspects of obtaining care is the next stage that an individual has to consider. It has been generally acknowledged that factorssuch as income, health insurance, transport, extent and quality of care may impede or enable access to health services (Adamu, 2011; Ahmed et al., 2010; Alam et al., 2015; Golden et al., 2012). It has also been noted that community economic disadvantages such as unavailability of health services, poverty and area development may

also impede access to health care (Achia et al., 2015; Alam et al., 2015; Aremu et al., 2011; Chitalu et al., 2013; Makate and Makate, 2017).

Perceived risk/need based characteristics outlined in Andersen's Behavioural Model refer to health status or illness and its severity as perceived by the individual (Andersen 1995). Although, environment, predisposing and enabling factors contribute to the factors that influence health seeking behaviour, Andersen believes that they may not be enough for the actual use, until an individual sees the benefits of using the services. Based on the author's hypothesis, the perceived risk or need for health services depends on how people view their own general health and functional state, as well as how they experience symptoms of illness, pain, and worries about their health and whether or not they judge their problems to be of sufficient importance and magnitude to seek professional help (Andersen, 1995). This postulation is closely related to the Health Belief model which holds that people act if they perceive they are susceptible to a particular health problem, or if they believe in effectiveness of the new behaviour (Iriyana, 2007).

2.3.5 Limitation of the Theoretical Frameworks

In view of the impact of HIV/AIDs on health care services utilization, the reviewed health behaviour theories have some limitations. The social ecological theory incorporates two or more levels of influence, permitting an examination of health problems at both individual and community levels. But its limitation is that it places more emphasis on environmental factors. On the other hand, the behavioural models of health place more emphasis on individual analysis, not taking in account environments in which these individuals live. Notwithstanding their limitations, all models identified important factors that influence utilisation of health care services. This study argues that, although all the factors identified by these models may be necessary, there is a lack of a conceptual framework that unifies a number of seemingly disparate factors including the effects of HIV on health care service utilisation. Although the decision to seek health is usually influenced by one's socio-economic status and illness characteristics, factors to do with HIV such as awareness and HIV prevalence may also influence service utilisation. In the literature, studies of SRH have not given much consideration to the independent impact of HIV/AIDs on service utilization. In view of the shortcomings of the frameworks reviewed, the following analytical framework is proposed for studying determinants of HIV and sexual and reproductive health care in Zimbabwe.

2.4 General analytical framework

This study advances the theoretical frameworks discussed above in yet another important way by including HIV factors within the determinants of SRH service utilization. Specifically, the conceptual framework incorporates HIV as either an enabling or disabling factor for health care services utilization. The lack of frameworks that emphasise structural and environmental factors, for example the social implication of HIV/AIDS in the community, has resulted in the conceptualisation of a conceptual framework (fig 2.2) that merges Health behavioural, AIDS Risk reduction Model, Social ecological and Heath Belief Models. Whereas, there maybe empical associations between public policy and external factors with health seeking behaviour, these are not addressed in the current study due to data limitations, hence represented with dashed arrows in figure 2.2. However, as can be seen in fig 2.2, predisposing factors (individual and community level) mediate the enabling factors such as access to media, HIV awareness and financial wherewithal to enable access to health care services.

The framework recognises that community variables including, place of residence (urban and rural), community mass media exposure, community poverty, HIV prevalence and HIV stigma have independent effects on health care service utilisation. The community characteristics may also influence the enabling factors. For example, individuals resident in communities with a higher employment rate are more likely to be rich, enabling them to have easier access to financial resources, a major factor that is known to be a key determinant of health services utilisation (Ahmed et al., 2010; Bond et al., 2005) such as access to financial resources and media. Given the paucity of knowledge on how HIV may facililates or impedes service utilisation in Zimbabwe, the conceptual framework has also included HIV awareness and HIV related stigma. HIV awareness has been identified as a key determinant service utilisation, for example individuals with higher HIV awareness were found to be more likely to use a condom (Agha, 2002), initiate antenatal care early (Jayleen et al., 2016); and test for HIV (Godif et al., 2015). Recognising the importance of The Health Belief model (Iriyana, 2007) and the AIDS Risk reduction Model (ARRM) developed by Catania (1990) discussed in section 2.3.1 and 2.3.2, respectively, perceived risk/need factors are also included in the conceptual framework. Based on the above-mentioned models, an individual must have a set of beliefs in order to change behaviour. For example, if individuals believe that they are at risk of HIV infection, they will take action in order to prevention HIV infection (Iriya, 2007). On the other hand,

Catania (1990) suggested that high levels of distress over HIV/AIDS that blunt emotional states may facilitate or hinder the labelling of one's behaviours.

The developed framework enabled this thesis to measure the relative contribution of each of the relevant cluster of variables (i.e individual and community level predisposing, enabling and perceived/need factor) including HIV, that influence access to HIV and SRH integrated services. Based on the literature reviewed, I believe that integrating the Health beliefs (HBM), Socio-ecological (SEM), AIDS Risk Reduction Model (ARRM) and Behavioural model provides a good conceptual model to investigate the complexity of people's health seeking behaviour. It has been suggested that, an interaction with specific context will influence someone's decision either to accept or decline to use health services (Catania, 1990). Therefore, inclusion of SEM perspectives in the final analytical model of the thesis enhances our understanding of every possible context in the system and it helped in giving substantial information towards factors that influence HIV and SRH services utilisation. Figure 2.2 below depicts the overarching conceptual framework for this thesis which will be modified to suit relevant subsequent chapters.

Figure 2-2: Analytical framework for the study of HIV and SRH service utilisation



Source: Author's Construct

3 CHAPTER THREE: DATA AND METHODS

This chapter provides a general description of study population, design, data and the analytical methods used in the study. More details about the description and methods for specific outcomes are provided in each relevant chapter. The chapter is divided into three sections. The first section covers general description of study design, data sources and sampling. The second section, introduces study variables (i.e., outcome variables, covariates and HIV variables) as well as an overview of the limitations of data sources on HIV and SRH care services in Zimbabwe. The final section describes the basic statistical methods used i.e multilevel logit models and multilevel multinomial models, as well as modelling specifications and parameter estimations.

3.1 Study area and population

Zimbabwe is a land locked country in Southern Africa that shares borders with Zambia, Mozambique, Botswana and South Africa.By 2015, the country had an estimated population of 13 million people of which 48% are males and 52% being females (ZIMSTA, 2015). The total fertility rate is 3.1 per women and life expectancy at birth is currently estimated at 58 years of age (ZIMSTA, 2015). According to ZNASP (2015-2018), Zimbabwe has one of the highest HIV prevalence at 15.2% among adults aged 15-49 years of age (17.7% among females and 12.3% males). The majority of Zimbabweans have attained some education. Overall, 94% of males and 91% females have ever attended school. About 68% of the population live in rural areas (ZDHS, 2015), and close to 66% of the population live below the poverty line (ZIMSTS, 2012). Zimbabwe has experienced many economic and political problems in recent years and the unemployment rate is estimated to be close to 90% (Duri et al., 2013).



Figure 3-1: Study area and population

3.2 Study Design

The thesis is based on secondary analysis of cross-sectional data from Zimbabwe Demographic and Health Surveys (ZDHS) conducted in 2005/06 and 2010/11. The datasets were as a result of household based surveys, which in Zimbabwe, have been running every five years since 1988. All eligible women aged 15-49 and men aged 15-54 years resident in each sampled household were included in the surveys. There are three types of data that are collected in the Demography and Health surveys and these are Household, Women's and Men's Questionnaires. These questionnaires solicit information on a range of demographic and health indicators including, history of HIV testing, use of family planning and maternal health services. The 2005/6 and 2010/11 ZDHS, were the first two of a series of on-going surveys to include HIV testing. The repeated nature of DHS surveys, along with the availability of HIV test data that can be linked to individual-level survey data, provides a unique opportunity for a population-based study of factors associated with the HIV/AIDS epidemic in Zimbabwe. The DHS HIV testing protocol undergoes a rigorous ethical review process (ICF Macro, 2010), providing for informed, anonymous, and voluntary testing of women and men of reproductive age. The protocol allowed for the linking of the HIV results to the background socioeconomic and demographic data collected in the individual questionnaire, ensuring that any information that could potentially identify an individual was destroyed before linking the 2005/6 and 2010/ 11 ZDHS datasets.

3.3 Sampling Design

The sampling framework used for the 2005-06 and 2010-11 ZDHS was based on the updated 2002 Population Census framework (ZDHS, 2005/6-10/11). Both samples were selected in two stages. The first involved selecting clusters from a nationally representative sample framework. In the second stage, all adults aged 15-49 (women) and 15-59 (men) in the selected households were targeted for household interviews. This resulted in a total of 18078 eligible women aged 15-59 years being interviewed for the two surveys; 8 907 in 2005-06 and 9171 in 2010-11. For men, 7175 were interviewed in 2005-06 and 7480 in the 2010-11. The response rate for individual surveys was 96 percent on average. The data from DHS are in hierarchical form, and Table 2 presents the hierarchical nature of the two ZDHS datasets that were used in the study. The table shows that at the lower level (level 1) are individuals who are nested within units at a higher level (household: level 2) and households are again nested within units at the next level (cluster: level 3). In this study, clusters are described as communities and they are the primary sampling units from which households are selected. Details of the sampling design and data collection procedures for the surveys are available in the individual 2010 DHS reports.

Survey Year	2005-6 ZDHS		2010-11 SDHS	
Clusters	403		406	
Households	9778		9756	
Individuals	Men (n=7175)	Women (n=8907)	Men (n=7480)	Women (n=9171)

Table 3-1: Samples showing ZDHS Hierarchical pooled datasets

Source: Zimbabwe Demography and Health surveys (2005/6-2010/11)

3.3.1 Ethical consideration

Secondary data, mainly the 2005/6 and 2010/11 ZDHS, was used for the data analysis. The DHS had already taken ethical issues into consideration such as the assurance of voluntary participation of respondents, anonymity and confidentiality. Permission for the use of the data in this thesis was obtained from ICF Macro Inc., USA.

3.3.2 Data limitations

The data has advantages and limitations, which may have affected the study results. The main advantages and disadvantages are discussed below. Application of ZDHS data offers several advantages. Firstly, the data included respondent's HIV-test results, which offers a unique opportunity for exploring the association between HIV status and sexual and reproductive health behaviour (Magadi and Magadi, 2017). Before the introduction of voluntary HIV testing during the ZDHS (2005/6) survey, there was a paucity of information about HIV-serostatus available in Zimbabwe to allow researchers to investigate the effect of HIV prevalence and knowledge of HIV status on health care service utilization. Linkage of HIV testing results to DHS surveys was designed to rectify this lack of knowledge (Magadi, 2016).

The data has some limitations too, that readers and policy makers need to consider in this thesis. Both 2005-06 and 2010-11 ZDHS data were collected retrospectively. This method of data collection may have led to recall bias, given that the events in question took place five years preceding each of these surveys (Titaley et al., 2010). Uthman et al (2015) noted that people remember fewer details as time passes.Accordingly, during the interviews, individuals may have forgotten or may have not accurately recalled events, for example, when they first visited antenatal services. They may also have forgotten events such as their date of birth, number of sexual partners or which facility they attended for skilled delivery (Aremu et al., 2011). Furthermore, because of lower level of education in rural areas there may have been misunderstanding of some questions and concepts leading to biased response and misreporting of some concepts (VU, 2005).

Variables such as asset-based wealth index used in the surveys were only proxy indicators, hence it may not produce accurate results similar to those obtained from direct measures of income in other datasets where such data are reliably collected (Aremu et al.,2011). In addition, the surveys did not capture direct questions for HIV stigma, awareness and HIV perceived risk of infection. To get a clear picture of the level of HIV stigma, awareness and perceived risk it would have been better for the survey to ask direct questions such as in the 2007 Malawi Demographic and health surveys in order to minimize information and estimation measurement bias during the imputation of these variables (Sambisa, 2008). Another limitation was the inability to control for the availability and quality of health care services. For example, the

women's questionnaire has exposure questions for maternal health care and Family planning services, but lack useful questions about availability and quality of these services. Previous studies have shown that the presence of quality health facilities and availability of drugs have great impact on utilization of health care services (Gabrysch et al., 2011; Hope et al., 2014). It is therefore possible that the absence of data on the availability of health care services (i.e mainly number of health facilities) and quality of health care services (i.e availability of health care providers and drugs) that are provided in different community settings may have effects on the variations of health care service utilization. In order to achieve a clearer picture of contextual influences on health care service utilization, future surveys should include measures of presence of health care infrastructure, availability of HIV treatment and other HIV intervention programs.

Although it would have been desirable to incorparate geographical data on public policy i.e national and regional policy, this has been hindered by lack of such data in the DHS. This limits the extent to which differences in HIV and SRH service utilisatiion across the regions would be attributed to the characteristics of the region themselves or differences between the types of individuals living in different regions in Zimbabwe. Oseni et al (2018) found striking variations in health service utilisation acorss different provinces in Zimbabwe and have suggested that response/policies should vary at the regional level to ensure that the often diverse needs of populations acrocc the country are met and incorporated into planning the HIV and SRH response. Future studies on HIV and SRH integrateed services need to provide insights into spatial and regional variation in HIV and SRH service utilisation.

On the question on HIV testing in various regions, there were reasonably high response rates and no clear systematic patterns that can cause bias. However, we must be cautious when interpreting the results for specific subgroups (for example condom users and those with multiple sexual partners that have some degree of non-response rates in the surveys). This may have the effect of under-representing condom users and those who have multiple sexual partners.

Another interesting bias is that poorer HIV positive individuals die quicker than those who are better off because they lack financial wherewithal needed for health care services and treatment (MacPhail et al., 2007). As a result, HIV positive respondents may have over- represented subgroups of population who are better off economically (Magadi and Desta, 2011). Given that the surveys excluded people who lived in army barracks, hospitals, police camps, boarding schools and other institutional arrangements, there could have been some under sampling of employed people who are generally knowledgeable about health issues and healthier than unemployed people (Carneiro and Howard, 2011). Fourthly, there are a number of variables relating to health facilities such as distance, staffing, facilities and equipment and cultural beliefs and practices that could confound the results obtained in the analysis as the data set used did not collect them. Another important limitation is that because of the cross-sectional nature of the data, the relationship between outcomes and community factors could work in the other direction where constraints of health care supply maybe due to lack of incentive of providers to enter deprived areas rather than the constraints of demand driven by for example, community wealth, HIV awareness or stigma (Delprato, 2016). Indeed, strong arguments for causality effects on determinants of health care service utilisation would require a panel data structure. Notwithstanding the limitations above, I believe that since the questionnaire used has already been validated internationally, limitations discussed above have no significant bearing on the results of the study outcomes.

3.3.3 Missing data

The 'exclude case pairwise' option was used to exclude cases only if they had truly missing data that was required for specific analysis. However, the cases were included in the analysis for which they had relevant information.

3.4 Introduction to multilevel modelling analytical methods

The thesis adopts a multilevel modelling analytical method. For several years, great efforts have been devoted to the study of public health focussing on relationships between characteristics of community where people live and health outcomes. Based on sociological theories and literature, there seems to be an agreement that individuals' health and behaviour is shaped not only by individual risk factors but also by the structure of the social environment in which they live, creating widespread acceptance that health varies across places and communities (Ngome and Odimegwu, 2014; Magadi, 2011; Uthman, 2008). In the history of public health and places, the focus on community variation in health outcomes has always been there. Fundamentally, in the mid 1940s, public health and epidemiological investigations of diseases were ecological (Diez-Roux, 1998, 2002) being interested in the association between health outcomes with environmental and community characteristics. A chief example of ecological investigation is that of John Snow's study of cholera in London in which he concluded that the geographical setting was the key to spread of cholera (Diez-Roux, 1998).

In contrast with ecological investigations, modern research on health outcomes has also focused more on individual level factors rather than environmental factors (Achi and Matero, 2015; Angles et al., 2005; Aremu et al., 2011; Bond et al., 2005; Magadi and Desta, 2011). One reason for the shift is the increased prominence of chronic diseases, with research focusing mainly on behaviour and biological factors responsible for chronic diseases (Leyland et al., 2003). Another reason for the shift concerns the 'ecological fallacy' which occurs when the associations found at the group level are inferred to the individual level when in truth, such association may not be existing (Golden et al., 2012). Considering the 'ecological fallacy, there has been a renewed interest in investigating the effects of context on health outcomes, which seeks to bring public health research back towards structural and environmental influence on health outcomes (Diez-Roux, 2002). Many experts have always been content that health outcomes may be affected by contextual effects associated with community characteristics (e.g Angles et al., 2005; Aremu et al., 2011; Bond et al., 2005; Magadi and desta, 2011). Furthermore, it has been suggested that variations in health outcomes may be because of compositional effects whereby, for example, particular types of individuals, who are more

ssusceptible to poor health outcomes due to their individual characteristics, are clustered in the same community (Goldstein and Rasbash, 1996).

Instinctively, individuals within the same area tend to be more similar in health status than individuals from different areas (Aremu, 2011). It has been argued that the clustering of individuals within communities may lead to a correlation of health outcomes for individuals within the same community, which demonstrates the shared experiences of individuals in the same area (Delprato, 2016). This correlation may lead to violation of the assumption of independence required for regression techniques, which in turn leads to underestimation of standard errors (Magadi et al., 2007). In addition, the finding of differences and relationships when they do not exist may be more likely to happen (Jones, 1991). Most statistical methods assume that observations in the dataset are independent of each other, but when groups of observations share some common features or have information at different levels (e.g., individual and community); they are no longer independent of each other (Vu. 2005).

Data that fall into hierarchies can be analysed using multilevel models, which account for the dependence of outcomes of individuals within the same community (Chitalu et al., 2013). The DHS dataset has a hierarchical structure with individuals living in households and households clustering within communities, thus such kind of data violates the assumption of independence of ordinary (linear) or logistic regression models (Diez-Roux, 2002). In this thesis, a multilevel modelling technique is used to account for the hierarchical nature of the data, and to be able to estimate community level effects on the outcome variables. In addition, the technique was used to remove the effect of clustering in order to obtain valid point estimates for the parameters and standard errors for point estimates: and also, to study the effects of variables that operate at different levels of the hierarchy and how the variability in the outcome variable is distributed across the levels of the hierarchy (Vu, 2005). Multilevel models employed in this thesis allowed the total variation in the response, which is measured at the individual level, to be partitioned into variation attributable to the individual level and variation that is attributable to differences between communities (Goldstein and Rasbash, 1996). According to VU (2005), the advantage of this model is that the researcher simultaneously estimates the individual and community level effects on the outcome variable. It has been argued that not only do multilevel models overcome the ecological fallacy, but they also overcome the individualistic fallacy

when the association found between, for example maternal health outcomes and individual characteristics are inferred to the community level, when it is not true (Golden et al., 2012).

3.4.1 Multilevel Logistic Model

A two-level logistic regression model for binary data and a multinomial logistic regression model for data with unordered responses were employed in the study. These consist of two sub-models at level 1, and level 2, with individuals (level 1) nested within communities (level 2). In this study, individual and household variables were considered as individual level variables. The reason for this was that the average number of individuals in a household as contained in the dataset was small, thus the household could not be considered as a separate level of analysis. Here we are interested in identifying the factors associated with HIV testing, antenatal HIV testing, timing and number of ANC visits; place of delivery and delivery with a professional birth attendant as outcomes. The analysis further examined the extent of between community variations in care service use of these outcomes.

Single-level logistic regression

A general single-level model for binary (1, 0) response data can be represented as follows:

 $f(\pi_i) = \beta_0 + \beta_{1i} x_{1i} + \ldots + \beta_{m i} x_{m i}$

Where f (π_i) is some transformation of the probability π_i , namely the *link function* For the logit link,

 $f(\pi_i) = \text{Log}\left[\pi_i/(1-\pi_i)\right]$

where the quantity $\pi_i/(1-\pi_i)$ is the odds of event occurring for the individual, i.We use the logit link $f(\pi_i)$, which is a function that models the odds that an individual uses HIV and SRH outcomes in Zimbabwe.

B₀ is the constant; $\beta_{1 \text{ to}} \beta_{m}$ represent coefficients of the explanatory variables x_{1} to x_{m} for individuals as subscripted with *i*. In this case, x_{1} to x_{m} are variables such as individual's age, sex, marital status etc.

The individual-level model may be extended to a two-level random intercept model, for individual 1 within the community j. This allows for inclusion of community effects on the odds of using health care services.

Logit $(\pi_{ij}) = \beta_{0j} + \beta_1 x_{ij}$ $\beta_{0j} = \beta_0 + u_{0j}, \quad u_{0j} \sim N(0, \sigma^2_{u 0})$

Here the intercept consists of two terms: a fixed component β_0 and a community-specific component, the random effect u_{0j} . The random effects represent the variation of HIV and SRH service utilization for individuals from different communities. *Xij* is the matrix of fixed (observed) covariates corresponding to ith level-1 unit in the jth level-2 unit. Given that the study has two levels, where level 1 is individual and household, level 2 is community.

3.4.2 Multilevel Multinomial logistic regression Model

Chapter 7 adopted multinomial regression model for analysis of contraceptive methods choice among women who know their HIV status. The multinomial model can be built in stages as follows:

Stage 1-Unordered multinomial model

Using a logit link, the unordered single-level multinomial model for contraceptive methods choice (categorical response variable with four categories) can be written succinctly as follows:

$$\log\left[\frac{\pi_{i}^{(s)}}{\pi_{i}^{(t)}}\right] = \beta_{0}^{(s)} + \beta_{1}^{(s)}x_{i} + \cdots, s = 1, \dots t - 1$$

Where, s superscript (s=1, ..., t-1) denotes each response with response specific intercepts and slopes. In the model, the same explanatory variable appears in each of the t-1 contrasts. From the above equation, the probability for choosing a particular contraceptive methods $\pi_i^{(s)}$ (s=1,..., t-1) can be obtained as follows:

$$\pi_{i}^{(s)} = \frac{\exp(\beta_{0}^{(s)} + \beta_{1}^{(s)}x_{i})}{1 + \sum_{k=1}^{t-1}\exp(\beta_{0}^{(k)} + \beta_{1}^{(k)}x_{i})}$$

The probability of being in the reference category $_{t}$ is the remainder from 1 minus the summed probability of all other categories:

$$\pi_i^{(t)} = 1 - \sum_{k=1}^{t-1} \pi_i^{(k)}$$

Stage 2: a model with hierarchical structure for women nested in communities

The second model sees individual women as being nested within communities. The general equation of the random intercepts two-level multinomial Logistic regression model used for analysis of determinants of contraceptive methods choice therefore takes the form:

$$log\left[\frac{\pi_{ij}^{(s)}}{\pi_{ij}^{(t)}}\right] = \beta_0^{(s)} + \beta_1^{(s)} X_{1ij} + \beta_2^{(s)} X_{2ij} + \ldots + \beta_k^{(s)} X_{kij} + u_j^{(s)}, \ s=2,3.$$

where: $\pi_{ij}^{(s)}$ denotes the probability of using a particular contraceptive method (s) (i.e. long-term methods=2, condom=3 or other=4) for woman,*i*, in the jth community ; $\pi_{ij}^{(t)}$ is the probability of using hormonal methods (t=1 for hormonal methods used as reference category) for woman *i*, in community j; $\beta_0^{(s)}$ are the regression intercepts/constant for contraceptive methods s; $X_{(1-k)ij}$ are 1-k explanatory variables defined at woman or community level; $\beta_{(1-k)}^{(s)}$ are the associated usual regression parameter estimates for contraceptive methods choice (s); and $u_j^{(s)}$ are the community-level residuals for contraceptive methods choice (s). These are assumed to be normally distributed with mean zero and variance $\sigma^{2(s)}_{u}$. The community random effects may be correlated across contraceptive methods choice categories: covariance ($u_j^{(s2)}, u_j^{(s3)}$) = $\sigma^{(s2, s3)}_{u}$, (s2=long-term, s3=condom; and s4=other methods). (Rasbash et al, 2016).

3.5 Fixed and Random Effects

The models were estimated initially by IGLS and then by Markov chain Monte Carlo (MCMC) estimation with an initial burn-in of 500 simulations followed by 5k simulations. Fixed effects are the level-1 and level-2 covariates, expressed as odd ratio (OR) for binary outcomes and as relative risk (RR) for the multinomial outcome, both at 95% confident interval (95% CI). The random effects are the measures of variances in health services utilization and contraceptive methods choice across communities (level-2). The ratio of the variance at the community level to the total variance is called inter-class correlation coefficient (ICC), which is a tool of measuring the degree of homogeneity within units such as households or communities. According to Field (2009) the ICC measurements are used to gauge whether contextual variables have any effect on the outcome. The results of random effects are expressed as variance partition coefficient (VPC), which is in this study are equal to ICC. The models were

estimated in MLwin 2.32 (Rasbaschet al.2015); by bayesian methods using Markov chain Monte Carlo (MCMC) estimation (Browne. 2014).

3.5.1 Inter –class correlations

For binary outcome, the partition of variance between different levels is not interpreted the same way as in the case of a linear model. Therefore the, VPC analysis will be calculated based on the linear threshold model which converts the individual level variance from the probability scale to the logistic scale, on which the community variance will be expressed (Chitalu et al, 2013). This means that by using the linear threshold model, the unobserved individual outcome variance follows the logistic distribution with individual level variance σ_e^2 equal to $\pi^2/3$ (equal to 3.29). In this case, the VPC corresponds to the ICC, which is a measure of general clustering of the individual outcome of interest in the community.

The ICC is calculated as follows:

 $P=\sigma_{U}^{2}/(\sigma_{U}^{2}+\sigma_{\epsilon}^{2})$ Where:

P is the intra-class correlation coefficient (*ICC*). σ_U^2 in the variance at the community level. $\sigma_{\varepsilon}^2 = 3.29$ represents the fixed individual variance(Goldstein, 2003).

3.5.2 Proportional change in Variance (PCV)

Community differences in health service utilisation may be attributed to contextual influences or differences in the individual composition of the communities (including unobserved community characteristics) (Field. 2009). Thus, while adjusting for the individual characteristics in the multilevel models, some part of the compositional differences will be taken into account to explain some differences in communities as observed in the empty model. The equation for the proportional change in the community variance is:

 $PCV_1 = (V_{n-1} - V_{n-2})/V_{n-1}$

Where V_{n-1} is the community variance in the empty model and V_{n-2} is the community variance in the models including individual characteristics or community characteristics.

3.5.3 Parameter estimation

Before fitting the models, we have to specify details about estimation procedure to be used. For both binary and multinomial response models, different procedures have been implemented in *MLwiN* for the estimation of multilevel models that have categorical responses: quasi-likelihood methods (MQL) (Goldstein, 1991; Goldstein and Rasbash, 1996) and Penalized Quasi Likelihood (PQL) (Breslow and Clayton, 1993, Laird, 1978), 1st or 2nd order) For most of the analysis I shall used the quasi-likelihood methods, starting with the default procedure 1st order MQL and extending to 2nd order PQL on convergence. Both orders can be used; however, 2nd order is preferred because it is an improved approximation procedure (Rasbash et al., 2005). For the analysis the quasi-likelihood methods was used, starting with the default procedure 1st order quasi-likelihood methods (MQL) (Goldstein, 1991; Goldstein and Rasbash, 1996) extending to Penalized Quasi Likelihood (PQL) (Breslow and Clayton, 1993, Laird, 1978), 1st or 2nd order on conversion.

3.6 Summary

Chapter three introduced methods and data to be used, and this was followed by analytical methods including binary and multinomial logistic regression models. Chapter four, five, six and seven began the analysis of data and chapter seven and eight presented findings and conclusion of the study.

4 CHAPTER FOUR: FACTORS ASSOCIATED WITH HIV TESTING IN ZIMBABWE

4.1 Introduction

4.1.1 Background context

A national HIV testing program was implemented a decade ago in Zimbabwe, providing HIV treatment to millions` of HIV-positive individuals, many of whom previously struggled with the illness due to unknown HIV status. By 2015, Zimbabwe sought to half new HIV infection prescribed by the World Health Organisation's strategic plan (WHO, 2016). Although it has been a daunting task, adult HIV prevalence in Zimbabwe has almost halved from its peak of nearly 30% around 1997 to around 13.7% in 2011 (ZIMSTAT, 2012). HIV related deaths have also been reduced by over 60% as a result of a successful HIV testing, treatment and support programme (ZIMSTA, 2012). However, despite government's efforts to stem the HIV scourge, Zimbabwe remains one of the countries in Sub-Saharan African still burdened with HIV epidemic (WHO, UNICEF, 2014). The current aim in the post-2015 era is to end the AIDS epidemic by 2030. To achieve this, the government, in collaboration with its partners have set up the 90-90-90 strategic milestones to be achieved by 2020. The milestones means that by 2020, 90% of all people living with HIV will know their HIV status, 90% of all people with diagnosed HIV infection will receive sustained antiretroviral therapy and 90% of all people receiving antiretroviral therapy will have viral load suppression (UNAIDS 2014). HIV testing has been used as an entry point for both HIV prevention and treatment; and an early detection of HIV can add 15 years to a person's life span (Wringe et al., 2008). People have to get tested and know their HIV status to enable them to seek treatment and to choose preventative strategies. For this reason HIV testing services need to be available and accessible to all people to enable easy utilisation (Kayeyi et al., 2014).

To ensure targets are met, Zimbabwe needs to improve HIV testing coverage, but scaling up HIV testing coverage; currently estimated at 56% can be daunting, given that HIV testing is voluntary. The success of the HIV self-testing programme will require a better understanding of the factors that influence people's HIV testing behaviour, which this study seeks to achieve.

4.1.2 Factors associated with HIV testing

Existing studies have identified a range of factors to be associated with HIV testing, including: place of residence, gender, marital status, and socio-economic status (Mitchel at al., 2010; Stephenson et al., 2013). Extant literature from Sub-Saharan African countries indicates that gender has been an important determinant of HIV testing behaviour. Specifically the studies found that females were more likely to get tested for HIV than males (Godif et al., 2015; Mwangi et al., 2014), due to reasons such as frequent access to health care services through maternal care (Peltzer et al., 2009).

Besides gender, marital status has been found to be strongly associated with HIV testing and findings have been consistent across gender. Individuals who are divorced, widowed and married tend to have higher odds of HIV testing than never married individuals (Lepine et al et al., 2014; Obermeyer et al., 2013; Weiser et al., 2006). For example, married individuals were found to be 2.54 times more likely to be tested for HIV than never married individuals in South Africa (Venkatesh et al., 2011). Similarly, a study in Ivory Coast (Jean et al., 2013) found that never married individuals were 66% less likely to be tested than married people.

HIV testing is also influenced by enabling factors such as wealth status, education, HIV awareness and media exposure. Several studies have found that poorer individuals were less likely to test for HIV than those from wealthy households (Mwangi et al., 2014; Weiser et al., 2006; Jean et al., 2013). It has been noted that poverty is likely to be associated with lack of education, and lack of education implies that messages regarding HIV testing are often inaccessible (Hendriksen et al., 2009; Mitchel at al., 2010). Education is a key indicator of socioeconomic status and more educated individuals are more likely to have better healthdue to their ability to have better access to health information and to understand and respond to such information. Socioeconomic challenges may also decrease the likelihood of HIV testing, due to lack of financial resources that would enable individuals to have access to health care services (Bond et al., 2005; Musheke et al., 2013; Obermeyer et al., 2013; Staveteig et al., 2013).

The effect of HIV/AIDS factors (e.g stigma and discrimination, risk perception of HIV infection, risky behaviour, HIV awareness, or knowing someone who died of HIV/AIDS) on

HIV testing may depend on an individual's predisposing factors such as age, education, marital status, social and economic status (Murire et al., 2014; Jacobi et al., 2013; Wringe et al., 2008). These interactions may lead to an increased risk perception of HIV infection which may in turn decrease the willingness to utilise health care services that are linked to HIV (Chirawu et al., 2010; Oshi et al., 2007). The risk perception of contracting HIV may either increases or decreases the likelihood of using health care services (Akwara et al., 2003; Kayiki and Forster, 2011; Muchini et al., 2010). An example of this phenomenon could be that, since HIV prevalence is reported to be positively associated with higher socio-economic status and education (Magadi and Desta, 2011) individuals with higher educational attainment and those who are wealthier may perceive themselves to be at risk of HIV infection, and therefore have recourse for HIV testing, compared to those who are less well-off (Jean et al., 2012; Obermeye et al., 2013).

There is also evidence of a positive association between HIV awareness and HIV testing (Weiser et al., 2006) and awareness and education are highly correlated (Sambisa.2008). However, Lepine et al (2014) have argued that HIV awareness may be negatively associated with HIV testing if the awareness affects risk perception and associated behaviours. Firstly, because individuals who have a good understanding of HIV prevention and transmission methods could be less likely to adopt risky behaviour and thus may perceive themselves to have less need for HIV testing as they think they are less at risk (Izugbara et al., 2009). Secondly, people who overestimate their likelihood of getting infected through their poor health knowledge could have higher odds of HIV testing thinking they are more at risk. This notion resonates with Musheke et al (2013) who assert that the effect of HIV knowledge on HIV testing may be heterogeneous in the population. For instance education may determine access and exposure to HIV information and will affect the way this information is used to modify attitudes towards HIV testing (Mwangi et al., 2014).

Also, many studies have linked HIV infection with risky sexual behaviours, including multiple sexual partners and limited ability to negotiate safer sex, which may lead to lower rates of condom use and disclosure of HIV status (Agha, 2012; Akinyemi et al., 2010; Rosenberg et al., 2013). All of these may lead to an increased risk perception of HIV infection, leading to lower or higher uptake of HIV testing. In line with the effects of risk perception on healthcare utilisation, studies have shown that perceived susceptibility as a result of risky sexual behaviour

exacerbates risk perception of HIV infection, in turn leading to underutilisation of health care services (Kayiki and Forster, 2011; Martin-Herzet al., 2006). On the other hand, existing literature also suggests that perception of being at risk of a disease is a prerequisite for behaviour change (Turan et al., 2008), and that perceived high risk of infection causes people to take precautionary measures including HIV testing and knowing the results in order to reduce the risk of getting the virus (Lepine et al., 2014; Jean et al., 2012). According to Health Behavioural models (Iriya et al., 2007) perceived susceptibility to a particular health problem (e.g., am I at risk of HIV?), perceived seriousness of the condition (e.g., witnessing the death or illness of a close friend or a family member due to HIV) may move an individual to take necessary action or change their behaviour (Iriyana et al., 2007). Therefore, knowing someone who died because of HIV is also a significant cue to action to an individual to test for HIV (Meiberg et al., 2008).

At the community level, the likelihood of HIV testing may increase for individuals who reside in communities with higher HIV prevalence, high poverty, high risk perception of HIV infection, or within easy access of HIV intervention programmes, compared to individuals living in other communities (Bond et al., 2005; Mitchel et al., 2010; Murire et al., 2014; Uthman et al., 2010). While HIV/AIDS stigma may exist within an individual as a result of imagined fear due to perceived risk of HIV infection based on self-assessment (Nyblade et al., 2001), stigma can also be observed and exerted from the outside world (health care providers, family, community, and friends). Either form of stigma may lead to more people underutilising health care services (Sambisa, 2008).

Since risky sexual behaviour may be linked to poverty in the community, people living in poor communities may perceive themselves to be at a high risk of HIV infection, and this may reduce the likelihood of uptake of HIV testing services in such communities (Hendriksen et al., 2009). Furthermore, availability and quality of health care services, for example availability of family planning and HIV testing services in the community, may motive individuals to seek help (Stephensen et al., 2013).

The determinants of HIV testing have received considerable research attention. However, different studies have come up with different positions and it is challenging to reconcile existing findings and use them to inform our knowledge about HIV testing in Zimbabwe. Most

available literature has focused on the association between HIV testing and individual-level factors (Jean et al., 2012; Stephensen et al., 2013; Sambisa, 2008; Takarinda et al., 2014). However, implications regarding community influences on HIV testing has received limited attention. Also, most studies have tended to analyse the association of predisposing factors such as age, gender, marital status etc, without taking into account the enabling factors such as resources and enabling environment that facilitate individual's behaviour. Working from the premises that community inequalities in access to health care services is as significant as individual level factors, it is essential and worthwhile to investigate and understand the effects of both individual and community-level determinants of HIV testing in the context of Zimbabwe.

4.2 Conceptual framework

The trajectories to HIV testing first need to be explained before undertaking the analysis. In this regard, Andersen (1995) provided a useful analytical framework for examining determinants of HIV testing. This theoretical framework is grounded in the notion that utilization of health care services is dependent upon contextual situations, which in turn influences individual circumstances. The framework is conceptualised based on the predisposition, enablement and need for health care use (Andersen, 1995). What is vital in this approach is its emphasis on the joint effects of community and individual level factors that influence health care service utilisation. First, the level of HIV testing uptake is affected by predisposing factors which usually includes personal attributes of an individual such as demographic characteristics (i.e age, gender, marital status, religion, ethnicity and others). Secondly, predisposing factors are assumed to operate through enabling and perceived need factors and these comprise of contextual and sometimes personal circumstances, representing the ability to use health care services. In this study, enabling factors are composed of an individual's income, access to media, and awareness of HIV, observed HIV stigma, HIV disclosure and confidentiality concerns (Angotti et al., 2009; Mashuba et al., 2015; Lepine et al., 2014). Thirdly, enabling factors are assumed to facilitate or inhibit individuals from accessing HIV testing services. However, whether or not an individual has a means ofaccess to health care services, there must be a need for them taking that action (Iriyana et al., 2007). The need factors pertain to the perceived and evaluated assessment of one's health status which may compel the need for seeking health care services (Bond et al., 2005). For that reason, perceived/need factors such as whether an individual had engaged in risky behaviour such as

having multiple sex partners, or had history of sexually transmitted diseases and whether one knew someone who had died of HIV were also considered in the framework. The conceptual framework in Figure 4.1 provides a schematic display of the perceived determinants of HIV testing in Zimbabwe.





Source: Authors' construct.

The accumulated effects of predisposing, enabling and need factors govern one's trajectory to HIV testing behaviour. Given the background of growing inequality in access to health care services in Zimbabwe, issues of community influence on HIV testing comes at the fore. A growing need exists for information on the implication of both individual and community level effects on HIV testing. Such information will help the HIV prevention programs identify communities with high concentrations of individuals who are likely to have been ever tested for HIV, as well as sub-populations that are not likely to be tested (Mwangi et al., 2014). Guided by this conceptual framework, this chapter aims to understand individual and

community-level factors associated with HIV testing. Specifically, the study aims to identify the predisposing, enabling and perceived risk or need factors associated with HIV testing with particular focus on gender differences

4.3 Data and Methods

4.3.1 The Data

This is a secondary data analysis study based on the 2005-6 and 2010-11 Zimbabwe Demographic and Health surveys (ZDHS), the first two nationally representative surveys to include HIV testing. The surveys were designed to provide national estimates of HIV in the population, including HIV testing coverage. The analysis is based on respondents aged between 15-59 years residing in 402 clusters, and the sample size was 32384 of which 27.8% of males (n=14587) and 43.7% of females (n=17797) were ever tested for HIV. Overall, about 93% of those who tested for HIV received their test results. The individuals responded to a questionnaire survey asking questions related to their HIV testing history, socio-economic and demographic background and health indicators such as the individual's sexual behaviour, HIV related knowledge, attitude and behaviour and media exposure. The sample excluded the institutional population, which includes individuals living in hospitals, prisons and other institutions.

4.3.2 Outcome variables

The key outcome variables of interest were whether the respondent was ever tested for HIV and received results. For each outcome variable, possible responses were 'yes' or 'no' if an individual had ever been tested for HIV or have received results or not.

4.3.3 Explanatory Variables

We identified a range of independent variables based on the conceptual framework described above, their presence in existing empirical literature and the sample distribution in the ZDHS datasets. The explanatory variables were clustered into specific categories; individual and community-level predisposing, enabling and perceived/need factors.These categories were considered to give the best representation of community and individual characteristics that are associated with HIV testing behaviour. Community-level factors such as poverty, HIV awareness, and stigma and media exposure were imputed from relevant individual -level factors.

4.3.4 Individual-level socio-economic and demographic predisposing variables

While inclusion of wealth, occupation, education and media exposure may indicate the manifestation of economic position of an individual (Obermeye et al., 2013; Uthman et al., 2010) demographic factors such as gender, parity, religion, marital status and age are considered important because they may capture behavioural factors such as the perception of risk, acceptance of testing and sexual practice (Kayeyi et al., 2014). The following are factors which were included at the individual level.

(*a*) Gender: gender plays a key role in health seeking behaviour and there is evidence from existing literature suggesting that women, due to reasons such as frequent access to health care services through antenatal care (ANC), are more likely to test and receive HIV test results than men (Peltzer et al., 2009; Lepine et al., 2014). The variable was used to compare utilisation of HIV testing services between males and females. This is a binary measure coded 1 if female and 0 if males.

(*b*) Age: The ZDHS sample included females (aged 15-49) and males (aged 15-59) of reproductive age. Age was recorded into five categories: 15-24; 25-29; 30-34; 35-39; 40+. The variable wasused to compare HIV testing behaviour between different age groups.

(*c*) Marital status: This variable captures differentials in HIV testing behaviour between marital status categorises. It was recoded into four categories: never-married, married, widowed and divorced.

(*d*)Type of place of residence: This variable captures HIV testing behaviour of individuals by urban / rural residence.

(*e*)Level of education: This was one of the key variables, which captured socio-economic characteristics of individuals. The variable was recoded into three categories: primary, secondary and higher educational attainment.

(f) Religion: Catholic and Protestant followers tend to have higher odds of health care service utilisation. This behaviour has been interpreted as a consequence of greater social diversity within mainstream religious groups that facilitates change of behaviour towards health care services utilisation (Hartwig et al., 2006; Mashuba and Hemalata, 2015;Luginaah et al., 2005). A categorical variable based on religious affiliation classified into: Catholic, Protestant, Pentecostal and Apostolic categories is included in the analysis

(g) Occupation: Economic dependency has been one of the major barriers to women's control over their health behaviour in developing countries. However, studies have shown that a woman's occupation is an important factor that influences her access to health care services. This is due to the fact that women who are working and earning money will have greater autonomy and control over financial resources, and thus they are more able to pay for health care services (Jean et al., 2012). A categorical variable based on occupation is also included in the analysis

4.3.5 Enabling factors

Another explanatory domain in the conceptual framework relates to enabling factors. These are factors which promote or inhibit use of HIV testing services and they include enabling resources and environmental factors.

(*a*) Household wealth: This variable was used as a measure of economic wellbeing. The index was constructed by DHS using principal component analysis to assign indicator weights based on household ownership of assets (Rutstein, 2008). This was a key variable of interest, and five wealth quintiles (poorest, poorer, middle, richer, and richest) were used in analysing the relationship between wealth and HIV testing behaviours.

(b) Media exposure: This variable captures how individuals' exposure to media was associated with their HIV testing behaviour. Exposure to HIV/AIDS information through mass media may lead to high levels of awareness, which may in turn influence self-assessed risk of HIV infection and the need to test for HIV (Weiser et al., 2006). This variable was measured by three

items relating to frequency of watching television, reading newspapers and listening to radio. An additive scale was imputed and classified into three categories: 'low, medium and high.

(c) AIDS awareness index: This variable was derived from eight HIV related questions which consist of prevention and misconceptions about transmission. Aids awareness index, was classified into three categories: low: if participants answered three or less questions correctly; medium: if answered up to five questions correctly; and high: if answered 6-8 questions correctly.

(d) HIV related stigma: This referred to the respondent's attitude towards HIV developed from four questions of HIV related stigma, namely; if the respondent' would buy vegetables from a vendor with HIV', 'can care for a relative with HIV', 'would want an HIV-positive teacher to continue teaching', and 'would want others to know if a family member became infected with HIV'. The variable was classified into three categories: low: if not experienced any of the stigma indicators, medium: if experienced 1-2 of the stigma indicators and high stigma: if experienced 3-4 stigma indicators.

(e) Observed stigma: Like in Sambisa (2008), this study uses this variable to assess individual's attitude towards people living with HIV. People act upon what they see happening around them (Chirawu et al., 2010). People may have been discouraged from being tested for HIV if they observed people living with HIV being discriminated against in their communities or families. In a study by the Government of Zimbabwe (ZNASP, 2011-2015), a significantly number of people living with HIV reported that they had been forced to change residency or denied accommodation. We therefore, constructed observed an'stigma variable' from the following set of prompt items; whether participant knows someone suspected to have HIV/AIDS who has been denied health services, knows someone suspected to have HIV/AIDS who has been verbally abused. For the observed stigma variable, the additive scale was split into two categories: yes (1 or more) or no (0)

(f) Disclosure and Confidentiality: Not much attention has been given to HIV status disclosure and confidentiality dynamics within private and public places. People do not worry about how to disclose their HIV sero-status only, but they also worry about how their health information is shared (Kalichman et al., 2003). Mashuba and Hemalata (2015) pointed out that people living with HIV may have to disclose their HIV status to their family members, health care workers, employers, religious leaders, counsellors and community members. The disclosure and issues with how health information is going to be shared if results come out positive can act as a barrier to access to HIV testing services. Disclosure and confidentiality concern was a variable measured by a single item; 'would want others to know if a family member became infected with HIV and coded 'yes'if someone reported that they would want others to know if a family member became infected, and 'no' if they didn't want others to know.

4.3.6 HIV and risky sexual behaviour variables

Finally the perceived need factors were considered. The conceptual framework suggests that for someone to take an action on his/her health there must be a perceived risk and a need to do so (Iriyana et al., 2007). Here the perceived need factors are operationalized by whether someone feels he/she is at risk of HIV infection because of their history of sexually transmitted diseases (STI), multiple sex practice or knowing someone who died of HIV. Literature suggests that people get tested for HIV following the Illness or death of people they know (Kalichman et al., 2003). Each perceived need variable was coded 'yes or no'

(a) Condom used in the last sex encounter. This is a binary variable. Those who had not used condom takes the value of 0, and those who have used take the value of 1.

(b) Multiple sex partnership: This also is a binary variable, which takes the value of zero if an individual did not have sex or had sex with only one partner and takes the value of one if an individual had sex with more than one sex partner in the last 12 months preceding the survey.

(c) History of STI: This is a binary variable. Those who have never had STI take the value of 0, and those who have ever had STI takes the value of 1.

4.3.7 Methods of analysis

A two-level logistic regression model for binary response was employed. Individuals who responded to the questionnaire were considered as level-1. Other studies which investigated
multilevel determinants of HIV testing behavior considered household (Mitchel et al., 2010) and couples (Lepine et al., 2014), as levels of analysis. This study did not consider household as a level of analysis because the average number of individuals in a household who have been ever tested for HIV as contained in the dataset was too small to be considered as a level of analysis. The study did not also consider couples; this was premised on the fact that never married people are less likely to be tested for HIV than married, divorced or widowed people (Agha, 2012; Weiser et al., 2006). Therefore, in this study of the determinants of HIV testing, level-2 of analysis was represented by a cluster, which is defined as a community.

4.3.7.1 Multilevel logistic models

We recognize that communities may share similar socio-economic and demographic characteristics, resources and experiences (Mitchel et al., 2010) and it is therefore, reasonable to assume that residents of one community maybe more similar to each other with respect of their HIV testing behaviour. With such background knowledge, it can be argued that variations in HIV testing in Zimbabwe may not have been as a result of individual characteristics alone, but of the effects of community effects in which they live as whole (Godif et al., 2015; Uthman et al., 2017). For that reason, this study used two level logistic regression models to investigate the community and individual level factors that influence HIV testing behaviour. The basic form of two-level random intercepts logistic regression model used can be expressed as:

Logit
$$(\pi_{ij}) = \text{Log} [\pi_{ij}/(1-\pi_{ij})] = \beta_0 + \beta X'_{ij} + u_{0j}$$

Var $(u_{0ij}) = \sigma^2 u_0$

Where (π_{ij}) is the probability of having ever been tested for HIV for an individual i, in the jth community. β_0 is the regression intercept, X'_{ij} is the vector of covariates defined at individual or community level; β is the associated vector of usual regression parameter estimates, and is shared by all communities, while the random effect u_{0j} is specific to community j.

Before including any explanatory variables in the models, it was crucial to know how much between community variations was there in the propensity to test for HIV. To assess this, we look at the estimated value of σ^2_{u0} , which is the variance of theu_{0j} terms. We use a threshold model approach which measures proportions at group level: $\rho {=} {\sigma_U}^2 {/} \left({\sigma_U}^2 + {\sigma_\epsilon}^2 \right)$

Where σ_U^2 is the total variance at the community level, $\operatorname{and} \sigma_{\varepsilon}^2$ is the total variance at the individual level. For the multilevel logistic regression model, the level-1 residuals are assumed to have a standard logistic distribution with mean zero and variance $\sigma_{\varepsilon}^2 = \pi^2 / 3$ where π is the constant 3.14159. Thus, $\rho = \sigma_U^2 / \sigma_U^2 + 3.29$ (Goldstein, 2003).

4.3.7.2 Modelling Approaches

Since our analysis is based on hierarchical data given the ZDHS multi-stage sampling design, it is necessary to use techniques that consider the possible dependence of individuals clustered in the community (Goldstein, 2003). Conventional regression analysis techniques assume that individual observations are independent from one another (Magadi and Desta, 2011). If this assumption is violated, estimates of the regression coefficients can be biased and standard errors may be underestimated. Multilevel regression techniques make it possible to take into account the possible dependence of the outcome variable between people in the same community (Goldstein, 2003). Four models were fitted in the analysis. Model 0 was an empty model with year of survey as the only co-variate. This was fitted to decompose the total variance between individual and community level. Background demographic and socio-economic factors were then added in Model 1. The next model (Model 2) was composed of background demographic/socio-economic factors and HIV and risky sexual behaviour; enabling and need factors. The final model (Model 3) took account of community factors.

Measures of fixed effects, that is the effects of individual-level and community-level determinants of HIV testing were reported in terms of odds ratios, whilst the measures of variation (random effects) were expressed in terms of intra-class correlation coefficient (ICC). The estimation procedure was based on quasi-likelihood methods, starting with the default procedure 1st order Quasi-likelihood Methods (MQL) and extending to 2nd order Penalized Quasi Likelihood (PQL).The 2nd order estimation was preferred because it is an improved approximation procedure (Rasbash et al., 2005).

4.4 Results

4.4.1 Descriptive and bivariate analysis

Cross-tabulation analysis, including chi-square tests were undertaken to assess the distribution of individuals who were ever tested for HIV by predisposing, enabling and need factors. Relevant means or proportions were imputed for community-level variables. The proportions of individuals who were ever tested for HIV and received their results are presented in table 4.1, based on the 2005/6 and 2010/11 ZDHS.

4.4.1.1 Percentage distribution of individuals ever tested for HIV and those who received the results

Table 4.1 presents the proportion of individuals who ever tested for HIV and the proportion that received results among those who were ever tested for HIV in the 2005-6and 2010-11 surveys. The coverage of HIV testing was significantly higher for females (26%) versus males: (18%), p<0.005) in the 2005-6 survey. By 2010-2011 the coverage of HIV testing increased for both males and females to 62% for females: versus males: 38% (p<0.005). Among those who ever tested for HIV, 89% of males and 85% of females received results in the 2005/6 survey; these percentages increased to 93% for males and 96% for females in the 2010-11 survey.

G	Proportion who have ever been tested for HIV							
Survey year		Males	Females					
	% ^a	n ^b	% ^a	n ^b				
2005-6	17.7	7107	25.7	8849				
2010-11	37.6	37.6 7480		8948				
All	27.9 14587		43.7	17797				
Proportio	on who received	results among those v	vho have ever be	een tested for HIV				
Survey year		Males	Females					
	% ^a	n ^b	% ^a	n ^b				
2005-6	88.6	88.6 1258		2274				
2010-11	92.8	92.8 2812		5503				
All	91.5	91.5 4070		7777				

Table 4-1: Trends in HIV testing coverage and receipt of results, 2005/6-2010/11 ZDHS

^aWeighted; ^bunweighted cases

4.4.1.2 **Distribution of HIV testing by background characteristics**

A bivariate analysis was used to assess the association between each independent variable set out in the framework and the study outcomes. The Chi-square tests showed association between key variables and HIV testing, but not with receipt of HIV test results. The sample size of those who did not receive results after an HIV test was too low to allow for a meaningful multivariate multilevel logistic regression analysis. Therefore, the determinants of receipts of HIV test results were excluded from further analysis. Given that over 90% of those who tested for HIV collected their results (ZDHS, 2010/11), policies needs to focus mainly at increasing uptake of HIV testing in Zimbabwe.

The results in Table 4.2 indicated that there was an association between HIV testing and the following factors: age, place of residence, marital status, educational attainment, occupation, household wealth, exposure to media, and HIV awareness. The highest proportion (58%) was observed among females in the age group (25-29); compared to males (38%) in the same age group. For both males and females, HIV testing coverage was lowest among individuals below the age of 20 years, followed by those aged 40 years and above. Because of disparities in social and economic development between urban and rural areas, place of residence was also a key factor in determining access to HIV testing services.

A higher proportion of people who ever tested for HIV were from urban than rural areas for both males and females. With respect to marital status, for females, the highest coverage of HIV testing was observed among those who were married (52%), whilst for men; it was among those who were widowed (41%). As would be expected, HIV testing uptake has increased by educational attainment for both males and females. For example only 19% of males and 36% of females with primary education were ever tested for HIV, compared to 53% and 65% males and females with higher educational attainment, respectively.

	Proportion ever been tested for HIV by background characteristics						
Background	Males		Females			All	
characteristics						Cases ^a	
	% ^a	n ^o	<i>P</i> -value	% ^a	n ^D	<i>P</i> -value	
Age groups*			< 0.001			< 0.001	
15-24	16.7	6533		35.4	7717		14250
25-29	37.7	2218		57.9	3089		5307
30-34	36.3	1838		51.6	2470		4308
35-39	37.1	1462		48.0	1859		5321
40+	34.6	2536		36.5	2660		5196
Residence*			< 0.001			< 0.001	
Urban	34.6	4992		47.9	6571		11563
Rural	24.4	9595		41.3	11224		20819
Marital status*			< 0.001			< 0.001	
Never married	17.7	6724		22.9	4695		11419
Married	36.7	7154		51.9	10529		17683
Widowed	41.1	180		46.5	1231		1411
Divorced	34.2	529		49.1	1340		1869
Education level *			< 0.001			< 0.001	
Primary	18.8	4176		35.5	6065		10241
Secondary	29.6	9523		46.9	11083		20606
Higher	52.8	888		64.5	647		1535
Religion*			< 0.001			< 0.001	
Roman catholic	26.9	6837		42.2	4754		11591
Protestant	29.4	2202		41.3	3734		5936
Pentecostal	33.6	1944		47.2	3343		5287
Apostolic sect	25.8	3604		44.4	5964		9568
Occupation*			< 0.001			< 0.001	
Unemployed	18.5	4634		40.7	10278		14912
Professional	52.1	1056		72.6	1796		2852
Manual work	46.0	2121		70.8	729		2850
Agriculture	36.4	1842		58.6	1218		3060

Table 4-2: Distribution of HIV testing by background socio-economic and demographic characteristics

^aWeighted; ^bunweighted cases

4.4.1.3 Distribution of HIV testing by enabling and need Factors

Table 4.3 shows differentials in HIV testing based on enabling and need factors, including household wealth, media exposure, HIV/AIDS awareness, observed stigma, and HIV test results disclosure concern. The proportion of individuals who were ever tested for HIV was higher among those from wealthier households. For example, 21% of males from poorest households versus 39% from richest household reported ever being tested for HIV. Whilst, HIV testing was higher among individuals who reported having higher media exposure or HIV awareness, it was lower among those who reported having observed HIV stigma and those with HIV disclosure concerns.

Variable	Proportion ever been tested for HIV by enabling and need factors						
	Males			Females			
	% ^a	n ^b	<i>P</i> -value	% ^a	n ^b	<i>P</i> -	All Cases
						value	
Wealth *			< 0.001			< 0.001	
Poorest	20.7	2513		38.1	3227		5740
Poorer	21.3	2638		39.6	3143		5781
Middle	24.3	2712		42.9	3165		5877
Richer	30.1	3475		48.6	3925		7400
Richest	39.1	3249		46.9	4335		7584
Media exposure*			< 0.001			< 0.001	
Low	24.3	9446		44.1	12396		22742
Medium	37.8	3245		48.2	2662		5907
High	28.7	1877		34.4	1804		3681
HIV awareness	*		< 0.001			< 0.01	
Low	19.2	661		23.8	1117		1778
Medium	33.8	7694		51.6	10174		17868
High	21.6	6232		34.7	6504		12736
Observed stigm	na*		< 0.01			< 0.01	
No	16.5	5148		47.3	14782		19930
Yes	22.4	9439		30.6	3013		12452
Disclosure concern*			< 0.001			< 0.001	
No	28.7	8226		44.3	9982		18208
Yes	26.9	6361		42.9	7813		14174

Table 4-3: Distribution of testing by enabling and need factors

Continued

Need/risk perception factors							
History of STI*			< 0.001			< 0.001	
No	27.7	14204		43.3	17248		31452
Yes	35.2	383		57.2	547		930
Condom use*			< 0.001			< 0.001	
No	36.0	12057		42.1	16374		28431
Yes	37.1	2530		62.4	1421		3951
Multiple sexual	Multiple sexual partners*		< 0.001			< 0.001	
No	31.2	7501		47.4	11698		19199
Yes	24.5	6576		36.7	5931		12507
Know living with HIV*			< 0.001			< 0.001	
No	29.9	11089		47.3	14313		25402
Yes	21.7	3490		30.7	3261		6751

Table 4-3 continued

^aWeighted; ^bunweighted cases.

Based on the Health belief and Aids Risk Reduction models (Catania, 1990, Iriyana, 2007) individuals who engage in multiple sexual partnerships, have history of (STI), and do use condoms or those who knew someone who had died of HIV may consider themselvesto be at a high risk of HIV infection. Therefore, they may perceive themselves as having a need to test for HIV (Lepine et al., 2014). In our sample, a higher proportion of individuals (35% of males and 57% of females) who mentioned that they have a history of STI had been tested for HIV than those who did not have a history of STI (28% of males and 43% of females). Furthermore, a higher proportion of individuals who reported having used condoms in their last sexual intercourse encounter had been tested for HIV compared with individuals who did not use condoms (e.g. 62% versus 42% for females). For both males and females, those who reported having multiple sexual partners and those who knew someone who died of HIV were less likely to report having been tested for HIV than their counterparts without multiple partners or who did not know anyone who had died of HIV/AIDS, contrary to expectation.

The results from bivariate analysis suggested a significant association between variables considered in the framework. However, given that bivariate associations can be influenced by confounding factors, a multivariate analysis that simultaneously takes into account these effects was used to accurately establish the independent predictors of HIV testing behaviour (Snijders and Bosker, 1999).

4.4.2 Multivariate multilevel Analyses

Multivariate analyses here start with combining the data for males and females across the two surveys to assess gender differences. Although the association patterns observed in bivariate analysis were relatively consistent with gender, it is reasonable to recognise that the determinants of HIV testing among males and females are likely to be different (Godif et al., 2015). For this reason, multivariate analysis is presented separately for males and females to explore the differences in predictors of HIV testing by gender. Results for the combined sample are given in Annex (4.1), while tables 4.4 and 4.5 present estimates for females and males respectively. There is evidence of a significant gender disparity in HIV testing, with females being about twice as likely to be tested (average OR=1.8) as compared to males of similar characteristics (See Annex 4.1).

4.4.2.1 The Determinants of HIV testing among females

Table 4.4 presents the odds ratios (ORs) associated with HIV testing for females. With all variables excluded, the results show females were 5.4 times more likely to have been HIV tested in 2010/11 than in 2005/6 survey. When all predisposing variables were included in Model 1, the odds of HIV testing substantially increased to 6.9. Females from rural areas were 36% less likely to test for HIV than those from urban areas. As females grow older, they have a tendency to avoid HIV testing. For example, females aged between 30-34 years and those who were 40 years and above were 31% and 65% less likely than young individuals aged 15-24 years to test for HIV, respectively. Married females were 2.1 times while those who were widowed and divorced were 2.4 and 1.8 times more likely, respectively, to test for HIV compared with those who never married. Also females who were professionals and those with manual jobs had 1.4 times higher odds of HIV testing than those who were unemployed. Again, the likelihood of HIV uptake increased significantly with birth order. For instance, birth order 2 was associated with 5.4 times higher odds of HIV testing than birth order 0. Furthermore, individuals with secondary or higher educational attainment were found to be 1.8 and 3.6 times more likely to have been tested for HIV compared with those with primary or no educational attainment. Compared with the empty model 0, the inclusion of individual level variables in Model 1 increased the intra-community correlation to 6.8%, with approximately 21.4% of proportional change in variance unexplained.

Parameters Model 1 Model 2 Model 3 AOR 95%CI AOR 95% CI AOR 95% CI Survey (2005-6) 2010-11 5.39 [4.83-6.02]* 6.89 [6.08-7.79]* 6.91 [6.11-7.82]* Residence (urban) 0.74 [0.67-0.83]* 0.96 [0.82-1.11] 1.08 Rural [0.91-1.29] Age groups (15-24) 25-29 0.93 [0.83-1.05] 0.93 1.02 [0.91 - 1.14][0.30 - 1.05][0.61-0.79]* [0.54-0.71]* 30-34 0.69 0.62 0.62 [0.54-0.71]* 35-39 0.51 [0.39-0.53]* [0.39-0.53]* [0.44-0.59]* 0.46 0.46 40 +0.35 [0.30-0.40]* 0.31 [0.27-0.36]* 0.31 [0.26-0.36]* Marital status (never married) Married 2.12 [1.85-2.43]* 2.23 [1.94-2.57]* 2.23 [1.94-2.57]* Widowed 2.43 [2.00-2.95]* 2.40 [1.97-2.92]* 2.39 [1.97-2.91]* Divorced 1.76 [1.47-2.11]* 1.71 [1.45-2.10]* 1.74 [1.45-2.09]* Occupation(unemployed) Professional 1.41 [1.24-1.63]* 1.34 [1.17-1.54]* 1.34 [1.17-1.54]* Manual work 1.36 1.34 [1.11-1.63]* 1.41 [1.17-1.70]* [1.12-1.64]* Agriculture 1.03 [0.88-1.19] 0.98 [0.84-1.15] 0.98 [0.84 - 1.14]Religion (Catholic) 1.10 [0.99-1.22]1.05 [0.94 - 1.17][0.94-1.17] Protestant 1.04 Pentecostal 1.08 [0.97 - 1.21]1.05 [0.94-1.17] 1.04 [0.93-1.17] 0.93 Apostolic [0.85 - 1.02]0.96 [0.87 - 1.06]0.96 [0.87-1.05] Education (primary) 1.78[1.63-1.94]* [1.43-1.71]* 1.55 [1.42-1.70]* Secondary 1.56 2.79 [2.24-3.47]* [2.24-3.46]* Higher 3.60 [2.91-4.44]* 2.78 **Enabling and perceived/need factors** Wealth (poorest) 1.05 [0.93-1.19] [0.91-1.17] Poorer 1.03 Middle 1.20 [1.06-1.37]* [0.99-1.31] 1.14 Richer 1.24 [1.07-1.45]* 1.15 [0.97 - 1.36]1.16 Richest 1.27 [1.05-1.53]* [0.95-1.42] Media exposure (low) Medium 1.21 [1.09-1.35]* 1.21 [1.09-1.34]* [1.42-1.97]* [1.42-1.97]* High 1.67 1.67 HIV/AIDS Awareness(low) Medium 1.59 [1.31-1.93]* 1.59 1.31-1.93]* High 2.04 2.04 [1.72-2.43]* [1.71-2.43]*

 Table 4-4: Average odds of HIV testing from multilevel logistic regression models (95% confidence intervals are given in square brackets)-females

Continue

Table 4-4 Continued

Observed AIDS stig	ma(no)					
Yes			1.07	[0.90-1.27]	1.06	[0.90-1.26]
HIV disclosure conc	ern(no)					
Yes			0.82	[0.74-0.90]*	0.81	[0.74-0.89]*
Knows someone died with Aids(no)						
Yes			1.23	[1.08-1.41]*	1.23	[1.08-1.41]*
History of STI (no)						
Yes			1.55	[1.27-1.89]*	1.55	[1.27-1.90]*
Community factors						
Poverty –gm					0.73	[0.56-0.94]*
Random Variance (SE)		0.23(0.025)*		0.22 (0.024)*		0.218(0.024)*
(VPC)=ICC (%)		6.8		6.3		6.2

VPC=Variance Partition Coefficient, ICC=intra-cluster correlation, *Statistical significance at 5% level p<0.05

Model 2 controlled for the enabling, perceived and need factors. The results showed that, although slightly reduced, the background characteristic factors remained significantly associated with HIV testing when enabling, perceived and need factors were controlled for. However, controlling for model 2 factors substantially explained the effect of place of residence such that place of residence was no longer significantly associated with HIV testing for females. Enabling factors such as wealth, media exposure and HIV awareness were found to be associated with HIV testing for females. Being wealthier was associated with increased odds of HIV testing. For instance, females in the richest quintile households were 1.3 times more likely to test for HIV than those from the poorest households. Similarly, those with higher levels of media exposure and HIV awareness had higher odds of HIV testing compared with their counterparts with lower levels of media exposure or HIVAIDS awareness. The results relating to perceived and need factors suggest that having disclosure concerns was associated with reduced odds of being tested for HIV by 18% compared with females with no disclosure concerns. On the other hand, having a history of STIs and knowing someone who died of HIV was associated with increased odds of HIV testing. The results provide no evidence of a relationship between HIV testing and observed stigma among females.

In the final model (Model 3), poverty in the community was the only community-level variable found to be significantly associated with HIV testing for females. Living in a community with

a high proportion of individuals living in poverty was associated with reduced odds of HIV testing. The estimates for household wealth diminished after we controlled for community poverty such that household wealth was no longer significantly associated with HIV testing.

The results of the variance components model (i.e the empty model) and other models in Table 4.4 suggest that most of the variation in HIV testing for females was at the individual level. However, there was also some variation at the community level as indicated by the significant random variance in reporting of ever being tested for HIV across communities. As shown by the variance partition coefficient (VPC), the intra-class correlation coefficient (ICC) was estimated at about 6-7%, even after controlling for individual and community level factors in Table 4.4. Thus, about 6% of the total unexplained variation in HIV testing could be attributed to unobserved community-level effects with the remaining 94% of unexplained variation attributable to individual –level factors.

4.4.2.2 The Determinants of HIV testing among males

The results for males (Table 4.5) slightly differed from those for females. As with females, the odds of HIV testing for males were higher in 2010/11 than in 2005/06, but the odds of HIV testing for males were lower by 3.03 points compared with the odds for females between the same periods (OR=5.60 females versus OR=2.57 males). This suggests that the increase in uptake of HIV testing was greater for women than men. In comparison with female's estimates, some results for males also differ when we controlled for background characteristics. It is noticeable that males residing in rural areas were about 1.2 times more likely than those from urban areas to have ever been tested for HIV. This finding differs with those of females which suggested that rural residence was associated with lower odds of HIV testing than urban residence. Results also revealed that being older is associated with reduced odds of HIV testing by 32% for males, while being widowed was associated with increased odds by 47% compared to being never married. Unlike females, being a manual worker or working in agricultural sector was associated with increased odds of HIV testing for males.

Demonsterne]	Model 1]	Model 2		Model 3
Parameters	AOR	95% CI	AOR	95% CI	AOR	95% CI
Survey (2005-6)						
2010-11	2.57	[2.28-2.91]*	2.92	[2.53-3.37]*	3.04	[2.63-3.53]*
Individual level fac	tors					
Residence (urban)						
Rural	1.16	[1.01-1.33]*	1.17	[1.02-1.34]*	1.13	[0.99-1.30]
Age groups (15-24)						
25-29	0.30	[0.24-0.38]*	0.37	[0.29-0.48]*	0.37	[0.29-0.48]*
30-34	1.04	[0.85-1.27]	1.04	[0.85-1.27]	1.04	[0.85-1.27]
35-39	0.93	[0.76-1.14]	0.93	[0.76-1.14]	0.93	[0.76-1.14]
40+	0.94	[0.76-1.15]	0.94	[0.76-1.16]	0.94	[0.76-1.15]
Marital status (never	married)				
Married	0.68	[0.59-0.77]*	0.68	[0.59-0.77]*	0.57	[0.49-0.68]*
Widowed	1.47	[1.06-2.03]*	1.47	[1.06-2.03]*	1.23	[0.87-1.74]
Divorced	1.00	[0.82-1.22]	1.00	[0.82-1.22]	0.77	[0.62-0.96]*
Occupation(unemplo	oyed)					
Professional	1.18	[1.00-1.39]	1.17	[0.98-1.38]	1.16	[0.98-1.38]
Manual work	1.37	[1.20-1.58]*	1.35	[1.17-1.55]*	1.35	[1.18-1.55]*
Agriculture	1.20	[1.03-1.39]*	1.18	[1.02-1.37]*	1.20	[1.03-1.40]*
Religion (Catholic)						
Protestant	1.21	[1.05-1.40]*	1.25	[1.09-1.45]*	1.26	[1.09-1.45]*
Pentecostal	0.96	[0.84-1.09]	0.99	[0.87-1.12]	0.99	[0.87-1.13]
Apostolic	0.84	[0.74-0.95]*	0.85	[0.75-0.96]*	0.85	[0.75-0.96]*
Education (primary)						
Secondary	1.89	[1.29-2.76]*	1.82	[1.24-2.66]*	1.88	[1.28-2.75]*
Higher	3.03	[2.01-4.57]*	2.90	[1.92-4.37]*	3.00	[1.98-4.54]*
Enabling and perce	eived/nee	d factors				
Wealth (poorest)						
Poorer			1.01	[0.88-1.18]	1.02	[0.88-1.19]
Middle			1.16	[1.01-1.35]*	1.18	[1.02-1.36]*
Richer			1.25	[1.06-1.47]*	1.26	[1.07-1.49]*
Richest			1.62	[1.34-1.96]*	1.65	[1.36-2.00]*
Media exposure (low	v)					
Medium			1.12	[1.01-1.25]*	1.12	[1.01-1.25]*
High			1.53	[1.35-1.73]*	1.54	[1.36-1.74]*

Table 4-5: Average odds of HIV testing from multilevel logistic regression models (95%confidence intervals are given in square brackets)-males

Continued

Table 4-5 continued

Condom used (no)							
Yes			1.63	[1.43-1.86]*	1.63	[143-1.86]*	
Knows someone died of Aids(no)							
Yes			1.26	[1.11-1.43]*	1.26	[1.11-1.44]*	
History of STI (no)							
Yes			1.46	[1.15-1.85]*	1.45	[1.14-1.84]*	
Community level fa	Community level factors						
Stigma –gm					0.55	[0.34-0.87]*	
Random effects							
Variance (SE)		0.082 (0.019)*		0.08(0.019)*		0.08 (0.019)*	
(VPC)=ICC (%)		2.4		2.5		2.5	

VPC=Variance Partition Coefficient, ICC=intra-cluster correlation, *Statistical significance at 5% level p<0.05

Also differing from females, the odds of being tested for HIV was 21% higher for individuals belonging to the Pentecostal church and 16% lower for those belonging to the Apostolic faith, compared with those from Roman catholic church for males, but not for females. Education was found to be positively associated with HIV testing, with those with higher educational attainment exhibiting increased odds of HIV testing for both males and females.

Model 2 controlled for enabling, perceived and need factors. Comparing females and males, both wealth and media exposure were found to be associated with HIV testing. Like for females, being wealthier and having exposure to media were associated with increased odds of HIV testing for males. Similarly, males exposed to media messages were more likely than their counterparts with less media exposure to have tested for HIV. Among perceived and need factors that were associated with HIV testing were condom use, knowing someone who died of HIV and history of STI. The results suggest that those who did use condoms in their last sex encounter were 1.6 times more likely to have been tested for HIV than those who did not use condoms. Meanwhile, those who reported knowing someone who died of AIDS and those with a history of STI. Circumcised males were about 1.3 times more likely to have been tested than those not circumcised. In the final model 3, the results showed that living in a community with a high proportion of individuals with stigma was associated with reduced odds of HIV testing. Other community level factors such as poverty and media exposure were not associated with

HIV testing for males. Controlling for community-level stigma in Model 3 considerably reduced the estimates for place of residence such that rural residence was no longer significant for males.

The results for different models in (Table 4.5) shows that most of the variation in HIV testing for males was at the individual level, but there were also some variation at community level as indicated by significant community-level random variance. The variance partition coefficient (VPC) (i.e. the intra-cluster correlation coefficient (ICC) was estimated at about 2.5%, which was the proportion of total unexplained variability in HIV testing that could be attributed to unobserved community level effects. This implies that most (i.e. about 97.5%) of the unexplained variation in HIV testing was attributable to unobserved individual-level factors.

4.5 Discussion, conclusion and recommendation

4.5.1 Key findings

The main objectives of this study were to identity individual and community-level factors associated with HIV testing in Zimbabwe, and establish gender disparities. Overall the results show that, HIV testing is improving as indicated by higher odds in the 2010/11 than in 2005/6 surveys. This finding applies to both males and females, but the main difference lies in the degree females were more influenced to test for HIV in the 2010/11 survey. This may be that more females have been responding to calls for HIV testing during maternal health care services than males. Although, males have not been a focal target for HIV testing during antenatal care, recent initiatives have encouraged men to accompany their partners to the health facility for HIV testing during pregnancy.

The two objectives were addressed by applying multilevel logistic regression models to separate samples by gender to identify and compare individual and community-level determinants of HIV testing among women and men. Measures of individual-level predisposing factors (i.e gender, age, educational attainment, marital status, household wealth, media exposure) showed significant associations with HIV testing. As expected, higher odds of having been tested for HIV were observed among females than males. Several studies conducted outside Zimbabwe, reported similar results to those found here with reference to differences in HIV testing between sexes. For example, a study conducted by Weiser et al

(2006) in Botswana and Mitchel et al(2010) in ten Southern African countries are comparable to the results found here, showing a significantly higher uptake of HIV testing for females than males. This may be explained as follows: First, it is possible that women with little control over their sexual activities of their partner's and more vulnerable to infection by their partners may have perceived themselves to be at a high risk of HIV infection and consequently this may have acted as a cue for HIV testing (Martin-Herz, 2006). The second explanation could be the fact that females are more likely to be exposed than males to attend health care services through maternal health care (Jean et al., 2012; MacPhail et al., 2009). As such, the disparity may have been as a result of success of the increased testing uptake during antenatal care.

The person's age was found to be a significant predictor of health care service utilisation. The effect of age on HIV testing behaviour observed was consistent with other studies; although the study by Weiser et al (2006) suggested that coverage of HIV testing begins to increase after 40 years. This study found lower odds of HIV testing among individuals over 30 years for females and those between the ages of 25-29 for males. This could be that older adults may have already experienced feeling of isolation due to illness or loss of someone they know who died of HIV (Takarinda et al., 2014). Having an HIV diagnosis may increase that sense of isolation. Older people may also face unique issues such as being widowed or divorced and are dating again, but they may have less knowledge of HIV and are less likely to protect themselves than younger people. Although studies have shown that older people visit health care services more frequently than younger people, they are less likely than younger people to discuss their sexual health with the doctors, who in turn may be less likely to ask older people about these issues. Older women who are less worried about getting pregnant may have less of a perceived need for HIV testing since they will not pass on HIV to unborn babies (Elul et al., 2009; Wekesa et al., 2014;Nakayiwa et al., 2006). We did not find evidence of lower uptake of HIV testing among males over 30 years of age. Further studies are needed to examine why being older for males was not associated with HIV testing.

We found that being married, divorced and widowed were associated with higher odds of HIV testing than being never married for females, but not for males. The finding that these females are more likely to test for HIV than those who never married has also been noted in previous studies (Mwangi et al., 2014; Jean et al., 2012; weiser et al., 2006). This pattern is likely to relate to the fact that never married females perceive themselves to be at lower risk of HIV;

and thus see no need to test for HIV (Chirawu et al., 2010; Taegtmeyer et al., 2006). It may also be that those married ordivorced may have had an opportunity for HIV testing during marriage and with partner involvement during ANC visits (Theuring et al., 2009). This study did not come to the same conclusion with other studies in regards to HIV testing for males based on marital status (Jean et al., 2012; Stephenson et al., 2013). The study found that married and divorced males were less likely to have been tested compared to those who were never married. This could indicate that married and divorced males may be relying on proxy testing if their partners were evertested for HIV during antenatal care (Morrill and Noland, 2006).

Several studies have found significantly higher odds of HIV testing among individuals with higher educational attainment than those with primary education (Jean et al., 2012; Stephenson et al., 2013; Weiser et al., 2006). The results presented here further confirm that the odds of having been tested for HIV were significantly higher among individuals with secondary or higher educational attainment. This finding applies to both males and females, but the main difference lies in the degree to which males were influenced by education. Disparities in HIV testing by education may be caused by differences in awareness of the importance of testing, access to testing centres and riskier sexual behaviour (Mlay et al., 2008). People with higher education may have better access to health care services generally, and particularly more uptake of antenatal care among females with higher education (Sherr et al., 2007).

Wealth may make it easy to afford payment of transport to go to testing centres consultation fees since health care services are paid for in Zimbabwe (Loewenson et al., 2012). The results observed here show increased odds of having been tested for HIV among the wealthier than the poorer for males, and no association between wealth and HIV testing for females after controlling for community wealth/poverty. This finding is consistent with previous studies from similar settings in Sub-Saharan African countries (Jean et al., 2012; Lepine et al., 2014) which found levels of HIV testing increasing with wealth for males. Several hypotheses may explain this outcome. Firstly HIV prevalence has been found to be positively correlated with socio-economic status in many sub-Saharan African countries (Magadi and Desta, 2011).Thus, individuals (particularly males) who are wealthy may perceive themselves more at risk, and therefore have a higher recourse to HIV testing compared to the poor (Jean et al., 2012). Another explanation could be that HIV testing promotion and programmes are failing to reach the most deprived populations (Chirawu et al., 2010). Thirdly, the adverse living conditions

associated with low socio-economic status may itself constitute a barrier to access to HIV testing (Wringe et al., 2008) and that the wealthier may have the opportunity to access health service easily and to choose a health service which they are comfortable with. The fact that household wealth is more important for males, while community wealth is more important for females may suggest that most males are economically empowered and have more resources enabling them to access healthcare services than females, while females rely on public health care which is cheaper or free in Zimbabwe.

Exposure to messages about HIV testing and counselling or campaigns about condom use or abstinence can be an important factor in motivating individuals to adopt HIV testing (Weiser et al., 2006). Several studies have found positive associations between HIV testing and higher media exposure (Mitchel et al., 2010; Mwangi et al., 2014; Peltzer et al., 2009). This study found similar results and suggests that being exposed to media (i.e listening to radio daily, watching and reading newspapers frequently) increased the odds of having been tested for HIV for both males and females. This might reflect the fact that individuals who are exposed to mass media may have the opportunity to learn the benefits of HIV testing, where and how HIV counselling and testing is given, and this subsequently improves their awareness towards HIV related knowledge and stigma.

Females who were exposed to HIV stigma and HIV awareness were also more likely to have been tested than those who were not. The possible reason could be those who had better knowledge or comprehensive knowledge about HIV/AIDS are more likely to know how HIV/AIDS is transmitted, the prevention mechanism and the benefit of HIV testing (Kayiki et al., 2011). The positive association between exposures to stigma could be as a result of success of HIV awareness programs aimed at reducing stigma in the communities. There is no evidence suggesting an association between HIV testing and HIV stigma and awareness for males.

The association between most of the sexual behaviour factors and HIV testing conform to what might be expected. For both males and females, the odds of having been tested for HIV were higher among individuals, who used condom during their last sexual contact, and those who had multiple sex partners. Several studies have found that people are more likely to be tested if they have risky sexual behaviours, such as having multiple sex partners and inconsistent condom use (Martin-Herz et al., 2006; Sherr et al., 2007). In regards to HIV knowledge and

awareness, the highest estimates for HIV testing coverage are observed among individuals with medium to higher HIV/AIDs awareness, echoing the results found in the previous studies (Mitchel et al., 2010; Obermeyer et al., 2013).

At the community level, community poverty, HIV-related stigma and media exposure were found to be associated with HIV testing. Community level risky sexual behaviour and HIV awareness were not associated with HIV testing, consistent with findings from previous studies (Agha, 2012; Bond et al., 2005; Lepine et al., 2014). In the general population, it was living in the communities with higher proportions of individuals with HIV-related stigma and media exposure that were associated with HIV testing. Living in a community with higher levels of media exposure was positively associated with HIV testing. Meanwhile, the results suggest that living in communities with higher proportion of individuals with HIV-related stigma was negatively associated with HIV testing in the general population. Similarly, the results showed that males living in communities with higher levels of stigmatisation were less likely to have reported ever testing for HIV. An explanation of this behaviour could be that men may be reluctant to know their HIV status if they feel that by doing so they become defined in the community as ill or weak, or feel that they can be denied opportunities such as employment (Jurgensen et al., 2012). Living in a community with high levels of poverty was associated with reduced HIV testing for females, but not for males. Overall, there is no evidence of an association between HIV testing and other community level factors (i.e community education and risky sexual behaviour).

Finally, the analysis focused on ascertaining the intra-community correlation to obtain a clear picture of between-community variation in HIV testing. The study found significant community variations in HIV testing, partly attributed to the individual level factors. As shown in tables 4 and 5, the random part showed that about 6-7% of the total variation in HIV testing among females (and 2-3% for males) in Zimbabwe was attributed to differences across the communities. The results were fairly consistent with the literature where the proportion of total variation in HIV testing attributed to community level factors ranged from 4 to 15 percent (Lepine et al., 2014; Stepheson et al., 2013). As in other countries, the community level variation obtained supports the theory that some differences in health outcomes are attributed to the community characteristics in which the individuals live (Stephenson et al., 2013; Uthman

et al., 2017). The study sheds light beyond the contribution of individual characteristics to the determinants of HIV testing in Zimbabwe.

4.5.2 Study limitations

The data used in this study came from two nationally representative surveys with household and individual response rates of 98% and 97% respectively. The missing data was lower than 5%, thus not likely to have affected the estimates in the study. However, the data had some weakness which might have affected the results. Firstly, the study did not control for the availability of HIV health care services at the community level. The ZDHS does not provide data to distinguish whether the number of people having an HIV test is limited by the availability of testing services or whether the testing services were underutilised and why. There is a possibility that inclusion of factors regarding health care services would have helped to explain some of the variance in the model. While knowing one's own HIV status is a proxy for a person having received counselling, the indicator does not provide information on the quality of the counselling services. In countries like Zimbabwe, where scaling up of HIV testing services is happening; population-based surveys conducted every few years will not capture annual progress. The information in the survey was self-reported, so to some extent under-reporting of socially unacceptable behaviours and attitudes (such as stigma) and over reporting of socially desirable behaviours were likely. In Zimbabwe, HIV testing has been heavily promoted as a responsible thing to do, so it is possible that some people may say they have been tested when in fact they were not. Another limitation was that the HIV testing and risk measures did not include assessments of time since the behaviours occurred, not allowing this study to examine whether people who were recently tested or who recently engaged in risky sexual behaviour differed from practising these behaviours less or more recently. Although, variables for the females such as 'problems of getting money, permission and transport to go to a health care centre are related to HIV testing behaviour, it would be a mistake to use them to make a general claim about their relative importance using variables available in ZDHS because most of these questions were not completed by the respondents. Otherwise the representation of the whole country is a major strength of this study as it allows generalization of findings for the country as a whole, for both genders.

4.5.3 Conclusion and way forward

Overall the multilevel results showed little variations in HIV testing across communities in Zimbabwe. The variations were mainly explained by individual level variables such as background characteristics rather than enabling, perceived need factors that were included in the models for both sexes. At the community level, variables found to be contributing to the variations in HIV testing differed by gender. It was found that community poverty was negatively associated with HIV testing for females, while community stigma was important for males. Controlling for community level factors did not have much effect on community variation in HIV testing. Therefore policies should focus on predisposing and enabling factors in order to improve HIV testing in Zimbabwe. For males' household wealth and HIV factors such as risky sexual behaviour are critical for improvements in HIV testing. Given that household wealth was a significant predictor of HIV testing for males, the introduction of financial incentives may stimulate males to access HIV testing services by providing compensation for transport cost and opportunity cost of time associated with accessing HIV testing services. Monetary incentives could reduce barriers and stigma as it may provide a broader reason for going to testing centres. More importantly perhaps for both sexes, awareness creation on HIV counselling and testing service utilization should focus on avoiding stigmatizing and discriminatory behaviours, so that clients can develop positive attitudes towards people living with HIV/AIDS. Furthermore, introducing couple-oriented HIV counselling and testing services where men accompany their spouses to HIV screening during pregnancy may help increase the HIV testing uptake for males. While offering HIV testing to everyone normalises HIV testing to everyone, raising awareness amongst several populations about the risk of HIV infection, it is believed that a substantial reduction of HIV burden is contingent upon increasing uptake of prevention of mother to child HIV transmission through improvement of key dimensions of HIV and SRH integrated services (Druce et al., 2006). The subsequent chapters focus on determinants of maternal health and family planning services, the global aim of these services being to contribute to the reduction of HIV burden by increasing uptake of (PMTC) programs.

5 CHAPTER FIVE: DETERMINANTS OF HIV AND ANC SERVICES UTILISATION.

5.1 Introduction

In the previous chapter, factors which could potentially increase uptake of HIV testing in the general population were investigated and tested using data from both males and females. As females are disapprortionately affected with HIV, including hgher risk of transmitting HIV to infants if they are HIV positive, it is of interest to understand factors associated with uptake of HIV and antental care (ANC) services among pregnant women. The current chapter contributes to the dialogue on prevention of mother to child HIV transmission through the use of antenatal HIV testing, timing and frequency of ANC visits. Included in the first sections of the chapter are literature on the magnitude of mother to child HIV transmission, determinants of intervention programs and conceptual framework relating to utilisation of HIV and ANC services. In the later sections, data, methods and results from the analysis of each indicator are presented. It concludes with a discussion on the results, including the effects of HIV on these services.

5.2 Background

In Sub-Saharan African (SSA) countries, new HIV infection in infants persist unabated notwithstanding the fact that antenatal HIV testing and prevention of mother-to child-transmission (PMTCT) interventions are available with a potential to reduce MCTC to as low as 1–2% (Ganich et al., 2015). There are approximately 3.2 million children aged 0-14 living with HIV in Sub-Saharan African countries (WHO, UNAIDS, UNICEF, 2011). Although declining from 390000 in 2009 an estimated 200000 children in SSA were newly infected with HIV in 2013 (UNAIDS, 2014). It has been estimated that, in the absence of antenatal HIV testing, the likelihood of HIV passing from mother-to-child is as high as 30% to 45% (Moodley et al., 2011; USAIDS, 2014). Approximately 90% of new HIV infected cases in infants in Sub-Saharan Africa countries are attributable to mother to child HIV transmission (MTCT) during pregnancy, labour, delivery or breastfeeding (Wangaet al., 2016), and MTCT is becoming a major contributor to HIV burden in SSA countries (Jayleen et al., 2016). UNAIDS (2014)

PMTCT intervention in SSA countries since 2010. PMTCT services are designed for maternal health HIV diagnosis, treatment, information and education on maternal health and delivery care and breastfeeding among pregnant women in order to reduce new HIV infection in infants (Townsen et al., 2008), and the importance of antenatal HIV testing is more apparent in SSA given that the region is characterised by high mother to child HIV infection.

In spite of the benefits accrued from PMTCT intervention in reducing HIV infection in infants, SSA remained the region with the largest proportion of MTCT (UNAIDS, 2014). Lack of and access to ANC (Alam et al., 2015; Fotso et al., 2009), low uptake of HIV testing among pregnant women (Murugugi et al., 2008; Jayleen et al., 2016) HIV related stigma (Church et al., 2013; Gausset et al., 2012; Turan et al., 2012), late ANC booking (Adekanle and Isawuni, 2008; Munjoma et al., 2010) inadequate ANC visits (Rutaremwa and Kabagenyi.,2016), compounded by low socio-economic status (Ahmed et al., 2010; Car et al., 2012) have all contributed to the slow pace of PMTCT coverage and increased HIV new infection in infants.

It has been argued that ANC has a potential to increase antenatal HIV uptake and other PMCT services (Car et al., 2012). When PMTCT programs were first introduced, they were provided in stand-alone clinic programs (Church et al., 2013). However, a number of studies reported the inadequacies of this model (Kalichman and Simbayi, 2003) leading to integration of HIV and ANC services in late 2000. The rationale for integration of HIV and ANC services is to increase the coverage of PMTCT services as pregnant women who use maternal health care services can also benefit from PMTCT services and vice-versa (Obure et al., 2015; Sweeney et al., 2012). Against this backdrop, the WHO guidelines recommend routine HIV testing for all pregnancy related visits (WHO, 2014). Studies have shown that integration of HIV and ANC has the potential to reduce missed opportunities for key intervention including HIV testing, provision of ART, education in breasfeeding and adherence to treatment (Druce et al., 2006; Wiegert et al., 2014). Indeed, studies have reported an association between service integration and an increased uptake of antental HIV testing (Car et al., 2012; Hope et al., 2014). Furthermore, implementation of PMTCT as a part of ANC may help reduce stigma among HIV positive women during their routine visits to ANC (Turan et al., 2015). It is generally recommended that prenatal visits start early in the first trimester and continue at regular intervals throughout the pregnancy; as well as having antenatal HIV testing to mitigate potential mother to child HIV transmission (Asefa and Beyene, 2013). It has been argued that women who start ANC early and those who frequently attend maternal health care are more likely to have been HIV tested compared to those who initiate late or attend few visits (Sahlu et al., 2014).

While available evidence tends to support integrated services as critical in response to MTCT, there is surprisingly little evidence of increased uptake of ANC services and antenatal HIV testing. In most SSA countries uptake of ANC services including an ANC visit in the first trimester and the proportion of women who accumulate a recommended four visits to allow for HIV routine testing are still very low (UNAIDS, 2014). There is urgent need to identify HIV positive women within integrated services and enrol them in the PMTCT intervention program for treatment, disease monitoring and education on breastfeeding (Johnson et al., 2012). While uptake of ANC care has increased in the past decade, only 38% of pregnant women in SSA countries received antenatal HIV counselling and testing and the majority have late booking or infrequent ANC (WHO, UNICEF et al., 2013). Although ANC services have been visited by the majority of pregnant women at least once during pregnancy in 2015, only about 37% pregnant women attended at least four ANC visits and half of them in the second trimester in SSA (Jayleen et al., 2016). This is worrisome given that integration of HIV and RH care services has been promoted as the panacea for increasing uptake of PMCT (Kennedy et al., 2010), yet there is scant evidence in regards to increased uptake of HIV testing during pregnancy. In the literature, there is very limited and non-genelizable evidence of improved uptake of ANC in the first trimester (Alam et al., 2015; Golis et al., 2013; Houweling et al., 2007); four or more ANC visits (Asefa and Beyene, 2013); and antenatal HIV testing in SSA countries (Staveteig et al., 2013).

While it is difficult to measure the efficacy of integrated services on health outcomes, standalone studies on ANC HIV testing, timing and frequency of ANC visits have been able to shed light on the influence of predisposing socio-demographic factors. Late booking of ANC and infrequent ANC visits have been consistently linked to predisposing and enabling factors such as place of residence, age, marital status, unwanted pregnancy, religion, parity, economic status, education and access to media (Ahmed et al., 2010;Bertrand et al., 2006; Golis et al., 2013; Jayleen et al., 2016). It has been a constant finding that pregnant women living in urban areas are more likely to have antenatal HIV testing, early ANC initiation and frequent ANC services than those from rural areas (Ochako et al., 2016; Rutaremwa and Kabagenyi, 2016;Titaley et al., 2010). Due to geographical inequalities in SSA countries, Alam et al (2015) reached the conclusion that pregnant women from urban areas enjoy easier access to health care service, including information, education on HIV and resources, enabling them easier access to health care services than women from rural areas. Alam et al's assumption that low economic status is associated with low use of health care services seem to be well-founded because women with low economic status tend to reside in rural areas where access to resources, employment, economic activities, health information and education on HIV are low (Gross et al., 2012).

5.3 Factors associated with HIV and maternal healthcare: evidence from available literature

Studies investigating the influence of women's age on antenatal HIV testing, timing and frequency of ANC visit consistently found out that older women are more likely to have an antenatal HIV test (Jayleen et al., 2016; Rutaremwa and Kabagenyi, 2016), initiate ANC in the first trimester and to have at least four or more ANC visits (Achia and Mageto, 2015; Beauclair et al., 2014; Gupta et al., 2014; Magadi, Madise and Rodriques, 2001). This is natural because young women can generallybe expected to have less knowledge and experience about child birth, while older mothers may have accumulated knowledge of risks associated with child birth (Ngome and Odimegwu, 2014). Other studies have noted that older women's ANC seeking behaviour is shaped by awareness of dangers associated with HIV/AIDS; particularly that HIV can be transmitted to a child during pregnancy as well as their knowledge of existence of PMTCT (Asefa et al., 2013; Birhane et al., 2015; Byamugisha et al., 2010; Sama et al., 2017). Older women may also have better resources enabling them access to health care services (Aremu et al., 2011; Tarekegn et al., 2014). An article by Kayeyi et al (2014) reported that young and never married pregnant women are either less likely to attend ANC or to attend late and infrequently due to lack of knowledge, lack of power to take decisions, lack of money, or societal norms which may forbid teenage pregnancy. Another critical issue surrounding maternal health care among young pregnant women in SSA countries is pregnancy wantedness and acceptance by the community (Magadi et al., 2007). Studies in Zambia and Uganda showed that pregnant teenagers were more likely to experience violence from parents, be expelled from school, and to be stigmatised (Denison et al., 2008; Kabagenyi et al., 2016). The fact that antenatal care services in SSA are not often oriented to the needs of teenagers makes the health risks associated with their pregnancies and childbearing more pronounced than those of older women (Matthews et al., 2009; Renju et al., 2010). Unintended pregnancies have also been shown to be associated with low use of maternal health care such as not having ANC in the first trimester because of delay in recognising or acknowledging pregnancy (Magadi et al., 2000). There is on-going debate on the relative importance of parity as opposed to maternal age in access to HIV and ANC services. Higher order parity has been found to be strongly associated with uptake of ANC services (Ochako et al., 2011; Jayleen et al., 2016). Beaclair el at (2014) found that pregnant women were more likely to present themselves late for prenatal care and make fewer ANC visits in their second than first pregnancies.

Marital status of pregnant women is also an important factor in both HIV testing and ANC seeking behaviour. Heemelar et al (2014) have noted that married women are more likely to use ANC and HIV testing services during pregnancy if they make more decisions in the household, if they were educated and if they regularly watch television or read newspapers. This postulation agrees with Woldemicael and Tenkorang (2009) who suggested that married women may have improved risk perception and better decision-making power. It was also postulated that married women may have a lower perception of the risks of acquiring HIV infection, and thus are more likely to have adequate HIV and maternal health care, assuming that their partners would be mutually faithful (Keogh et al., 2009).

Besides background demographic factors such as parity, marital status and age, individual socio-economic characteristics such as household wealth, mother's education, religion, mass media and social status have also been found to be associated with use of HIV and ANC services in SSA countries (Ahmed et al., 2010; Gustav et al., 2014; Jayleen et al., 2016; Jesmin et al., 2013; Kevany et al., 2012). In particular, it has been argued that education and media offer women an expanded role in family decision making and control over resources (Makate and Makate, 2017).

While low economic status may be associated with rural residence, an increasing number of studies have found that pregnant women with low economic status, in general, are significantly more likely to receive poor ANC (no visit in the first trimester, antenatal HIV testing; and make less than four ANC visits), compared with pregnant women with higher economic status (Adamu, 2011; Fotso et al., 2009; Gabrysch et al., 2011). In the literature, pregnant women from wealthier households are more likely to initiate early and have more frequent ANC visits.

Several theories have been proposed to explain this association. Ahmed et al (2010) suggested that this may be due to the fact that pregnant women from wealthier households have financial ability to afford cost of health care services. A related hypothesis holds that pregnant women from wealthier households are more likely to be educated and women who are educated are more likely to have access to information and enjoy greater freedom to seek health care without seeking permission of their husbands even in situations of HIV epidemic (Fotso et al., 2009; Houweling et al., 2007). Although, not apparent, it has also been suggested that pregnant women who have low socio-economic status do not perceive many dangers to pregnancy itself, but they perceive risk happens during labour and delivery (Asefa et al., 2013). This claim seems to be supported by a qualitative study from South Africa which found that pregnant women from poor households did not perceive many hazards to pregnancy; rather they perceive risk happened during labour and child birth (Birhane et al., 2015).

Mass media is another important determinant of HIV and ANC services and it can reach a wide range of audience. Several authors have documented the role of mass media in disseminating information on health related issues through radio, television, newspapers and magazines (Gustav et al., 2014; Jesmin et al., 2013). In recent years, HIV prevention and SRH promotion information has been disseminated through posters, mobile phones and the internet (Underwood et al., 2014). Employment status has also been linked to HIV and ANC service utilisation. Studies have also noted that employment status was associated with timing of and frequency of ANC visits (Maruva et al., 2014; Kevany et al., 2012). They have noted that non-working mothers were more likely to have antenatal HIV testing; or initiate ANC early and frequent ANC more than those in employment. It has been hypothesised that employed women do not accept HIV tests because they view positive tests as devastating, and this may lead to a number of consequences including loss of job, stigma and discrimination and difficulties in getting an insurance policy at work (Chevo and Bhatasara, 2012).

In the literature, a number of perceived risk/need factors relating to HIV sero-status, knowing someone who died of HIV, perceived risk of infection, perceived lack of confidentiality, HIV awareness and stigma were also found to be associated with HIV and ANC services (Asefa et al. 2013; Church et al., 2013; Killam et al., 2010; Sahlu et al., 2014; Sama et al., 2017, Tenthani et al., 2015; Turan et al., 2015). Some studies seem to suggest that being personally affected by HIV, especially having seen a close person die of HIV, may be associated with higher

perceived risk of HIV infection and perception of personal vulnerability leads to adopting protective behaviour (Yeatman et al., 2009; Sahlu et al., 204). Therefore, being HIV positive or knowing someone who died of AIDS is expected to influence utilisation of HIV and maternal health services as a protective measure against mother to child HIV transmission (Maruva et al., 2014). As Kalembo et al (2014) alluded, depending on the past sexual behaviour and knowledge of how HIV is transmitted, individuals construct their own probabilities of being infected-affecting their health seeking behaviour.

Although it is expected that integration of HIV and maternal health will potentially reduce HIV-related stigma and discrimination, there is considerable literature showing that pregnant women are still vulnerable to HIV-related stigma and discrimination at home, in the health care centres and communities after diagnosis (Theuring et al., 2016). It has been suggested that fears and experiences of stigma or discrimination are potential reasons why some pregnant women avoid HIV and maternity health care services altogether i.e refusal of antenatal HIV testing, delay and infrequent ANC visits (Buzdugan et al., 2015). It has been shown in multiple studies from SSA that men can positively influence female health decision-making and utilization of HIV and ANC services, for example the uptake of HIV testing and PMTCT measures (Farquhar et al., 2004). One of the drawbacks to HIV and ANC service utilisation relates to low couple attendance in ANC and HIV testing services (Byamugisha et al., 2011; Kalembo et al., 2013; Mohlala et al., 2011; Theuring et al., 2016).Fewer males are involved in maternal health care in most SSA countries, and a pregnant woman is usually the first family member to be tested for HIV and Yargawa and Bee (2015) have noted that she may be blamed for bringing HIV if she discloses her HIV-positive status to her partner.

Antenatal HIV testing requirements may also increase the potential for involuntary unwanted disclosure of HIV status, heightening fears of experiencing stigma and discrimination in health facilities (Turan et al., 2009). Multiple studies from various countries report that perceived lack of confidentiality decreases the likelihood of HIV and maternal health care service utilisation (Angotti et al., 2009; Hardon et al., 2012; Mashuba and Hemalata, 2015; Mucheto et al., 2011; Sahlu et al., 2014). This is because, in most SSA countries, hospital records and information are recorded in hand-held notes, and the notes may be seen by friends and family members, thereby inadvertently discouraging pregnant women from utilising HIV and maternal health integrated services (Demisse et al., 2009; Washington et al., 2015). While it has yet to be fully

recognised, unmitigated HIV-related stigma and discrimination present a missed opportunity to identify HIV positive women and enrol them into PMTCT.

5.4 HIV and maternal healthcare in Zimbabwe

Zimbabwe with its high maternal mortality ratio (MMR) of 960 deaths per 100,000 live births, is among the top 40 countries in the world with the highest maternal mortality (ZIMSTAT and ICFI, 2015); and child morbidity as a result of mother to child HIV transmission (WHO et al 2012). An estimated 790,000 women are living with HIV in Zimbabwe, and mother-to-child transmission accounts for 28% of all new HIV infections in children aged 0-14 years (ZIMSTAT, 2015). Although, the number of new infections in this age group has fallen from 12,000 in 2010 to 4,900 in 2015, mother to child HIV transmission is still unacceptably high (ZIMSTAT, 2015). To avert new infections in infants, detection of maternal infection early in pregnancy through HIV testing and access to antiretroviral prophylaxis is crucial (Buzdugan el a., 2015). To ensure the prevention of mother to child HIV transmission (PMTCT), Zimbabwe is scaling up integration of HIV programs into existing family planning, maternal and child health care services, through the Focussed Antenatal Care (FAC) program (ZNASP 2011-2015). The rationale is to increase the availability and access to HIV testing services among pregnant women through antenatal care services (Wiegert et al., 2014). The FAC framework stipulates that; pregnant women should have at least four ANC visits, with the first visit being in the first trimister (WHO, 2014, Titaley et al 2010). The purpose of an ANC visit in the first trimister includes early screening of infectious dieseaes such as HIV/AIDS for the positive mothers to begin treatment to prevent mother to child HIV transmission during pregnancy or delivery (Heemelar et al 2014). As discussed in chapter one, stand alone models of health care services are insufficient in increasing uptake of HIV testing coverage and PMCT services (Fetene et al., 2010). This inadequacy led the WHO to establish guidelines recommending routine HIV testing among all pregnant women visiting health facilities for all pregnancy related care in SSA. As a result of this initiative, HIV diagnosis is often offered to pregnant women during their first ANC visit, which is usually part of an integrated maternal and child health service (Maruva et al., 2014) in Zimbabwe.

Although, the 2010/11 ZDHS has reported that 90% of pregnant women in Zimbabwe had at least one antenatal care (ANC) visit, only 59 % had both received counselling and accepted an HIV test as part of ANC care. In addition, 67 % had attended the recommended four or more

ANC visits, and 19 % had an ANC visit in the 1st trimester. The reported rates for maternal health care service utilisation in Zimbabwe remain far too short of the 90% targets adopted in the country (ZIMSTAT, 2015). In addition, the attainment of the United Nations global plan to reduce the mother to child HIV transmission rate to 5%, and decreasing the paediatric HIV infection by 90%, as well as increasing ANC coverage to 90% by 2020 (ZIMSTAT, 2015), can be hard to obtain. Despite the overall increase in ANC coverage and uptake, antenatal HIV testing remains low and unevenly distributed and MCTC of infants remains unacceptably high in Zimbabwe. The low uptake of PMCT greatly undermines the aim of reducing the MTCT rate to 5% by 2020 in Zimbabwe. It is therefore imperative to identify the factors responsible for delay in ANC visits, use and non-use of HIV testing offered as part of antenatal care and frequency of ANC visits in order to fast track the attainment of future targets such the 90-90-90 targets by 2020 and 95-95-95 targets by 2030, each of the 90s representing the anticipated coverage on HIV testing, treatment of the positive cases and viral suppression respectively (UNAIDS, 2014). Knowledge of enabling and inhibiting factors may help to inform the Zimbabwean government and other concerned local and international stakeholders to devise programmes and policies that will help to scale up the antenatal service utilization and thereby enhance the PMTCT programmes in the country. However, without sufficient information on which population groups are less likely to use the services, the broader aim to increase coverage of HIV testing and enrolling pregnant women who are found to be HIV position in PMTCT program will remain elusive and new HIV infections in infants will continue in Zimbabwe. Therefore, understanding and addressing barriers against ANC and HIV testing could significantly contribute to the main goal of increasing uptake of PMCT services.

Previous population studies on HIV and maternal health services have largely focused on individual level predisposing factors in specific components of integrated care services in Zimbabwe (Makate and Makate, 2016; Maruva et al., 2014;Muchabaiwa et al., 2012). Others are community-based studies-for example, (Kevany et al., 2012) have evaluated influence of household income on demand for maternal health care in rural areas; and (Chirwa et al., 2013) have focused on the impact of user fees on access to maternal health care services. The overall drawback of these studies has been the use of single level analytical techniques that ignore clustering and the hierarchical structure of data for individuals living in different communities (Aremu et al., 2011). There is evidence showing that women's access to HIV and maternal health is not only related to individual choice, but it is also to a large extent depends on social

contexts that influence individual's HIV and maternal health seeking behaviour (Olalekan and Uthman, 2010; Lugalla et al., 2012). The context within which individuals live matters because the broader environment not only provides opportunities to learn and reassurance, but also generates general norms regarding activities within the community (O'Meara et al., 2013). Quantitative studies have shown that HIV factors such as awareness and stigma can be influenced by social grouping in the community (Chiu et al., 2008; Lugalla et al., 2012; Olalekan et al., 2010). Thus, multilevel modelling can be used to account for factors at the individual and community levels simultaneously and provide a more robust understanding of the factors associated with HIV and maternal health service utilisation in Zimbabwe (Rasbash, 2004). A better understanding of community level factors that influence HIV and ANC service utilisation would contribute to the large effort to design multilevel, effective and sustainable interventions (Underwood et al., 2014). This study therefore aims to address this evidence gap by explicitly examining both individual and contextual level determinants of HIV and ANC services, by implementing a multilevel methodological approach (Magadi and Desta, 2011). As more women become infected, mother-to-child transmission (MTCT) of HIV continues to be a major challenge in Zimbabwe underscoring the need to understand the correlates of each of the components of HIV and ANC integrated services from truly nationally representative data.

Using the adapted Andersen's framework discussed in chapter 2 and in section 5.1 this chapter seeks to explore predisposing, enabling and perceived risk/need factors that may influence uptake of HIV and ANC services within the integrated model of care. The objectives of the study are:

- To identify individual and community socio-economic, demographic and HIV factors associated with use of HIV and ANC integrated services i.e (a) timing of first ANC visit, (b) antenatal HIV testing; and (c) four or more ANC visits among pregnant women in Zimbabwe.
- To examine national variations in HIV and ANC services utilisation.

The study aims to provide an overall picture of the determinants of HIV and ANC integrated services, which is useful for informing the design of multilevel HIV intervention programs.

5.5 Analytical framework

The framework of Andersen discussed in chapter 2 guides selection of factors that could affect utilisation of both HIV and maternal health care services. The model is conceptualized based on the predisposition, enablement and need for health care use (Andersen, 1995). Fig 5.1 and Table 5.2 present the modified framework and selected variables for this chapter. Based on literature reviewed, a number of predisposing factors have been selected, ranging from individual demographic factors (e.g age. marital status, religion, place of residence, year of survey, parity, education, media exposure) to community HIV factors that predispose pregnant women (i.e HIV awareness, stigma and prevalence). The enabling factors include the logistic and resource aspects of obtaining care which comprise of household income, woman's bargaining power, employment status, availability and accessibility of health care facilities. The HIV and psychological aspects of enabling factors include (HIV stigma, awareness, disclosure and confidential concern). The need factors relate to both the perceived and evaluated assessment of one's health status which may inform the need for seeking health care (Lepine et al., 2014). In line with other studies, the need factors in this chapter are conceptualized as the need for HIV testing (Demisse et al., 2009; Lepine et al., 2014). HIV status and knowing someone who died of HIV are considered as a proxy for the perceived risk need. Based on the literature reviewed, no previous study has applied this conceptual framework to examine the determinants of HIV and maternal health integrated services in Zimbabwe. However, the framework has been widely applied on survey data and in various settings elsewhere (Rutaremwa and Kabagenyi, 2016). The conceptual framework in figure 5.1 below shows three sets of factors that are linked to the outcomes of this study.





Predisposing factors	Enabling factors	Need/percived risk factors
Individual factors	Resources and autonomy	Ever been tested for HIV
Year of survey	Household wealth	Knows someone died of HIV
Place of residence	Employment status	HIV status
Women's age	Accessibility problems	
Marital status	Women decision making	
Religion	HIV and Psychological	
	influence	
Parity	Confidentiality concern	
Media exposure	HIV awareness	
Mother's education	HIV stigma	
Wanted Child		
Community influence		
HIV prevalence		
HIV related stigma		
HIV awareness		

 Table 5-1: List of variables related to each set of variables shown in Figure 5.1 above

5.6 Methods and data

5.6.1 Data source

The chapter is based on secondary analysis of existing data (ZDHS 2005/6 and 2010/11) surveys. The availability of HIV test data that can be linked to individual-level survey data provides a unique opportunity for a population based study of factors associated with the HIV/AIDS epidemic in Zimbabwe. For the purpose of analysis of this chapter, data from males are excluded. A total number of 18078 women were interviewed in the two surveys. To construct the sample for analysis from merged ZDHS datasets, the selection was initially restricted to all women who had given birth during the past five years (DHS code; V208>0) preceding the surveys. This way of selection excluded women who did not give birth within the specified period and it generated an unweighted sample of 8471 women who gave birth to a total of 10620 children. This selection procedure was adopted in order to ensure compatibility with other related studies (Achia and Mageto, 2015; Ochako et al., 2011). To avoid clustering of multiple births and to reduce the likelihood of recall bias on sensitive questions (Uthman et al., 2015), the sample was further limited to the last birth in the five years preceding the surveys (DHS code; BIDX\$01 = 1). This restriction gave a final sample size of 8471 women with at least one live birth in the five years preceding the surveys for the analysis. Of these, (92.4%, n=7879) had at least one antenatal care visit, whilst (7.6%, n=648) did not have, representing near universal access to antenatal care service (table of percentages not shown). From this sample, dependent and independent variables of interest were extracted. The following section deals with the study outcomes of this chapter.

5.6.2 Measures of outcome variables

Based on literature reviewed, there are three problematic maternal health care indicators within integrated HIV and health services in Zimbabwe (Maruva et al., 2014; Mashumba and Hemalata 2015; Muchabaiwa et al., 2012; Mucheto et al., 2009). These are whether the pregnant woman had; i) late ANC booking; ii) lack of antenatal HIV testing; and, iii) infrequent ANC visits.

Timing of antenatal care

The first measure of outcome relates to whether or not the pregnant woman had her first ANC visit in the first trimester during the last pregnancy. The measure was based on a recode of DHS variable M13\$M. Definition of first trimester ANC visit (All mothers who had an ANC visit between 1 to 13 weeks of pregnancy were coded '1' and having no ANC visit or having a first ANC visit during the second or third trimester of pregnancy were coded '0'. The coding categorisation adopted is consistent with a previous study (Manzi et al., 2014) which defined 'delayed ANC' as having no ANC visit or having the first ANC visit during the second trimester. The standard definition of 'first trimester' is first three months in pregnancy, which is up to week 12 or 13 based on the recommended maternal and perinatal health care guidelines (WHO, 2012).

HIV testing during ANC

The second outcome variable is Antenatal HIV testing. In the DHS, pregnant women who attended ANC for the most recent birth responded to these questions; (a) were you tested for AIDS virus during any of the ANC visits?, if yes,(b)did you get results of the AIDS test?. Women who answered 'yes' to both questions and those who answered 'yes' to question (a) are classified as having been tested for HIV during ANC. Women who were not tested answered 'no' to both questions (Staveteg et al., 2013). The measure was based on a (DHS code; V840-1) and women who were tested for HIV are coded '1' and those who were not are coded '0', indicating a dichotomous response variable.

Number of antenatal care visits

The third outcome measure was constructed based on whether or not the mother had at least 4 or more ANC visits during the last pregnancy. The number of antenatal care visits a woman attends as shown in previous studies is important for allowing routine HIV testing and monitoring viral load in the blood among HIV positive pregnant women. The lack of retesting during pregnancy represents a missed opportunity to identify women who have recently acquired HIV infection and have an increased risk of MTCT (Sama et al., 2017). The measure of frequency of ANC visits was based on a recode of variable M14, in which women who had

at least 4 ANC visits are coded '1', and women who had less than 4ANC visits and those who reported not having ANC visit coded as '0'. The ZDHS asked the respondents: How many times did you receive antenatal care during this pregnancy? The responses enabled a classification of the number of times antenatal care was attended. Manzi et al (2014) have argued that a total number of visits may better reflect coverage rather than quality of services, which is the goal of this chapter.

5.6.3 Key independent variables

Based on the conceptual framework, literature reviewed and data available, a range of independent variables were selected for this chapter. The variables are classified into individual and community-level predisposing, enabling and perceived/need factors. All variables included in the models are from the previous chapter except for HIV related factors. However, categories of some variables in the individual and enabling categories had fewer cases; and to ease comparison, these were collapsed and re-coded. The recoded variables were educational attainment, women's age, marital status, religion, media exposure and household wealth. The coding categories were consistent with the categorisation of the same variables in previous studies on HIV and maternal health care in Zimbabwe (Makate and Makate, 2017; Maruva et al., 2011; Muchabaiwa et al., 2012; Ngome and Odimegwu, 2014). For the original coding refer to ZDHS questionnaires (ICF International 2011). Individual predisposing and enabling factors such as parity, wanted child, and women's autonomy were not included in chapter 4, but were controlled for in this chapter. The additional factors were defined as follows:

(a) Parity

The variable parity V201 (total number of children ever born) was recoded into three categories with 1 equals to lowest birth order and 3 plus as the highest birth order. Birth order is expected to have a u-shaped association with HIV and maternal service utilisation since the first births and higher birth orders have an increased risk of death during pregnancy (Fotso et al., 2009).

(b) Wanted child

The variable (V367) provides information on whether or not the woman wanted the last child. The wanted last child variable was recoded into three categories: a) wanted child then -1, b) wanted later -2, and c) not wanted -0.

(c)Women's autonomy

Also, consistent with other studies (Matthews and Gubhaju, 2004), the women's decision making variable was derived from five questions (V743A-V743E) asking about decision making. The five questions were; final say on health care (V743A)' final say on large purchases (C743B), final say on daily needs (C742C), final say on visitors (V743D); and final say on food to be cooked (V743E). The mother was coded as 1 - (makes decision alone), if they were the sole decision maker; 2 - (jointly) if made the decision jointly with a partner or another person; or 0 - (husband only) if they did not contribute to decision. A single variable problem of access was created from a combination of five questions (V467A through to V467E) asking about the problems women face when seeking health care services. The variable was coded '0' for no perceived problems and '1' for perceived problems.

5.6.4 HIV factors

The HIV variables were selected carefully based on previous literature (Sambisa, 2010; Lepine et al., 2014; Mashumba and Hemalata 2015). The HIV-related stigma and comprehensive HIV awareness variables are not available in the ZDHS dataset. The literature reviewed which used the HIV related stigma and comprehensive awareness has used an additive procedure to generate a single variable from a set of questions from the survey (Magadi, 2016).

(i) Comprehensive HIV awareness:Comprehensive HIV awareness about transmission of HIV was measured based on the following: If a pregnant woman correctly identified the four main modes of transmission of HIV (sexually, blood transfusion, infected sharps and MTCT), if she rejects common local misconceptions (i.e. HIV/AIDS can be transmitted through mosquito bites, supernatural means, and by physical contact e.g. hugging, handshakes); and if she is aware that a healthy looking person can have HIV. Each correct response to the above questions was scored 1 point, otherwise 0. In total, the maximum obtainable score is 8 points. Participants with 0-4 correct responses were considered to have a low HIV awareness; 5-6 medium knowledge on transmission of HIV; and 7-8 were considered to have high level of comprehensive HIV awareness. Therefore, the variable was classified into three categories: low, medium and high.
(ii) **HIV related stigma:** This referred to the respondent's attitude towards HIV developed from five questions of HIV related stigma, namely; if the respondent' would buy vegetables from a vendor with HIV', 'can care for relative with HIV', 'would want an HIV-positive teacher to continue teaching', and 'would want others to know if a family member became infected with HIV'. The variable was classified into three categories: low: if not experienced to any of the stigma indicators, medium: if experienced 1-2 of the stigma indicators and high stigma: if experienced 3-4 stigma indicators.

Other HIV variables like 'HIV status, knowing someone who died of HIV, disclosure concern and previous HIV testing' adopted the original coding from the DHS survey. The previous HIV test variable was not used among the determinants of the antenatal HIV testing outcome because of high multicolinearity between the two variables; however, it was among the determinants of number of ANC and timing of visits. The binary previous HIV test variable relates to whether the woman had ever been tested for HIV before. The variable was coded '1' if ever been tested and '0' for otherwise. The 'disclosure and confidential concern' was derived directly from the question 'would you want others to know if a family member became infected with HIV'. Similarly, the 'knowing someone with HIV/AIDS variable was also derived directly from the DHS question. Knowing someone who died of AIDS is hypothesized to be positively associated with HIV testing uptake, since women would be aware of the potential risks of HIV transmission and benefits of testing (Godif et al., 2015). At the community level, variables relating to HIV related stigma, HIV prevalence and HIV awareness were derived from relevant individual level data and were created and grand mean centred during the multivariate modelling.

5.6.5 Methods of analysis

Like in chapter 4, this chapter adopts a multivariate binary logistic regression analysis, with community at level 2 and last birth in the last 5 years at level 1 for all three outcomes.

The logit of the probability of outcome can be modelled as follows: $f(\pi_{ij}) = \text{Log} [\pi_{ij}/(1-\pi_{ij})] = \beta_0 + \beta_1 X_{1ij} + \beta_2 C_{2ij+\mu_{0j}}$

Var $(u_{0j}) = \sigma^2 u_0$

Where (π_{ij}) is the probability of an outcome for a pregnant women in the j_{th} community. The intercept β_0 is shared by all communities; β s denoted the fixed coefficients, where X and C are a vector of covariates at the individual and community levels variables, respectively; while $\mu_{0j} \sim N(0, \sigma^2 j)$ shows the random effects for the j_{th} community. The random intercept is shared by all individuals from the same community (defined by primary sampling unit) and serves as an indirect control for community factors not included in the models that may affect the HIV and maternal health integrated service utilisation.

Before allowing for any explanatory variables in the models, it was crucial to know how much between community variation was there in the propensity for utilisation of HIV and maternal health services in the study. To assess this, we look at the estimated value of $\sigma^2_{u\,0}$, which is the variance of the u_{0j} terms. We use a threshold model approach which measures proportions at group level:

 $P{=}{\sigma_U}^2{\!/}\left({\sigma_U}^2+{\sigma_\epsilon}^2\right)$

Where σ_U^2 is the total variance at community level, and σ_{ε}^2 is the total variance at individual level. For the multilevel logistic regression model, the level-1 residuals are assumed to have a standard logistic distribution with mean zero and variance $\sigma_{\varepsilon}^2 = \pi^2 / 3$ where π is the constant 3.14..., Thus: $\sigma_U^2 / \sigma_U^2 + 3.29$ (Goldstein, 2003).

Proportional change in variance was derived from:

$PCV = V_A - V_B / V_A * 100$

Where: V_A =variance of the initial model, V_B =variance of the succeeding model.

The Deviance information criterion (DIC) was used as measure of the goodness fit of the models. A lower DIC value in subsequent models indicates a best fit of the model (Aremu et al., 2011).

5.7 Results

The results are presented based on univarite, bivariate and multivariate logistic regression analysis. To gain an initial sense of the structure of variables, a univariate description of the study respondents was done followed by bivariate analysis. Then, the parameter estimates were determined by a multilevel multivariate logistic regression analysis. The percentage distribution and proportions in outcomes found here cannot be directly compared with the estimates in the ZDHS final reports. This is because the denominators which were used to produce estimates in the ZDHS reports included multiple births within the 5 years preceding the surveys rather than the last birth in the ZDHS used here. As stated earlier, factors that determine the utilization of HIV and maternal health integrated services in this chapter are organized into three categories as predisposing, enabling and perceived need factors.

5.7.1 Proportions of mothers who gave birth in the last 5 years preceding the surveys

Table 5.2 shows the proportions of mothers who gave birth within the last five years preceding the ZDHS 2005/6 and 2010/11 surveys used in this chapter. The number of respondents totalled 4074 in 2005/6 and 4397 in 2010/11. Of these, on average 75% gave birth to one child in both years. About a quarter of mothers gave birth to two of more children within the same periods.

Table 5-2: Percentages	of mothers	who gave	births in	the	last 5	years	preceding	the
surveys, 2005-6; 2010-11	ZDHS							

	Prop	Proportions of births in the last 5 years (n=8471)									
	ZDH	IS 2005-6	ZDHS 2010-11								
No. of births	% ^a	n ^b	% ^a	n ^b							
1	74.8	3007	75.6	3315							
2+	25.2	1067	24.4	1082							
Total	100	4074	100	4397							

^aWeighted; ^bunweighted cases.

5.7.2 Description of Study Population by Background predisposing factors

This section describes the general characteristics of women who gave birth to their last child in the two samples. As shown in table 5.3, about 69% of births occurred in rural areas, compared to 31% in urban areas for both surveys. Across the age groups, most of the women who gave birth were within the age range of 25 to 34 years of age. Over 80% and 60% were married or had secondary education respectively. Fewer births occurred among single women i.e never married, divorced or widowed. The majority were HIV negative, made household decisions jointly with their partners, unemployed, had wanted a child then, and were from poorer households, with multi- parity and low media exposure. There was a slight increase in the proportion of HIV positive women who gave birth from 12.9% in 2005/6 to 14.3% in 2010/11. Almost two thirds of women perceived that money is a big problem in accessing medical care, while more than one quarter perceived that transport and distance were big problems. Less than half of the women (45%) were in the two poorest wealth quintiles, while the lowest proportion was in the richest wealth quintile. More than two thirds (71%) of women in the sample were from large households, while more than three quarters (89%) were from male headed households.

We whether	ZDH	HS 2005-6	ZDHS	ZDHS 2010-11		
variable	% ^a	n ^b	% ^a	n ^b		
Residence						
Urban	31.3	1136	31.2	1339		
Rural	68.7	2938	68.8	3058		
Age groups						
15-24	39.2	154	36.6	1592		
25-34	45.0	1813	46.5	2073		
35+	15.8	672	16.9	732		
Marital status						
Never married	5	217	4.1	247		
Married	81.4	3334	85.1	3669		
Widowed	4.9	19	2.7	130		
Divorced	8.7	332	8.1	351		
Women's education						
No/Primary	39.3	1648	32.8	1480		
Secondary/Higher	60.7	2426	67.2	2917		
Husband's education						
No/Primary	66.1	2653	74.2	3123		
Secondary/Higher	33.9	1421	25.8	1274		

 Table 5-3: Proportion of last birth in the last 5 years by selected socio-demographic factors

Continued

Religion				
Catholic/others	28.0	1179	23.0	1107
Protestant	20.7	830	13.7	597
Pentecoastal	16.6	646	19.2	799
Apostolic	34.7	1419	44.1	1894
Employment status				
No	63.5	2559	68	2889
Yes	36.5	1515	36	1508
Household Wealth				
Poor	63.5	2584	63.2	2692
Rich	20.3	842	20.9	965
Richest	16.2	648	15.9	740
Wanted child				
Then	63.2	2623	66.7	2921
Later	20.2	879	20.6	1134
Not Wanted	16.6	564	12.6	340
Parity				
1	30.1	1202	29.5	1301
2	25.6	1033	27.9	1225
3+	44.0	1839	42.6	1871
Decision-making				
Huband alone	23.7	966	30.9	1359
Women alone	29.9	1219	11.1	490
Jointly	46.4	1889	57.9	2548
Media exposure				
Low	37.1	1567	31.9	1431
Medium	29.4	1192	49.7	2138
High	33.4	1315	18.8	828
HIV sero-status				
Negative	87.1	3549	85.7	3767
Positive	12.9	525	14.3	630

Table 5.3 Continued

^aWeighted; ^bunweighted cases.

5.7.3 Percentage distribution of women who had HIV test, had 4 ANC visits and ANC in the first trimester

Delivery of integrated HIV and maternal health services in Zimbabwe has resulted in a substantial increase in antenatal HIV testing uptake and greater improvement in frequency of ANC visits, but not for appropriate timing of ANC. As shown in table 5.4 and figure 5.2, from the 2005/6 to 2010/11 survey periods, ANC HIV testing uptake increased dramatically

(percentage differences, 26.5%); and those who had four or more ANC visits (percentage difference 4.2%) compared with their counterparts. Regrettably, there was a marked decrease in the timing of ANC visits, with percentage differences of -10.6% of proportion of women reporting not having ANC in the first trimester between the two surveys. Although about 56% of women were reported to have had antenatal HIV testing in 2010/11, four or more ANC visits (44.5%) and ANC in the first trimester (37.5%) were relatively lower.

 Table 5-4: Percentage distribution of HIV and reproductive health care outcomes in the samples, 2005/6 and 2010/11 ZDHS

Outcome verichle	ZDHS	2005-6	ZDHS 2010-11		
Outcome variable	% ^a	n ^b	% ^a	n ^b	
ANC HIV TEST					
Yes	29.4	1195	55.9	2494	
No	70.6	2879	44.1	1903	
Had 4+ ANC visits					
Yes	40.3	1721	44.5	1993	
No	59.7	2353	55.5	2344	
Timing of ANC visits					
1 st trimester	48.1	1878	37.5	1610	
Late or no visit	51.9	2196	62.5	2727	
Total	100	4074	100	4337	

^aWeighted;^bunweighted cases.

Figure 5-2 : Percentage distribution of HIV and reproductive health care outcomes in the study population by survey year



Source: Zimbabwe Demography and Health Surveys (2005-6; 2010-11)

5.7.3.1 Bivariate analysis of timing of ANC, antenatal HIV testing and frequency of ANC visits by predisposing factors

Table 5.5 presents the bivariate association between selected individual predisposing factors and HIV and maternal health care indicators (antenatal HIV testing; timing of; and number of ANC visits). The results show that pregnant women in rural areas have better timing of; and have an increased number of ANC visits than urban pregnant women. About 43.5% (p<0.000) of pregnant women in the rural areas had their first ANC visit in the first trimester and 44.4% had at least four ANC visits, significantly higher proportions compared with 40.1% and 38.2% in urban areas, respectively. While, urban women were late initiators of ANC visits and infrequent visitors, they were more likely to be acceptors of antenatal HIV testing. About 50% (p<0.000) of urban women were offered and accepted HIV testing compared with 40.2% in rural areas. Given that HIV testing during pregnancy is low in rural areas, it means that a number of new babies were exposed to HIV during pregnancy in these areas. Although the proportions of access to maternal health care are lower than 50% across all age groups, women under the age of 24 years are late attendees of ANC, but have higher proportions having four or more ANC visits and acceptance of HIV testing during pregnancy than women in other age groups. Antenatal HIV testing was much lower among pregnant women over 35 years. The proportions with antenatal HIV testing and that of women who had at least four ANC visits were higher among married and never married women. Never married pregnant women exhibited the lowest proportions having an ANC visit in the first trimester, whilst those who are widowed are the earliest in initiators of ANC care. The formerly married i.e widowed and divorcee women are also the least acceptors of antenatal HIV testing.

The majority of women who had antenatal HIV testing and those who initiated ANC in the first trimester had secondary or higher educational attainment, while this was not the same for number of ANC visits. A significant proportion of pregnant women with primary and lower educational attainment were more likely to have four or more ANC visits than those with secondary or higher educational attainment (46.5% versus 40.2%). Regarding religious affiliation, the differences were not statistically significant for ANC timing and antenatal HIV testing, but women affiliated with apostolic faith groups are late attendees to antenatal care. Observed also is a higher level of ANC visits in the first trimester among pregnant women who reported that they wanted the child then, compared to later or not wanted. The results suggest no statistical difference in antenatal HIV testing and number of ANC visits among women who reported wanting the child then, later or not wanted. HIV and maternal health care service utilisation, increases with an increase in media access for antenatal HIV testing and timing of ANC, but this observation was not so for the number of ANC visits. The majority of women who had four and more ANC visits are in a category of women who have low access to media. Birth order parity is not associated with timing and number of ANC visits, but antenatal HIV testing was found to be more common among women with low birth order parity of one pregnancy.

Variables % ANC 1st *P*-value % HIV-*P-value* % 4+ *P*-value All cases^b ANC trimister^a ANC tests^a visits^a .001 Residence .001 .001 40.6 49.6 38.2 2475 Urban 43.5 40.2 44.4 5996 Rural .002 .001 .003 Age groups 15-24 46.9 39.2 48.3 3176 25-29 44.2 42.6 43.2 2338 30-34 45.7 43.3 40.8 1548 1409 35 +41.4 32.3 41.6 Marital status .002 .004 .001 Never married 34.2 44.8 48.8 464 Married 42.7 44.1 42.6 7003 Widowed 30 40.4 321 48 43.7 39 39.5 Divorced 683 Women's education .001 .000 .001 40.8 33.3 46.5 3128 No/Primary 48.7 5343 Secondary/Higher 43.6 40.2 .021 .011 .029 Religion Catholic/others* 40.2 45.4 43.1 2286 43.9 44.8 43.2 1427 Protestant 45.7 Pentecoastal 45.0 40.2 1445 Apostolic 39.3 43.6 44.7 3313 Wanted child .001 .332 .011 Then 44.6 43.6 42.7 5544 37.3 43.4 45.6 2013 Later 37.8 40.6 47.8 914 Not Wanted Parity .248 .001 .723 43.6 48.0 43.2 2503 1 2 41.4 47.0 44.4 2258 3+ 37.7 44.1 3710 41.5 Media exposure .001 .001 .001 38.2 35.9 49.2 2998 Low Mediun 42.0 48.6 42.7 3330 High 47.4 45.1 38.6 2143

Table 5-5: Bivariate analysis of timing of ANC, antenatal HIV testing and frequency ofANCvisits by predisposing factors, 2005/6 and 2010/11 ZDHS

^aWeighted; ^bunweighted cases.

5.7.3.2 Bivariate analysis of the use of antenatal HIV testing, timing and number of ANC visits by enabling and need factors

The first part of Table 5.6 describes bivariate analysis of the use of antenatal HIV testing services, timing and number of ANC visits by enabling and need factors (HIV and Socio-economic factors). The results show that previous HIV testing was associated with timing of ANC and HIV testing during ANC care, but not with number of visits. About 44.5% (p<0.000) of women who had been previous tested for HIV had their first ANC visit in the first trimester, compared with 38.9% of their counterparts. Over three quarters 75.5% (p<0.000) of women who had a previous HIV test were also tested for HIV during pregnancy, compared with only 24.5% of their counterparts. HIV sero-status was significantly associated with antenatal HIV testing, whereas, this was not so for timing and frequency of ANC. Although not so pronounced, four or more ANC visits were common among pregnant women with low HIV awareness.

HIV stigma was significantly associated with four or more ANC visits and antenatal HIV testing, but not with timing of ANC visits. Paradoxically, the majority of pregnant women who had antenatal HIV testing were in a category of individuals with high stigma 74.8% (p>.000), while four or more ANC visits was more pronounced among women with low to medium stigma. It is also noted that knowing someone who died due to HIV was an important factor of timing of ANC and having a subsequent antenatal HIV test. Fewer pregnant women who reported knowing someone who died of HIV had an ANC visit in the first trimester (40.8% versus 44.9%, p<0.000), compared with those who reported not knowing someone who died due to HIV. A large proportion of those who reported knowing someone who died due to HIV, 72.2% (p<0.000) were tested for HIV during antenatal care visits, compared with their counterparts.

The second part of Table 5.6, describes bivariate analysis of the use of antenatal HIV testing services, timing and number of ANC visits by socio and economic enabling factors. The results show that employment status was significantly associated with HIV and maternal care service utilisation. Employed women have a higher proportion of initiating ANC in the first trimester (45.2% versus 40.2%), but are less likely to have antenatal HIV test (40.6% versus 44.7%,

p<0.000) and four or more ANC visits (42.7% versus 44.6%, p<0.000) than unemployed women. The majority of pregnant women who had antenatal HIV testing and those who initiated ANC in the first trimester were from wealthier households. However, four or more ANC visits were more common among women from poorer households compared with their counterparts. The results also show that decision-making in the household only matters in respect of HIV testing during ANC. HIV testing was more among pregnant women who jointly make decisions with their husbands/partners or when their husbands make decision alone. Women who reported having no big problem on physical access to health care facility were more likely to have their ANC visit in the first trimester or have an antenatal HIV test. Surprisingly, the majority of the women who reported having a big problem in accessing health care were more likely to have four or more ANC visits.

ANC visits by enabling and need factors, 2005/6 and 2010/11 ZDHS										
% ANC-	P-value	% HIV-	P-value	% 4+	P-value	All				
1st		ANC		ANC		cases ^b				
trimister ^a		tests ^a		visits ^a						
HIV and enabling and perceived risk/need factors										
IV test	.001		.001		.410					
38.9		24.5		43.7		3550				
44.5		75.5		44.0		4921				
tatus	.015		.003		.450					
41.5		42.5		43.9		7316				
45.1		47.0		43.7		1155				
ness	.447		.001		.003					
44.4		57.2		46.3		2180				
43.0		64.4		44.3		3765				
47.7		55.1		39.6		2526				
e with HIV	.000		.001		.240					
40.8		72.2		44.2		2476				
44.9		36.5		43.2		5995				
	by enabling % ANC-1st 1st trimister ^a enabling and IV test 38.9 44.5 tatus 41.5 45.1 ness 44.4 43.0 47.7 with HIV 40.8 44.9	by enabling and need % ANC- 1st trimister ^a P-value 1st trimister ^a P-value enabling and perceived 01 38.9 .001 38.9 .015 44.5 .015 41.5 .015 45.1 .015 44.4 .43.0 47.7 .000 40.8 .44.9	by enabling and need factors, 20 % ANC- 1st trimister ^a P-value % HIV- ANC tests ^a enabling and perceived risk/need IV test .001 38.9 24.5 44.5 75.5 tatus .015 41.5 42.5 45.1 47.0 ness .447 44.4 57.2 43.0 64.4 47.7 55.1 with HIV .000 40.8 72.2 44.9 36.5	by enabling and need factors, 2005/6 and 2 $\%$ ANC- 1st trimisteraP-value NC testsa $\%$ HIV- ANC testsaP-valueIV test.001.00138.924.5.00144.575.5.00341.542.5.00341.542.5.001anss.447.00144.457.243.064.447.755.1with HIV.00040.872.244.936.5	by enabling and need factors, 2005/6 and 2010/11 ZDH $\%$ ANC- 1st trimisteraP-value $\%$ HIV- ANC 	by enabling and need factors, 2005/6 and 2010/11 ZDHS $\%$ ANC- 1st trimisteraP-value $\%$ HIV- ANC testsaP-value $\%$ 4+ ANC visitsaP-value $\%$ 4+ ANC visitsaP-valuemabling and perceived risk/need factorsP-value% 4+ ANC visitsaP-valueIV test.001.001.41038.924.543.744.575.544.0tatus.015.003.45041.542.543.945.147.043.7ness.447.001.00344.457.246.343.064.444.347.755.139.6with HIV.000.001.24040.872.244.244.936.543.2				

Table 5-6: Bivariate analysis of timing of ANC, antenatal HIV testing and frequency ofANC visits by enabling and need factors, 2005/6 and 2010/11 ZDHS

Continued

HIV Stigma	ı	038		.001		.001		
Low	44.0		36.5		44.2		1459	
Medium	40.8		44.0		47.9		1566	
High	44.9		74.8		39.6		5446	
Resources and access to health care service (enabling factors)								
Employmen	nt Status	.001		005		.001		
No	40.2		61.3		44.6		5448	
Yes	45.5		57.5		42.7		3023	
House weal	th	.001		.001		.001		
Poor	41.2		54.9		47.0		5276	
Rich	40.3		67.0		43.1		1807	
Richest	48.9		70.7		32.9		1388	
Accessibilit problems	y	.001		.001				
No	44.9		69.1		38.8		2585	
Yes	40.8		55.9		46.1		5886	

Table 5-7 Continued

^aWeighted; ^bunweighted cases.

5.7.4 Multilevel logistic regression analysis

In the previous sections, bivariate analysis shows that the majority of the selected predisposing, enabling and need factors are associated with HIV and ANC service uptake. However, HIV factors such as previous HIV testing, knowing someone who died of HIV, and HIV-sero status were found not be associated with the number of ANC visits. In addition, HIV sero-status, awareness and stigma were also associated with antenatal HIV testing. Following important findings in the bivariate analysis, multivariate multilevel logistic regression analysis was used to estimate the factors that influence ANC HIV testing, timing and number of ANC visits. Separate analysis was done for each of the HIV and ANC service integrated outcomes. A total of four models were estimated for each outcome. The first model (model 0) was an intercept only model, so called empty model, with year of survey, as a covariate. This empty model is useful in showing if there is sufficient variance at the community level. All the individual predisposing factors were entered simultaneously in model 1. In model 2, enabling, need/perceived factors are added and adjusted for. The community level factors were then

included in model 3. All the fixed effects are expressed as odds ratio (OR) in the multivariate models. A 95% confidence interval is used (95%CI). The random effects were expressed as variance partition coefficient (VPC), also known as intra-class correlation coefficient (ICC) and proportional change in variance (PCV). The multilevel fitness was tested using deviance information criterion (DIC). The multilevel multivariate analysis was performed in MLwin statistical software.

5.7.4.1 Multilevel logistic regression parameter estimates for ANC timing visit

As previously noted in table 5.2, only 37.7% of pregnant women reported having an ANC visit in the first trimester during the 5 years period preceding the 2010/11 ZDHS, down from 48% in the previous survey. Table 5.7 presents the parameter estimates for timing of ANC in Zimbabwe. The results in the empty model show that most of the variation in ANC visit in the first trimester was at the individual level (96%). However, there was also some variation at the community level as indicated by significant variance in the log odds of reporting being ever tested for HIV across communities (estimated= 0.144, S.E= 0.026, i.e Z-value of 5.5>2.58 at 1% sig, level). Based on the estimated variance partition coefficient (VPC) also known as the intra-class correlation coefficient (ICC), the intra-community correlations was estimated at about 4.2%. This indicates the extent of the variability in early ANC visits that that could be attributed to community level factors. Confirming the results from bivariate analysis, the results from an empty model suggests that women in 2010/11 were 44% less likely to have ANC in the first trimester than in 2005/6.

In comparison with model 0, the variation in ANC visit in the first trimester remained significant across the communities in model 1 (estimated= 0.137, S.E= 0.026, i.e Z-value of 5.3>2.58 at 1% sig, level). The ICC remained stable at 4%. This showed that the inclusion of individual level predisposing factors did not help explain much of the between community variance in ANC visits in the first trimester. However, the proportional change in variances (PCV) in the odds of ANC visit in the first trimester of 4.8% across the communities was explained by individual level predisposing factors considered. This therefore, means that part of the clustering of ANC visit in the first trimester in the communities, albeit small, was as a result of the composition of the communities by individual predisposing factors.

The variables found to be associated with ANC in the first trimester in model 1 included place of residence, women's age, marital status, wanted child, parity and media exposure. As would be expected, women who reside in urban areas were 1.39 times (OR=1.39, 95% CI=1.23-1.58) more likely to have an ANC visit in the first trimester than rural women. Also, women above 25 years of age had higher likelihood of having an ANC visit in the first trimester, compared to young women in the age group 15-24. Similarly, women who were married, divorced or widowed were more likely to have an ANC visit in the first trimester, and this was more pronounced among those who were divorced (OR=1.70, 95% CI= 1.51-2.34), compared to never married pregnant women. However, the estimates indicated an inverse relationship between wanting the child with the timing of ANC visit. Women who reported that they wanted the child later (OR=0.78, 95% CI=0.69-0.87) and the child not wanted (OR=0.68, 95% CI=0.60-0.79) had lower odds of an ANC visit in the first trimester compared to those who wanted the child now. In the same vein, women with higher order parity, 2nd birth (OR=0.75, 95% CI=0.75-0.86); and 3rd plus birth (OR=0.66, 95% CI=0.59-0.77) were also less likely to have their initial ANC visit in the first semester. When other individual predisposing factors were controlled for in model 1, there was no evidence of an association between women's education and religious affiliation with timing of ANC in Zimbabwe.

Model 2 added enabling and needs factors. After adjusting for factors in model 2, the variation in ANC visit in the first trimester remained significant and stable across the communities. Also, the estimates for predisposing factors remained almost the same as in model 1, except for women affiliated with the protestant religious group. The results show that this group of women have 1.22 times the odds of initiating ANC in the first trimester compared with women who are affiliated with a catholic religious group. The estimates for enabling and need factors show that richer women (OR=1.38, 95% CI=1.23-1.69); and the richest (OR=1.23, 95% CI=1.09-1.39) were more likely to have an ANC visit in the first trimester, compared to poor women. Being in employment increased the odds of having an ANC visit in the first trimester by 1.20 times. The likelihood of having an ANC visit in the first trimester was also found to be positively associated with media exposure. It was found that pregnant women who had low (OR=1.28, 95% CI=1.14-1.44) and high (OR=1.34, 95% CI=1.19-1.53) media exposure were more likely to have had their initial ANC visit in their first trimester compared to those who were not exposed to media at all. Women's autonomy and access to health care have not been found to be associated with ANC in the first trimester.

There is no evidence of an association between HIV factors (i.e sero-status, awareness, and stigma) with timing of ANC. However, those who reported not having confidentiality concern and knowing someone who died due to HIV were 1.10 and 1.61 times more likely to have an ANC visit in the first trimester than their counterparts. Similarly, those who reported having previous HIV test before pregnancy were 1.50 (times (OR=1.50, 95% CI=1.33-1.69) more likely than those who had a test to have an ANC visit in the first trimester. Again, after adjusting for model 2 factors, the variation in timing of an ANC visit in the first trimester remained stable.

Finally, model 3 introduced community level factors. In model 3, the community level variation in ANC visits in the first trimester was attenuated, but became significantly lower than in models 0-2. The variance partition coefficient indicated that intra-community correlations were 3.6%. The proportional change in variance in the odds of an ANC visit in the first trimester across communities (10%) was explained by the community factors, indicating that part of the clustering of ANC visit in the first trimester within communities is due to contextual factors. The ICC of 3.6% indicate that there was a slight difference in timing of ANC for mothers at the community level

Women who lived in communities with higher HIV prevalence were 1.84 times (OR=1.84, 95% CI=1.64-2.74) more likely to have visited antenatal care service in the first trimester than those from communities with low HIV prevalence. Other community level factors considered were not associated with timing of ANC. Lower values of the deviance information criterion (DIC) in successive models were indicative of the fit of the multilevel model in explaining the variation in the odds of ANC visit in the first trimester by individual and community predisposing factors, and enabling and need factors.

Parameters		Model 1		Model 2		Model 3	
	OR	95% CI	OR	95% CI	OR	95%CI	
Predisposing factors							
Survey 2010	0.56	[0.50-0.64]*	0.59	[0.52-0.68]*	0.68	[0.60-0.79]*	
Place of residence (run	ral)						
Urban	1.39	[1.23-1.58]*	1.61	[1.43-1.91]*	1.61	[1.44-1.91]*	
Age groups(15-24)							
25-29	1.36	[1.21-1.55]*	1.33	[1.18-1.51]*	1.32	[1.17-1.50]*	
30-34	1.53	[1.36-1.80]*	1.47	[1.31-1.73]*	1.47	[1.30-1.73]*	
35+	1.47	[1.31-1.75]*	1.41	[1.26-1.69]*	1.39	[1.24-1.67]*	
Marital status(never m	narried						
Married	1.47	[1.30-1.83]*	1.44	[1.28-1.80]*	1.45	[1.29-1.81]*	
Widowed	1.70	[1.51-2.34]*	1.66	[1.46-2.26]*	1.68	[1.49-2.31]*	
Divorced	1.58	[1.40-2.05]*	1.53	[1.36-1.99]*	1.54	[1.37-2.01]*	
Women's education (p	orimary)					
Secondary	1.19	[1.07-1.32]	1.03	[0.92-0.78]	1.03	[0.91-1.16]	
Religion(Catholic)							
Protestant	1.19	[1.06-1.44]	1.22	[1.08-1.47]*	1.21	[1.08-1.46]*	
Pentecoastal	1.13	[1.01-1.36]	1.12	[1.00-1.35]	1.11	[0.99-1.34]	
Apostolic	0.98	[0.87-1.16]	1.01	[0.89-1.19]	0.99	[0.88-1.31]	
Wanted child (now)							
Later	0.78	[0.69-0.87]*	0.79	[0.70-0.88]*	0.78	[0.69-0.87]*	
Not wanted	0.68	[0.60-0.79]*	0.69	[0.61-0.80]*	0.67	[0.60-0.79]*	
Parity							
2	0.75	[0.75-0.86]*	0.76	[0.67-0.86]*	0.76	[0.68-0.87]*	
3+	0.66	[0.59-0.77]*	0.67	[0.60-0.79]*	0.68	[0.60-0.79]*	
Wealth (poor)							
Medium			1.38	[1.23-1.69]*	1.39	[1.24-1.70]*	
Rich			1.23	[1.09-1.39]*	1.22	[1.08-1.38]*	
Employment status(N	lo)						
Yes			1.20	[1.08-1.33]*	1.19	[1.07-1.33]*	
Media exposure (Low)						
Medium			1.28	[1.14-1.44]*	1.28	[1.14-1.44]*	
High			1.34	[1.19-1.53]*	1.35	[1.20-1.54]*	

 Table 5-8: Average odds ratios of ANC visit in the 1st trimester from multilevel logistic

regression models (95% confidence intervals are given in square brackets)

Continued

raole o / commaca						
Decision making (hus	sband)					
Women alone			0.96	[0.80-1.16]	0.96	[0.80-1.16]
Jointly			0.95	[0.80-1.12]	0.94	[0.80-1.11]
Physical access to a he	ealth fac	cility problem ((no)			
Yes			0.93	[0.82-1.04]	0.91	[0.81-1.03]
Pre-HIV test (no)						
Yes			1.50	[1.3-1.69]*	1.50	[1.34-1.69]*
HIV sero-status (nega	tive)					
Positive			1.13	[0.98-1.29]	1.06	[0.92-1.22]
HIV awareness (low)						
Medium			1.13	[0.99-1.28]	1.12	[0.99-1.27]
High			0.98	[0.86-1.13]	0.99	[0.86-1.13]
HIV stigma(low)						
Medium			0.97	[0.82-1.14]	0.97	[0.82-1.15]
High			1.17	[0.97-1.40]	1.15	[0.96-1.38]
Confidential concern	(yes)					
No			1.10	[0.98-1.2]*	1.09	[0.98-1.20]*
Knows someone livin	g with]	HIV (no)				
Yes			1.61	[1.43-2.]*	1.62	[1.44-2.22]*
Contextual Factors						
HIV prevalence in con	nmunity	у			1.84	[1.64-2.74]*
HIV Awareness in con	mmunit	у			0.81	[0.53-1.25]
HIV Stigma in comm	unity				1.07	[0.71-1.61]
Random effects para	meters					
Community-level						
Variance (SE)	0.14	[0.03]*	0.13	[0.03]*	0.123	[0.026]*
(VPC)=ICC (%)		4				
(PCV)(%)		4.8		4		3.6
Model fit statistics				0		10
Cluster		679		679		675
DIC		8293		8293		8255

Table 5-7 continued

*Statistical significance at 5% level P < 0.05.

5.7.4.2 Multilevel logistic regression parameter estimates of antenatal HIV testing uptake

The results in the empty model 0 in (Table 5.8) showed that most of the variation in antenatal HIV testing was at the individual level. However, there was also some variation at the community level as indicated by significant variance in the log odds of reporting ever being

tested for HIV across communities (estimated= 0.291, S.E= 0.037, i.e Z-value of 7.9>2.58 at 1% sig, level). Based on the estimated variance partition coefficient (VPC), the intracommunity correlation was estimated at about 8%. This indicates the extent of the variability in antenatal HIV testing that could be attributed to community level factors. Like in Table 5.7 above, the likelihood of having antenatal HIV testing was 3.77 times (OR=3.77, 95% CI=3.32-4.29) higher in the 2010/11 ZDHS than in 2005/6 ZDHS.

All the individual predisposing individual level factors are included in model 1. The variables found to be associated with antenatal HIV testing in model 1 included place of residence, women's age, marital status, women's educational attainment and parity. While, the odds of having an antenatal HIV test were increasing in the general population, pregnant women from urban areas were 48% less (OR=0.52, 95% CI=0.43-0.63) likely to be tested for HIV, compared with their rural counterparts. The results also show that age of women, marital status and parity were inversely correlated with antenatal HIV testing. The estimates suggest antenatal HIV testing decreases with age, such that women from age groups, 25-29, 30-34 and over 35 years of age were less likely to have been tested for HIV during ANC by 29% (OR= 0.71, 95%CI=0.62-0.81]; 31% (OR=0.69, 95% CI=0.56-0.70); and 57% (OR=0.43, 95% CI=0.36-0.52), compared with young women in the age group 15-24, respectively. Within marital status categories, divorced women were 26% less likely to have an antenatal HIV test (OR=0.74, 95% CI= 0.57-0.96), compared with never married pregnant women. Other marital status categories i.e being currently married and widowed were not found to be significantly associated with antenatal HIV testing. Similarly, women in their 2nd order parity were 28% less (OR=0.72, 95% CI=0.58-0.90) likely to have antenatal HIV testing. It was only having secondary or higher educational attainment that was found to be increasing the odds of antenatal HIV testing by 1.43 times (OR=1.43, 95% CI=1.28-1.61), compared with women with primary or no education. The estimates suggest no evidence of an association between religious affiliation and wanted child with antenatal HIV testing in model 1.

In comparison with model 0, the variation in antenatal HIV testing remained significant across the communities (estimated-0.266, S.E= 0.036, i.e Z-value of 7.4>2.58 at 1% sig, level). As judged by the variance partition coefficient, the ICC was 7%. The proportional change in variances (PCV) in the odds of HIV testing during ANC of 12.5% across the communities was attributed to individual level predisposing factors. This therefore, means that the composition

of individual characteristics did explain part of the nesting of HIV testing during an ANC visit within the communities.

Model 2, added enabling resources and HIV need factors to model 1. When model 2 factors were controlled for, pregnant women in the 2010/11 ZDHS continued to have antenatal HIV tests more than in the 2005/6 survey. There was no suggestive evidence of an association between place of residence and antenatal HIV testing in model 2 and 3. As in model 1, the likelihood of antenatal HIV testing decreased by an increase in age in model 2. A similar pattern was also observed regarding women's marital status. Being wealthier was found to be associated with increased odds of HIV testing during ANC visit in model 2. The estimates suggest that women in the rich and richest quintiles were 1.45 (OR=1.45, 95% CI=1.11-1.91) and 1.67 (OR=1.67, 95% CI=1.16-2.41) times more likely to have an antenatal HIV test, compared with poor women. This behaviour regarding the association between antenatal HIV testing with household income continued into model 3. The likelihood of antenatal HIV testing was also found to be significantly associated with media exposure. It was found that pregnant women who have medium to high media exposure were more likely to have antenatal HIV testing than those with low media exposure. For example, women who had high media exposure were 1.30 times (OR=1.30, 95% CI=1.06-1.61) more likely to have an antenatal HIV test, compared to those with low media exposure. Likewise the association between wealth and antenatal HIV testing, women with higher media exposure continued to have an antenatal HIV test even after community level factors were controlled for. It was found that the odds of HIV testing during ANC were on average 14% lower among women reported as having problems in accessing a health care centre in model 2 and 3, compared to those said it was not their main problem. Employment status and decision-making in the household were not significantly associated with antenatal HIV testing.

To explore the effects of HIV on antenatal HIV testing, model 2 also controlled for HIV factors. The estimates show that HIV factors such as sero-status, awareness and confidentiality concern were not associated with antenatal HIV testing across the models. However, knowing someone who died due to HIV increased odds of antenatal HIV testing. The estimates suggest that women who knew someone who died of HIV were on average 1.24 times more likely to have an antenatal HIV test compared with their counterparts. Compared to model 1, the variation in HIV testing during an ANC visit remained significant across the communities

(estimated = 0.25, S.E = 0.035, i.e Z-value of 7.14>2.58 at 1% sig, level). The inclusion of resources and HIV enabling and need factors appear to have no effect on community variation in antenatal testing as evidenced by 0% proportional change in variance.

Community level variables including in model 3, were HIV stigma, HIV awareness, and HIV prevalence. The outputs of the fixed effects from model 3 are approximately the same as the previous models 1 and 2. There was no evidence of association between HIV prevalence and stigma in the community with antenatal HIV testing, but women residing in communities with high HIV awareness were 2.08 times (2.08, 95% CI=1.05-4.11) likely to have been tested for HIV compared with their peers who lived in communities with low HIV awareness. Compared to model 2, the final ICC remained at 7% across the communities, with 0% proportional change in variance observed. The successive reduction in Deviance Information Criterion (DIC) from Model 0-3 demonstrates a better fit model over every previous one in table 5.7.

Doromotors		Model 1		Model 2	Model 3		
Farameters	OR	95% CI	OR	95% CI	OR	95% CI	
Survey 2010	3.77	[3.32-4.29]*	3.10	[2.66-3.62]*	1.87	[1.58-2.21]*	
Place of residence (1	ural)						
Urban	0.52	[0.43-0.63]*	0.83	[0.60-1.13]	0.97	[0.68-1.39]	
Age groups(15-24)							
25-29	0.71	[0.62-0.81]*	0.70	[0.61-0.80]*	0.68	[0.60-0.75]*	
20-34	0.69	[056-0.7]*	0.65	[0.55-0.78]*	0.65	[0.55-0.77]*	
35+	0.43	[0.36-0.52]*	0.43	[0.35-0.52]*	0.42	[0.35-0.51]*	
Marital status(never	married	1					
Married	1.10	[0.85-1.32]	1.03	[0.83-1.29]	1.02	[0.82-1.28]	
Widowed	0.75	[0.54-1.06]	0.73	[0.52-1.02]	0.47	[0.47-0.93*]	
Divorced	0.74	[0.57-0.96]*	0.73	[0.56-0.95]*	0.73	[0.56-0.95]*	
Women's education	(primar	·y)					
Secondary	1.43	[1.28-1.61]*	1.38	[1.23-1.55]*	1.32	[1.17-1.48]*	
Religion(Catholic)							
Protestant	1.17	[0.96-1.43]	1.13	[0.90-1.42]	1.13	[0.90-1.41]	
Pentecoastal	1.13	[0.91-1.38]	1.03	[0.81-1.30]	1.01	[0.80-1.28]	
Apostolic	1.15	[0.97-1.36]	1.23	[1.02-1.49]*	1.23	[1.02-1.49]*	
						Continued	

Table 5-9: Average odds ratios of ANC- HIV testing from multilevel logistic regression models (95% confidence intervals are given in square brackets)

Continued

Table 5-8 continued

Wanted child (now)						
Later	0.88	[0.75-1.04]	0.81	[0.68-0.98]*	0.81	[0.67-0.98]*
Not wanted	1.15	[0.93-1.43]	1.16	[0.90-1.49]	1.16	[0.90-1.49]
Parity				[0000000]		
2	0.72	[0.58-0.90]*	0.90	[0.73-1.11]	0.90	[0.73-1.11]
3+	0.88	[0.75-1.04]	0.78	[0.61-0.99]*	0.78	[0.61-1.00]
Wealth (poor)						
Rich			1.45	[1.11-1.91]*	1.35	[1.02-1.78]*
Richest			1.67	[1.16-2.41]*	1.53	[1.05-2.24]*
Employment status	(No)					
Yes			0.86	[0.73-1.01]	0.85	[0.72-1.00]
Media exposure (Lo	w)					
Medium			1.35	[1.12-1.63]*	1.34	[1.10-1.64]*
High			1.30	[1.05-1.61]*	1.28	[1.03-1.60]*
Distance to health fa	acility p	oroblem (no)				
Yes			0.86	[0.75-0.98]*	0.85	[0.75-0.97]*
HIV sero-status (neg	gative)					
Positive			1.08	[0.88-1.31]	1.10	[0.89-1.35]
HIV awareness (low	/)					
Medium			0.96	[0.80-1.15]	0.91	[0.76-1.10]
High			0.83	[0.68-1.02]	0.79	[0.64-0.98]
HIV stigma(low)						
Medium			1.41	[1.14-1.73]*	1.36	[1.10-1.68]*
High			1.11	[0.87-1.42]	1.07	[0.3-1.38]
Confidential concern	n (no)					
Yes			0.94	[0.81-1.10]	0.94	[0.81-1.09]
Knows someone with	th AIDS	(no)				
Yes			1.23	[1.02-1.49]*	1.24	[1.02-1.51]*
Contextual Factors	5					
HIV Awareness in c	commun	ity			2.08	[1.05-4.11]*
Random effects para	ameters					
Community-level						
Variance (SE)	0.266	[0.036]*	0.25	[0.035]*	0.241	[0.034]*
(VPC)=ICC (%)		8		7		7
(PCV)(%)		12.5		0		0
Model fit statistics						
Cluster		679		679		679
DIC		8461		8461		8461

*Statistical significance at 5% level P < 0.05.

5.7.4.3 Multilevel logistic regression parameter estimates of at least four ANC visits

At 67%, the rate at which women use antenatal defined as making at four and more visits-is moderately high in Zimbabwe. However, bivariate analysis indicates that there are variances in ANC visits within populations. In the multivariate logistic regression analysis, based on the result in the empty model 0, the variance in the odds of having four or more ANC visits was significant across the communities (estimated= 0.144, S.E= 0.026, i.e Z-value of 5.5>2.58 at 1% sig, level). Based on the estimated variance partition coefficient, the ICC was estimated at 4.2%. This suggests that about 96% of the variability in having four or more ANC visit was attributed to individual level factors. Pregnant women were 1.19 times (OR=1.19, 95% CI=1.08-1.32) more likely to have four or more ANC visits in ZDHS 2010/11 than in the ZDHS 2005/6.

Adjusting for individual predisposing factors in model 1, the estimates suggests that place of residence, women's age, marital status, educational attainment, parity and wanted child were associated with four or more ANC visits in integrated HIV and RH services. The estimates show that number of ANC visits is largely dependent on the place of residence. The positive slope means that women in urban areas are 1.24 times (OR=1.24, 95% CI=1.10-1.40) more likely to have four or more visits than their rural counterparts. The negative slopes for age mean that middle aged women (25-29) were 22% less likely to attend four or more ANC visits than those of younger age (15-24). Meanwhile, those aged (30-34) years are 29% less likely to have four or more ANC visits. Similarly, older women above 35 years were 37% less likely to have four or more ANC visits. Across marital categories, married people, those who were widowed and divorced were 29%, 33% and 39% less likely, respectively to attend four or more ANC compared with those never married women. Educational attainment is another important factor that works alone in predicting the number of ANC visits. Unexpectedly, women with secondary or higher educational attainment are comparatively less likely to have four or more ANC visits by 25%, compared to those with no education. Whether pregnant women will attend four or more ANC visits or not also is dependent extensively on whether the current pregnancy was wanted then or later. Women who reported wanting the child later or not wanted were 1.15 and 1.24 times more likely than those who wanted the child now to have four or more ANC visits. There is also a considerable amount of effect of birth order parity over number of ANC visits in Zimbabwe. Women with in parity 2 and 3 groups are 1.21 and 1.33 times more likely

than those in parity (0-1) to have attended ANC four or more times. In comparison with an empty model, the variation in having four or more ANC visits across the communities was small as indicated by a 3.4% intra-community correlation and the proportional change in variance of 0%.

Model 2 included resources and HIV enabling factors. The inclusion of enabling factors made the effects of place of residence redundant throughout the models. However, similar patterns regarding other predisposing factors in model 1 are also observed in model 2. The estimates show that household wealth, media exposure, women's autonomy and confidentiality concerns are important inhibiting factors for decreasing number of ANC visits. Quite interestingly, it has been found that women from the richer households and those with higher exposure to media were on average 23% less likely to have four or more ANC visits. Confidentiality concern about personal HIV related issues is also a vital factor in decreasing the odds of four or more ANC visits. Women who reported that they have issues with confidentiality regarding HIV are 11% less likely to have four or more ANC visits than those who reported they were not concerned. Women's autonomy, physical access to a health facility and antenatal HIV testing successfully increased the odds of four or more ANC visits. Women who made decisions regarding household purchases and their own health care alone or jointly with their husbands were 1.26 and 1.23 times more likely to have four or more ANC visits compared with women whose decisions are made by their husbands alone. Also women who had been offered and accepted antental HIV testing were 1.33 times more likely to have attended ANC four or more times. Likewise, women who reported having no problems with physical access to health care services were 1.20 times more likely to have ANC visits compared with their counterparts. There is no suggestive evidence of an association between four or more ANC visits with employment status and HIV factors (i.e sero-status, awareness, stigma and knowing someone who died due to HIV). The pattern observed in model 1 and 2 remained the same in model 3 when community level factors were adjusted for, except for household wealth which does not have any effects anymore. At the community level, HIV awareness, stigma and HIV prevalence in the community are not important factors for the number of ANC visits. Across the three models the ICC remained low at 3%, suggesting that all individual factors included in the models accounted for about 97% of the variability in four or more ANC visits in HIV and RH integrated services in Zimbabwe.

Parameters	Ν	Iodel 1	Model 2		Model 3	
	OR	95% CI	OR	95% CI	OR	95% CI
Predisposing factors	5					
Survey 2010	1.19	[1.08-1.32]*	1.83	[1.48-2.26]*	1.82	[1.45-2.29]*
Place of residence (rural)					
Urban	1.24	[1.10-1.40]*	0.82	[0.67-1.00]	0.82	[0.67-1.00]
Age groups(15-24)						
25-29	0.78	[0.69-0.89]*	0.78	[0.67-0.89]*	0.78	[0.68-0.90]*
20-34	0.71	[0.61-0.84]*	0.73	[0.61-0.87]*	0.74	[0.62-0.88]*
35+	0.63	[0.53-0.75]*	0.64	[0.53-0.78]*	0.65	[0.53-0.79]*
Marital status(never	married					
Married	0.71	[0.58-0.87]*	0.62	[0.47-0.81]*	0.62	[0.47-0.81]*
Widowed	0.67	[0.50-0.92]*	0.76	[0.54-1.07]	0.76	[0.54-1.07]
Divorced	0.61	[0.48-0.79]*	0.66	[0.50-0.87]*	0.66	[0.50-0.87]*
Women's education	(primary)					
Secondary	0.85	[0.76-0.94]*	0.89	[0.79-1.01]	0.90	[0.80-1.02]
Religion(Catholic)						
Protestant	0.20	[0.86-1.14]	1.07	[0.91-1.26]	1.07	[0.91-1.27]
Pentecoastal	0.94	[0.81-1.08]	0.99	[0.84-1.17]	0.100	[0.84-1.18]
Apostolic	0.98	[0.87-1.09]	1.09	[0.95-1.24]	1.09	[0.95-1.25]
Wanted child (now)						
Later	1.15	[1.04-1.29]*	1.18	[1.05-1.33]*	1.19	[1.05-1.34]*
Not wanted	1.24	[1.07-1.45]*	1.30	[1.09-1.54]*	1.29	[1.09-1.53]*
Parity (0-1)						
2	1.21	[1.06-1.37]*	1.22	[1.06-1.40]*	1.21	[1.05-1.39]*
3+	1.3	[1.14-1.6]*	1.34	[1.13-1.59]*	1.34	[1.13-1.58]*
Wealth (poor)						
Medium			1.03	[0.89-1.19]	1.09	[0.92-1.29]
Rich			0.77	[0.64-0.91]*	0.82	[0.66-1.02]
Employment status	(No)					
Yes			0.96	[0.86-1.08]	0.96	[0.86-1.07]
Media exposure (Lo	owl)					
Medium			0.82	[0.72-0.93]*	0.82	[0.72-0.93]*
High			0.78	[0.67-0.90]*	0.77	[0.67-0.89]*
Decision making (h	usband)					
Women alone			1.26	[1.04-1.54]*	1.27	[1.04-1.55]*
Jointly			1.23	[1.02-1.45]*	1.22	[1.02-1.45]*

 Table 5-10: Average odds ratios of at least four ANC visits from multilevel logistic

 regression models (95% confidence intervals are given in square brackets)

Continued

Distance to health facility problem (yes)						
No			1.20	[1.06-1.36]*	1.20	[1.06-1.36]*
ANC-HIV test (no)						
Yes			1.33	[1.21-1.46]*	1.32	[1.20-1.45]*
HIV sero-status (negative)						
Positive			1.01	[0.87-1.18]	1.03	[0.88-1.21]
HIV awareness (low)						
Medium			0.90	[0.79-1.03]	0.92	[0.80-1.05]
High			0.97	[0.84-1.12]	0.98	[0.85-1.14]
HIV stigma(low)						
Medium			0.10	[0.84-1.18]	0.99	[0.3-1.18]
High			0.96	[0.79-1.17]	0.96	[0.78-1.17]
Confidential concern (no)						
Yes			0.89	[0.80-0.99]*	0.89	[0.80-0.99]*
Knows someone with AIDS (no)						
Yes			0.94	[0.80-1.10]	0.94	[0.80-1.09]
Contextual Factors						
HIV awareness in community					0.81	[0.53-1.25]
HIV stigma in community					1.07	[0.71-1.61]
HIV prevalence in community					0.75	[0.42-1.35]
Random effects parameters						
Community-level						
Variance (SE)	0.116	[0.024]*	0.11	[0.028]*	0.113	[0.028]*
(VPC)=ICC (%)		3.4		3.4		3.3
(PCV)(%)		0		0		0
Model fit statistics						
Cluster		679		679		675
DIC		8461		6841		6808

Table 5-9 continued

*Statistical significance at 5% level P < 0.05.

5.8 Discussion

The main aim of this chapter was to provide an overall picture of the determinants of HIV and maternal health integrated services as measured by three indicators i.e antenatal HIV testing, timing and frequency of ANC visits. It examined the individual and community level predisposing, enabling and need factors, including the effects of HIV-related factors such as HIV sero-status, previous test, awareness, stigma, confidentiality concerns and knowing someone who died due to HIV/AIDS. Overall, the results of the analysis showed a low level of uptake of HIV and maternal health integrated care, only 37% of pregnant women had antenatal care in the first trimester; 44% were not HIV tested during pregnancy; and 45% had

less than four ANC visits. The results also indicated that utilization of HIV and maternal health integrated services was clustered within and across communities, although the variability was small. As one would expect, individuals from the same community were homogeneous (Diez Roux, 2002). This was expected in this study given the tribal, cultural and religious diversities in Zimbabwe, which to some extent; serve as markers of attitudinal identities, socio-economic status and health beliefs (Kayeyi et al., 2009). After controlling for all variables selected in the study, the variation in ANC timing, antenatal HIV testing and number of ANC visits across the communities were 3.6%, 7% and 3.3% respectively. This suggests that over 90% of the total unexplained variation in utilisation of HIV and maternal health integrated services was as a result of unobserved individual level factors. Less than 10% of the total unexplained Variation could be attributed to unobserved community level factors. This finding was consistent with other studies which indicated that over 90% of the factors that influence utilisation of HIV and maternal health care services are attributed to individual rather than community level characteristics in southern Africa (Gabrysch et al., 2008; Ngome and Odimegwu, 2014).

The analysis shows that uptake of ANC visits in the first trimester was decreasing, whilst antenatal HIV testing and frequency of ANC visits increased significantly between 2005/6 and 2010/11. The individual predisposing and enabling factors relating to socio-economic status appear to be more important than community level factors for HIV and maternal health integrated services, although the results reveal mixed patterns across the three outcomes. Most individual predisposing factors associated with timing of ANC, antenatal HIV testing and number of visits are largely consistent with patterns observed in other SSA countries (e.g., place of residence, age, marital status, education, parity, wanted child and religion). However, unique patterns have emerged regarding uptake of each outcome. In particular, the results show that urban residence is a key predictor for timing of ANC, but not for antenatal HIV testing and number of ANC visits. Urban residence was found to be positively associated with timing of ANC care and the results resonate with the situation in SSA countries including in Zimbabwe. This is mainly due to inequalities in access to health care services between urban and rural residence (Gross et al., 2012). Women resident in urban areas may have been enjoying easier access to health care services, information and education on timing of ANC visits through media (Takarinda et al., 2016) than those residing in rural areas. Although access to HIV and maternal health services is free in Zimbabwe, there are several challenges that women face in these services in rural areas. These challenges include staff shortages, drug stock-out, poor infrastructure, low access to media and poverty (Chirwa et al., 2013; Kevany et al., 2012). In fact, WHO (2012) reported that 98% of drugs and health care provider's salary were donor funded in rural areas and due to poor salaries and working conditions (Maruva et al., 2014) have noted high staff turnover, negligence and uncaring attitude among health care staff working in rural hospitals. The finding that urban residence is positively associated with timing of ANC visits is therefore plausible and directly relates to poor health services delivery in rural areas. This is consistent with most of the literature which suggests that women are more likely to have trust in health care system due to the increased quality of care (Achia and Mageto, 2015; Alam et al., 2015; Adam, 2011; Jean et al., 2012; Ochako et al 2011; Titaley et al., 2010). Although, the findings in this study seem to be reliable, the relationship between place of residence and maternal health service utilisation needs further investigation in future studies, taking into account the different nature of deprivation within urban areas (Matthews et al., 2010). It has been suggested that the urban poor have very similar or even less progress in terms of access to health care than in rural areas due to urbanisation (Usman, 2016). More recent evidence (Usman, 2016) reveals that health and social services in urban areas have not kept pace with urban population growth. Over the past few years, large-scale migration from rural to urban areas in Zimbabwe has led to a proliferation of slums and informal settlements in many urban areas (Chatiza and Mlalazi, 2009). As noted by Gwauya (2013) high fertility in urban areas, especially among poorer groups, has further boosted city populations. The author stated that, cities are not only becoming larger, but they are becoming more inequitable, with large impoverished and marginalised settlements springing up often in close proximity to relatively wealthy existing communities.

Some studies have suggested that urbanisation is changing the face of poverty and marginalisation (Fotso et al., 2008; Matthews et al., 2005; Shah et al., 2009); and that poor and marginalised urban populations compare unfavourably within urban dwellers, particularly migrants from rural areas and slum dwellers, who may have low rates of service utilisation as low as rural populations. Matthews et al (2005) have pointed out that urban population dynamics are changing rapidly in many countries-leading to the possibility of an erosion of the urban health advantage. In countries like Ghana, Ethiopia, Kenya, Liberia and others, Matthews et al (2010) have reported that a facility birth rate of over 80% for urban rich coexists with a rate of around 20% for women in the poorest quintile. A scoping study in Ghana, byAdongo et al (2014) found that women living in informal settlements and slum areas are faced with limited

health services from the public sector and in such circumstance, small and unregulated forprofit making clinics fill that gap. The authors noted that health care services from profit making clinics are often unreliable, relatively very expensive and do not provide full and quality care. However, much of the existing literature has tended to ignore the inequities among and within urban dwellers and has focused instead on simple average differences between urban and rural areas, indicating that most nations experience substantially better health care services in urban than in rural areas (Fotso et al., 2008; Matthews et al., 2005).

The results also revealed mixed effects of women's age on timing of ANC, antenatal HIV testing and number of ANC visits. It was found that being over 25 years was positively associated with timing of antenatal care, but inversely associated with antenatal HIV and number of ANC visits. These general patterns were consistent with those observed in previous studies based on DHS data from sub-samples related to timing of ANC (Ochako and Gichuh, 2016), antenatal HIV testing (Semali et al 2014); and number of ANC visits (Gupta et al 2014). It was noted in the literature that maternal age is often presented as a proxy for accumulation of experience and knowledge, including the knowledge about HIV and the complications following late ANC booking (Weizer et al., 2006; Peltzer et al., 2009; Jean et al., 2012). This hypothesis agrees with the findings in this study. One explanation of this phenomenon could be that older women's sense of urgency regarding timing of ANC visit is shaped by awareness of the dangers associated with pregnant complications or negative experience resulting from previous pregnancies (Ngome and Odimegwu, 2014). It may also be that younger mothers may be less experienced in child birth or may lack economical resources and maybe cared by families, which may make it difficult for them to report early pregnancy (Jean et al., 2012). Another potential explanation could be that younger pregnant women do not know the gestation period at which they should initiate ANC. This postulation was supported by Ochako and Gichuh (2016) who stated that not knowing the right gestation age among young pregnant women partly explains why women initiate late their first ANC visit. Aside, due to culture, religious and health care provider attitude which stigmatise teenage pregnancies, young people may not want to be seen at a health facility seeking maternal health care services, which is usually tailored to cater for older and married women in Zimbabwe (Murire et al, 2014). Other studies reported that young and never married pregnant women are either less likely to attend or attend late or infrequently due to lack of power to make decisions

and they are also likely to experience violence from parents or stigmatised in the community (Kabagenyi et al., 2016).

The findings that older women are less likely to have antenatal HIV testing do not lend credence to the previous findings that indicate that acceptance of an antenatal HIV test increases with a woman's age. Previous studies have noted that older women have higher odds of accepting antenatal HIV testing due to accumulated knowledge on HIV (especially that HIV can be transmitted to a child during pregnancy as well as their knowledge of existence of prevention of mother to child HIV transmission (PMTCT) programmes (Semali et al., 2014). Although, the knowledge of need for PMCT has increased remarkedly over the years among older women in Zimbabwe (Murire et al 2014), the results in this current study could not find solace with older women's behaviour. The finding is unclear and perhaps requires a qualitative investigation. However, a simpler explanation could be that, older women may have previously given birth to presumably healthy children and thus see no reason for HIV testing in their future pregnancies. This is consistent with (Ochako and Gichuh, 2016)'s finding that pregnant women who had previously given birth to presumably healthy children were less likely to optin to voluntary HIV testing in the subsequent pregnancy. Another possible hypothesis is that, the low levels of antenatal HIV testing among older women shows that the campaigns to promote PMTCT programmes have not been effective among this group (Mtumbuka et al, 2012). There is a need to develop new methods of communication in order to encourage antenatal HIV testing among older women. Similarly, older women were less likely to have at least four ANC visits in this study. This finding supports (Mason et al., 2015)'s observation that HIV screening and being given a registration card was the main motivation for women to attend ANC so that they could receive medication and delivery in hospital in case of emergency. Once a woman is registered she stops visiting health facilities for fear of associated costs and time (Moindi et al 2015). Some women dislike frequent hospital examination during pregnancy, especially examination by use of fingers (Shiferaw et al., 2013). These reasons were similarly important in this study, and the hypothesis is that most pregnant women relied on traditional birth attendant (TBA) for their subsequent ANC after an initial registration and presumably HIV testing (Maruva et al., 2014).

Women's marital status has also seemingly shown paradoxical associations with HIV and maternal health care service utilisation. However, in line with the findings of many studies (for

example Semali et al, 2014) this study found positive associations between being married and better timing of antenatal care. The findings particularly that never married women -usually young teenagers were less likely to have an ANC visit in the first trimester have important implications for prevention of mother to child HIV transmission (PMCT) programs. This can certainly increase mother to child HIV transmission, given that about 70% of teenagers in Zimbabwe are not aware of their HIV status (ZIMSTAT, 2012). This result might reflect the fact that maternal health care in Zimbabwe is geared towards older and married women; and it is important to provide information and raise awareness of PMCT programs to prevent further mother to child HIV testing and number of ANC visits does not conform to what might be expected. For both outcomes, the levels of HIV testing during pregnancy and number of ANC visits were lower among previously married women (widowed and divorced), compared with never married and married women.

The study results reveal that educational attainment is a key factor for antenatal HIV testing, but not for timing and frequency of ANC visits somehow makes sense in the context of HIV, and supports the notion that educated women are more likely to realise the benefits of antenatal HIV testing. Extant literature reported multiple potential pathways that can explain why maternal education is associated with antenatal HIV testing. These include an increased knowledge of the benefits of preventative health care and awareness of risk (e.g PMCT), higher receptivity to new health care information, and socialisation with formal services outside the home, access to financial resources, and better communication with a partner, increased selfworth and confidence (Asefa et al., 2013; Birhane et al., 2015; Byamugisha et al., 2010; Maruva et al., 2014; Sama C-B et al., 2017).

In regards to parity, women with higher order parity tend to have reduced odds of maternal health care services utilisation (Ochako, 2016). This study found decreased odds of an ANC visit in the first trimester and an increased likelihood of having four or more ANC visits among women with a higher birth order. The fact that pregnant women with a higher order birth were less likely to initiate ANC in the first trimester maybe a reflection of lack of time, limited resources in the household, constraints related to their other responsibilities for their other children; and negative perceptions resulting from a previous pregnancy, if the woman was in her second or higher order pregnancy (Jayleen et al., 2016; Tarekegn et al., 2014). Higher order

parity may also be a reflection of a lack of access to family planning, traditional attitudes and lower economic status, which apparently maybe associated with lack of access to maternal health care services (Achia and Mageto, 2015, Beauclair et al., 2014). The results revealed that women were more likely to increase ANC visits with each additional child. This may be that women who have had more children may be in a more stable relationship and may be more confident of the processes of antenatal care and are more prepared to increase frequency of ANC visits (Bbaale, 2011).

Enabling and perceived need factors were also found to be major determinants of HIV and maternal health integrated health care in Zimbabwe. In particular, household wealth was found to be associated with timing of ANC visits and antenatal HIV testing. Extant literature has shown that being wealthier is associated with an increase in the likelihood of maternal health care utilisation (Babalola and Fatuisi 2009; Tarekegn et al., 2014). The finding is expected, women from wealthier households are more likely to be able to afford `health care services and their associated costs such as transport (Moindi et al., 2015; Semali et al 2014, Lepine et al 2014). Another possible explanation is the fact that since higher household wealth could beindicative of a higher number of educated household members; women from such households may have higher access to information and education on HIV and its implications (Ahmed et al., 2010). Employment status was found to be associated antenatal HIV testing, while there was no evidence of an association between employment and timing and number of ANC visits. The lower HIV testing during pregnancy among employed women raises issues of accessibility of HIV testing services and time. One potential explanation could be that, although Zimbabwe Employment policy protects people living with HIV against abuse and discrimination at the workplace, employed women do not accept HIV testing because they still view a positive test as devastating, and may lead to a number of negative consequences including loss of job, stigma and discrimination and difficulties in getting insurance (Sambisa, 2008). The reason for the absence of an association between employment and timing and number of ANC visits may be explained by the factor that even if women were employed, the decision on her health care may be made by her husband (Woldemicael and Tenkorang, 2009). However, decision-making for women was found not to be associated with either of the outcomes in this study.

The results show mixed patterns regarding enabling and perceived need factors relating to HIV/AIDS. There is no evidence of a significant association between HIV awareness and HIV sero-status with the three outcomes, once important background socio-economic and demographic as well as other HIV/AIDS factors are controlled for. This confirms earlier findings based on bivariate analysis in Table 5.6 suggesting HIV awareness and sero-status are not associated with timing and number of ANC visits. In addition, the results in the bivariate analysis suggesting an association between HIV awareness and sero-status maybe explained by differences in the background characteristics. However, the finding that higher HIV awareness at a community level is associated with increased odds of antenatal HIV testing resonates with the findings of Buzdugan et al(2015), which are suggestive of a stronger and wider campaign on PMCT programs in the communities (Ngome and odimegwu, 2014)

There is evidence of an association between HIV stigma and antenatal HIV testing, but not with timing and number of ANC visits. The result that pregnant women with medium to high levels of stigma are more likely to have antenatal HIV testing is unexpected. The logical explanation for this situation may suggest acceptance of HIV/AIDS and that pregnant women want to protect their infants from HIV, never mind their experience with stigmatisation. The results also indicate that confidentiality concern is a key factor for timing and number of ANC visits, whilst it is not for antenatal HIV testing. Pregnant women with no confidentiality concerns were more likely to initiate ANC early and to have four or more ANC visits. This may be that pregnant women who have had ANC in the first trimester may have been tested for HIV and are more confident of their HIV status enabling them to frequent ANC. The fact that antenatal HIV testing is not associated with confidentiality concern may be that women are given private individual counselling and space to make a private decision without staff and group interference in Zimbabwe (Perez et al., 2004). Actually, a study by Shetty et al (2010) suggests that the quality of care including confidentiality concerns shapes the women's decision to use health care services. The findings from this study suggest that the content of ANC is an important determinant of antenatal HIV testing and that improving health care service delivery through training and retention of qualified staff can encourage the overall use of maternal health services (Schnippel et al., 2015). This study found strong associations between timing and number of ANC visits and previous-HIV testing and counselling. The pretest HIV counselling which included information on prevention of mother to child HIV

transmission in the DHS could be argued to be an enabling factor for early ANC attendance, in view of the fact women may be aware of their test results (Asefa and Beyene, 2013).

Knowing someone living or who died of AIDS was associated with an increased likelihood of ANC in the first trimester and antenatal HIV testing. The results also revealed that pregnant women living in communities with high HIV prevalence rates are more likely to initiate ANC in the first trimester. This may suggest that personal acquaintance with AIDS may be a pointer to behavioural change to avoid mother to child HIV transmission and knowing someone who died of HIV is common in communities with higher HIV prevalence rates (Magadi and Desta 2011).

5.9 Recommendations and way forward

Overall the multilevel results showed little variations in HIV and maternal health care integrated services across communities in Zimbabwe. The variations were mainly explained by individual level variables such as background characteristics and enabling/ perceived need factors that were included in the models for all outcomes. At the community level, variables found to be contributing to the variations in HIV testing differed by HIV prevalence and awareness. It was found that community HIV prevalence was positively associated with timing of ANC visits, while community HIV awareness was important for antenatal HIV testing. Controlling for community level factors did not have much effect on community variation in HIV testing. Therefore, policies should focus on predisposing and enabling factors in order to improve HIV and maternal health care in Zimbabwe. For timing of ANC visits and antenatal -HIV testing, household wealth, access to media, employment status and HIV factors such as confidentiality concern, knowing someone who died as a result of HIV are critical, whilst women's decision making role, previous HIV test and confidentiality concern were more important for number of ANC visits. More importantly perhaps for both antenatal HIV testing and number of ANC visits, the introduction of financial incentives may stimulate pregnant women to access HIV and maternal services by providing compensation for transport cost and opportunity cost of time associated with accessing HIV testing services.

6 CHAPTER SIX: HIV AND DELIVERY CARE

6.1 Introduction

This chapter places particular interest on women's utilisation of delivery care, especially since professional delivery care is essential for prevention of mother to child HIV transmission which is a major contributor to HIV new infections. The aims of the chapter are twofold: first and primarily to identify individual and community level determinants of institutional and professional delivery care; and this includes hitherto unnoticed independent effects of HIV. Firstly, literature on the impact of HIV on delivery care is presented. This is followed by data and methods used in the analysis, while the last part presents the results and discussion of the findings.

6.1.1 Background

While antenatal care is an important strategy to reduce mother to child HIV transmission, child birth with a medically-trained health professional, which is most likely to be available if a pregnant woman gives birth in an institution is an additional strategy in prevention of mother to child HIV transmission (Buzdugan et al., 2015; Turan et al., 2008). Institutional delivery presents an opportunity for a repeat HIV test and for education on PMTCT if the mother is found to be HIV positive during labour and delivery (Buzdugan et al. 2015). HIV testing at labour and delivery may also present an opportunity to identify HIV positive mothers who may have had false HIV negative results and those who were not tested during antenatal care (Heemelaar et al., 2014). One study has reported that pregnant women who test HIV negative during antenatal care may feel reassured that neither they nor their infants are at risk of HIV infection (Moodley et al, 2009). However, HIV can be acquired during pregnancy and postpartum and women who sero-converted would not be detected unless a repeat testing is conducted throughout child and maternal care processes (Johnson et al., 2012; Weiser et al., 2006). The lack of professional delivery care can represent a missed opportunity for retesting and identification of pregnant women who have recently acquired HIV infection (Moodley et al., 2011). It has also been noted that the intrapartum events such as duration of membrane rupture, preterm births, invasive procedures during labour and delivery are crucial factors that

may increase mother to child HIV transmission since this is period where the risk is highest (Heemelar et al., 2014), thus it is important for pregnant women to deliver in an institution and with a professional delivery attendant. There is evidence that professional delivery assistance enhances the effectiveness of PMTCT intervention as women who give birth with a professional health care attendant have additional benefit of using ARVs meant to be taken around the time of delivery (Nkonki et al., 2007; Sibanda et al., 2013; Washington et al., 2015).

While, the need for increasing PMTCT uptake through delivery care continues, mounting evidence demonstrates an urgent need to examine barriers to institutional and professional delivery care. There is evidence indicating that uptake of delivery care still lags behind other HIV and maternal integrated health care indicators, and skilled care at birth remains elusive for many pregnant women in many SSA countries (Buzdugan et al. 2015). Against a backdrop of a frail health care system (characterised by a shortage of skilled professional attendants, increasing costs of care, poverty and inaccessibility of health care among other factors), home care delivery and child birth administered by traditional birth attendants and family members remain a lifeline for most pregnant women in many Sub-Saharan African countries (Zere et al.,2011). Studies have shown that factors such as culture, rural residence and low socioeconomic status of pregnant women constraint the effectiveness of PMTCT programs (Ahmed et al., 2010; Aremu et al., 2011; Arthur, 2012; Makate and Makate, 2017) as these factors force pregnant women to stay away from health facilities when they are due to give birth. In particular, access to delivery care is severely compromised for many low income pregnant women in SSA countries who sometimes have to travel a long distance to arrive at the nearest health care centre (Gabrysch et al., 2011).

Perhaps, some of the key barriers relate to how HIV factors such as HIV-related stigma, discrimination and lack of comprehensive knowledge of HIV among pregnant women (Turan et al., 2011) impact on delivery care. Regardless of economic status, often pregnant women are placed in an untenable position in which they have to choose between immediate threats to their social, economic and physical well being if their HIV status is disclosed and the future health benefits for themselves and their child in accessing delivery care (Rochat et al., 2006). Researchers such as Hardon et al (2012) pointed out that due to lack of attention to privacy and confidentiality in most SSA countries, HIV pregnant women may experience stigma and discrimination during delivery care. In particular, literature reviewed reveal that high levels of

fear, experience of HIV-related stigma and discrimination in the community and health facilities may negatively affect all women, not just those who know they are HIV positive (Turan et al., 2008).

Studies of HIV and antenatal care revealed that HIV related stigma could be inhibiting the effectiveness of delivery care as women who test positive want to keep such information from relatives and others who play an active role during child birth (Mucheto et al., 2011). Culturally, in most SSA countries, husbands, friends and relatives tend to be present during delivery (Bisika, 2008), and for women who desire to keep their status secret, the tendency is to avoid returning to the facility where they were diagnosed, preferring a home delivery or somewhere their status is not known (Turan et al., 2012). A qualitative study in Ethiopia reported that most HIV positive women do not give birth where their status in known, so that relatives will not become suspicious when they and their infants are given special medication shortly after birth (Shiferawet al., 2013). The authors reported that fewer pregnant women returned to deliver their babies in facilities they were diagnosed with HIV and that the majority of those who gave birth were not enrolled for antenatal HIV testing in the same facilities. In Tanzania, Theilgaard et al. (2011) reported that pregnant women recognised the importance of PMTCT and most were accepting antenatal HIV testing, but they were desperate to conceal their positive status from close family members, compelling them to stay away from health facilities where they were diagnosed and to give birth at home. A study in Kenya revealed that pregnant women who held negative stigma views about people living with HIV were less likely to deliver in a health facility (Turan et al., 2012). In another study, HIV positive women who had disclosed their HIV status to anyone were more likely to deliver in a facility than HIV positive women who did not disclose to anyone (Buzdugan et al., 2015).

6.1.2 Rationale for the study of HIV and delivery care in Zimbabwe

The primary health care clinics in Zimbabwe provide both HIV and maternal health care services to pregnant women. However, safe assisted delivery and essential HIV care cannot be provided for women opting for home deliveries (Maruva et al., 2014). Proper obstetric care during labour and delivery constitutes important prerequisites for reducing mother to child HIV transmission (Rutaremwa and Kabagenyi. 2016). This is particularly serious in Zimbabwe where about 93% of pregnant women attended ANC clinics, but the institutional delivery
averaged 77% according to the latest ZDHS (2014/15). The question that may require an answer is why more women attend ANC clinics and yet they continue to deliver at home?. This may be important in explaining the disparities between the number of ANC bookings and institutional delivery in Zimbabwe. In rural areas, a number of initiatives to improve uptake of delivery care have been implemented, including construction of more health facilities closer to the communities, setting up maternal waiting home shelters where pregnant who might not get to the hospital in time for delivery can be accommodated at the clinic during their final stages of their pregnancy (Makate and Makate, 2016). During their stay, pregnant women are offered an HIV test and are given information about HIV and pregnancy, as well as breastfeeding for those who are HIV positive (Maruva et al., 2014). Partly, due to the effects of HIV-related stigma and discrimination, and lack of comprehensive knowledge of HIV, many pregnant women in rural Zimbabwe prefer home delivery with help from family members or traditional birth attendants (Buzdugan et al., 2015). This is worrisome because over 90% of women in Zimbabwe are aware that HIV can be transmitted during labour and delivery and that PMTCT can prevent HIV in infants (ZDHS, 2014/15). Given that delivery care may help reduce mother to child HIV transmission, it is imperative for this chapter to identify factors influencing women's preference for home delivery rather than institutional delivery in Zimbabwe. Based on the research results, recommendations will address the question why delivery care is lagging behind ANC uptake. The results will also address women's preference so that the number of home deliveries could decrease. This may help enhance PMTCT programs in Zimbabwe. Given that an increased uptake of delivery care enhances effective utilisation of HIV and maternal health integrated services, the main objective of this study is to identify individual and community predisposing, enabling and need factors that influence institutional and professional delivery.

To achieve this aim the specific objectives are:

- Identify individual and community level determinants of delivery care; this includes hitherto unnoticed independent effects of HIV factors, and to investigate their relative contribution in explaining the disparities between institutional and professional delivery; and to
- To compare factors associated with institutional and professional delivery care

6.2 Methods and data

The chapter applies multilevel logistic regression models to data from the 2005/6 and 2010/11 Zimbabwe Demographic and Health Surveys. The sampling and procedures for data collection and methods of analysis are discussed in chapter 2-6. In this chapter, the sample size was first restricted to all women who tested for HIV. To avoid clustering of multiple birth and reduce recall bias (Lepine et al., 2014; Mumah et al., 2014), the sample was further restricted to the last birth in the five years preceding the surveys. This yielded a sample size of 8471 women with at least one live birth. In order to see the changes that occurred between the survey years, the data was further restricted by year of survey resulting in a split sample size of 4074 and 4397 for the 2005/6 and 2010/11 ZDHS, respectively. Measures of fixed effects, that is the effects of individual-level and community-level determinants of HIV testing were reported in terms of odds ratios (OR), whilst the measures of community variation (random effects) were expressed in terms of intra-class correlation coefficient(ICC). The estimation procedure was based on quasi-likelihood methods, starting with the default procedure 1st order Quasilikelihood Methods (MQL) and extending to 2nd order Penalized Quasi Likelihood (PQL). The 2nd order estimation was preferred because it is an improved approximation procedure (Rasbash et al., 2005).

6.2.1 Outcome variables

There were two main outcome variables in this chapter. These were whether or not a pregnant woman had delivered in a health institution and whether they were attended by a professional birth attendant. In addition, the study also examined two composite indicators, which combines attendant and place of delivery. There are questions of relevance to delivery care that women are asked during the DHS survey. Women aged between 19-49 years in selected households were asked a series of questions on delivery care for each live birth they reported occurred in the past 5 or 3 years. The specific questions on delivery care (ORC Macro, 2010) are:

Who assisted with the delivery of [NAME OF CHILD]? Anyone else?

Where did you give birth to [NAME OF CHILD]?

The DHS questions allowed for multiple responses to be recorded for the place of delivery and the person who attended to delivery.

6.2.1.1 Institutional delivery care variable outcome

The binary outcome measure was constructed based on whether or not the mother had delivered in a health facility. All the facilities were included in this indicator-hospitals, clinics, health centres and maternal homes. The measure was based on a recode of DHS variable (M15\$1), where those mothers who delivered at any form of health institution eg. Government or private hospitals were coded 1 and those who delivered elsewhere were coded '0'

6.2.1.2 Professional delivery

This is a standard international indicator of delivery care. The term 'health professional' rather than skilled attendant was used because the surveys recorded the professional label used by the women to identify their attendants and provide no indication of the individual's level of practical skills. As noted above, DHS recorded the attendants named by the women, but for the purpose of this analysis the indicator is defined by the most qualified attendant reported. For example, if a woman mentioned a midwife and a doctor in attendance, the delivery will be classified as attended by a doctor. Therefore, in this study a health professional is defined as midwife, doctor and a nurse. Given this working definition, the binary outcome measure for 'health professional' was constructed based on whether or not the mother had delivery attended by a doctor (M3A\$1), a midwife (M3B\$1), or a nurse (M3C\$1). An additive index with scores ranging from 0-3 was coded '0' no 'professional delivery', and '1-3', coded 'profession delivery'. This way of coding was important because providing skilled attendants for delivery care, along with the equipment, drugs and supplies necessary for effective management of obstetric complications has been advocated as the single most important factor in preventing maternal deaths (WHO, 2010).

6.2.2 Explanatory variables

The explanatory variables used in this chapter were adapted fromAndersen (1995) framework and are contained in figure 6.1 below. As noted in the previous chapters, the main ideology behind the framework is that the use of health care services is a function of contextual, predisposing, enabling or impeding and need factors. Based on the reviewed literature, theory and data availability, variables were selected and thereby operationalized as shown in Table 5.1 in the previous chapter. Some of the variables were re-coded for the purpose of this study. The original coding of the data is shown in the 2005/6 and 2010/11 ZDHS questionnaire. Consistent with other studies (Magadi and Desta, 2011; Takarinda et al, 2016) variables such as media exposure, HIV knowledge and AIDS stigma related variables were constructed from a set of correlated indicators using a principal component analysis (PCA). The PCA is a statistical technique that is used to condense a number of variable dimensions without significant loss of relevant information (Lepine et al., 2014). At community level, apart from HIV prevalence, two variables describing the community were newly imputed and added to the dataset. The variables were average HIV awareness and stigma in the cluster. In the analysis, the later variables were represented by a two-category variable split at the median across all communities, indicating high or low levels of the characteristic. The variables were coded '1' for communities with a high level; and '0' for low level of characteristics.

In addition to other factors as are operationalised in Table 6.1, three more enabling factors were added in this chapter. These were (i) timing of ANC visit and (ii) HIV testing during pregnancy. These variables were exploratory were a proxy for use of HIV and maternal health integrated services. However, the rationale for including these variables stems from literature reviewed which suggested a high correlation between antenatal and delivery care without adjusting for the effects of HIV (Jayleen et al., 2016). Utilisation of antenatal care is expected to reflect several things including the pregnant women's access to information about HIV, knowledge of mother to child HIV transmission, attitude towards ANC and potential awareness of pregnancy complications (Kevin et al., 2014; Muruva et al., 2014; Orne-Gliemann et al., 2006). Often, pregnant women who get in contact with maternal health care services are given information about pregnancy complications, mother to child HIV transmission and HIV prevention programs; and they are also encouraged to test for HIV, delivery in a hospital and with a professional birth attendant (Tenthanet al., 2015). It is therefore expected that women who do have access to ANC and those who are offered and accept an HIV test during pregnancy are more likely to use delivery care than their counterparts. The three variables are described below:

a) Timing of first ANC visit

Proxy for timing of prevention of mother to child HIV transmission program

This variable is a proxy for a need for an early HIV test during pregnancy. There is evidence indicating vertical HIV transmission can occur during pregnancy (Mugurungi et al., 2008;

Townsen et al., 2008). In the 2005-11 ZDHS, women were asked about methods of HIV transmission from mother to child and prevention methods. About 91% of respondents reported knowing that HIV can be transmitted from mother to child during pregnancy, delivery and breastfeeding (ZDHS, 2014/15), 90% knew that drugs can prevent mother to child HIV transmission. This reflects a high level of (MTCT) awareness in Zimbabwe and pregnant women are expected to initiate ANC earlier in order to get an HIV test with subsequent enrolment into PMCT programmes if a woman is found to be HIV positive. This variable therefore can be used to predict timing of an enrolment into prevention of mother to child HIV transmission programme. The variable is coded '1' if the women initiated ANC care in the first trimester and '0' for all others; including those who did not attend any ANC.

b) Number of ANC visits

The study included number of ANC visits since adequate utilization of ANC is a strong predictor of health institutional delivery in the literature (Gupta et al., 2014). In addition, the number of antenatal care visits a woman attends is important for allowing routine HIV testing and monitoring viral load in the blood among HIV positive pregnant women (Maruva et al., 2014). The lack of retesting during pregnancy represents a missed opportunity to identify women who have recently acquired HIV infection and have an increased risk of MTCT (Samaet al., 2017). In addition, pregnant women who do not frequent ANC have fewer opportunities to learn about PMCT programs (Buzdugan et al., 2015). The measure of frequency of ANC visits was based on a recode of variable M14, in which women who had at least 4 ANC visits are code '1', and women who had less than 3 ANC visits coded as '0'.The ZDHS asked the respondents: How many times did you receive antenatal care during this pregnancy? The responses enabled a classification of the number of times antenatal care was attended.

c) Antenatal HIV testing

This variable represents the actual need to prevent mother to child HIV transmission and to start taking drugs if a women was HIV positive. In the ZDHS, pregnant women who attended ANC for their last birth and were offered an HIV test responded to the question, I don't know the results, but (a) were you tested for HIV virus during any of the antenatal visits? If yes, (b) did you get the results of HIV test? Respondents who reported 'yes' to these two questions were classified as having been tested for HIV and know their results during ANC (Staveteig et

al., 2013; ICF international, 2014). Women who were tested and know their results were coded '1' and those who were not tested coded '0' indicating a binary response variable.

Predisposing	Enabling Factors	Perceived/need	Outcome
Community level HIV stigma HIV awareness HIV Prevalence	Household wealth Access to health facility Employment status Women's autonomy	Knows someone died of AIDS ANC-HIV testing	Delivery Care
Individual level Age Place of residence Marital Status Religion Education Parity Wanted Child	Access to media HIV awareness HIV stigma HIV Disclosure	HIV sero-status Confidentiality concern	

Figure 6-1: Analytical framework for the analysis of determinants of delivery in Zimbabwe

Source: Authors' construct

Table 6-1:	: Summary	description	of study	variables
------------	-----------	-------------	----------	-----------

Variable	Operation Definition
Institutional delivery	Deliveries that occurred in a health facility: coded '1' if delivery
	occurred in any form of health institution eg. Government or private
	hospitals and 0' for deliveries that occurred elsewhere.
Professional	Deliveries that were attended by a (doctor, midwife or nurse): coded
Delivery	'0' for no 'professional delivery', and '1' for 'profession delivery'
Individual predispos	ing factors
Residence	Women's residence, coded as 1=rural, and 0=urban
Women's age	Categorised into three age-groups: 15-24, 25-34 and 35+
Women's education	Educational attainment classified into: No education and primary coded
	'0', and secondary and higher '1'
Marital status	Classified into four groups: never married, currently married, widowed
	and divorced/separated
Religion	Religious affiliation, classified into: Catholic/Traditional, protestant,
	Pentecoastal, apostolic
Wanted last child	Three categories: wanted then, later and not wanted
Parity	Total number of children ever born, classified into three groups: 1, 2
	and 3+

Continued

Table 6-1 continued

Enabling and perceiv	ved risk/need factors
Media exposure	A composite summary index derived from information on frequency of
	reading newspapers, listening to radio and watching TV by
	respondents. The additive scoresranging from 0-1 were coded '0'-'low',
	'2-3' was '1' 'medium' and 4-5 was '2' 'higher'
Household wealth	DHS composite wealth index (Rustein et al 2004), based on household
	possessions and amenities, reclassified into three groups, poor, rich and
	richest, based on ODHI (2013) classification of poverty in Zimbabwe
Perceived distance to	The survey question asked women if distance was a problem for them
health facility to get	to get any medical help from health institutions. The responses were
medical help	coded as: '1'=yes big problem, '0'=not big problem
Employment status	Coded 1=if respondent is employed; 0=otherwise
Woman's autonomy	The relevant DHS questions (mv739 to mv743b)on who makes the final
	decision on own health care elicited categorical responses, 1=Wife
	alone 2=jointly, 3=Husband alone/ Others
Frequency of ANC	The measure of frequency of ANV visits was based on a recode of
visits	variable M14, in which women who had at least 4 ANC visits are code
	'1', and women who had less than 3 ANC visits and those who reported
	not having ANC visit coded as '0'
ANC-HIV testing	The binary variable was constructed based on whether or not the
	pregnant women had been offered and accepted HIV test for the last
	birth during their ANC visits. The measure was based on a recode of
	variable V840, in which a woman who had been tested for HIV as part
	of ANC was coded '1' and '0' otherwise.
HIV sero-status	Coded as 1=if respondent is HIV-positive;0=otherwise
HIV stigma	A composite index derived from a set of four DHS questions on
	HIV/AIDS stigma. The resulting score is classified as '0' low, '1-2'
	medium and '3-4' high
HIV awareness	A composite index derived from a series of questions on knowledge of
	how HIV is transmitted and ways to avoid infection. The resulting
	index is classified into quartiles and the higher the index, the higher the
	knowledge.
Knows someone	Coded as 1 = if respondent personally knows someone who died of
with HIV	HIV/AIDS; 0=otherwise
Confidentiality	Coded 1=if the respondent want HIV in the family to remain secret;
concern	0=otherwise
Contextual community	y/cluster HIV factors
HIV prevalence	Proportion of individual in cluster who are HIV-positive
HIV/AIDS	Average HIV awareness in the cluster
awareness	
HIV/AIDS stigma	Average HIV/AIDS stigma in the cluster

6.3 Results

6.3.1 General trends in institutional and professional delivery care

Table 6.2 shows the extent of changes in delivery care between the two surveys. The bivariate analysis for deliveries that were attended by a health professional by place of delivery and professional deliveries by place of birth are presented in tables 6.3 and 6.4 below. The results indicate that institutional delivery care remained low, and have declined between 2005/6 and 2010/11. About 68% of pregnant women had institutional delivery in 2010/11, down from about 71% in 2005-6, a decrease of about 3%. Equally, professional delivery decreased slightly from about 68% in 2005/6 to 67% in 2010/11. Although table 6.2 suggests that the proportion of health facility deliveries in 2010-11 (67.6%) is about similar to health professional delivery (67.4%), the two are not identical. For example, about half of the home deliveries (49.4%) were attended to by a health professional and almost (18.6%) of professional deliveries occurred outside a health facility (Tables 6.3 and 6.4). Thus it is important to examine factors associated with institutional delivery separately from factors associated with a professional delivery attendant.

 Table 6-2: Percentage Distribution of institutional and professional care delivery in

 Zimbabwe, 2005/6 and 2010/11 ZDHS

Outcome variable	ZDHS 2005-6		ZDHS 2010-11		
Place of delivery	% ⁰	n ^b	⁰⁄₀ª	n ^b	
Home	29.5	1251	32.4	1411	
Institution	70.5	2823	67.6	2986	
Professional					
Yes	68	2803	67.4	2985	
No	31.2	1271	32.6	1412	
Total	100	4074	100	4397	

^aWeighted; n^b:unweighted cases

6.3.2 Delivery attended by a professional delivery attendant and institutional delivery

While women's use of a professional birth attendant and delivery in an institution are expected to take place together, use of professional birth attendant can also occur in the community or at home (Gabrysch et al., 2009; Mutumbuka et al., 2012). Similarly, a delivery in an institution can occur without a professional health delivery attendant. Results in figures 6.3 and 6.4 are

presented in order to identify births that were attended by a health professional (doctor, midwife or nurse) and place of delivery. The bivariate results in table 6.3 indicated that about 2662 births occurred at home, of which about 49% (n=1316) were attended by a professional health attendant, suggesting that about 51% of births that occurred at home were attended by traditional birth attendants, family members or friends. While about half of the women who gave birth at home were attended by a professional health attendant, almost all (99%; n=5809) of those who gave birth in an institution were attended to by a professional health attendant. The results also showed that 95% (n=1346) of deliveries that occurred outside a facility were not attended to by a health professional, while only 5% of institutional deliveries were not attended by a health professional. About 19% of births delivered by a health professional occurred at home.

Figure 6-2: Percentages of birth deliveries that were attended by a health professional by institutional delivery



Source: Derived from Zimbabwe Demographic and Health Surveys (2005-6; 2010-11)



Figure 6-3: Percentages of health professional birth deliveries by institutional delivery

Source: Derived from Zimbabwe Demographic and Health Surveys (2005-6; 2010-11)

6.3.3 Bivariate association between institutional and professional delivery by individual level predisposing factors

Table 6.5 presents the proportion of women who gave birth in an institution and those who were attended by a professional health care provider (doctor, nurse, and midwife) according to their background predisposing factors. There are clear similarities between institutional and professional delivery and the trends for both outcomes match each other across the key factors as expected. This pattern was expected because births delivered at home are usually more likely to be delivered without assistance from a professional provider, whereas births delivered at an institution are more likely to be attended by a trained health professional. There was a strong association between institutional and professional delivery and background predisposing factors, with the exception of marital status. According to the results about, 91% of pregnant women gave birth in an institution and were also attended by a health professional in urban areas. Fewer women in rural than urban areas gave birth in an institution (57.7%) and only (53.3%) of these were attended by a professional health provider, suggesting that 4.4% of births that occurred in an institution in rural areas were not attended by a professional health provider.

This number may be as a result of pregnant women arriving at the institution very late and in labour, however this will be discussed further in the chapter. Age is significantly associated with both outcomes. Across the age groups, the lowest coverage of institutional and professional delivery was observed among older women aged 35 and above, which was averaging 60% for both outcomes, followed by those in the age group 15-24 years. The highest proportion of institutional and professional care uptake was observed among women aged 25-34 years and was averaging 70% for both outcomes. As expected, institutional and professional delivery were more common among women with secondary or higher educational attainment 77% (p<0.000) than those with primary or low educational attainment which was about 50% for both outcomes. Pregnant women affiliated with the apostolic faith group were less likely than others to have institutional and professional delivery, whilst a considerable majority of the women who had institutional and professional delivery were belonging to pentecoastal and protestant faith groups. The proportions of women who had institutional and professional delivery were higher among women who reported wanting the current pregnancy compared with those who reported wanting the child later or not wanted. Meanwhile, parity had a negative association with institutional and professional delivery. The negative association is shown by constant decrease in the proportion of institutional and professional delivery with an increase in the number of children ever born. For example, for women with one pregnancy about 77% gave birth in an institution compared to women with higher order birth of 3 or more pregnancies (59%). Meanwhile, the majority of women who reported having fewer than four ANC visits (71%) gave birth in an institution and were attended by a professional health provider compared with those who reported having four or more visits (66%) on average for both outcomes.

Table 6-3: Bivariate analysis of institutional and professional delivery by predisposing

individual factors

Variables	% Hospital delivery % ^a	<i>P</i> -Value	% Skilled attendant % ^a	<i>P</i> -value	All cases ^b
Residence		.001		.001	
Urban	91.3		91.3		2475
Rural	57.7		53.3		5996
Age groups		.001		001	
15-24	66.4		66.5		3176
25-34	70.1		69.8		3886
35+	60.9		59.8		1409
Marital status		.510		.247	
Never married	69.7		70.0		464
Married	67.2		67.0		7003
Widowed	68.6		66.7		321
Divorced	65.2		63.7		683
Women's education		.001		.001	
No/Primary	49.9		50.0		3128
Secondary/Higher	77.7		77.0		5343
Religion		.001		.001	
Catholic/others*	63.4		63.3		2286
Protestant	79.4		79.6		1427
Pentecoastal	80.4		80.1		1445
Apostolic	58.8		58.1		3313
Wanted child		.001		.001	
Then	69.3		69.4		5554
Later	62.0		62.3		2013
Not Wanted	65.9		61.7		904
Parity		.001		.001	
1	76.7		76.7		2503
2	70.1		70.4		2258
3+	59.0		58.0		3710
No.of ANC visits		.001		.001	
4 plus	65.5		65.7		4757
No	71.0		71.4		3714

^aWeighted; ^bunweighted cases

6.3.4 Bivariate association between institutional and professional delivery by enabling factors

Table 6.6 presents the association between institutional and professional delivery by different enabling and perceived risk/need factors. The factors in the table are classified into two groups. The first group relates to employment status, household wealth, women's autonomy, media exposure and access to a health facility as key enabling factors for institutional and professional delivery care. The second group relates to the association between a range of HIV factors with institutional and professional delivery. In the first group, only three variables relating to household wealth, physical access and media were significantly associated with delivery care. The highest proportion of women who had institutional and professional delivery were from the richest households (about 94%), followed by those from middle income households and the least were from poor households. Similarly, institutional and professional delivery care was more common among women who reported having higher media exposure and those with no problem accessing a health care facility. Meanwhile, the bivariate results indicate that women's autonomy and employment status are not important factors for delivery care. With respect to HIV enabling and perceived risk/need factors, the results show that institutional and professional delivery care were highest amongst women who reported having had antenatal HIV testing, those who were not concerned about confidentiality issues, with medium HIV awareness and stigma and those who did not know someone who died of HIV.

Table 6-4: Bivariate association between institutional and professional delivery by

Variables	% hospital delivery	<i>P</i> -Value	% Skilled delivery	<i>P</i> -value	All
	% ^a		% ^a		cases ^b
Resources, acces	s, autonomy (enabling	<u>g factors)</u>			-
Employment statu	15	.032		.016	
No	66.4		65.9		5448
Yes	68.7		68.5		3023
House Wealth		.001		.001	
Poor	55.3		54.7		5276
Rich	83.3		83.3		1807
Richest	93.6		93.8		1388
Media exposure		.001		.001	
Low	53.4		52.7		2998
Mediun	72.7		72.7		3330
High	77.5		77.2		2143
Distance to a heal	th facility problem	.001		.001	
No	80.3		79.9	l	2585
Yes	61.6		61.2	l	5886
HIV and enablin	ng and perceived risk/	need facto	rs		<u> </u>
ANC- HIV test		.001		.001	
No	55.6		55.0		3550
Yes	75.9		75.7		4921
HIV sero-status		.128		.070	
Negative	66.9		66.3		7316
Positive	68.7		68.8		1155
HIV awareness		.001		.001	
Low	67.0		67.1		2180
Medium	73.7		73.1		3765
High	57.9		57.5		2526
Knowsome who c	lied of HIV	.009		.008	
Yes	65.1		64.8		2476
No	68.1		67.8		5995
Confidential conc	ern	.001		.001	
Yes	64.7		59.4		4388
No	69.9		76.5		4083
HIV related Stign	na	.001		.001	
Low	65.0		65.8	1	1459
Medium	74.9		73.5		1566
High	65.5		65.1		5446

enabling and perceived risk/need factors

High65.5aWeighted;bunweighted cases

6.3.5 Multivariate multilevel analysis

A series of multilevel logistic regression models were estimated to investigate the determinants of delivery care and to estimate how much variation in institutional and professional delivery existed between communities. An empty model was initially specified to assess the baseline variation in the dependent variable including establishing a benchmark value for Deviance Information Criterion (DIC). In the second stage of analysis, three models assessed whether institutional and professional delivery care were associated with predisposing factors (model 1); enabling factors (model 2); and community factors (model 1). In the final stage of analysis, separate models were built for each year. This was in order to evaluate the trends and to identify factors that may have been attributable to change in the probability of having institutional and professional delivery care between the surveys. The analytical strategy of specifying sequential models (Valpine, 2012) employed here allowed for an evaluation of relative contribution of each model in explaining variance in delivery care utilisation as well as allowing an evaluation of DIC-value comparison of the predisposing individual level, enabling, need and community level models.

6.3.5.1 Multilevel logistic regression parameter estimates for institutional delivery

Results presented in table 6.7 uneartheddisparities in the use of institutional and professional delivery among pregnant women in Zimbabwe. Results in the empty model show that women were 21% less likely (OR=0.79, 95% CI=0.69-0.91) to have delivered in an institution in ZDHS 2010/11 than in 2005/6. The results also show that there is variation in the likelihood of giving birth in an institution across the communities, and this variation was significant as indicated by significant variance in the log odds of reporting giving birth in an institute across communities (estimated= 1.199, S.E= 0.097, i.e Z-value of 12.36>2.58 at 1% sig, level). Based on the estimated variance partition coefficient (VPC), the intra-community correlation was estimated at about 27%. This significant variation indicates the extent of variability in institutional delivery which is attributed to unobserved community factors. This means that 73% of the variability in institutional delivery was as a result of the composition of individual level characteristics.

Regarding individual predisposing factors in model 1, the results show that place of residence, maternal age; marital status, educational attainment, religion, wanted child and parity are significantly associated with institutional delivery. Women in rural areas are 82% less likely (OR=0.18, 95% CI=0.15-0.22) than in urban areas to have an institutional delivery. Significantly, more women aged 25-29 years (OR=1.71, 95% CI=1.46-2.00) and those over 35 years (OR=1.67, 95% CI=1.36-2.05) were more likely than young women aged between 15-24 years to have an institutional delivery. Marital status was also found to be positively associated with place of delivery. Being married or widowed increased the odds of institutional delivery by 1.38 and 1.88 points respectively. The greatest odds of likelihood of institutional were observed among women with secondary or higher educational attainment. The estimates in this group indicated that women were 2.07 times (OR=2.07, 95% CI=1.84-2.32) more likely to have institutional delivered, compared with women with primary or no education. Being a member of Protestant and Pentecostal religious groups increased the odds of delivering in hospital by 1.67 and 1.45 times than women of catholic religion affiliation. Higher order birth decreases the odds of giving birth in a facility. For example, when a pregnancy is third or higher order, women were 64% less likely to give birth in facility than the zero or first pregnancy. Meanwhile, women who reported wanting a child later or not wanted were 23% and 20% less likely to give birth in a facility compared with those who reported wanting the child now. The inclusion of individual level predisposing factors decreased the community variance to 17% with the proportional change in the odds of delivering in hospital of 37% across communities explained by individual level predisposing factors.

Model 2 included enabling and HIV perceived risk/need factors. All enabling factors included in this model were found to be associated with institutional delivery. The results indicated a high correlation between institutional delivery and household wealth and the likelihood of giving birth in an institution increasing by the increase in household income. Women in the rich wealth quintile were 2.66 times (OR=2.66, 95% CI =1.94-3.66) more likely to give birth in an institute compared with women from the poorest households. Similarly, a woman from the medium wealth quintile was 1.65 times (OR=1.65, 95% CI=1.36-2.01) more likely than poorer women to have institutional delivery. Also women with medium to high levels of media exposure were 1.23 and 1.26 times more likely than women with low media exposure to have delivered in hospital. Women who reported having problems in accessing a health facility were 21% less likely to have institutional delivery than their counterparts. Perhaps, one of the influential enabling factors relates to frequency of ANC visits. Significantly, women who reported having four or more ANC visits were 4.03 times (OR=4.03, 95% CI=3.22-5.04) more likely than those who reported they have four or less ANC visits to deliver in an institution.

Fewer HIV factors in model 2 appear to have influence an institutional delivery, antenatal HIV testing, confidentiality concern and knowing someone who died of HIV/AIDS in particular. The odds of institutional delivery were 2.78 times (OR=2.78, 95% CI=2.43-3.17) higher for women who were offered and accepted an HIV test during antenatal care, compared with their counterparts. Women who have confidentiality concerns and those who knew someone who died due to HIV were 1.19 and 1.25 times more likely than their counterparts to have institutional delivery. Having higher HIV stigma minimally decreased likelihood of giving birth in an institution by 3%. In model 2, employment status, women's autonomy, HIV status, and HIV awareness are not associated with institutional delivery care. After adjusting for all enabling and perceived risk/need factors, the odds of institutional delivery decreased substantially to 57% (OR=0.43, 95% CI=0.34-0.54) in the 2010/11 compared with ZDHS 2005/6. The model 2 variables reduced community level variation to 12%, with about 29% of proportional change in variation explained by enabling and HIV perceived risk/need factors.

At the community level, there is no evidence of an association between HIV stigma, awareness and prevalence in the community and delivery care. The pattern of an association between institutional deliveries observed in model 1 is also observed in model 2 and 3. After adjusting for all the independent factors individual predisposing factors (i.e place of residence, age, marital status, education, wanted child, religion) were significant explanatory variables for disparities in institutional delivery. These independent contributions of individual predisposing factors to institutional delivery were preserved after inclusion of all other factors. Enabling factors including household wealth, media exposure, access to health care and timing of ANC visits appear to be key determinants of institutional delivery, while on the other hand HIV factors including not having antenatal HIV testing, having confidentiality concern constraints and knowing someone who died because of HIV drive women to deliver in an institution. The final ICC remained 12%, suggesting that 88% of unexplained variation in institutional delivery is at individual level factors. The successive reduction in Deviance Information Criterion (DIC) from Model 0-3 demonstrates a better fit model over every previous one. Table 6-5: Average odds ratios of institutional delivery from multilevel logistic regression

Parameters		Model 1		Model 2		Model 3	
	OR	95% CI	OR	95% CI	OR	95% CI	
Predisposing factors							
Survey 2010	0.79	[0.69-0.91]*	0.43	[0.34-0.54]*	0.41	[0.32-0.53]*	
Place of residence (u	ırban)						
Rural	0.18	[0.15-0.22]*	0.39	[0.30-0.50]*	0.39	[0.30-0.50]*	
Age groups(15-24)							
25-29	1.71	[1.46-2.00]*	1.68	[1.43-1.97]*	1.68	[1.43-1.97]*	
35+	1.67	[1.36-2.05]*	1.77	[1.43-2.19]*	1.77	[1.44-2.19]*	
Marital status(never	married						
Married	1.38	[1.07-1.79]*	1.39	[1.03-1.88]*	1.39	[1.03-1.88]*	
Widowed	1.88	[1.29-2.74]*	1.75	[1.19-2.57]*	1.75	[1.19-2.57]*	
Divorced	1.10	[0.81-1.49]	1.11	[0.81-1.52]	1.11	[0.81-1.52]	
Women's education	(primar	y)					
Secondary	2.07	[1.84-2.32]*	1.69	[1.50-1.91]*	1.69	[1.50-1.91]*	
Religion(Catholic)							
Protestant	1.67	[1.39-2.01]*	1.52	[1.26-1.84]*	1.53	[1.26-1.84]*	
Pentecoastal	1.45	[1.20-1.74]*	1.30	[1.08-1.57]*	1.30	[1.07-1.57]*	
Apostolic	0.92	[0.81-1.05]	0.94	[0.82-1.07]*	0.94	[0.82-1.07]	
Wanted child (now)							
Later	0.77	[0.68-0.88]*	0.78	[0.68-0.89]*	0.78	[0.68-0.89]*	
Not wanted	0.80	[0.66-0.95]*	0.77	[0.64-0.92]*	0.77	[0.67-0.92]*	
Parity							
2	0.59	[0.50-0.70]*	0.60	[0.50-0.71]*	0.60	[0.50-0.70]*	
3+	0.36	[0.30-0.44]*	0.38	[0.32-0.47]*	0.38	[0.32-0.47]*	
Wealth (poor)							
Medium			1.65	[1.36-2.01]*	1.64	[1.35-2.00]*	
Rich			2.66	[1.94-3.66]*	2.66	[1.94-3.65]*	
Employment status((No)						
Yes			0.93	[0.83-1.06]	0.94	[0.83-1.07]	
Media exposure (Lo	w)						
Medium			1.23	[1.07-1.40]*	1.23	[1.07-1.40]*	
High			1.29	[1.11-1.51]*	1.29	[1.10-1.51]*	

models (95% confidence intervals are given in square brackets)

Continued

Table 6-5: continued

Decision making (h	usband)					
Women alone			0.98	[0.80-1.20]	0.98	[0.80-1.20]
Jointly			0.95	[0.80-1.14]	0.95	[0.80-1.13]
Perceived distance p	oroblem	(no)				
Yes			0.79	[0.69-0.91]*	0.79	[0.69-0.91]*
Timing of ANC visi	ts (no)					
1 st trimester			3.85	[3.05-4.85]*	3.83	[3.14-4.67*
No. of ANC visits (no)					
4 plus			4.03	[3.22-5.04]*	4.14	[3.28-5.21]*
ANC-HIV test (no)						
Yes			2.78	[2.43-3.17]*	2.77	[2.43-3.16]*
HIV sero-status (neg	gative)					
Positive			0.99	[0.85-1.17]	1.01	[0.86-1.19]
HIV awareness (low	/)					
Medium			1.15	[1.00-1.32]	1.14	[0.99-1.31]
High			0.85	[0.74-0.99]	0.84	[0.72-0.98]
HIV stigma(low)						
Medium			1.06	[0.87-1.28]	1.04	[0.86-1.27]
High			0.97	[0.79-1.19]*	0.95	[0.77-1.17]*
Confidential concer-	n (yes)					
No			1.18	[1.05-1.32]*	1.18	[1.05-1.32]*
Knows someone with	th AIDS	(no)				
Yes			1.25	[1.05-1.49]*	1.25	[1.05-1.49]*
Contextual Factors	5					
HIV awareness-gm	-				1.22	[0.73-2.04]
HIV stigma-gm					1.26	[0.71-2.24]
HIV prevalence-gm					0.68	[0.31-1.46]
Random effects pa	rameter	ſS				
Community-level						
Variance (SE)	0.663	[0.074]*	0.465	[0.062]*	0.447	[0.061]*
(VPC)=ICC (%)		17		12		12
(PCV)(%)		37		29		0
Model fit statistics						
Cluster		679		679		675
DIC		6882		6882		6882

Model 0: empty model (survey year only included), ICC=27%

6.3.5.2 Multilevel logistic regression parameter estimates for professional delivery

Table (6.8) presents the odds ratios (ORs) for professional delivery. The results in the empty model show that there is a variation in the likelihood of having a health professional delivery across communities. The variation was significant (estimated= 1.219, S.E= 0.131, i.e Z-value of 9.3>2.58 at 1% sig, level). Based on the estimated variance partition coefficient (VPC), the intra-community correlation was estimated at about 27%. This significant variation indicates the extent of variability in professional delivery attributed to unobserved community factors. This means that 73% of the variability in professional delivery was as a result of the composition of predisposing individual level characteristics. Similar patterns observed in institutional delivery are also observed in professional delivery. With all variables excluded, the results show that pregnant women were 20% less likely to have been attended by a professional in 2010/2011 than in 2005/2006 survey.

When all individual predisposing factors were included in model 1, the results show that pregnant women from rural areas were 83% less likely to have professional delivery than those from urban areas. Part of the rural disadvantage is explained by enabling factors, but even after controlling for enabling factors, rural residence is still associated with 62% lower odds of institutional delivery than urban residence. Significantly, older women aged 25-29 and 35 and above were 1.77 and 1.86 times more likely than young women aged 15-24 years to have a health professional delivery, respectively. Women with secondary or higher educational attainment were found to be 2.05 times more likely to have a professional health delivery compared with those with primary or no educational attainment. Furthermore, widowed women were 1.48 times more likely than never married to have a professional delivery. Women affiliated to the Pentecostal religious group were 1.48 times more likely than women in other groups to have professional health care delivery. Conversely, women who reported wanting the child later or not wanted were on average 24% less likely to have professional delivery than those who wanted the child now. Higher order birth decreased the odds of professional birth delivery. For example when a pregnancy is third or higher order parity, women were 66% less likely to have a professional health delivery than the first pregnancy. In comparison to model 0, the ICC decreased to 14 from 27% and the PCV was 26%, suggesting that the clustering of health professional delivery across communities was not as a result of the composition of the community by individual predisposing factors.

Enabling factors such as household wealth, employment status, women's decision making, media exposure, and accessibility to a health care facility, timingof ANC visits and HIV perceived risk/need factors including antenatal HIV testing, HIV sero-status, HIV-related stigma, awareness, confidentiality concern and knowing someone who died because of AIDS/HIV were all added in model 2. The results suggest a high correlation between household wealth and professional delivery. Relative to women from poor households, the odds of having professional delivery were 1.70 and 2.76 times higher for women from medium to rich households, respectively. Similarly, having four or more ANC visits increases significantly the odds of professional delivery. The results reveal that women who had four or more ANC visits were 5.90 times more likely than those who did not to have professional delivery. However, women who reported having a big problem accessing health care were 19% less likely to have professional delivery compared with those who have easier access to health facilities. While individual predisposing and enabling factors appear to have greater influence in professional delivery, HIV perceived risk/need factors including antenatal HIV testing, awareness, confidentiality concern and knowing someone who died of AIDS/HIV are also key determinants of delivery care. Unlike in institutional delivery, HIV awareness is found to be associated with professional delivery, but not with HIV stigma. The results suggest that pregnant women with higher HIV awareness were 16% less likely than those with low awareness to have professional delivery. The effects of other HIV factors on professional delivery are comparable with institutional delivery. Antenatal HIV testing is significantly associated with professional delivery, such that those who were offered and accepted HIV testing were 2.75 times more likely than those who did not to have a professional delivery. Meanwhile, those who reported knowing someone who died of AIDS/HIV and those with no confidentiality concern about their personal HIV-related issues were 1.19 and 1.23 times more likely to have professional delivery compared with their counterparts. Compared to ICC in model 0 and 1, the ICC in model 2 decreased further to 12% and PCV from 30% to 26%. This suggests that 12% of variance in professional health delivery across the communities is explained by unobserved factors at the community level and 88% is explained by variance as a result of unobserved individual level factors. There is no evidence of an association between community level factors included in model 3 with professional delivery. This suggests that HIV awareness, stigma and prevalence in the communities where these pregnant women reside do not influence their utilisation of professional delivery. The inclusion of model 3 factors did not have much effect on the ICC which remained stable at 12%.

Table 6-6: Average odds ratios of professional delivery from multilevel logistic regression
models (95% confidence intervals are given in square brackets)

Parameters	Ν	Aodel 1	Model 2		Model 3	
Farameters	OR	95% CI	OR	95% CI	OR	95% CI
Predisposing factor	rs					
Survey 2010	0.80	[0.69-0.92]*	0.46	[0.36-0.57]*	0.44	[0.34-0.56]*
Place of residence (urban)						
Rural	0.17	[0.14-0.21]*	0.38	[0.29-0.49]*	0.38	[0.29-0.49]*
Age groups(15-24)						
25-29	1.77	[1.52-2.07]*	1.74	[1.49-2.04]*	1.74	[1.49-2.04]*
35+	1.86	[1.51-2.28]*	1.98	[1.60-2.45]*	1.99	[1.61-2.45]*
Marital status(never	married					
Married	1.28	[0.98-1.66]	1.27	[0.94-1.72]	1.27	[0.94-1.72]
Widowed	1.56	[1.06-2.26]*	1.43	[0.97-2.10]	1.42	[0.97-2.09]
Divorced	0.94	[0.69-1.28]	0.95	[0.69-1.30]	0.95	[0.69-1.30]
Women's education	(primary	r)				
Secondary	2.05	[1.82-2.31]*	1.68	[1.48-1.89]*	1.67	[1.48-1.89]*
Religion(Catholic)						
Protestant	1.74	[1.45-2.09]8	1.59	[1.32-1.92]*	1.59	[1.32-1.92]*
Pentecoastal	1.48	[1.23-1.79]*	1.33	[1.10-1.61]*	1.33	[1.10-1.61]*
Apostolic	0.93	[0.81-1.06]	0.94	[0.82-1.08]	0.94	[0.82-1.08]
Wanted child (now)						
Later	0.75	[0.66-0.85]*	0.76	[0.66-0.87]*	0.76	[0.66-0.87]*
Not wanted	0.74	[0.61-0.88]*	0.71	[0.59-0.85]*	0.71	[0.59-0.85]*
Parity						
2	0.59	[0.50-0.69]*	0.59	[0.50-0.70]*	0.59	[0.50-0.70]*
3+	0.34	[0.29-0.41]*	0.36	[0.30-0.44]*	0.36	[0.30-0.44]*
Wealth (poor)						
Medium			1.70	[1.40-2.07]*	1.70	[1.40-2.07*
Rich			2.76	[2.00-3.80]*	2.76	[2.00-3.80]*
Employment status	(No)					
Yes			0.96	[0.51-1.09]	0.97	[0.85-1.10]

Continued

Media exposure (Lo	wl)					
Medium			1.21	[1.06-1.39]*	1.21	[1.06-1.39]*
High			1.28	[1.09-1.50]*	1.28	[1.09-1.50]*
Decision making (hu	usband)	1.				
Women alone	·		0.95	[0.78-1.17]	0.96	[0.78-1.17]
Jointly			0.98	[0.82-1.17]	0.98	[0.82-1.17]
Perceived distance p	roblem (no)				
Yes			0.83	[0.72-0.95]*	0.83	[0.72-0.95]*
Timing of ANC visi	ts (no)					
1 st trimester			4.03	[3.22-5.04]*	4.14	[3.28-5.21]*
No.of ANC visits (r	10)	1.				
4 plus			5.89	[4.68-7.46]*	5.90	[4.82-7.24]*
ANC-HIV test (no)						
Yes			2.73	[2.37-3.10]*	2.75	[2.41-3.13]*
HIV sero-status (neg	gative)					
Positive			0.99	[0.84-1.16]	0.10	[0.85-1.18]
HIV awareness (low	r)					
Medium			1.13	[0.98-1.30]	1.13	[0.98-1.30]
High			0.84	[0.72-0.97]*	0.84	[0.72-0.97]*
HIV stigma(low)						
Medium			1.03	[0.85-1.25]	1.01	[0.3-1.23]
High			0.91	[0.74-1.11]	0.89	[0.73-1.10]
Confidential concern	n (Yes)					
No			1.19	[1.06-1.33]*	1.19	[1.06-1.33]*
Knows someone wit	h AIDS	(no)				
Yes			1.23	[1.02-1.45]*	1.21	[1.02-1.45]*
Contextual Factors	5					
HIV awareness-gm					0.73	[0.33-1.51]
HIV stigma-gm					1.07	[0.60-1.92]
HIV prevalence-gm					1.06	[0.75-2.13]
Random effects par	rameters	5				
Community-level						
Variance (SE)	0.758	[0.097]*	0.544	[0.084]*	0.526	[0.083]*
(VPC)=ICC (%)		19		14		14
(PCV)(%)		30		26		0
Model fit statistics						
Cluster		406		406		406
DIC		4395		4276		4276

Table 6-6 continued

Model 0: empty model (survey year only included), ICC=27%

6.4 Discussion

The main objectives of the chapter were to identify individual predisposing and community level determinants of institutional and professional delivery care focusing on predisposing, enabling and perceived risk/need factors (including the effects of HIV); and to compare factors associated with institutional and professional delivery care. Firstly, while interpreting the findings, the reader should bear in mind the potential limitation of the data used in the analysis. This relates to the problem of causality since the cross-sectional nature of the data makes it impossible to determine the time sequence of key events of interest, that is, whether the use of institutional and professional delivery preceded various risk factors. The cautionary note that correlation does not equal causation applies here. Overall the results show that institutional and professional delivery are not improving as indicated by lower odds in the 2010/2011 than in 2005/2006 surveys. This may be due to a number of factors relating to general decline in health care systems over the past two decades as a result of dollarization, lack of financial/human capital and material resources (Loewenson et al., 2012). Dollarisation was the shift from using the Zimbabwe dollar, whose value had collapsed due to hyper-inflation, to the US dollar as the national currency in 2009 (Makate and Makate, 2017). Dollarisation resulted in a multifaceted health care crisis due to a shortage of hard currency and these included a health care staff exodus, high hospital fees, shortage of drugs, and closure of important health care institutions and health promotion programs, including key maternal health waiting shelters for pregnant women which forms part of the maternal health care delivery in rural areas (Chirwa et al. 2015, Zimsta, 2011).

While the general decline observed in institutional and professional delivery care utilisation maybe associated with a general collapse of the economy and the health care system, results from this study have revealed variations in institutional and professional delivery among different populations. Measures of individual predisposing factors (i.e place of residence, age, marital status, religion, education, wanted child and parity) showed significant association with institutional and professional delivery care. The results suggest that on average the odds of women having institutional delivery were higher in urban than rural areas. This observation applies to professional delivery too. This relationship does not change when enabling and perceived risk/need factors are controlled for, suggesting that institutional and professional

delivery care are on average 61% lower for pregnant women in rural areas than those from urban areas. The evidence presented here supports the view that geographical accessibility might influence delivery care by predicting how fast and easily maternal health can be reached for primary delivery care or in the case of obstetric complications (Claudia, 2013). Almost all studies that investigated place of residence have reported lower delivery care in rural areas and among women with longer distances to a health facility (Faye and Niane, 2011; Moyer and Mustafa, 2013; Rockers et al., 2009; Sanga et al., 2017; Sharma et al., 2015). Place of residence might also be a proxy of socio-economic status as the urban population is often wealthier and better educated (Gregson et al., 2011). However, there is a suggestion that even when controlling for poverty and education, health in urban areas is better than in rural areas (Matthews et al., 2010). It was hypothesised that a proximity to affluence neighbourhood, which is more evident in urban than rural areas, may help to sustain neighbourhood social organization which in turn positively affects health (Golden et al., 2012). The authors have stated that the presence of affluent members of society may attract the attention of politicians, and government agencies, and help to win development in the community including maternal health services for delivery care. This may also provide a strong base for civic and other community-based activities such as HIV/AIDS campaign encouraging facility delivery (Bertrand et al., 2006). Thus, socioeconomic heterogeneity, one of the hallmarks of urbanisation, may bring benefits such as health care and education within the reach of everyone including the more disadvantaged urban residents (Shetty et al., 2010). The picture of an urban health advantage in Zimbabwe becomes more relevant when considering that pregnant women in rural areas must walk a substantial distance for delivery care, requiring significantly longer travel times to reach care than their counterparts in urban areas (Buzdugan et al., 2015). It was reported that, even if the women can afford to pay for delivery care in rural areas in Zimbabwe, they cannot always get to the facility for delivery care, because distance makes it difficult to get to the facility when they need health services (Loewenson et al., 2012).

The age of pregnant women is another key individual predisposing factor that has been found to be associated with both institutional and professional delivery. The disparity by age groups resonates with other previous studies (Ahmed et al., 2010; Buzdugan et al., 2015; Gustav et al., 2014; Jayleen et al., 2016), confirming that being young is associated with low odds of institutional and professional delivery compared to older women. Such disparity has been attributed to older women having greater autonomy in accessing maternal health care than

young women (Moyer and Mustafa, 2013). Higher odds of institutional and professional delivery for older women may also mean that older women have benefited enormously from information and education campaigns as well as community mobilization efforts aiming to increase utilization of HIV and maternal health care services (Loewenson et al., 2012). As a result of stigma and discrimination against adolescent pregnancy, young mothers may fail to attend ANC care, which benefits pregnant women by receiving healthinformation such as pregnancy complications, mother to child HIV transmission, breastfeeding and the need for facility delivery (Murire et al., 2012), health education in Zimbabwe has tended to emphasize certain sexual and maternal health practices, including no sexual engagement at an early age (McCoy et al., 2014), and the emphasis on 'no early sex' campaign may have inadvertently prevented teenage mothers from accessing delivery care (Maruva et al. 2014). Generally, access to maternal care for young women can be difficult because they are despised and stigmatised and thus some prefer to deliver at home because they anticipate a negative provider attitude (Renjuet al., 2010).

The results show that being married and widowed were associated with higher odds of institutional delivery than never married women, but not for professional delivery. The finding that these women are more likely to have institutional delivery has been noted in other studies (Aremu et al., 2011; Buzduganet al., 2015; Eijik et al., 2006) This finding is likely to relate to stigma and discrimination that never married (usually teenagers) experience when they access maternal health care services (Ngome and Odimegwu, 2014). Researchers such as McCoy et al. (2015) and Perez et al. (2008) have pointed out that the health care system in Zimbabwe is tailor-made to suit married and older women and adolescentpregnancy is associated with wayward behaviour and is frowned upon even in health facilities in Zimbabwe, and this can act as a barrier to institutional delivery. Some studies have indicated that being married improves socio-economic status of a woman (Mucheto et al., 2011; Sibanda et al., 2013), making it easier for this group of women to pay for costs associated with institutional delivery care. Therefore, the finding that married women were more likely than never married women to have institutional delivery may relate to their improved household income (Jayleen et al., 2016). Being widowed was also found to be more influential for institutional delivery, probably via its influence on female autonomy and status (Kohler et al., 2014). Although widowed mothers are more likely to be living in female-headed households and may be in an

unfavourable economic position, they may enjoy greater autonomy than those who are married (Magadi, 2011). The finding that widowed women were more likely to have institutional delivery may mean that widowed women in Zimbabwe enjoy freedom of choice and have greater autonomy than never married women, who usually live under the control of their parents (Houweling et al.2007). The results suggestthat while marital status was associated with institutional delivery, there was no evidence of an association between marital status and professional delivery once other individual and community predisposing, enabling and perceived/need factors are controlled for. This may relate to poor quality of health care services delivery and this supports evidence presented in Table 6.1 indicating that 5% institutional delivery was not attended by a professional birth attendant in Zimbabwe. Muchabaiwa et al (2012) reported that pregnant women deliver their babies in a facility with no help of a professional assistant because of a lack of staff and equipment or they arrive very late in labour due to distance or problems of getting transport to health centres.

Education builds awareness of HIV and prevention methods (Chiu et al., 2008). There is evidence of better utilisation of health care services among individuals with higher educational attainment than those with primary or no education (Kayeyi et al., 2014; McCoy et al., 2014; Mitchel et al., 2010). The observed higher odds of institutional and professional delivery among women with secondary or higher educational attainment presented here corroborates with previous studies. This may be as a result of differences relating to higher income, better exposure to media and urban residence among women with higher educational attainment than those with no education , all of which are associated with institutional and professional delivery (Aremu et al., 2011; Babalola et al. 2009; Mopembeni et al., 2007; Wanjira et al., 2011). Women with higher educational attainment may also have better knowledge of HIV transmission modes and are more likely to deliver in an institution to protect their infants against mother to child HIV transmission if they know their HIV positive status or if they perceive themselves to be at risk of HIV transmission (Stephensen et al., 2013).

Religion is often considered as a cultural predisposing background factor and is thought to influence beliefs, norms and values in relation to access to health care service. Traditional religious groups including Catholics and the apostolic faith may avoid institutional and professional delivery due to cultural requirements of privacy; or because of specific requirements around delivery position or having to undergo health screening including HIV

(Sahlu et al., 2014). Therefore the findings that women who are affiliated to protestant and Pentecostal faith groups were more likely to use delivery care services maybe credible, because they are more liberal and receptive to modern health care service, including antenatal HIV testing (Luginah et al., 2005).

Consistent with other studies (Fotso et al., 2009; Mrisho et al., 2007) the results show that women with lower birth order were more likely than women with higher birth order to have institutional and professional delivery care. This could be as a result of the fact that women with high parity relied on their experience from a previous pregnancy and therefore feel no need for care-seeking (Ngome and Odimegwu, 2014). Alternatively, some might have experienced difficulties in accessing delivery care due to constraints related to their other responsibilities for their other children and financial resources (Anyait et al., 2012).

This study found remarkable influence of enabling and HIV perceived risk/need factors on both institutional and professional delivery. Enabling factors relating to household wealth, media exposure, frequency of ANC visits and antenatal HIV testing are largely associated with higher odds of institutional delivery. These findings persisted even after taking into account HIV perceived risk/need and community level factors. This observed behaviour also applies to professional delivery. Significantly, household wealth was found to be one of the major enabling factors in institutional and professional delivery when all other factors are controlled for. For example, women from the richest households were 2.66 and 2.76 times more likely to have institutional and professional delivery, respectively. This result is expected since pregnant women are required to pay user fee and additional out of pocket payments for transport and medical supplies in Zimbabwe (Takarinda et al, 2016), inhibiting women from poor households from institutional delivery (Ahmed and Mukanga et al., 2010). The fact that the odds of home delivery are higher for women from poor households may also suggest that women of low socioeconomic status fear bad treatment at health facilities. Makate and Makate (2016) have pointed out that health workers often treat impoverished patients harshly compared with the treatment they give to richer and educated women. This finding means that better socioeconomic status enables pregnant women to overcome their socioeconomic barriers to institutional and professional delivery care such as transport, distance, lack of education and media exposure (Ahmed et al., 2010; Magadi et al., 2000), all of which are key factors that enable women to have access to health care services.

Not having access to media also impedes institutional and professional delivery (Tarekegn et al., 2014). Frequently, listening to radio, reading newspapers and watching television enables women to learn more about delivery risks, including mother to child HIV transmission during delivery as well as the availability of PMTCT programmes (Kevany et al., 2012; Sibanda et al., 2013). The result that women with higher levels of access to media were more likely to use delivery care could probably be due to high levels of access to information on importance of giving birth in a facility, location of such facilities, HIV awareness and the availability of PMTCT programmes among pregnant women in Zimbabwe, particularly those in urban areas. The study also found a negative association between delivery care and access to health facilities. Previous studies have suggested accessibility problems (distance, time and cost of transport to the nearest facility) influence delivery care (Faye et al., 2011; Fotso et al., 2009) and ability to pay for transport costs may be related to household wealth and decision-making (Titaley, 2010). The fact that problem of access to delivery care continued after household wealth and decision-making are controlled for suggests women's socio-economic vulnerability could result in home-based delivery. It could also be a result of impromptu labour, as some women do not know the actual date of delivery, especially among young mothers, and this along with money, distance and transport problems may act as a barrier to delivery care (Gabrysch et al., 2011).

Perhaps the most important finding in this chapter relates to the impact of antenatal care on delivery care. The results that those women who had four or more ANC visits and those who were tested for HIV during pregnancy were more likely to have institutional and professional delivery than their counterparts indicate that an integration of HIV and antenatal care maybe a critical pathways institutional and professional delivery uptake. The study findings show that ANC in the first trimester, four or more ANC visits and antenatal HIV testing were important factors enabling a women access to institutional and professional delivery, probably due to HIV information and education on pregnancy they receive during ANC visits (Ahmed et al., 2010). A study in Ghana found that women who frequent ANC were more likely to deliver in a facility than those who do not present at all or have infrequent visits as a result of their accumulation of knowledge of prevention of mother to child HIV transmission. In other words, access to antenatal care enables expecting mothers to learn more about delivery care, MTCT

and associated programmes. The findings did support the hypothesis that HIV and ANC services can enhance delivery care.

The study had postulated, from the existing Health Belief Model (Iriyana, 2007), that measures of individual- level HIV perceived risk and need factors relating to HIV sero-status, awareness, stigma, confidentiality concern and knowing someone who died of HIV would affect women's behaviour in seeking institutional and professional delivery. Although, one might expect the odds of institutional and professional delivery care to differ by HIV-sero status, since previous studies have suggested that HIV positive women are more likely to be more aware of PMTCT due to frequent contact with health providers (Buzdugan et al., 2015), the results provided evidence that HIV-sero status does not have an effect on institutional and professional deliver care, once important individual-predisposing and enabling factors, as well as other HIV factors are controlled for. Given the widespread campaign advising HIV positive mothers to deliver in a health facility, one would expect HIV positive mothers to give birth in a facility at least if they know their HIV status (Uwakwe et al., 2016). However, in addition to their economic disadvantages, HIV positive pregnant women face further barriers to delivery care relating to stigma and discrimination (Jayleen et al., 2016; Turan et al., 2012). If they have not yet disclosed their HIV positive status to the community, HIV positive women may be reluctant to deliver in an institution to avoid involuntary disclosure, stigma and discrimination (Mucheto et al., 2011). The results found here that pregnant women with HIV's confidentiality concerns are less likely to have delivery care corroborates with the assertion that non-disclosure of HIV status is a key barrier to institutional and professional delivery (Gourlay et al., 2016; Turan et al.,2012). However, the result that HIV sero-status is not significantly associated with delivery care is not surprising given that significantly about 44% of pregnant women were not tested for HIV during antenatal care and were not aware of their HIV-sero-status.

There is no evidence of a significant association between HIV/AIDS stigma and professional delivery, but pregnant women with higher HIV/AIDS stigma are less likely to have an institutional delivery. Conversely, HIV/AIDS awareness is not an important factor for institutional delivery, but pregnant women with high levels of HIV/AIDs awareness are less likely to have a professional delivery. The results that higher HIV stigma is associated with low uptake of institutional delivery is consistent with previous studies (Buzdugan et al., 2015; Turan et al., 2012). Assuming that HIV-stigma is consistent with women's HIV-sero status,

pregnant women who find they are HIV negative during ANC or if they hold negative attitudes about people living with HIV, may not want to deliver in a facility in order to avoid the assumption that they are HIV positive or they can simply consider themselves at low risk of HIV transmission during labour and delivery (Martin-Herz et al., 2006). The results that pregnant women who hold higher perceptions of HIV-related stigma are delivering at home in Zimbabwe is not good news because pregnant women remain at risk of HIV-infection throughout pregnancy (Heemelar et al., 2014), thus more efforts should be focused on reducing HIV-related stigma among women who test HIV-negative during ANC. For those who test HIV-positive during ANC, other considerations, including her perceptions of the risk of mother-to child transmission of HIV or whether or not she has disclosed her HIV status to her husband and family, may be at the fore (Mlay et al., 2008; Turan et al., 2015). The importance of avoiding unwanted disclosure of HIV status in HIV-positive women's decisions about where to give birth has been reported in other studies (Manzi et al., 2014; Rochat et al., 2006). Other studies suggested that some HIV positive women prefer a home delivery or somewhere their status is not known to avoid involuntary disclosure (Mucheto et al., 2011). The fact that pregnant women who personally know someone who died of AIDS were more likely to have institutional and professional delivery than their counterparts of similar background who had no personal acquaintance with AIDS victims suggests that personal acquaintance with AIDS may have led to taking appropriate action to prevent mother to child HIV transmission (Magadi and Desta, 2011). Despite considerable random variation at the community level, the three community-level factors included in the models were not significantly associated with both institutional and professional delivery, possibly due to the fact that the majority of people in the communities are now aware of HIV and that HIV is now a treatable disease (Kayeyi et al., 2009). A qualitative study by Musheke et al (2013) found that pregnant women who have received information about HIV during antenatal care have reduced negative attitudes towards people living with HIV and need to prevent HIV transmission to their infants outweighs their perceptions of discrimination against HIV positive women in the community (Mason et al., 2015).

One of the specific objectives of this study was to compare the determinants of institutional and professional delivery care. A comparison of determinants of institutional and professional delivery suggests that overall risks factors among women who had institutional delivery are more or less similar, but with some exceptions (see appendix 6.1). One of the notable

differences pertains to the odds of professional delivery which are much greater among older women compared with institutional delivery. For example, pregnant women who are over 35 years of age are 1.77 times more likely to have institutional delivery than younger women below the age of 24 years, women above 35 years of age have 1.99 times higher odds for institutional delivery. More telling than other factors was the fact that marital status (i.e being married or widowed) was positively associated with institutional delivery, but not with professional delivery. Besides age and marital status, the biggest variation between institutional and professional delivery is observed among women who had four or more ANC visits. Although, pregnant women who have four or more ANC visits are more likely to have both institutional and professional delivery, the results show that more pregnant women in this group gave births with a professional assistant than have institutional delivery. Women who had four or more ANC visits were 5.90 and 4.14 times more likely to have professional and institutional delivery, respectively, with 1.76 points differences in favour of professional delivery. HIV stigma is associated with low uptake of institutional delivery, but not with professional delivery. The observed disparities underscore the need for program specific response in order to increase uptake of both institutional and professional delivery, given that the data indicates that professional delivery uptake is lagging behind institutional delivery care.

6.5 Conclusion

The multilevel results show significant variations in institutional and professional delivery care across the communities in Zimbabwe. About 12% and 14% of the total variation in institutional and professional delivery are attributed to community-level factors. These variations are partly explained by individual and community characteristics. Controlling for community HIV prevalence, awareness and stigma, the effect of individual predisposing, enabling and perceived risk factors remain the same for both institutional and professional delivery care. This may suggest that HIV factors relating to community prevalence, stigma and awareness do not explain the differences in delivery care that are observed. The models suggest that predisposing individual level factors (i.e place of residence, age, marital status, education, religion, parity and age at marriage) were the superior and important factors for predicting the institutional and professional delivery and did explain most of the variance, whilenthere was also a reasonable amount of variance explained by enabling factors (household wealth, media exposure, access to health care, HIV testing during pregnancy, timing of ANC, HIV stigma, AIDS awareness). The results provide further evidence that uptake of antenatal care services i.e

ANC in the first trimester, ANC HIV testing and four or more ANC visit are key important factors for delivery care. Given that having ANC anvisit in the first trimester, four or more visits and having ANC HIV testing were significantly associated with institutional and professional delivery, encouraging women to initiate ANC care and HIV testing during pregnancy may enable pregnant women to have institutional and professional delivery. More importantly perhaps, focus should be on reducing involuntary HIV status disclosure to families and friends who accompany pregnant women during labour and delivery. Asking permission from pregnant women to allow families and friends during delivery care may reduce stigma associated with involuntary disclosure. Finally, the study found little evidence of the impact of integrating HIV with delivery care, except the negative effectof HIV stiga on institutional delivery which may be as a result of pregnant women being concerned with how their information relating to HIV is held.

7 CHAPTER SEVEN: KNOWLEDGE OF HIV STATUS AND CONTRACEPTIVE METHODS CHOICE AMONG WOMEN IN ZIMBABWE

7.1 Introduction

Having argued in the previous chapters that HIV has an impact on maternal health care services; it is also possible to make a link between this chapter and the whole argument that HIV has an impact on contraceptive methods choice. This chapter thereforeexamineswhether knowledge of HIVserostatus has an impact on contraceptive methods choice among women. This is crucial for policy and may help by improving access to effective family planning for women who know their HIV status and do not wish to become pregnant; and to promote long-acting reversible and dual contraceptive method use strategies among women who require condoms for HIV/STI prevention. The chapter begins with background and literature review on contraceptive methods choice, and then moves on to explain the methodology that has been used, arguing that multinomial regression analysis is the most suitable approach to provide answers to the question of contraceptive method choice among women who know their HIV sero-status in Zimbabwe. The last part of the chapter presents results, discussion and conclusion.

7.2 Background and Objectives

The impact of knowledge of HIV sero-status on pregnancy intention and contraceptive choice is attracting considerable research interest in high-HIV burden SSA settings in this era of widespread antiretroviral treatment availability. The high rates of reported unplanned pregnancy and poor contraceptive choice in Zimbabwe underscores an urgent need for better access to effective contraceptive methods for women who know their HIV sero-status (ZNFPC, 2016). While there is suggestive evidence that knowledge of HIV sero-status in some parts of SSA countries is associated with more frequent condom use for disease prevention and effective contraceptive methods for unplanned pregnancy, little is known about contraceptive choice among women who know their HIV status in Zimbabwe. With an increasing rate of HIV testing and receipt of test results, it is possible to examine whether knowledge of HIV serostatus has an impact on contraceptive method choice among women, and this may help by improving access to effective family planning for women who know their HIV status and do not wish to become pregnant; and to promote long-acting reversible and dual contraceptive method use strategies among women who require condoms for HIV/STI prevention in Zimbabwe.

The Zimbabwe Demography and Health survey (2010) indicated that 38% of women were ever tested for HIV and of this 96% received their HIV test results. Women who know their HIV status have better contraceptive decision-making regarding safe sex and choice of birth control methods that prevent both HIV transmission and unplanned pregnancies (Church et al., 2014, Johnson et al., 2009). Thus, the decision on contraceptive methods choice is paramount, particularly in countries with higher HIV prevalence rates such as Zimbabwe. Women who choose not to use effective contraceptive methods are at risk of unplanned pregnancies and HIV transmission (Habte and Namasasu, 2015). While a woman's socio-economic situation may influence desire to have more or fewer children (Askew and Berer, 2003), HIV may have a role in the reproductive preferences in countries with high HIV prevalence. Magadi and Agwanda (2010) have stated that HIV may have a role in the reversal of reproductive preference for women who perceive themselves to be at risk of HIV infection because they anticipate giving birth to a child who may be at risk of mother to child HIV transmission. Such women, including those who have personal acquaintance with HIV victims, particularly if they witnessed death of a child as a result of perceived HIV are more likely to adopt more permanent contraceptive methods to avoid unintended pregnancy (Laher et al., 2009). They may also use condoms if they think they are at risk of HIV transmission themselves (Magadi and Magadi, 2017). However, behaviours may also be dependent on whether or not a woman knows her HIV sero-status (Taulo et al., 2009). While extant literature has shown a positive relationship between women's knowledge of their HIV positive status and increased contraceptive use (Elul et al., 2009, Mayondi et al., 2016, Johnson et al., 2009, Keogh et al., 2012), it is not yet clear to what extent women who know their HIV positive status may have contraceptive needs that differ from those who are HIV negative in the Zimbabwean context. There is also a lack of information regarding groups of women after HIV diagnoses which are more likely to use effective contraceptive methods to prevent unplanned pregnancy or condoms to prevent HIV transmission. Literature reviewed showed that knowing your HIV status after diagnosis may temporarily alter women's fertility intention in favour of delaying or limiting birth (Hoffman et al., 2008; Heys et al., 2009). Johnson et al. (2009) analysed data from eight countries in SubSaharan Africa (SSA) and found that women who knew their HIV positive status were more likely to limit childbearing by increasing contraceptive use. Similar to Johnson et al. (2009) studies by Elul et al. (2009) and Heys et al. (2009) found that HIV positive women were less likely to desire for more children than HIV negative women; hence, the odds of using longterm methods was higher among HIV positive women. However, in Rwanda, contraception use among both HIV positive and HIV negative women desiring to avoid pregnancy was low (Elul et al., 2009) suggesting that future needs for family planning for both HIV positive and HIV negative women is high. In Zimbabwe, where 13% of women are infected with HIV (ZDHS, 2010/11) and antiretroviral treatment (ART) coverage is high (77%) (UNAIDS, 2014), access to a variety of contraceptive options is important because HIV positive women may change their childbearing plans in response to learning their HIV status (Habte and Namasasu, 2015) and in response to HIV treatment (Farahani et al., 2014). In fact, from a reproductive health-based and rights-based perspective, all women should have access to their methods of choices for avoiding unwanted pregnancies, particularly for HIV positive women who have particular needs for contraception to avoid unintended pregnancy, prevent the transmission of the infection to their partners and to eliminate the risk of mother-to-child HIV transmission (McCoy et al., 2014). Despite the protective effect of contraception against adverse sexual and reproductive health outcomes among HIV positive women, HIV negative women are also expected to prevent unwanted pregnancies and HIV infection (Mumah et al., 2014). Prevention strategies among HIV negative women include condom use, sexual abstinence, and the use of most effective methods including long-term methods (Mumah et al., 2014). However, to sum it up, all women are simultaneously at risk of both unintended pregnancy and HIV infections (Reynolds et al., 2013). Therefore, all contraceptive methods should be made available to all women such that they make appropriate choices and align their choices with their reproductive intentions (Mayondi et al., 2016).

Typically, the rationale for choosing a particular contraceptive method is dependent upon its motivation, effectiveness, availability and whether or not it will be used for disease prevention such as HIV and STIs (Mumah et al., 2014). The choice of these methods may also vary by stage of a woman's reproductive life course (Warren et al., 2013) and it is postulated that women use more effective methods such as permanent methods if they have completed their fertility desires (Reynolds et al., 2008). A study by McCoy et al (2014) found out that women who had not reached their fertility intention and were sterilised regretted doing so. Due to the
varying effectiveness of different contraceptive methods, contraceptive decision-making is a critical health behaviour that directly affects an individual's risk of unplanned pregnancy and HIV infection (Yeatman, 2009). Ideally in the context of high levels of HIV infection, a dual methods approach (i.e combining condoms with highly effective contraceptive methods) should be promoted (UNAIDS, 2012). Using a condom alone may pose a challenge due to high failure rate for pregnancy and HIV prevention (Bogale et al., 2010) and this may be due to low levels of correct and consistent use (Elul et al., 2009). Like condoms, the contraceptive pill method was found not to be reliable due to poor adherence (Church et al., 2014). Several longitudinal studies of HIV positive women found higher use of short-acting methods (Hoffman et al., 2008; Nanda et al., 2011), but their contraceptive use was not sustained (Hoffman et al., 2008). Hence, unintended pregnancy could be high in Zimbabwe, where the available methods are mainly short-acting requiring daily adherence (Moyandi et al., 2016).

Long-term methods such as implants and sterilisation are credited for eliminating the need for daily or pre-act intercourse adherence that is normally associated with the use of pill and condoms (McCoy et al., 2014; Warren et al., 2013) but may not be appropriate for HIV positive women who may change their fertility intention due to improved antiretroviral treatment after they have HIV post-test counselling (Hoffman et al., 2008). On the contrary, long-acting reversible contraceptives are highly effective in the long-term and useful for women who want to limit childbearing (Kaida et al., 2011). Examples of long-term methods are the intrauterine device (IUD) and the birth control implant. Such long-term methods offer women some form of reproductive control that does not require sexual negotiation that is required with condom use (Matthews et al., 2013). However, long-term methods are less accessible than condoms and other less effective methods (abstinence, withdrawal) which do not require a prescription from a health provider (Church et al., 2014). Also, in the context of HIV, hormonal methods such as IUD are not frequently provided in the public health sector due to safety concerns, but Curtis et al (2009) and Stringer et al (2007) concluded that hormonal contraceptive methods are not associated with disease progression and HIV transmission to uninfected partners. Another problem that is associated with the use of long-term methods relates to provider skills in the administration of implants and other permanent contraceptive methods. Previous studies found that long-term methods are sometimes not recommended by healthcare providers or accessible to women with HIV due to limited knowledge and skills of health care providing HIV services and lack of operational guidelines (Moodley and Cooper, 2014).

Literature reviewed suggests that the woman's knowledge of her own HIV status plays a significant part in the choice of contraceptive methods (Church et al., 2014; Johnson et al., 2009; Mumah et al., 2014). Although some studies found no difference in family planning unmet needs for both HIV positive women and HIV negative women, the decision for contraceptive choice among HIV positive women may be complex and can be affected by many factors. While some studies find no difference in methods choices between HIV positive and HIV negative women (Elul et al., 2009), others suggest high condom use among women who know their HIV positive status following a diagnosis (Mayondi et al., 2016; Johnson et al., 2009; Heys et al., 2009), suggesting a desire to prevent HIV transmission to their partners (Mayondi et al., 2016). Mumah et al. (2014) analysed data from 13 countries in SSA countries and found that hormonal method use was high among women who knew their HIV positive status. Condom use was found to be three times higher among HIV positive women than HIV negative women in Zambia (Elul et al., 2009), while injectables were more popular among HIV positive women in Tanzania (Keogh et al., 2012).

In Zimbabwe, mitigating the HIV epidemic through the family planning programme has been one of the government's long-term goals (UNAIDS, 2012). Specifically, there is an increasing funding and scaling up of the HIV and family planning integrated program at the health centres across the country. The outreach efforts target hard to reach women who may be at risk of HIV and unintended pregnancies (Washington et al., 2015). Women who visit family planning health care centres are offered options from which they choose their most appropriate contraceptive methods. Other programmes such as Prevention of Mother-to-Child Transmission Programmes (PMTCT); National Behaviour Change Programme (i.e Abstinence, Be faithful, and Condom use (ABC); and Voluntary Counselling and Testing Services programs were also implemented with the aim to reducing HIV infection. These programs run health campaigns promoting condom use and encouraging HIV testing. They also promote the use of more effective contraceptive methods such as long-acting contraceptives.





Source: Zimbabwe Demography and Health Surveys (2005/6; 2010/11)

Despite the government's effort in promoting the use of family planning services, the use of long-term contraceptive methods and condoms is very low in Zimbabwe. As shown in fig 7.1 above, it is estimated that 73% of women using contraceptives in Zimbabwe are currently using hormonal contraceptive methods (pill and injectables), 18% are using condoms (male and female condom), 5% using long-term methods (mainly implant/norplant and sterilisation); and other methods (4%). This is worrisome, because in the context of HIV, women are encouraged to use condoms concurrently with other effective methods for dual protection i.e prevention of HIV and unwanted pregnancy (Mumah et al., 2014). The low use of condoms and use of less effective contraceptive methods has been identified as an important factor for high rates of unplanned pregnancy and HIV in Sub-Saharan African countries (Church et al., 2014, Johnson et al., 2009, Mumah et al., 2014). Relative to hormonal and long-term methods, studies have noted low typical condom use, perhaps due to the fact that it must be worn at every sexual encounter and requires male cooperation (Cleland and Ali, 2009). Despite the benefits of long-term methods and condom use, these methods are less frequently used in Zimbabwe.

Studies have argued that choice of contraceptive methods is dependent on socioeconomic status or community environment depending on the level of analysis. On the role of social and economic development, studies have shown that women from more affluent communities will adopt smaller family sizes and seek out permanent contraceptive methods when they achieve their fertility desires (Kakaire et al., 2010; Heys et al., 2009; Laher et al., 2009). On the contrary, women from poor communities may react slowly to societal changes that govern fertility behaviour (Nattabi et al., 2009). Given that HIV prevalence is said to be higher among richerpeople than poorer people, it may be that some factors that influence contraceptive methods choice among HIV positive women may differ from HIV negative women (Mumah et al., 2014).

While several studies investigating the behaviour of women who know their HIV status towards choosing contraceptive methods have been conducted in Sub-Saharan African countries (Magadi, 2016; Habte and Namasasu, 2015), they have not broken down the methods into clusters according to their relative strength in preventing HIV infection and unintended pregnancy. To gain more in-depth knowledge of the determinants of contraceptive choices among women who know their HIV status, this study assumes the decision-making processes of family planning choices is a two-way process. A woman who knows her HIV serostatus is confronted with choice between pregnancy prevention and HIV prevention, hence the choice of four contraceptive clusters in fig 7.1. The limitation of previous studies addressed here is the assumption that contraception choice can be a yes or no answer, without taking into account the possibility of parallel branches of contraceptive methods available (Johnson et al., 2009). The strength of the current study is that it assumes that the choices of parallel contraceptive methods might be influenced by different determinants. Conditionally on choosing contraceptive methods, the study assumes that if a woman opts for hormonal methods, her second options consist of either long-term, condom or other methods (Mayondi et al., 2016). This way it is possible to compare the differences in the profiles of women who make different contraceptive methods choices. To shed more detailed light on what is happening in Zimbabwe regarding contraceptive methods choice after HIV diagnosis, the study used multilevel multinomial logistic regression and by using this technique, the study was expected to separate individual from community determinants of each particular contraceptive methods choice in Zimbabwe.

The study fits in with the research on contraceptive choices among HIV positive women which has been expanding since its beginning in the 2000s, for example (Magadi, 2016, Mayondi et al., 2016, Johnson et al., 2009). This study therefore investigates:

- The association between knowledge of HIV sero-status and contraceptive method choice
- Other (individual and community level) determinants of contraceptive method choice

7.3 Data and Methods

7.3.1 Data source

The data come from the 2005-6 and 2010-11 Zimbabwe Demographic and Health surveys, which included HIV testing. A total of 18078 women were interviewed in the two surveys and 13028 were eligible for HIV test. For the purpose of this chapter, all women who reported having been previously tested for HIV, received test results, and were currently using any form of contraceptive methods were included in the analysis. Women who had been declared infecund, those who were pregnant, and non-respondents to the survey's HIV test were excluded from the sample. This yielded a final sample of 3757 women of whom 13.5% (506) were confirmed HIV positive and 86.5% (3251) HIV negative. The final sample represents women who knew their HIV serostatus and were using contraceptives at the time of the surveys. To create the analytical file, HIV serostatus data was merged with the woman's individual file which also contained household variables. Women who were not using any methods of contraceptives were excluded to avoid modelling the differential decision processes of choosing to adopt contraception and choosing a specific contraceptive method (Mayondi et al., 2016) and this is in line with Heys (2009) who stated that inclusion of non-users in the models obscures the variation in factors associated with use of each type of contraceptive method. Therefore, the models in this study are largely driven by the decision to adopt any form of contraception.

7.3.2 Methods of analysis

The analysis of the determinants of choice of each particular contraceptive method was initially carried out by bivariate and later by multinomial logistic regression analysis. To begin with, it is plausible that some factors that drive knowledge of contraceptive methods and hence demand for a particular method may operate at the community level. For example, cultural beliefs around childbearing tend to be localised. Also contraceptive methods choice in the same community may be correlated due to unobserved factors such as the availability of specific methods or community's perception of specific methods (Magadi and Curtis 2003). In this case, a multilevel multinomial logistic regression modelling technique for measuring the separate influence of individual level correlates and community effect was used in order to account for unobserved community level determinants of contraceptive methods choice. The multilevel multinomial logistic regression applied two-level models that take into account the woman-level 1 and the unobserved community-level 2 effects. These models estimated the probability of choosing a particular method holding other factors constant. Details of the multilevel multinomial logistic regression used are provided below.

7.3.2.1 Multivariate multilevel multinomial logistic regression analysis

The multinomial regression analysis adopted here expands from the binary logistic regression discussed in chapters 4-6. We have seen how logistic regression models were used in a situation where we have a binary response outcome, for example whether or not an individual has been tested for HIV (ref chapter 4). Where we have more than two response categories we make use of multinomial models. In this chapter the response variable has more than two responses and the model was built in stages. The model sees individual women as being nested within communities. The general equation of the random intercepts two-level multinomial Logistic regression model used for analysis of determinants of contraceptive methods choice therefore takes the form:

$$log\left[\frac{\pi_{ij}^{(s)}}{\pi_{ij}^{(t)}}\right] = \beta_0^{(s)} + \beta_1^{(s)} X_{1ij} + \beta_2^{(s)} X_{2ij} + \dots + \beta_k^{(s)} X_{kij} + u_j^{(s)}, \ s=2,3,4$$

where: $\pi_{ij}^{(s)}$ denotes the probability of using a particular contraceptive method (s) (i.e. long-term methods=2, condom=3 or other=4) for woman*i*, in the jth community ; $\pi_{ij}^{(t)}$ is the probability of

using hormonal methods (t=1 for hormonal methods used as reference category) for woman i, in community j; $\beta_0^{(s)}$ are the regression intercepts/constant for contraceptive methods s; $X_{(1-k)ij}$ are 1-k explanatory variables defined at woman or community level; $\beta_{(1-k)}^{(s)}$ are the associated usual regression parameter estimates for contraceptive methods choice (s); and $u_j^{(s)}$ are the community-level residuals for contraceptive methods choice (s). These are assumed to be normally distributed with mean zero and variance $\sigma^{2(s)}_{u}$. The community random effects may be correlated across contraceptive methods choice categories: covariance ($u_j^{(s2)}, u_j^{(s3)}$) = $\sigma^{(s2, s3)}_{u}$, (s2=long-term, s3=condom; and s4=other methods). (Rasbash et al. 2016).

7.3.2.2 Fixed and Random Effects

The model was estimated initially by IGLS and then by Markov chain Monte Carlo (MCMC) estimation with an initial burn-in of 500 simulations followed by 5k simulations. Fixed effects are the level-1 and level-2 covariates, expressed as relative risk (RR) and 95% confidence intervals (95% CI). The random effects are the measures of variances in contraceptive methods choice across communities (level-2). The ratio of the variance at the community level to the total variance is called inter-class correlation coefficient (ICC), which is a tool of measuring the degree of homogeneity within units such as households or communities. According to Browne (2014) the ICC measurements are used to gauge whether contextual variables have any effect on the outcome. The results of random effects are expressed as variance partition coefficient (VPC), which is in this study equal to the ICC. The models were estimated in MLwin 2.36 (Rasbaschet al.2016), by bayesian methods using Markov chain Monte Carlo (MCMC) estimation (Browne. 2017).

7.3.3 Outcome variables

7.3.3.1 Contraceptive choice

The dependent variable is the choice of the contraceptive methods. This comes from the ZDHS question which asks a woman; *which methods are you currently using*? This question solicited multi response (not using any, intrauterine devices, implants and sterilization, pills, condoms, spermicides, injectables, traditional methods). For the purpose of this study, those who reported not using any method were excluded. The remaining were classified into four categories: hormonal methods (injectables, pill); long-term methods (sterilisation, implants/ IUD); condoms (female and male condoms); and other methods (mainly withdrawal, abstinence). This grouping was informed by previous studies that suggested that hormonal methods are easily accessible (Bertrand et al., 2014; Johnson et al., 2009) while long-term methods are effective for HIV positive women and that condoms prevent both unplanned pregnancy and HIV infection (Habte and Namasasu, 2015). The 'other methods'- category was as a result of small sample sizes in other contraceptive methods used by women.

7.3.4 Key predictor variables.

The key predictor variable in this study is the women's HIV status, and knowledge of their HIV status. To create the key variable, HIV biomarker data collected in the surveys were combined with the individual file. The variable HIV status and knowledge of HIV status was created from a sample of women who responded to questions about their HIV testing history and whether they received their HIV testing results following previous testing. This was deemed important as it may have influenced the respondent's contraception behaviour (Kurewa et al., 2011). The two categories were: 1=HIV positive and knew the results; 0=HIV negative and knew their results.

7.3.5 Selection of control variables

Control variables were selected and classified into two groups relating to predisposing individual level factors such as woman's age, place of residence, marital status, religion, educational attainment of the woman, parity, household wealth, exposure to family planning messages, multiple sex patners; HIV factors (i.e awareness, stigma and knowing someone who died of HIV/AIDS); and predisposing community level factors such as HIV prevalence,

poverty and family planning media exposure. The choice of individual level factors was informed by previous studies on factors influencing contraceptive use. Because the topic of how community context influence contraceptive method choice is a relatively new area of research, there is presently a lack of theory to provide a framework for selecting community level variables included in the models. The variables were initially screened for problems of multicollinearity before being included in the regression models. These variables were selected for the following reasons:

7.3.5.1 Predisposing individual level factors

HIV awareness: HIV awareness may be associated with reduced desire to child bearing (want no more child) to prevent perceived risk of mother to child HIV transmission (Magadi and Agwanda and Obare, 2007). Therefore, it is postulated that women who know their HIV sero-status who have comprehensive knowledge about how HIV is transmission and prevented are more likely to use condoms and long-term methods to prevent both childbearing and HIV transmitted to their infants, particularly if they know their HIV positive status. The variable was derived from a series of questions on knowledge of how HIV is transmitted and ways to avoid infection. The resulting index is classified into quartiles and the higher the index, the higher the knowledge

HIV stigma: Motivation for preventing childbearing may stem from an individual's attitude towards pregnant women living with HIV (Johnson et al., 2009). Women may choose to stop child bearing if they observe stigma and discrimination towards mothers who are perceived to be HIV positive in the communities (Church et al., 2014). It is therefore expected that women who observe stigma and discrimination against people living with HIV are more likely to use more effective contraceptive methods to prevent child bearing. The variable is derived from a set of four DHS questions on HIV/AIDS stigma. The resulting score is classified as '0' low, '1-2' medium and '3-4' high

Knowing someone who died of HIV/AIDS: like stigma and discrimination, personal experience with death related to HIV/AIDS, particularly death of a child perceived to be HIV positive may affect whether or not the mother wants to have another child (Turan et al., 2008). In such circumstances, women who know their HIV status who have witnessed death

that can be attributed to HIV/AIDS may be more likely to adopt effective contraceptive methods to prevent child bearing. The variable is a single item derived from a DHS question relating to whether or not the respondent personally knows someone who has or died of HIV/AIDS. The variable was Coded 1=if respondent personally knows someone who has or died of HIV/AIDS; 0=otherwise.

Woman's age: This study proposes that an increased age comes with an increase in the use of more effective contraceptive methods such as long-term methods for delaying and limiting child birth (Johnson et al., 2009) and condoms for HIV prevention (Church et al., 2014; Wagner and Wanyenze, 2013). This hypothesis agrees with other studies which postulate that with increased age and an improved knowledge of HIV, there is an increased likelihood of choosing long-term and condom contraceptive methods (Reynolds et al., 2008). Due to the importance of knowledge of HIV and the availability of contraceptive options (including long-term and condoms), it is highly likely that older HIV positive women who know their HIV status may choose long-term methods and condoms relative to hormonal methods. For younger women, regardless of their HIV status and their knowledge of HIV status, research has shown that they may prefer reversible contraceptive methods in anticipation of future childbearing (Mumah et al., 2014) or condoms particularly to prevent both unplanned pregnancies and HIV infection (Nattabi et al., 2009). In the DHS, age is a categorical variable classified into five groups ranging from 15-49 years of age. However, for the purpose of this study, age is categorised into three groups: 15-24 coded as 1; 25-34 coded as 2; and 35 and above was coded as 3, this was due to the fact that some of the categories such as 15-19 and in the original data had few cases to allow for a meaningful analysis.

Parity: in the DHS, the number of children ever born is a continuous variable. Just as with the preceding variable, this variable was categorised into three groups: 0-1 births coded as 1; 2 births as 2; and 3 or more birth order as 3. Generally, the number of births a woman has is likely to result in the need to prevent further pregnancies (Magadi, 2016). Women with higher order births are more likely to seek out more long-term methods than women with lower order parity and more reliable methods are adopted after a subsequent childbirth regardless of HIV status (Gray et al., 2005 cited in Reynolds et al., 2008). It is postulated that, once a woman has had a first birth, the priority shifts towards birth spacing, primarily for economic reasons as more children affect caring for other offspring (Laher et al., 2009). The ability to delay and

space childbearing is crucial to women's social and economic advancement (Shelton and Burke, 2016); therefore, women with higher order parity are more likely to use irreversible methods.

Education: Extant literature suggests that increased education usually translates into the use of modern contraceptive methods (Bertrand et al., 2014). However, in the case of contraceptive choice, research have shown that because the knowledge required for contraceptive choice is highly specific and not widespread, it is possible that education may not predict contraceptive choice (Mumah et al., 2014). Also, due to the fact that family planning issues are not taught in the university in Zimbabwe, it is likely that graduates from secondary or higher probably do not have more knowledge of family planning issues than women who did not attend secondary or higher. This variable was created by regrouping educational attainment categories in the DHS into two categories; coded as '1' primary and below; and 2 secondary and above.

Marital status: Research suggests that marital status matters most in predicting women's contraceptive methods choice. Generally, married women may elect sterilisation only when they have a strong aversion to future births (Johnson et al., 2009). On the other hand single women, particularly young women who are ambivalent and uncertain of their future fertility desire are not likely to use long-term methods (Dasgupta et al., 2015). When HIV comes into play, the motivation for the choice of contraceptive methods becomes complicated regardless of the woman's marital status. For married women who know their HIV-positive status and who have not achieved their desired number of children, motivation to achieve pregnancy and to prevent HIV re-infection should be their primary concern (Church et al., 2014) leading to continued use of hormonal methods or other reversible methods and inconsistent use of condoms. Among married women who know their HIV-positive status and who have achieved their fertility desire, motivation to prevent both pregnancy and HIV re-infection should be their predominant concerns and this should lead to sterilisation and condoms (Johnson et al., 2009). However, research has shown that condom use within marriage is highly unlikely particularly among HIV positive women due to disclosure concern (Akinyemi et al., 2010), feeling of rejection (Laher et al., 2009), abuse and concern for partner compliance with condom use (Church et al., 2014; Rosenberg et al., 2013). For older divorced and widowed HIV positive women, the motivation of avoiding infecting sexual partners is their greatest concern (Babalola et al., 2012) and their goal is likely achieved with condom use, while pregnancy prevention is most likely to be achieved by using long-term methods. On the other hand, unmarried women usually don't plan for sexual encounters -they may engage in sexual intercourse, but do not want to think of themselves as sexually active, leading to condom use and traditional methods that are accessible (Do and Kurimoto, 2012). Marital splits and altered relationships may lead to renewed desire for offspring from new unions, and reversal of sterilisation contraceptive methods (Church et al., 2014).

Religion: Aside from their own HIV status, Catholic and Protestant followers tend to have higher likelihood to use modern contraceptive methods than traditional and apostolic faith followers (McCoy et al., 2014). This behaviour has been interpreted as a consequence of greater social diversity within mainstream religious groups that facilitates change of behaviour towards health care services utilisation (Muchini et al., 2010). While the conventional religious groups such as Catholic, Protestant and Pentecostal churches have a propensity of adopting modern contraceptive methods, Apostolic and traditional faith groups are more likely to use traditional methods such as withdrawal and abstinence rather than the effective methods (Singh et al., 2009) even after their HIV diagnosis, due to lack of exposure to available contraceptive methods and HIV messages that they do not embrace in their practice. A categorical variable based on religious affiliation is included in the analysis

Wealth: Higher income is believed to have a positive effect on contraceptive choice, possibly due to the fact that increased income allows for easier access to more expensive and effective methods such as a long-term methods (Dias and Oliveira, 2015). When HIV status and fertility desire were controlled for, (Dias and Oliveria, 2015) found women in the richest wealth quintiles in 13 countries in Sub Saharan African countries were more likely to choose long-term methods than women in the poorest wealth quintile. This is due to the fact that long-term methods are more expensive than short-term and are usually provided at clinics in Africa (Rademacher et al., 2016). Although research suggeststhat wealth initially increases risk for HIV acquisition it later becomes protective (Bertrand et al., 2014) such that HIV-positive women who are wealthier are more likely to afford the cost of adopting long-term methods, than poorer HIV positive women. This variable is important for assessing inequalities in the reported unmet need for particular contraceptive methods among women who know their HIV status. In the DHS, an index of household economic status is constructed based on a household's possession of consumer durables such as electricity, television, bicycle, and car.

The index also includes the quality of housing (roof, walls, and floor), as well as a household's access to amenities such as water, sanitation, and so on. The wealth index is a categorical variable, with the following five categories: lowest wealth quintile (poorest); middle (middle); and the highest wealth quintile (richest).

Family planning media exposure: This study incorporates the family planning messages exposure variable. Family planning messages incorporate HIV prevention as a motivation to use contraception. In Zimbabwe, one of the most marked changes that occurred coincided with ashift in family planning practice that has been the increase in the exposure to family planning messages through radio, TV and newspapers in the general population (ZDHS, 2010). It is hypothesised that higher levels of exposure to family planning messages are associated with a lower probability of traditional methods and higher probability of modern use. All other methods provide no protection as only condoms offer protection against HIV/AIDS transition (Stringer et al., 2007). In addition to exposure to general family planning messages through media channels, HIV positive women receive extra contraceptive counselling, particularly on persistent condom use (Habte and Namsasu, 2015). Therefore, it is also expected that exposure to the condom use media campaign will be associated with a lower probability of hormonal, long-term and others methods, and a greater use of condoms among HIV positive women who know their status. Also, the magnitude of the increase in exposure to family planning messages in Zimbabwe (ZDHS, 2010), coupled with the prevalence of HIV/AIDS prevention programmes such as Abstain, Be faithful, use of condoms (ABC) aimed at individual behavioural change (Kaida et al., 2011) have provided a compelling reason to study the effects of exposure to family planning messages on contraceptive method choice in Zimbabwe. This variable was derived to score a woman according to the number of family planning messages she received, indicating how familiar a woman is with family planning services. It is the sum of three variables indicating the woman's access to family planning messages through radio, television and newspapers. The variable was categorised into three groups coded as: 1=low, 2=medium and 3= high.

Place of residence: Research on contraceptive methods choice based on women's knowledge of her own's HIV status and their place of residence has not been prominent. However, irrespective of their HIV status women in urban areas were found to be more likely to use condoms than hormonal contraceptive methods compared with women in rural areas (Magadi, 2016). In the context of Zimbabwe, HIV prevalence is higher among women in urban areas (Chevo and Bhatasara, 2012), this also corresponds with high prevalence of HIV/AIDS prevention programmes such as voluntary, counselling and HIV testing and PMTCT (McCoy et al., 2014) in urban than rural areas. This study hypothesises that women who know their HIV positive status and are resident in urban areas are more likely to use condoms (for prevention of HIV); and long-term methods (for pregnancy delay) than hormonal methods used by women of the same status in rural areas. This is due to the fact that women in urban areas have more access to family planning services including information about HIV/AIDS prevention (Mumah et al., 2014). Respondents from urban areas are recoded as 1 and those from rural areas as 2.

Fertility desire: This variable assesses the women's future intentions on childbirth i.e limiting, timing and spacing (Dasgupta et al., 2015) after knowing their HIV status. Studies have shown that fertility desire may change after knowing their own HIV positive status (Mayondi et al., 2016). It is postulated that HIV positive women may not want any more pregnancies leading to an increase in the use of permanent contraceptive methods among this group (Johnson et al., 2009). Their reason to avoid pregnancy is the concern that pregnancy will further compromise their health and fear of transmitting HIV to children if they conceive (Habte and Namasasu, 2015). While some HIV positive women may want to avoid pregnancy, some may desire pregnancy; and therefore, may choose less effective methods such as traditional methods. Research suggests that women who know their HIV-positive status may want to increase women's fertility desire because they may have societal and familial pressure to have children (Nattabi et al., 2009), fear that the children they already have may die (Laher et al., 2009), reassurance that prevention of mother to child HIV transmission (PMTCT) programs reduce the risk of having a HIV positive child (Homsy et al., 2009), and concern that avoiding pregnancy might give suspicion about one's HIV status (Matthews et al., 2013). This study hypothesises that women who know their HIV positive status and want to limit childbearing are more likely to use long-term methods than hormonal methods, and child spacers are less likely to use long-term methods. This variable is important because it may predict whether contraceptive choice adopted by the woman was motivated by birth spacing, limiters or diseases prevention (Magadi and Magadi, 2017).

7.3.5.2 **Predisposing community level factors**

The community level analysis includes three explanatory variables, i.e the HIV prevalence in the community, the proportion of women in poverty and those with exposure to family planning messages in the community. These covariates captured some of the structural characteristics of respondents' communities. The table 7.1 below gives a list of the variables.

Description	VARIABLE							
Contraceptive method choice	Current contraceptive method choice, classified as:							
-	1=hormonal (pill/injectable); 2=condom; 3=long-term							
	(sterilization, IUD/implants); and 4=other methods.							
HIV status	Binary variable, coded as 1=if respondent was previously							
	tested for HIV and is HIV-positive; 0 if HIV-negative							
Knows someone who died of	Coded 1=if respondent personally knows someone who has or							
HIV/AIDS	died of HIV/AIDS; 0=otherwise							
HIV stigma	A composite index derived from a set of four DHS questions							
	on HIV/AIDS stigma. The resulting score is classified as '0'							
	low, '1-2' medium and '3-4' high							
HIV awareness	A composite index derived from a series of questions on							
	knowledge of how HIV is transmitted and ways to avoid							
	infection. The resulting index is classified into quartiles and							
	the higher the index, the higher the knowledge							
Demographic and backgroun	d socio-economic factors							
Respondent's age	Three age categories: 15-24, 25-34 and 35+.							
Parity	Number of children ever born, classified into three categories:							
	0-1, 2 and 3+							
Marital status	Marital status, classified as: never married, married widowed							
	and divorced							
Education	Highest educational attainment classified into: primary and							
	secondary/ above.							
Household wealth index	Household wealth index; coded as low =1; middle =2; rich =3;							
	richest=5.							
Desire for children	Nomore=1, within 2 years=2; later=3; and undecided=4							
Respondent's religion	Religious affiliation, classified into: traditional/Catholic,							
	Protestant, Pentecostal and Apostolic							
Exposure to Family planning	Additive index of three family planning messages received							
messages	radio, television and newspaper. Dummy variable, coded							
	1=low; 2=medium; and 3=high							
Community level factors								
Urban residence	Binary variable, coded as 1=urban or; 0=rural residence.							
Poverty	Proportion of women in poverty in the cluster							
HIV prevalence	Proportion of individual in cluster who are HIV-positive							
HIV/AIDS awareness	Average HIV awareness in the cluster							

 Table 7-1: Selected Covariates for Method Choice Estimation

7.4 Results

7.4.1 Distribution of contraceptive methods choice in Zimbabwe

Figure 7.2 shows the distribution of women who knew their HIV status (n=3757) by contraceptive methods choices. The results show that the use of hormonal contraceptive (pill and injectables) was predominant among women who knew their HIV status. Over 80% of women chose hormonal methods (injectable and pill) over long-term, condoms and other methods for both the 2005/6 and 2010/11 surveys. Although, the choice of hormonal methods was high, the rate fell by 3.7% from 83% in the 2005/6 survey. Other methods, i.e. long-term (sterilisation and implants), condoms, and others methods were not very popular in Zimbabwe. While, long-term methods and condom appeared to be increasing, the change was hardly discernible, particularly for long-term methods which increased marginally from 6.9% to 8%. Condom use also increased by almost 3 points from 5.9% to 9%. Other contraceptive methods were least used in this sample, in fact the proportions in this category have declined since 2005/6 as more and more women appeared to have shifted to long-term methods and condoms.



Figure 7-2: Distribution of contraceptive methods choice for the 2005-11 ZDHS (N=3757)

Source: ZDHS 2005/6; 2010/11.

7.4.2 Bivariate association between HIV status, background characteristics and contraceptive method choice

Table 7.2 also describes the bivariate analysis indicating association between demographic variables and contraceptive method choice for the combined surveys. Overall, hormonal methods were more popular among women from different backgrounds and including their HIV status compared with all other methods. Regarding HIV status, hormonal methods were the predominant methods of choice among both HIV-positive (80.7%) and HIV-negative women (82.2%), but long-term methods were higher among HIV infected women (8.4%) compared with those who were not infected (6.6%). Similarly, condom use was more popular among HIV-positive (8%) than HIV-negative women (5.9%). Furthermore, the use of other methods was higher among uninfected women (4.8% than infected women (2.9%). There are no discernable differences in the use of all four methods within HIV awareness, stigma and knowing someone who died due to HIV/AIDS categories, except for condom use which was more popular among women with high HIV stigma.

The use of hormonal methods was much higher in rural areas 84.9% compared with 76.1% in urban areas. Conversely, long-term and condom method were more common in urban than rural areas. The effects of age on methods choice show the tendency that the women below the age of 34 years of age were more likely to use hormonal methods than older women. For example, about 89% of women in their 20s chose hormonal methods than older women (67%) above 35 years of age. It appears that some women aged (35 years and above) may have later in their lives transitioned from using hormonal methods to long-term methods and condoms. Above all, long-term methods (5.7%), condom use (3.1%); and other methods (1.9%) were least preferred by younger women in their 20s, but condom use (14.1%), long-term methods (13.7%) and other methods (5.2%) were more predominant among older women aged 35 years and above. Women between 24 and 34 years of age were also likely to adopt long-term methods (7%) and condom use (7.7%) compared with younger women in their early 20s.

Whilst use of hormonal methods was higher among married women (85%) than those who were single (never married, widowed, and divorced), long-term methods (5.1%) and condom use (6.8%) were least preferred among this group of women. Meanwhile, long-term methods were more popular among never married women (40.8%), divorced (26%) and widowed

(21.8%) than married women. Condom use was predominant among widowed (22.7%) and divorced (13%) than never married (9.5%) and married (6.8%). Surprisingly, other contraceptive methods were more common among widowed women (10%) than any other marital category. Hormonal methods were higher among women affiliated with the a postolic faith group (85.9%) and Pentecostal (80.1%) faith groups than traditional (79.5%) and Protestant (75.5%) faith groups, but long-term methods and condoms were the least used among this group. Condom use was remarkably higher among protestant followers (12.5%) than all other faith groups and least preferred among apostolic faith group (4.2%). Meanwhile, the use of long-term methods was lowest among women with high order parity of 3 or more births (77.6%) compared with women with low birth order of one and two children (82.7% and 84.4%) respectively. The use of long-term methods and condoms were also higher among women with low order parity (10.1%) and (10.7%), respectively.

Hormonal methods were least used among the richest women (73.1%), but long-term methods (11.3%) and condoms (12.6%) were highest among this group of women. Whilst hormonal use was highest among the poorest (85.9%) and middle income women (82.5%) there was no difference in the use of condoms between the poorest and those who were in the middle income groups. Long-term methods were higher among women in the middle income (8.1%) than the poorest (5.3%). There were no marked differences regarding the use of other methods among women from different wealth categories. Hormonal method was highest among women with low family planning media exposure (83.3%) and was lowest among women with high exposure (72.8%), but long-term methods and condom use were more popular among women in the medium to high family planning exposure groups. Meanwhile, women with medium to higher family planning media exposure were more likely to use condoms and long-term methods. Long-term methods were generally predominant among women who reported wanting no-more children and those who wanted children later than two years from now than those who reported wanting a child now or within 2 years. Long-term methods and condom use were highest among women with more than three sexual partners, but hormonal method was predominant among women with few sexual partners. Education and place of residence were found to be not significantly associated with contraceptive methods choice.

		All cases ^b			
Background factors	Hormonal	Long-term	Condom	Others	
HIV Sero status*					
Positive	80.7	8.4	8.0	2.9	3251
Negative	82.8	6.6	5.9	4.8	506
HIV Awareness					
Low	81.8	8.1	7.2	2.9	307
Medium	80.0	8.8	8.3	2.9	2157
High	80.8	7.7	8.4	3.1	1293
HIV stigma					
Low	79.2	8.7	9.2	2.9	1331
High	81.1	8.1	7.7	3.1	2426
Acquaintance with H	IV victim				
Yes	79.1	8.2	8.3	2.9	3311
No	80.6	9.0	8.3	3.6	446
Place of residence					
Urban	76.1	10.6	9.8	3.6	1579
Rural	84.9	6.3	6.1	2.7	2178
Maternal age*					
15-24	89.3	5.7	3.1	1.9	1154
25-34	82.4	7.0	7.7	2.9	1740
35+	67.0	13.7	14.1	5.2	863
Marital status*					
Never married	48.3	40.8	9.5	1.4	186
Married	85.0	5.1	6.8	3.1	3251
Widowed	45.5	21.8	22.7	10.0	115
Divorced	60.6	26.0	13.0	0.4	205
Religion*					
Traditional/catholic	79.5	9.4	7.7	3.4	964
Protestant	75.5	9.2	12.5	2.8	753
Pentecostal	80.1	8.8	8.8	2.4	764
Apostolic faith	85.9	6.3	4.2	3.6	1276
Education					
Primary	82.4	8.2	6.4	3.1	966
Secondary	80.2	8.1	8.2	3.1	2791
Parity*					
1	82.7	10.1	4.7	2.5	1037
2	84.4	6.9	6.1	2.6	1119
3+	77.6	7.8	10.7	3.9	1601
Wealth*					
Poorest	85.8	5.3	5.9	3.0	1141
Middle	82.8	8.1	5.9	3.2	1625

Table 7-2: Bivariate associations between demographic variables and contraceptivemethod choice among women aged between 15-49 years of age in Zimbabwe

Continued

Richest	73.1	11.3	12.6	3.1	991
FP media					
exposure*					
Low	83.3	7.3	6.5	2.8	298
Middle	74.2	11.0	10.5	4.3	459
High	72.8	10.8	12.8	3.6	400
Desire for children*					
Want no-more	76.9	11.4	7.7	4.1	1628
Later than 2 years	81.1	12.2	3.2	3.5	361
Now within 2 years	85.6	3.5	9.0	1.8	1492
Undecided	80.8	8.8	6.7	3.7	400
Multiple sex partners	*				
1	86.1	4.5	6.8	2.6	2420
2	77.2	9.9	9.1	3.7	783
3+	61.5	24.0	10.1	4.7	554
All	81.0	8.1	7.7	3.1	3757

Table 7-2 continued

Chi-Square -*P<0.001

Source: Computed from Zimbabwe Demographic and Health Surveys, 2005-6; 2010-11.

7.4.3 Determinants of contraceptive methods choice among women who knew their HIV status in Zimbabwe

Table 7.3 reports the multilevel multinomial logistic models exploring the determinants of contraceptive method choice among women who know their HIV status. The reference category of the dependent variable method choices was hormonal methods, which was compared against long-term methods, condoms, and other methods. Based on the literature reviewed, the study had postulated that HIV factors (i.e awareness and stigma) would be strong predictors of adopting more effective contraceptive methods to prevent both unwanted pregnancy and HIV. However, the analysis of potential association between these factors and contraceptive method choice among women who know their HIV sero-status was not fully explored due to relative small proportions of women within the categories of HIV awareness and stigma who used condoms and other methods. Thus HIV awareness and stigma variables were not included in the multivariate analysis.

Time of survey was controlled for in all models because the survey year may reflect other unobservable factors influencing women's attitude or information about contraceptive methods and HIV. These factors may have independent effectson contraceptive methods choice even when the influence of factors listed above is controlled for. The results suggest that all women who knew their HIV status were more likely to use long-term methods (RR=1.99), condoms (RR=1.71), and less likely to use other methods (RR=0.59) rather than hormonal methods in 2010/11 compared with 2005/6. The relative risk of using condoms rather than hormonal methods was significantly higher for women aged below 34 years by (RR=2.16) compared with younger women below the age of 24 years of age. Similarly, women over 35 years of age were more likely than young women below 24 years of age to use long-term methods. (RR=4.24), condoms (RR=4.30), and other methods (RR=3.58) rather than hormonal methods. The relative risk of choosing long-term over hormonal methods was significantly lower for women who were married (RR=0.11), widowed (RR=0.33) and divorced (RR=0.34) compared with never married women. At the same time, married women were less likely to choose condoms than hormonal methods by (RR=0.24) compared to never married women. Religiosity also matters in decision-making about contraceptive methods choice for women who know their HIV status. Protestant religious affiliation showed higher relative risk of using condom (RR=1.63) than hormonal methods compared with traditional and Catholic affiliation.

Furthermore, relative to women who knew their HIV status and reported that they want nomore children, women whose fertility intention was to have another child later than two years were less likely to use long-term methods (RR=0.44), and more likely to use condoms (RR=3.12) than hormonal methods. Also, the relative risk of belonging to a group of women choosing long-term methods other than hormonal methods is significantly lower for higher birth order 2 by (RR=0.61) and birth order 3 plus by (RR=0.53) compared with women with zero or first birth. Meanwhile, the relative risk of using condoms rather than hormonal methods was higher among women with higher birth order of three plus pregnancies (RR=2.64) compared with women of low birth order. Additionally, being exposed to family planning messages increased the likelihood of condom use (RR=1.75) compared to hormonal methods. The relative risk of using long-term over hormonal methods was also significantly higher for women who knew their HIV status and had multiple sex partners compared with women who had only one sexual partner.

There was a clear wealth differential in methods choice; and the likelihood of using long-term methods and condoms over hormonal methods increased with wealth from richer to richest. The richest had the highest likelihood of using both long-term methods (RR=2.23) and

condoms (RR=2.10) compared to hormonal methods compared with the poorest. We have seen from the main effects that the richest women who knew their HIV status were more likely to use long-term methods and condoms than hormonal methods compared with the poorest. The question that can be posed here is whether this behaviour applies equally for wealthier HIV positive women who knew their HIV serostatus. There is reason to believe that wealthier HIV positive rather negative women are more likely to use some active form of contraceptive methods such as long-term methods which are likely to be more expensive (Dias and Oliveira, 2015). Specifying HIV positive status and wealth term resulted in substantial changes in contraceptive methods choice as compared to the simpler wealth main-effects in the model. With the interaction effect between wealth and HIV positive status on contraceptive methods choice disappeared, while that of condom use and wealth remained significant, albeit negatively than in the main effects. Compared with those in the poorest wealth category with HIV positive status, women in the middle and highest wealth category with HIV positive status were less likely to use condoms (RR=0.07) and (RR=0.09) than hormonal methods.

Although urban residence and higher levels of educational attainment were expected to result in better choice of contraceptive methods including condom use, the results suggest no evidence of an association between contraceptive methods choice and these two variables. When community level factors were controlled for, it was found that women from poor communities were less likely to use condoms by (RR=0.20) than hormonal methods compared with their counterparts from wealthier communities. A significant community effect is seen in the multilevel modelling, suggesting that unobservable community factors influence the choice of different types of contraceptive methods. As shown by the variance partition coefficient (VPC), the intra-class correlation coefficient (ICC) is estimated from 14-20 % for the three contraceptive methods choice even after controlling for individual and community level factors. Thus, about 14-20% of the total unexplained variation in contraceptive methods choice could be attributed to unobserved community-level effects with the remaining 80-86% of unexplained variation attributable to individual-level factors. An example of this variation is that of the choice between condom and hormonal methods which suggest that 20% of the total unexplained variation in the choice of hormonal methods versus condom use is attributed to community factors not included in the model. This result implies a higher degree of homogeneity in the use of hormonal methods within the communities, but does not shed light on whether the factors behind that homogeneity are cultural or service related.

Table 7-3: Results of the multilevel multinomial logistic regression models to explore associations between HIV status and contraceptive method choice (adjusted RR and 95% CI of RR)

Demonstrate (D. f)		Long-termVs		Condom Vs		Others Vs
Parameters (Ref)		Hormonal		Hormonal		hormonal
Main Effects	RR	95% CI	RR	95% CI	RR	95% CI
Survey (2005-6)						
2010	1.99	[1.38-2.87]*	1.70	[1.14-2.52]*	0.59	[0.37-0.93]*
Place of residence	(urban))				
Rural	1.68	[0.62-4.55]	0.25	[0.07-0.88]*	0.49	[0.13-1.93]
Age groups (15-24))					
25-34	1.38	[0.91-2.08]	2.16	[1.38-3.38]*	1.49	[0.80-2.79]
35+	4.24	[2.54-7.08]*	4.30	[2.53-7.30]*	3.58	[1.68-7.63]*
Marital status (Nev	er Mai	ried)				
Married	0.11	[0.07-0.17]*	0.24	[0.14-0.43]*	1.46	[0.55-3.90]
Widowed	0.33	[0.17-0.66]*	1.02	[0.45-2.29]	3.12	[0.84-11.64]
Divorced	0.34	[0.20-0.60]*	0.88	[0.42-1.74]	0.34	[0.04-2.59]
Education (primary	/)					
Secondary/higher	0.77	[0.55-1.09]	1.01	[0.72-1.42]	1.14	[0.68-1.91]
Religious(/catholic)					
Protestant	1.13	[0.75-1.69]	1.63	[1.13-2.36]*	0.74	[0.39-1.38]
Pentecostal	0.95	[0.64-1.42]	1.15	[0.78-1.69]	0.77	[0.41-1.46]
Apostolic	1.03	[0.69-1.53]	0.61	[0.42-0.89]	1.22	[0.73-2.04]
Fertility intention (no-mo	re)				
Now - 2 year	0.85	[0.52-1.38]	1.22	[0.66-2.25]	0.95	[0.46-1.97]
Later- 2 years	0.44	[0.30-0.66]*	3.12	[2.33-4.34]*	0.61	[0.35-1.06]
Undecided	0.72	[0.41-1.24]	1.22	[0.66-2.25]	0.83	[0.35-1.98]
Parity (0-1)						
2	0.61	[0.41-0.93]*	1.29	[0.84-1.98]	0.75	[0.42-1.36]
3+	0.53	[0.32-0.61]*	2.64	[1.64-4.26]*	0.72	[0.37-1.41]
Enabling factors						
Household wealth						
Middle	1.27	[0.75-2.16]	1.13	[0.68-1.87]	0.86	[0.40-1.87]
Rich	2.02	[1.15-3.53]*	1.06	[0.59-1.91]	1.90	[0.83-4.35]
Richest	2.23	[1.20-4.13]*	2.10	[1.09-4.07]*	0.91	[0.32-2.59]
Interaction effects	3					
HIV +*Wealth (pe	oor)					
Middle	0.10	[0.24-3.84]	0.07	[0.02-0.33]*	0.48	[0.06-3.80]
Rich	0.27	[0.06-1.18]	0.34	[0.11-1.06]	0.39	[0.06-2.63]
Richest	0.63	[0.15-2.54	0.09	[0.03-0.30]*	2.10	[0.33-3.39]

Continued

FP media exposure							
Medium	1.29	[0.85-1.94]	1.48	[0.97-2.25]		1.26	[0.65-2.46]
High	1.57	[1.00-2.44]	1.75	[1.13-2.73]*		1.39	[0.66-2.94]
Risk perceiption/r	need fa	ctors					
HIV- sero status (n	.0)						
Yes	1.67	[0.52-5.58]	3.13	[1.40-7.00]*	1.6	2	[0.35-7.48]
Know someone di	ied of H	IIV/AIDS (no)					
Yes	2.66	[1.30-5.46]*	1.53	[0.85-2.76]*	0.72	2	[0.37-1.42]
Multiple sex partners (one)							
2	2.06	1.48.2.86]*	1.31	[0.93-1.84]	1.4	5	[0.89-2.36]
3+	3.61	[2.61-5.00]*	1.36	[0.91-2.03]	1.3	2	[0.71-2.45]
Contextual factor	S						
Poverty	1.02	[0.48-2.15]	0.40	[0.17-0.92]*	0.7	9	[0.23-2.69]
Random effects parameters							
Community-level							
Variance (SE)	0.59	[0.141]*	0.867	[0.198]8			0.567 [0.483]*
(VPC)=ICC(%)		14		20			15

Table 7-3 continued

VPC=Variance Partition Coefficient, ICC=Intracluster Correlation Coefficient, *Statistical significance at 5% level p<0.05

7.5 Discussion and conclusion

The study sought to ascertain the effects of knowing HIV sero-status on contraceptive methods choice and to identify Individual and community level factors that are associated with contraceptive method choice after HIV diagnosis. While comparison of factors associated with contraceptive method choice between HIV-positive and HIV-negative women was one of the primary objectives, the analysis was not possible due to a small number of women with an HIV-positive test. However, the study did find an important association between knowledge of an HIV test and contraceptive method choice. It turned out that a known HIV positive status drives women to use condoms rather than hormonal methods. Women who knew their HIV positive status and were using contraception were more than three times more likely to use condoms rather than hormonal methods than women who knew they were HIV negative. The results are consistent with previous studies which found higher odds of condom use among women who knew they were HIV positive than HIV negative women, compared with hormonal contraceptive methods (Church et al., 2014; Habte and Namasasu, 2015; Heys et al., 2009; Johnson et al., 2009; Keogh et al., 2009; Mumah et al., 2014). The finding is a welcoming development; given that advocacy for condom use has been a major component of HIV prevention in Zimbabwe (Chevo and Bhatasara, 2011). One plausible explanation of higher use

of condoms may be that counselling received by women who knew their HIV positive status might have included contraceptive methodwhich mostly likely increased their knowledge of the benefits of condom use (Mayondi et al., 2016). Church et al., (2014) contend that increased condom use among HIV positive women results from programs focusing on condom use. Church et al., (2014) argued that even at a facility where other methods are available, providers consider condoms imperative for HIV positive women to prevent re-infection or onward transmission of HIV. This message was clearly understood by women who knew their HIV positive status; thus, infection concerns predominates over pregnancy concerns (Elul et al., 2009); and other methods are considered as aback-up (Church et al., 2014).

The result that wealthier women who knew their HIV status were more likely to use condoms and long-term methods may suggests that women with economic resources gain decisionmaking power within their relationships, and they use this power to insist on always using condoms, the highest level of protection (Church et al., 2014) or they have resources to afford the long-term methods which are very expensive in Zimbabwe. Poverty which correlates with education can also influence women's sexual behaviour by reducing access to information about safe sex practicing or by inhibiting their ability to put such information into practice; such that being poor, regardless of HIV status may inhibit women from accessing long-term methods and condoms (Akimenyi et al., 2010). This phenomenon helps explain the finding in this study that women who knew their HIV status and have high access to family planning messages were more likely to use condoms compared with those with low access to family planning methods. The argument here is that HIV-positive women who know their status and who are exposed to family planning messages from media are more likely to use condoms because the messages they receive may increase the knowledge of available contraceptive methods (Bankole et al., 2009). Mass media channels, radio, television, and newspapers for example, have been suggested to be vital sources of information about HIV/AIDS for ordinary people (Jesmin et al., 2013). Although, HIV/AIDS-related health promotion has mainly targeted high risk groups through media in Zimbabwe, the majority of women do live far away from health facilities or they have no means to buy newspapers and televisions. The most commonly cited source of information on family planning was radio, while television and newspapers were mentioned by a minority of women in rural areas (ZDHS, 2010/11). Thus, the results that poorer women who knew their HIV status and have low access to family planning messages are less likely to use long-term methods and condoms in Zimbabwe are credible. These findings can be used to guide health care managers to target radio over other alternatives as a means to reach the poorest women with family planning messages.

The study further explored the interaction between HIV status and wealth on contraceptive method choice for both HIV positive and HIV negative women. It became apparent thatbeing wealthy and knowing your own HIV status matters for contraceptive method choice. The results showed being wealthier and HIV positive status decreases substantially the odds of condom use. The low condom use among wealthier women who knew their HIV positive status maybe as a result of national programs targeting poorer HIV positive women (ZIMSTA, 2016) in Zimbabwe. Other interactions checked such as HIV status and media exposure, HIV awareness and educational attainment were not significant.

Other important background correlates of contraceptive method choice among women who knew their HIV status included fertility intention, age, parity, religion, marital status and multiple sex partners. The low use of long-term methods among women who reported wanting children later does not agree with other studies which noted an increase in long-term methods indicating a desire for child spacing among the women who know their HIV status (Mayondi et al., 2013). Although it would be interesting to compare fertility desires for HIV positive and HIV negative women, analysis of this relationship was not possible in MLwin due to small numbers of women who were undecided on their fertility desire. However, it was postulated that HIV positive women may not want any more pregnancies due to fears of compromising their health and fear of transmitting HIV to children if they conceive (Habte and Namasasu, 2015), leading to an increase in the use of permanent contraceptive methods. For those who know their HIV-negative status, the results may imply that those using long-term methods are child spacers or that they want to limit childbearing for economic reasons (Bertrandet al., 2014). Research into the plight of low-income households and large families provide evidence to support this understanding (Shelton and Burke, 2016).

The apparent higher inclination towards long-term methods rather than condoms among women with multiple sexual partners compared to those with single partners is not what is expected, given the widespread condom use campaign in Zimbabwe. This may indicate that part of the C (condom use) elements of ABC health messages, involving always using condoms

with multiple sexual partners has not been effective in Zimbabwe. Studies have shown that long-term methods do not protect against HIV, only condoms can (Church et al., 2014).

The finding that older women are more likely to use long-term methods and condom rather than a hormonal method is expected. Younger women are less likely to use any form of contraception (Mayondi et al., 2016). The use of any form of contraception rises to its peak in the age range 20-29 years as women start having children and use contraception for child spacing, while non-permanent methods such as the pill and the male condom are the most preferred methods (Nattabi et al., 2009). Beyond 30-44 years, as women get older, the likelihood of use declines, and tilts towards more permanent methods such as injections (Shelton and Burke, 2016). There is evidence in support of this phenomenon (Church et al, 2014; Wagner and Wanyenze, 2013) to agree with the hypothesis of this study that increased age comes with an increase in the use of more effective contraceptive methods such as longterm methods for delaying and limiting child birth. Other studies have postulated that with increased age and an improved knowledge of HIV comes with increased likelihood of choosing long-term and condom contraceptive methods (Anand et al., 2009). Due to the importance of knowledge of HIV and the availability of contraceptive options (including long-term and condoms), it is highly likely that older HIV positive women who know their HIV status may choose long-term methods and condoms relative to hormonal methods in Zimbabwe. For younger women, regardless of their HIV status and their knowledge of HIV status, research has shown that they may prefer reversible contraceptive methods in anticipation of future childbearing (Mumah et al., 2014) or condoms particularly to prevent both unplanned pregnancies and HIV infection (Nattabi et al., 2009).

The results also show that married, divorced and widowed women who know their HIV status were less likely to choose long-term methods and condoms compared with single women. This could reflect the growing popularity of hormonal methods among women and that there is a slow shift from hormonal methods use to long-term, condom and other methods in Zimbabwe. It could also be that for single women who know their HIV status and who have not achieved their desired number of children their motive was to achieve pregnancy (Church et al., 2014) leading to their continued use of hormonal methods which are reversible. The low use of condoms among married women was expected. Studies have shown that condom use within the marriage institution are highly unlikely particularly among HIV positive women due to

disclosure concern (Akinyemi et al., 2010), feeling of rejection (Sambisa, 2008), abuse and concern for partner compliance with condom use (Rosenberget al., 2013).

The finding that women with higher order births are less likely to use long-term methods does not agree with other studies. It has been suggested that once a woman has had a first birth, the priority shifts towards birth spacing, primarily for economic reasons as more children affect caring for other offspring (Heys et al., 2009). The ability to delay and space childbearing is crucial to women's social and economic advancement (McCoy et al., 2014), therefore women with higher order parity are more likely to use irreversible methods. Therefore, this study offers no plausible explanation as to why women with higher order parity who know their HIV status were less likely to use long-term methods in Zimbabwe.

Although, there is a discernable shift in contraceptive methods choice from hormonal methods to long-term and condoms, this study argues that the level of use of long-term methods and condoms is experiencing a far too slow pace of change to justify any claim that HIV testing, family planning and condom promotion campaigns have been successful in effecting change of behaviour among vulnerable populations in Zimbabwe. Although it is agreeable that condom use adoption requires substantial adjustments in attitudes towards procreation (Habte and Namasasu, 2015); or that to expect an abrupt change in behaviour, particularly when some warnings of the threat of AIDS and methods of prevention and treatment of HIV come from largely from developed countries, which are initially received with denial and suspicion in much of Africa (Kayikiet al., 2011), a faster pace of behavioural modification in adopting condom use among women who engage in sex with multiple partners knowing their HIV status would have been highly desirable in Zimbabwe.

The central conclusion in this study is that the success pervading HIV prevention efforts through condom use and long-term contraceptive methods in Zimbabwe cannot be justified and that the investment in condom use and marketing has been an unappreciated effort, particularly for women who know their HIV status. The fact that the majority of women who know their HIV status are still using mainly hormonal methods is not a good sign of any shift of contraceptive mix in order to protect both unwanted pregnancies and HIV. There is a need for health managers to design programs that explicitly define the HIV protection component of condom use, as some women may not understand the dual protection concept of condom use,

which is being widely promoted. Although dual use of condoms with other contraceptives was of particular interest, this was not used as a distinct category because of very small numbers. However, HIV testing and receiving HIV test results which is part of voluntary and counselling HIV test prevention program does have an effect on the choice of contraceptive methods.

8 CHAPTER EIGHT: CONCLUSIONS AND RECOMMENDATIONS

8.1 Introduction

This chapter summarises the contributions of this thesis and the possible impact on health policy. The central issue addressed is the problem of underutilisation of HIV and sexual and reproductive integrated services. This problem was addressed by identifying determinants of HIV testing, maternal health care and family planning service utilisation, which constitute HIV and SRH-integrated programs in Zimbabwe. The current integrated programme is a strategic intervention to increase uptake of HIV interventions through sexual and reproductive healthcare services. The conceptual framework adopted for the analysis of this thesis is a modified version of the framework by Andersen (1996) on determinants of healthcare-seeking behaviour, which proposes that health outcomes are dependent upon individual and community-level predisposing factors. This thesis is based on the 2005/06 and 2010/11 Zimbabwe Demographic and Health surveys. The main statistical methods used are multilevel binomial and multinomial logistic regression models. There are some limitations to this study that have potential implications for our findings. The first second limitation concerns the use of 2005/6 and 2010/11 DHS data. This thesis has acknowledged that these data are relatively old, and the information provided might not reflect the current situation and practices. In countries like Zimbabwe, where the scaling up of prevention of parent-to-child HIV transmission is happening, use of the 2015/16 DHS data would have helped capture more recent progress. However, the 2015/16 DHS data were not yet available at the time the analysis of this thesis was being undertaken. The second limitation relates to the cross-sectional nature of the DHS data analysed, which does not permit temporal ordering of the events of interests. Therefore, the patterns observed represent simple associations and cannot offer insights into cause-and-effect relationships. Furthermore, because DHS surveys rely on a self-reported method of data collection, inaccuracies can come up because of intentional deception, poor memory, or misunderstanding of questions. Finally, this study has focused on major determinants of antenatal HIV testing in Zimbabwe with respect to demand-related factors (i.e. cultural, demographic and socio-economic etc). However, as argued by Buzdugan et al. (2015), service-related factors are also important determinants of exclusion and need investigation. Future research should explore the impact of the providers' characteristics of HIV and ANCintegrated service delivery. Elements such as quality of care, type of provider, actual distance and attitude of staff need to be explored. Nevertheless, the representation of the whole country is a major strength of the data analysed in this study, as it allows generalisation of the findings nationally.

8.2 Summary of key findings

This thesis consists of four analytical chapters: the determinants of HIV testing; the determinants of ANC and delivery care; and knowledge of HIV status and contraceptive method choice in Zimbabwe. The analysis showed that these services were determined by approximately the same set of factors, thus, the key findings and objectives of chapters 4 to 7 are brought together (Table 8.2). By integrating the summary of findings, this thesis attempts to reveal and compare how determinants of the utilisation of HIV and SRH-integrated services interplay. The overall findings from this thesis are located within the general analytical framework outlined in Section 2.4 of chapter 2. Based on the analytical framework, this thesis' hypothesis is that a continuum of individual and community factors relating to 'predisposing', 'enabling' and 'need/perceived risk' factors (see fig 2.2 of chapter 2) are the key determinants of HIV and SRH service utilisation. The use of multilevel models has helped to better explain the different levels at which various factors influence outcomes, for instance, at individual and community levels in the case of population-based data. The multilevel results showed small variations in HIV and SRH service utilisation across communities in Zimbabwe. The variations were mainly explained by individual-level predisposing factors, such as background characteristics, and enabling factors, relating to socio-economic status as well as the perceived risk/need of HIV and/or the need for healthcare services, rather than community factors that were included in the models used throughout this research. Although some variables influenced the outcomes, there is evidence that some factors in the models were affecting the uptake of different HIV and SRH-integrated services differently. This means that policymakers have to be careful in terms of structuring the strategies to improve the HIV and SRH-integrated programme in Zimbabwe. For instance, a strategy that might improve delivery care may not necessarily have a significant impact on HIV testing, maternal healthcare service utilisation and the contraceptive choices of women who know their HIV status. Policymakers have to appreciate these differences and strategies should be targeted at specific components of HIV and SRH-integrated service utilisation rather than planning umbrella strategies.

Controlling for community-level factors such as HIV prevalence, awareness and stigma, the effect of individual 'predisposing', 'enabling' and 'perceived risk/need' factors remain largely

the same across the study outcomes under scrutiny. This may suggest that HIV factors relating to community prevalence, stigma and awareness do not explain much of the differences in HIV and SRH service utilisation that are observed. The models suggest that 'predisposing' individual-level factors (i.e. place of residence, age, marital status, education, religion, parity and age at marriage) were the most important factors for predicting HIV and maternal healthcare service utilisation and explained most of the variance, while there was also a reasonable amount of variance explained by 'enabling' factors (e.g. household wealth, media exposure, access to health care and HIV testing during pregnancy and timing of ANC for delivery care). The results provided further evidence that uptake of antenatal care services, such as ANC in the first trimester, ANC HIV testing and optimal ANC visits are key factors for delivery care. Given that having ANC visits in the first trimester, having four or more ANC visits and having ANC HIV testing were significantly associated with delivery care, encouraging women to seek ANC care and to have HIV testing during pregnancy may enable pregnant women to have institutional and professional delivery. More importantly perhaps, focus should be on reducing involuntary HIV status disclosure to families and friends who accompany pregnant women during labour and delivery. Asking permission from pregnant women to allow families and friends to be present during delivery care may reduce stigma associated with involuntary disclosure (chapter 6). Finally, the study found little evidence of the impact of integrating HIV testing services with delivery care, apart from the negative effect of HIV status awareness on institutional delivery which may be a result of pregnant women being concerned with how their information relating to HIV is held. The summary of integrated findings for chapters 4 to 6 is presented in table 8.1 below.

Parameters	HIV testing	1 st trimester ANC	ANC HIV Testing	4 + ANC visits	Delivery care	Method choice			
Individual predisposing factors									
Gender (males)									
Females	[+]	N/A	N/A	N/A	N/A	N/A			
Residence (rural)									
Urban	[+]	[+]	N/S	N/S	[-]	[*]			
Age groups (15-24)									
25-29	[-]	[-]	[-]	[-]	[-]	[*]			
30-34	[-]	[+]	[+]	[-]	[+]	[*]			
35+	[+]	[+]	[-]	[-]	[+]	[*]			
Marital status (never m	arried)								
Married	[+]	[+]	[-]	[-]	[+]	[*]			
Widowed	[+]	[+]	[-]	N/S	[+]	[*]			
Divorced	[+]	[+]	[-]	[-]	N/S	[*]			
Wanted child (yes)									
Later	N/A	[-]	[-]	[+]	[-]	[*]			
Not wanted	N/A	[-]	N/S	[+]	[-]	[*]			
Parity (0-1)									
2	N/A	[-]	N/S	[+]	[-]	[*]			
3+	N/A	[-]	[-]	[+]	[-]	[*]			
Enabling and perceive	ed/need facto	ors							
Wealth (poor)									
Richer	[+]	[+]	[+]	N/S	[+]	[*]			
Richest	[+]	[+]	[+]	[-]	[+]	[*]			
Education (primary)									
Secondary plus	[+]	N/S	[+]	N/S	[+]	N/S			
Media exposure (low)									
Higher	[+]	[+]	[+]	N/S	[+]	[*]			
Previous HIV test (no)									
Yes	N/A	[+]	[+]	[+]	N/S	[*]			
ANC HIV testing (no)									
Yes	N/A	N/A	[+]	[+]	[+]	N/A			
HIV/AIDS Awareness	(low)								
High	[+]	N/S	N/S	N/S	N/S	N/S			
HIV/AIDS stigma (no)									
Yes	[-]	N/S	[-]	N/S	[-]	N/S			
HIV disclosure concern	n(No)								
Yes	[-]	[-]	N/S	[-]	[-]	N/S			
Knows someone died o	f Aids(no)								
Yes	[+]	[+]	[+]	N/S	[+]	[*]			

 Table 8-1: Summary of the effects of individual and community factors of HIV and SRH outcomes

Continued

Table 8-1 continued

Timing of ANC visits	s (no)					
1 st trimester	N/A	N/A	N/A	N/A	[+]	N/A
No. of ANC visits (less than 4)						
4 or more	N/A	N/A	N/A	N/A	[+]	N/A
Community factors						
HIV/AIDS stigma	[-]	N/S	N/S	N/S	[-]	N/S
Media exposure	[+]	N/S	N/S	N/S	N/S	N/S
HIV awareness	[+]	N/S	[+]	N/S	N/S	N/S

(-) =decreases the odds of an outcome (+) = increases the odds of an outcome (*)=significantly correlated, N/S=not significant at 5% level, N/A = Not applicable

8.2.1 Association between individual-level predisposing factors and HIV and SRHintegrated health outcomes

The results presented here reveal that the individual-level 'predisposing' determinants of HIV and maternal health outcomes under scrutiny are comparable. The associations observed are largely consistent with patterns observed in other SSA countries whereby the healthcare service utilisation was determined by gender, place of residence, age, marital status, education, parity, child wantedness/fertility intention and religion. In particular, the findings are in agreement with studies relating to HIV testing in Ethiopia and Botswana (Godif et al., 2015; Obermeyer et al., 2013), timing and frequency of ANC visits in Tanzania and Kenya (Gross et al., 2012; Magadi et al., 2000), delivery care (Johnson et al., 2009) and contraceptive choices among women (both infected and uninfected with HIV). The observed gender disparity in HIV testing, discussed in chapter 4, supports findings from earlier studies suggesting that females are more likely than males to have an HIV test due to the fact that they are more likely to be exposed to healthcare services, probably through maternal healthcare and family planning services (MacPhail et al., 2009). This clearly calls for policy makers to consider couple-oriented HIV counselling and testing where men accompany their spouses to HIV screening during pregnancy. Couple-oriented HIV testing has the potential to help increase HIV uptake for males and reduce gender disparities. This may also help increase the uptake of the continuum of the HIV and SRH-integrated programme in Zimbabwe, including contraceptive choice. This is because couple-oriented HIV counselling and testing has been shown to improve dialogue and spousal communication about issues related to SRH and HIV (Kalembo et al., 2013) leading to increased service utilisation by couples.

The evidence put forward in this thesis suggests that urban residence is associated with a higher uptake of HIV testing and ANC visits in the first trimester. However, it is also associated with poor delivery care and low use of condoms and long-term contraceptive methods. There are no significant differences detected between antenatal HIV testing and frequency of ANC visits by place of residence. The observed patterns for place of residence and antenatal HIV testing and frequency of ANC visits contradict the results from earlier studies suggesting that women from urban areas are more likely to have antenatal HIV testing and to have more ANC visits due to easier access to ANC services, including prevention of mother-to-child transmission (PMTCT) services (Chapter 5). Taking key factors into account, we can attest that the lack of significance of place of residence to antenatal HIV testing confirms that PMTCT programmes are now achieving comparable HIV testing coverage in urban and rural areas. However, the fact that rural residents are not necessarily worse off, in terms of their access to HIV testing, compared to their urban counterparts of similar characteristics (Table 5.8, Chapter 5), may not necessarily suggest equity in healthcareaccess. Some urban sub-groups may require particular attention, taking into account the different nature of deprivation within urban areas (Matthews et al., 2010). It has been suggested that the urban poor have very similar or perhaps even lower levels of progress, in terms of access to healthcare, than in rural areas due to urbanisation (Usman, 2016). More recent evidence (Usman, 2016) reveals that health and social services in urban areas have not kept pace with urban population growth due to large-scale migration from rural to urban areas in Zimbabwe, leading to health inequities within urban areas.

The significant negative impact of rural residence on HIV and SRH outcomes observed in this study could be in accordance with the notion of urban health advantages which has been described in chapters 4 to 6. This could be a revelation of the challenges, such as the lack of education and information relating to HIV, that women in rural areas in Zimbabwe are facing, preventing their access to healthcare services. In order to fill the gap in education and address the lack of HIV-related information, health workers can be used to provide knowledge to village communities, using non-written methods of communication, such as pictures, for those who are illiterate (Chapter 5). Tweya et al. (2014) have associated increased household visits by health workers and a higher density of health committees with an increased knowledge relating to HIV among rural communities, resulting in a better timing of ANC visits and greater acceptance of HIV testing in Malawi.

The relationship between rural residence and contraceptive methods choice, discussed in chapter 7, is also worth mentioning because the low use of condoms and long-term contraceptives in rural areas may be the result of better access to hormonal contraceptive methods at the expense of condoms in rural areas. The fact that condom use is lower in rural areas ran counter to the aim of HIV and family planning integrated services which promote condom use for both HIV prevention and unwanted pregnancies (Chapter 7). Regardless of HIV status, policymakers are now aware of the need to design a programme aimed at increasing condom use in rural areas.

The results revealed mixed effects of age on HIV and SRH service measurement outcomes. It emerged that young individuals were, in general, less likely to use HIV and SRH services. One important observation to emerge from the analysis of chapter 7 was the fact that, despite their HIV serostatus, the relative risk of using condoms and long-term contraceptive methods rather than hormonal methods was significantly higher for women aged above 34 years compared with younger women below the age of 24 years of age. The limited choice of contraceptive methods available to young women has important implications for prevention of mother-tochild HIV transmission (PMTCT) programmes through condom use and long-term contraceptive methods. The low use of HIV and SRH services including condoms and longterm contraceptive methods among young people has been attributed to the fact that HIV and SRH healthcare services are geared towards older and married women in most SSA countries (Buzdugan et al., 2015). Although various policies exist aimed at improving HIV and SRH health outcomes for adolescents, like sex education in communities and in schools (Chevo and Bhatasara, 2012), for instance, they are not well implemented and do not ensure comprehensive coverage ofhealth services for young people. Further steps need to be taken to improve the health of young people which may include the provision of youth-friendly health services, the dissemination of information and the raising of awareness regarding the implications of HIV on their health. If adopted, these methods would improve the conditions and address the HIV and SRH health needs of young and unmarried women.

Furthermore, the results revealed that older individuals (over 35 years of age) are less likely to have ever been tested for HIV or to have antenatal HIV testing and four or more ANC visits. I attest that this could be due to the false perception of diminished HIV risk as a result of having been previously tested for HIV or previously given birth to presumably healthy children, both
of which are key factors for future utilisation of HIV and SRH-integrated services (Chapter 5). This is consistent with Ochako and Gichuh's (2016) finding that pregnant women who had previously given birth to presumably healthy children were less likely to opt-in to voluntary HIV testing in subsequent pregnancies. The finding that older women are less likely to use HIV and SRH-integrated services certainly implies that the campaigns to promote PMTCT programmes have not been effective among this group of individuals. While it is important to improve HIV and SRH service utilisation among youth, policymakers should also pay special attention to the improvement of health needs of older individuals, and older women in particular.

It has also been noted that older women were less likely to have at least four ANC visits throughout pregnancy, but more likely to have ANC visits in the first trimester. This finding supports Mason et al.'s (2015) observation that being given a registration card was the main motivation for women to attend ANC in the first trimester so that they could receive medication and deliver in hospital in case of emergency. As discussed in Chapter 6, hospital delivery is denied to women who do not have a registration card in Zimbabwe. The results of this research certainly imply that once a pregnant woman is registered and has received a card, presumably in the first trimester, she stops visiting health facilities for fear of associated financial costs as well as the opportunity costs of the time she would spend to do so (Moindi et al 2015). She would then rely on a traditional birth attendant (TBA) for her subsequent ANC after the initial registration and presumable HIV testing (Maruva et al., 2014). While the need to obtain a registration card may be an important reason for initiating ANC in the first trimester, the higher odds of institutional and professional delivery for older women observed in this thesis may mean that this group has benefited enormously from information and education campaigns as well as community mobilization efforts aiming to increase institutional and professional delivery (Loewenson et al., 2012). Although the majority of pregnant women have reported having ANC in the first trimester, there is a need to raise awareness on the importance of HIV testing and regular ANC visits, given their central role in HIV and SRH service delivery.

Marital status has shown paradoxical associations with HIV and maternal healthcare service utilisation. However, in line with the findings of many studies (for example, Semali et al., 2014) this study found positive associations between being married, divorced and widowed and better timing of antenatal care visits, institutional delivery and uptake of HIV testing. The findings,

particularly that unmarried women - usually young teenagers - were less likely to have ever been tested for HIV, have an ANC visit in the first trimester or institutional delivery, have important implications for prevention of mother-to-child HIV transmission (PMTCT) programmes. This can certainly increase mother-to-child HIV transmission among the youth. Thus, it is important to provide information and raise awareness of PMTCT programmes to prevent further mother-to-child HIV infection among unmarried young mothers in Zimbabwe.

It was found that women with higher-order parity and those reporting unwanted or mistimed pregnancies were less likely to use HIV and maternal healthcare services, but were more likely to use condoms. The fact that pregnant women with a higher-order birth parity were less likely to initiate ANC in the first trimester and delivery care may be a reflection of lack of time, limited resources in the household, constraints related to their responsibilities towards their other children and negative perceptions resulting from previous pregnancies, if the woman was in her second or higher-order pregnancy (Chapters 5, 6). Higher-order parity and mistimed or unwanted pregnancy may also be a reflection of a lack of access to appropriate family planning services (Chapter 7). Previous studies have suggested that a pregnancy that is mistimed or not wanted leads to non-utilisation of maternal healthcare services and worse choices regarding contraceptive methods (Muchabaiwa et al., 2012; Ochako and Gichuh, 2016). While the findings in this thesis uphold the conviction that a pregnancy that is mistimed or not wanted may lead to non-utilisation of maternal healthcare services (Nattabi et al., 2009), there is still a need, however, to identify women who get unwanted pregnancies in Zimbabwe to allow for targeted intervention. It is likely that if they are teenagers, they will be afraid of revealing that they are pregnant (Renju et al., 2010), or they may have a false perception of low risk of HIV transmission. The implication is that they will be left out of the HIV and SRH-integrated programme because of fear. In fact, it has been reported elsewhere in this thesis' analysis that teenagers are less likely to receive adequate HIV and SRH healthcare services compared with older women. Policymakers and programme designers may need to target women with higherorder parity and those who reported unwanted pregnancies, particularly young teenagers, as the priority of programmes aimed at increasing uptake of ANC in the first trimester. They should also intensify integration of family-planning services that aim to increase uptake of more effective contraceptive use, such as long-term methods, across sexual and reproductive healthcare delivery services (Chapter 7).

While policymakers and programme designers are now informed that individual-level predisposing factors must be the priority for programmes aimed at increasing uptake of HIV and SRH-integrated service utilisation, we must bear in mind that these factors are not the ultimate determinants of service utilisation. Individuals' predisposing demographic factors alone may not be enough for the actual use of health services (Framework 2.2). The individual and community enabling, perceived risk or need factors that were found to be associated with HIV and SRH service utilisation in Zimbabwe are presented and discussed in the following sections.

8.2.2 Association between individual and community-level social and economic enabling factors and HIV and SRH-integrated health outcomes

The study identified the remarkable influence of social and economic enabling factors on HIV and SRH-integrated health outcomes, confirming and strengthening my analytical framework which originally assumed that enabling factors relating to socio-economic status (SES) such as wealth, education and media can facilitate or inhibit individuals from accessing health care services (Chapter 3). The household socio-economic status was observed to be particularly important, influencing all the indicators of HIV and SRH-integrated services, with higher household socio-economic status being associated with improved HIV and SRH-integrated healthcare services. These findings are supported by various studies cited in chapters 4 to 7. In chapters 5 and 6, it was noted that poverty may decrease the likelihood of service utilisation due to a lack of financial wherewithal that would enable individuals to have access to maternal healthcare services. As was highlighted in chapter 5, pregnant women from poor households, especially in rural areas, are exempted from user fee payment. However, it cannot be ignored that the implicit cost of healthcare, outside of the user fees, charged for actually receiving the services may inhibit poor women from accessing maternal healthcare services. Women from poor households are less likely to be able to afford the costs associated with access to healthcare (e.g transport), as well as the fact that they may also have little access to information and education on HIV and its implications (Chapter 5). This means that even public health expenditure meant to help the poor ends up benefiting the rich as individuals from wealthier households are more likely to afford both user fee charges and transport costs (Muchabaiwa et al., 2012).

It was also interesting to note gender differences in the likelihood of HIV testing according to wealth in chapter 4. Household wealth was a more important factor for males, while community wealth or poverty was more significant for females. The gender disparity found here within the wealth factor may suggest that more males are economically empowered and have more resources enabling them to access healthcare services than females, thus, economic programmes that aimed at improving uptake of HIV and SRH-integrated programmes should be aimed at improving economic well-being of communities which in turn benefit women. In chapter 7, the importance of wealth in contraceptive choice by women with HIV status was examined. It turned out that, for females, being wealthier and HIV-positive was associated with decreased likelihood of condom use. Lower condom use among wealthier women who knew their HIV-positive status may be the result of national programmes targeting poorer HIV-positive women (ZIMSTA, 2016) in the country. It is important that policymakers put forward a standardised campaign for condom use across the country to ensure a homogenous type of healthcare services to be received by all individuals irrespective of where they live and their wealth status.

A number of economic policies to improve women's SES status can be recommended from these findings. Firstly, there is a need to intensify efforts to reach out to economically disadvantaged women in order to improve uptake of these services. Policymakers and programme managers should recognize that poverty reduction must be part of a greater societal approach that includes strategies to promote economic development for women. There is a need to invest resources in healthcare services that are affordable and accessible, especially those that promote antenatal and delivery care in disadvantaged communities. Secondly, the introduction of financial incentives may stimulate access to HIV and ANC integrated services by providing compensation for transport costs and the opportunity costs of time associated with access to these services for individuals. Monetary incentives can reduce both socio-economic barriers and HIV-related stigma as it may provider a broader reason for going to a health facility for HIV and SRH care (chapter 5). Acknowledging this allows better targeting of health policy and enables more accurate need-based planning. Thirdly, there is a need to increase the number of health facilities both in rural and urban areas which will likely ease the economic and geographical burden incurred by pregnant women seeking care. Efforts at bridging access to health care services by the government must start with providing the enabling infrastructure to bridge institutional determinants for intended users (Chapter 6).

The evidence from this study has also underlined the importance of educational attainment and access to media to HIV and SRH service utilisation. However, it was noted that while having higher educational attainment is an important key factor for HIV testing, antenatal HIV testing and institutional and professional delivery, the converse is true for the timing and number of ANC visits. Generally, education is a key indicator of socio-economic status and more educated people are more likely to have better health due to their ability to have better access to health information and to understand and respond to such information (Chapters 4-7). It is also possible that the disparities in HIV and maternal service utilisation by education may be caused by differences in awareness of the importance of testing, access to testing centres and riskier sexual behaviour (Chapter 4). Hence, as has occurred in other studies, health workers can be used to provide knowledge to villages which can be simplified, in the form of pictures, for those who are illiterate (Chapters 5, 6). Use of health workers may fill the void of formal education lacking in some sections of the population. A study in Ethiopia by Birhane et al. (2015) reported that many women acknowledged that their knowledge of PMTCT came from health workers. In addition to that, social groups, such as inter-governmental organisations' networking, mother's clubs and health committees can also be useful in promoting HIV and SRH-integrated services (Chapter 5).

There is also a need to invest in women's education in communities, particularly at the secondary and higher level in order to increase knowledge and awareness of HIV. Although, secondary educational attainment is near universal in Zimbabwe, there is a disparity in educational attainment between boys and girls, with girls lagging behind. Policymakers should adopt an education policy that compels parents to send their girl children to school (i.e. compulsory education). This can be achieved by enacting legislation that makes it illegal for parents not to send a girl child to secondary education. For those who cannot afford to pay school fees, a bursary support system should be established (chapter 5). In fact, the government's Basic Education Assistance Module (BEAM) programme was a step in the right direction and it should be continued. In order to make education more useful in respect to HIV and SRH-integrated healthcare service delivery, the school curriculum should include sex education, with emphasis on HIV and SRH. It is important to introduce HIV and SRH topics early so that young people can take control of their sexual lives by understanding dangers of risk behaviour and how to protect themselves against HIV transmission.

Access to information about HIV and SRH services can considerably increase the rates with which health services are being used. This study's findings illustrate a clear indication that exposure to mass media (i.e. newspapers, radio and television) plays a major role in informing women about HIV and SRH services. It was found that mass media is associated with use of all the components of HIV and SRH-integrated programmes, including that concerning contraceptive choice among women who know their HIV status. For instance, women who listened to the radio every day had higher odds of having ever been tested for HIV and antenatal care than those who listened less than once a week. Moreover, the likelihood of the use of institutional deliveries increases with media exposure, whereas women who are not exposed to media use maternal health services less frequently. Condom use among women who knew their HIV status was also higher among those who were exposed to media. Various studies have argued that exposure to mass media influences the use of HIV and maternal health services (Chapters 4, 5, 6). The finding that HIV and SRH service utilisation increases with increasing access to media might reflect the fact that individuals who are exposed to mass media may have the opportunity to learn the benefits of HIV testing, where and how HIV counselling and testing is given, and this subsequently improves their awareness towards HIV risk and HIV related knowledge. The results from this study would therefore provide valuable evidence and information to advocate for on-going community-based HIV and SRH media campaigns in Zimbabwe. The rationale for targeting interventions in communities with high levels of deprivation/disadvantage (i.e. poverty, low access to media etc.) is that it provides the most effective way of reaching families most in need, especially in a multi-cultural and diverse country like Zimbabwe (Chapter 5). However, community-based interventions have been criticized due to evidence that most of them were designed to be delivered in a local community or to respond to the variety of community needs, while few of them were actually designed to foster positive neighbourhood effects by treating the community itself as a factor effecting outcomes. Since my results showed that community media was not salient for the uptake of most of the outcomes, it can be suggested that the on-going electronic mass media (television, print and radio) should be scaled-up to include areas where it does not currently exit and should be strengthened in areas where it is weak.

8.2.3 Association between HIV 'enabling' and 'perceived risk/need'factors and HIV and SRH-integrated health outcomes

Our results suggest that HIV 'enabling' and 'perceived risk/need' factors that were influencing the different indicators of HIV and SRH service utilisation were not the same. However, both individual and community-level HIV awareness and stigma were found to be important for HIV testing, antenatal care HIV testing and institutional delivery, but not for ANC timing and frequency of visits or professional delivery. These associations were observed after considering the effects of education, implying that the patterns observed are not attributed to higher educational attainment among those with higher HIV/AIDS awareness or higher stigma. The patterns may be suggestive of a reasonable integration of HIV and reproductive information, education, and communication (IEC), such that individuals and communities with higher HIV/AIDS awareness or higher stigma are now endorsing HIV and SRH messages that tend to encourage HIV testing and institutional delivery. This pattern may also be a result of the success of HIV/AIDS awareness programs that are aimed at reducing stigma in the communities. The results that, in Zimbabwe, pregnant women who hold higher perceptions of HIV-related stigma are more likely to deliver at home is not good news because pregnant women remain at risk of HIV-infection throughout pregnancy (Chapter 6). Thus, more efforts should be focused on reducing HIV-related stigma among women who test HIV-negative during ANC. For those who test HIV-positive during ANC, other considerations, including her perceptions of the risk of mother-to-child transmission of HIV, or whether or not she has disclosed her HIV status to her husband and family, may be at the fore (Mlay et al., 2008; Turan et al., 2015). The importance placed on avoiding unwanted disclosure of HIV status in HIVpositive women's decisions about where to give birth has been reported in other studies (Manzi et al., 2014; Rochat et al., 2006). Other studies suggested that some HIV-positive women prefer a home delivery or to deliver somewhere their status is not known to avoid involuntary disclosure (Mucheto et al., 2011). It is therefore important that women who deliver at home be counselled to receive HIV testing when they first bring their infants in for vaccination or other postnatal care services (Fetene et al., 2010).

The fact that HIV/AIDS awareness and stigma in the community are important for HIV testing and institutional delivery suggests that those who were never tested for HIV and those who give birth at home would benefit from projects addressing issues of HIV awareness and stigma/prejudice in higher settings (Chapters 4, 5). Policymakers are now informed that the community should be an important target of their efforts to help these specific high-risk groups living in communities with high stigma and low HIV/AIDS awareness. Programme managers should use both electronic media (i.e. television, radio, internet, phone) and print media (i.e. leaflets, booklets and posters) to communicate and spread awareness, taking advantage of the widespread access to print and electronic media in Zimbabwe. More importantly perhaps, awareness raising should focus on avoiding stigmatising and discriminatory behaviours, so that pregnant women can develop positive attitudes towards people living with HIV/AIDS. This method of awareness raising can also help fill the gap of formal education, which is lacking in some sections of the population. It can also be used to influence decisions in cases where pregnancies are mistimed or unwanted as well as to change attitudes in the case of sociocultural enabling factors like religious affiliation.

This study found a strong association between timing of ANC visits and previous-HIV testing and counselling. It could be argued that pre-test HIV counselling which includes information on prevention of mother-to-child HIV transmission in the DHS is an enabling factor for early ANC attendance, in view of the fact women may be aware of their test results. Having previously tested for HIV before a pregnancy seems to have a greater impact on the utilisation of HIV and ANC-integrated care (Jayleen et al., 2016). Although knowledge of HIV sero-status was not controlled, due to fewer cases of HIV-positive women in the data, the behavioural changes among all women who had a previous HIV test, whether HIV-positive or not, has emerged as one of the key channels through which HIV and ANC uptake can be increased. While the study has not put forward a well-developed understanding of how knowledge of HIV-positive status from previous tests may affect utilisation of ANC, the results certainly imply that broader efforts at improving timing of ANC visits should be targeted at pregnant women with no previous HIV testing history. There is ample scope for the testing of all women in the relevant communities, which can complement the on-going national HIV testing programme in the country.

Confidentiality concern has also emerged as a significant barrier for increasing the uptake of all HIV and SRH service outcomes, but not for antenatal HIV testing. Pregnant women with no confidentiality concerns were more likely to have four or more ANC visits as well as delivery care. This may be that pregnant women who have had ANC in the first trimester may

have been tested for HIV and are more confident of their HIV status enabling them to attend further ANC (Lepine et al., 2014). The fact that antenatal HIV testing is not associated with concern may also be due to the fact that, in Zimbabwe, women are given private individual counselling and space to make a private decision without staff and group interference. However, as reported by Loewenson et al. (2002), infrastructural limitations, such as a lack of space and healthcare personnel, can reduce the effectiveness of service integration in Zimbabwe. This is in agreement with a study by Mucheto et al. (2011) which reported that providers felt that more rooms and staff in the health facility for simultaneous counselling on HIV and SRH would increase the likelihood of their clients receiving health services. This implies that stigma related to HIV in pregnancy still undermines the uptake of HIV and maternal healthcare. Policymakers in Zimbabwe are now informed that individuals with confidentiality concerns about how HIV information is handled are less likely to have HIV and SRH care. The policy on protecting confidentiality should therefore focus on individuals who report that they have concerns about privacy and how their information will be handled. Knowing either someone living with HIV or someone who died of AIDS was associated with an increased likelihood of HIV and SRH-integrated service utilisation. This may suggest that being personally acquainted with AIDS victims may be a catalyst for behavioural change, leading individuals to take protective measures, including condom use, to protect themselves against HIV infection. It can also have the effect of encouraging both infected and uninfected women to use long-term contraceptives to prevent unintended pregnancies (Chapter 7). Knowing someone who either is living with HIV or has died of HIV is common in communities where a HIV epidemic is more advanced, like in Zimbabwe.

Perhaps the most important finding reported in this thesis relates to evidence of antenatal care effects on delivery care. It was found that the majority of women who had institutional and professional delivery had their first ANC visit in the first trimester, had previously been tested for HIV as part of ANC and had had four or more ANC visits. These findings are consistent with the notion that access to antenatal care increases knowledge of risks associated with delivery care (Sanga et al., 2017; Sharma et al., 2015), including the fact that HIV can be transmitted during labour and delivery. The results imply that programme managers need to intensify efforts to link these services in order to reach as many women who are likely to have home-based deliveries as possible, through their contact with health services. Hence, HIV intervention programmes need to continue to convey information about the importance and

benefits of institutional and professional delivery for all women during pregnancy and to promote the idea that ANC, particularly antenatal HIV testing, is a standard part of delivery care.

8.3 Recommendations, policy and research implications

The analytical framework used for this analysis proposed that individual and community-level predisposing, enabling and perceived risk/need factors are associated with the use of integrated HIV and SRH services. Among the most important findings, one set of key factors which cut across all the analysis chapters is the important role of predisposing individual demographic factors relating to gender, age, and marital status, enabling factors, such as socio-economic status (i.e household wealth, education, media exposure), and, to some extent, HIV perceived risk/need factors like HIV awareness, stigma and sero-status.It is therefore important that policy and HIV intervention programmes focus mainly on predisposing and enabling factors in order to improve the use of HIV and SRH-integrated services in Zimbabwe. The existing Zimbabwe health policy and strategic documents gleaned are strongly focused on improving geographical accessibility, and primary healthcare policy on SRH services aims to have a health facility in every community. The third Zimbabwe National HIV and AIDS Strategic Plan for 2015-2018 (ZNASP III) further outlines that all of these facilities should be equipped to provide HIV prevention services. Therefore, health policy targets the quantity of HIV and SRH health facilities; however, increasing the number of HIV and SRH facilities might constrain the quality of improvement of an integrated model of healthcare delivery, unless we know which populations to target for increasing the uptake of these services. In the context of Zimbabwe, it is also questionable whether the ambitious large quantities of health facilities providing both HIV and SRH care can be supported in the next few years against the backdrop of a lack of financial and human capital, as discussed in this study and elsewhere. Although the policy and strategic plans are sound, more needs to be done in order to identify factors that influence uptake of HIV services, maternal healthcare and family planning (particularly condom use and long-term contraceptive methods).

First, we should note that HIV testing remains a critical service in the national response to the HIV epidemic in Zimbabwe. Therefore, without progress in improving HIV testing services, the policy of integration of HIV and SRH services will yield limited gains in HIV reduction. The observed gender disparities in determinants of HIV testing discussed in chapter 4 call for

a gender specific response. For males, household wealth and HIV factors such as risky sexual behaviour are critical for improvements in HIV testing. Given that household wealth was a significant predictor of HIV testing for males, the introduction of financial incentives may stimulate males to access HIV testing services by providing compensation for transport costs and opportunity costs of time associated with accessing HIV testing services. Monetary incentives could reduce barriers and stigma as they may provide a broader reason for going to testing centres. More importantly perhaps for both sexes, awareness creation on HIV counselling and testing service utilisation should focus on avoiding stigmatising and discriminatory behaviours so that clients can develop positive attitudes towards people living with HIV/AIDS. Couple-oriented HIV counselling and testing services, where men accompany their spouse to HIV screening during pregnancy, may help increase HIV testing uptake for males and reduce gender disparities. However, there are some references in the literature acknowledging that it may be difficult for pregnant women to invite their partners to accompany them to maternity care due to fear of stigma and violence (Byamugisha et al., 2011; Godif et al., 2015; Kalembo et al., 2013; Mohlala et al., 2011; Theuring et al., 2016; Yargawa, 2015). Policy, programmes or intervention to address the issue of partner involvement should be sought to involve male partners through awareness raising and to encourage index testing (household testing) for women found to be HIV-positive. An evaluation of index testing in Malawi has shown that household testing has a potential to increase male HIV testing uptake (Kalembo et al., 2013).

The role of HIV testing in maternal healthcare underscores the need to consider strategies to retain women in the HIV and maternal health intervention programme. Antenatal HIV testing was identified as a major contributor to adequate antenatal care, yet many pregnant women are lost to the health system after antenatal care HIV testing. Community mobilisation and homebased follow-up care, encouraging pregnant women lost in the system to give birth in a facility, may help increase uptake of PMTCT among women who test HIV-positive during ANC. Experience from various community projects indicates that, where there is low uptake of maternal healthcare at the facility, community health workers can take up the role of promoting antenatal care and facility delivery (Akinyi et al., 2015).

The coverage of HIV testing should be expanded and improved, intensifying the targeting of individuals and groups of people, such as older females and those who express confidentiality

concerns about the health information, and male non-condom users. Policies, programmes and interventions targeting older women and non-condom users need to develop new methods of communication that take into account some basic knowledge of how HIV is transmitted. It is also important that the Zimbabwe National Behavioural Change programme that was rolled out to young people and community-based HIV testing in rural areas should expand to include older females and non-condom users (ZNNP, 2015-2018). These types of programmes may also help reduce risky behaviour and help keep the spread of HIV in check. There is wide agreement that when people know their HIV status they make better decisions about sexual behaviour.

8.3.1 Poverty as a potential impediment to utilization of services

We have also noted that women of low social economic status or in deprived communities are less likely to use HIV and SRH care services in general, implying that there is ample room for the advancement of women and their position in the society. Policies to address the issue of poverty must be part of a greater societal approach that includes strategies to promote economic growth in all areas of the country, reducing unemployment, increasing wages, accessible and affordable quality SRH care and removal of other barriers that prevent economically disadvantaged vulnerable groups from gaining employment or engaging in economic activities. Alternatively, conditional cash transfers to women living in poor households may be implemented, although evidence in support of this strategy is limited. One study in Zimbabwe on the effect of cash transfers on women's health reported that pregnant women who participated in a cash transfer programme were twice as likely to have adequate antenatal and delivery care (Gregson et al., 2011). On the other hand, financial insecurity places many women in economically dependent relationships with men, limiting women's autonomy to make decisions, including those around HIV and SRH (Church et al., 2016). A higher level of poverty, particularly in the context of gender inequalities, undermines women's decisionmaking power, which in turn undermines their SRH, thus, in policy-making perspectives, women who are economically poor need to be supported.

A striking finding is that rural residents are not worse off than their urban counterparts of similar characteristics before other factors are considered. As urbanisation is on the rise, populations in the cities are growing faster than the infrastructure is, creating a lack of resources and healthcare in Zimbabwe. Hundreds of thousands of poor women live in slums, without

proper access to healthcare or health professionals. In the literature it is shown that slums are predominantly served by privately owned and unlicensed health facilities with limited skilled staff and equipment, whereas most of the formal health facilities are located outside the slums (Fotso et al., 2011). This means that individuals living in slums in Zimbabwe are less likely to have delivery care than women living in rural areas. Although some may argue that, as opposed to rural people, the urban poor have better access to healthcare due to the proximity of facilities, large healthcare disparities within urban areas would suggest that proximity to health facilities cannot be interpreted as equal access to services (Magadi et al., 2003). Many of the development projects surrounding improvement of HIV and SRH healthcare have been focused on rural areas in Zimbabwe because they lack resources in the way that urban areas do not (Makate and Makate, 2016). For example, there are maternity home shelters in rural hospitals designed to accommodate pregnant women who are closer to giving birth in rural areas, but not in the urban areas.

Recognizing that the poorest in urban areas often have no access to essential healthcare services due to various barriers, HIV and SRH intervention programmes should target women with low income and those living in slums in urban areas. Although the government affirms the rights of all women to maternal healthcare and officially endorse increased services for the poor, its own health system struggles with constant lack of funding to cover healthcare in slum areas (Chatiza and Mlalazi, 2009). The problem of funding may be solved by strengthening non-governmental organisations working in maternal health by providing technical support and allowing them to work without political interference. While, we report that place of residence is crucial for antenatal care in Zimbabwe, the role played by an antenatal HIV testing and optimal frequency of antenatal visits in ensuring delivery care cannot be underestimated. The government should ensure universal access to prenatal care services, irrespective of the ability to pay.

8.3.2 The need to target the youth

It was also reported in the study findings that young women are less likely to use health services than older women. Government policy should make known the importance of HIV testing, maternal healthcare and family planning in averting HIV-pregnancy related problems. There is a need to meet the health needs of adolescent girls and young women, particularly, by providing

friendly and easier HIV and SRH services. There are reports that many of the health facilities are not youth friendly and services for adolescents are poorly publicised in Zimbabwe (Buzdugan et al., 2015). In addition, there are also some reports indicating that service providers' attitudes and biases greatly undermine adolescent healthcare uptake (Sibanda et al., 2013) in Zimbabwe. Policy and intervention to address the needs of adolescents must consider making health facilities youth-friendly by allocating specific times or space to serve youth. Such programmes may get young people to use health facilities that provide peer-counselling, family planning, HIV testing, maternal health information and distribution of condoms have proved to be a success in increasing the use of SRH care services by young people in South Africa (Matthews et al., 2009, Renju et al., 2010) as well as in Botswana, Malawi and Mozambique (Underwood et al., 2014).

Additionally, counselling on sexuality and safer sex are needed to empower youth to make healthy decisions, including protecting themselves from unwanted pregnancies and STI/HIV. Policymakers may consider school and community based approach programmes. There is evidence that 'in-school' interventions provide convenient means for HIV and sexual reproductive health education and have been demonstrated to be effective in improving knowledge and attitudes, particularly concerning HIV and uptake of sexual and reproductive health services (Harrison et al., 2010). Therefore, the school curriculum should emphasise SRH topics that need to be introduced early so that young people can take control of their sexual lives by understanding the dangers of early sex and learning how to protect themselves. Another way of increasing HIV awareness among young people is to provide life-skills through anti-AIDs social clubs and sports. Although there has not been a national programme to evaluate community life-skills training programmes for young people, local area programming suggests young people from rural areas enrolled in the programme are receptive to HIV information in Zimbabwe (Chevo and Bhatasara, 2012). Thus, I argue this programme should be scaled-up to include areas where it does not exist, and should be strengthened in areas where it is weak.

8.3.3 HIV perceived risk/need factors

Among the HIV perceived risk/need factors, confidentiality concerns and knowledge of one's HIV status emerge as major predictors of HIV and sexual and reproductive service utilisation.

In Zimbabwe, confidentiality concerns and stigma are still widely perceived as socially dangerous. The inadvertent or improper disclosure of HIV status, which may result in discrimination and stigma, is a significant risk faced by people living with HIV. Pregnant women may be afraid that a provider may share the reason for their visit to the facility with their relatives (Anyait et al., 2012). Thus, policy should focus on protecting confidentiality of HIV-related information, the rights and obligations of people living with HIV with respect to disclosure of HIV status. Privacy and confidentiality by healthcare providers must be a top priority and possible legal action when unauthorised disclosure of someone's HIV must occur. This intervention may also apply to condom use and long-term contraceptive methods for women who know their HIV-positive status. This may be an important policy for addressing unmet needs in HIV and maternal health services in Zimbabwe. At the health facility, privacy and confidentiality may be increased by minor renovations of facilities, such as partitioning of rooms or adding doors, as well as changes in the practices of providers, like minimising interruptions during client visits, for instance. This notion finds support in the literature where it is argued that individuals are more likely to use health services if their right to confidentiality is guaranteed (Hardon et al., 2012; Turan et al., 2008). An enabling policy, rooted in a legal and social environment, should be a pre-requisite to successfully reduce HIV stigma and discrimination at health facilities which is central to the promotion of human rights in the context of HIV.

This thesis' findings further indicated that HIV prevention awareness programmes targeting women who know their HIV status, to encourage them to use condoms and long-term methods of contraception, are not working. There is a great need to re-strategise the campaigns on condom use, especially in rural areas and among women who are married. Similarly, policy on long-term contraceptive method use should target women in rural areas and those with higher-order parity. It is important that policymakers suggest policy that will ensure that campaigns for condom use will be intensified in rural areas, and that there be a standardisation campaign across the country to provide a homogenous type of service for all women, irrespective of their HIV status.

8.3.4 Knowledge of HIV status as trajectory to better choice of contraceptive methods

Chapter 7 ascertained the extent to which knowledge of HIV status would enable women to choose appropriate contraceptive methods to meet their reproductive goals and needs. The analysis indirectly contributes to debates on family planning and HIV-integrated programmes that are aimed at increasing condom and long-term contraceptive methods use for prevention of both HIV infection and unintended pregnancy. It is important to note that, whilst a majority of women (both infected and uninfected) are still using hormonal methods, some have shifted to the use of condoms, long-term and other methods. It turned out from the analysis that women who know they are HIV-positive (as determined by our proxy variable of receipts of HIV test results from the previous test) were more likely to use condoms than hormonal methods. The result that women who know they are HIV-positive are more likely to use condoms is a positive outcome and suggests that being HIV-positive is associated with a felt need to prevent HIV transmission to partners. Therefore, condom use among HIV-positive women reflects a higher perception of risk of HIV transmission rather than the need for limiting child-bearing. This notion is reflected in our results which suggest no differences in women's choices to use longterm methods, whether they know they are HIV-positive or know they are HIV-negative. However, the fact that there is no difference in the choice of long-term methods, with hormonal , suggests that the information about the use of long-term methods as preventive measures for unintended pregnancies has been more integrated in family planning services and is now increasingly widely accepted. As a result of these findings, policymakers and programme managers in Zimbabwe should give special attention to HIV-positive women's needs for condoms, while at the same time, continue to address the unmet requirements for long-term methods to prevent unintended pregnancies for both infected and uninfected women.

It is worrisome to note that shifting from hormonal methods to condom and long-term methods use is not taking place as a response to the wider campaigns encouraging women to use condoms and long-term methods to protect themselves against HIV infection, whether they are uninfected or already infected with HIV and wish to prevent unintended pregnancies. Policymakers and programme managers may need new strategies for scaling-up the integration of HIV prevention programmes within family planning services. However, taken together, the findings from this thesis suggest that integration of condom distribution in family planning services may already be functioning and can be scaled-up country wide. As condom distribution within family planning services continue, women are learning more about the importance of condoms in the prevention of HIV.

8.3.5 Final remarks

While, it is reported in this thesis that individual-level predisposing, enabling and HIV perceived factors are all important in the delivery of HIV and SRH services, there is also a need to develop preventative measures which are appropriate for different local contexts, targeting different age groups and sub-groups, particularly for delivery care. Although, the study highlights the key role played by both socio-economic, demographic and HIV factors on integrated services, there is a need to use mixed methods research to further understand the geographical differential distribution of individuals who use these services. There is also a need to base such studies on sound theories, at the design stage, for the purpose of providing reasonable explanations for the associations seen in these studies, especially at the communitylevel. There have been no studies conducted in Zimbabwe to test theories such as the Social Ecological Models of Health Promotion and Structural and Environment theories (Stokvol, 1996), which is important because the analysis of this study showed that some communitylevel factors are associated with HIV and SRH-integrated services, when individual-level factors are considered. Also in the mainstream HIV sector, governmental policies and nongovernmental organisations should take emerging new strategic information and knowledge, global trends, practices in human rights, and gender into consideration as they relate to HIV response.

Table 8-2: Summary implication for health policy in Zimbabwe

HIV testing for primary prevention of HIV-

- Engage in social mobilisation to reduce stigma and gender inequities
- Integrate information and education on PMTCT into all education, health care and HIV and AIDS messaging and programming.
- Increase the use of HIV counselling and testing (HCT) at all levels, with increased use of couples counselling and testing.

٠	As social and behaviour change communication cuts across a wide range of
	other HIV and AIDS interventions, coordination and harmonisation of
	prevention interventions will be critical.
•	Integration of Social and Behaviour Change Communication interventions
	in the work place, schools and community development programmes
HIV and maternal health care	
•	Facilitate access and promote regular attendance to antenatal care for
	pregnant women.
•	Intensifying education, awareness, and community mobilisation to
	generated demand for maternal health care services
•	Promote opt-out/routine HCT for pregnant women in antenatal care and for
	their male partners.
•	Provide home-based and community-based counselling on PMTCT for all
	pregnant women and their male partners,
•	Empower non-governmental organisations to implement delivery care in
	urban areas
Contraceptive choice	
•	Facilitate information and access at the community level to modern
	contraceptives to prevent unplanned pregnancy.
•	Promote the use of dual methods of contraception to include condoms to
	prevent transmission of HIV.
•	Establishment of condom distribution outlets and in particular friendly outlets
	for youth and key populations
•	Integration of condom education and distribution in other health and sexual
	reproductive services including family planning
•	Promotion of consistent and correct use of condoms among sero-discordant
	couples.
•	Promotion of male and female correct and consistent use of condom among
	key and most at risk populations.

In conclusion, the analysis shows that HIV and SRH service uptake did increase significantly between 2005/6 and 2010/11. However, there is evidence that some factors included in the models may have influenced the uptake of different HIV and SRH-integrated services differently. This means that policy makers have to be careful in terms of the structuring strategies to improve the HIV and SRH-integrated programme in Zimbabwe. For instance, a strategy that might improve delivery care may not necessarily have a significant impact on HIV testing, ANC and contraceptive choice of women who know their HIV status. While this thesis found little evidence of the impact of HIV on sexual and reproductive healthcare outcomes, the existing HIV initiatives such as HIV awareness campaign to provide HIV services to men and women in Zimbabwe need to be commended. However, there is an urgent need to build on their strengths and address their shortcomings. Plans to implement the set of recommendations listed above should be put in place and be part and parcel of the HIV and SRH-integrated programme. Given the findings and recommendations, it is hoped that this will strengthen the on-going HIV and SRH-integrated programme and help reduce the rate of MTCT of HIV in Zimbabwe.

Peer reviewed publications extracted from this thesis are listed below:

Gazimbi MM, Magadi MA (2018) Individual and Community-level Determinants of Antenatal HIV Testing in Zimbabwe. *Journal of Biosocial Science*; JBS-4363 (in press)

Gazimbi MM, Magadi MA (2017). A multilevel analysis of the determinants of HIV testing in Zimbabwe: Evidence from the demographic and health surveys. *HIV/ AIDS Res Treat Open J*. 2017; 4(1): 14-31. doi: 10.17140/HARTOJ-4-124

REFERENCES

Achia, T. N., Mageto, L.E. (2015) Individual and contextual determinants of adequate maternal health care services in Kenya. Women & Health, 55 (2), 203–226.

Adamu, H.S. (2011) Utilization of maternal health care services in Nigeria: An analysis of regional differences in the patterns and determinants of maternal health care use. Women and Health, 53; 7.

Adedimeji, A., Abboud, N., Merdekios, B., Shiferaw, M. (2012) A Qualitative Study of Barriers to Effectiveness of Interventions to Prevent Mother-to-Child Transmission of HIV in Arba Minch, Ethiopia. International Journal of Population Research, Article ID 532154, 7 pages <u>http://dx.doi.org/10.1155/2012/532154</u>

Admachak, S.E., Grey, T.E., Otterness, C., Katz, K., Janowitz, B. (2007) Introducing family planning services into antiretroviral programs in Ghana: An evaluation of a pilot intervention. Arlington: Family Health International, 1-36.

Adongo, P. B., Phillips, J. F., Aikins, M., Arhin, D. A., Schmitt, M., Nwameme, A. U., Binka, F. N. (2014) Does the design and implementation of proven innovations for delivering basic primary health care services in rural communities fit the urban setting: the case of Ghana's Community-based Health Planning and Services (CHPS). Health Research Policy and Systems, 12(1), 16.

Agadjanian. V., Hayford, S.R. (2009) PMTCT, HAART, and childbearing in Mozambique: an institutional perspective. AIDS Behav, 13(1), 103–12.

Agha S. (2012) Factors associated with HIV testing and condom use in Mozambique. Implications for programs: Reproductive Health journal, 9:20.

Ahmed, S., Creanga. A.A., Gillespie, D.G., Tsui, A.O. (2010) Economic status, education and empowerment: Implications for maternal health service utilization in developing countries. PLoS ONE, 5 (6)

Akinyemi, J.O., Awolude, O.A., Adewole, I, F, et al. (2010) Condom use among antiretroviral therapy patients in Ibadan, Nigeria. J Infect Dev Ctries, 4:495–502.

Akinyi, C., Nzanzu. J., Kaseje, D. (2015) Effectiveness of Community Health Workers in Promotion of Maternal Health Services in Butere District, Rural Western Kenya. Universal Journal of Medical Science 3(1), 11-18

Akwara.P.A. Mandise.N.J., Hinde.A. (2003) Perception of Risk of HIV/ AIDs and Sexual Behaviour in Kenya. J.Biosoc. Sci (35), 385-411

Alam, N., Hajizadeh. M., Dumont .A. Fournier. P. (2015) Inequalities in Maternal Health Care Utilization in Sub-Saharan African Countries: A Multiyear and Multi-Country Analysis. PLoS ONE, 10(4).

Albarracín, D., Gillette, J. C., Earl, A. N., Glasman, L. R., Durantini, M. R., and Ho, M. H. (2005) A test of major assumptions about behavior change: A comprehensive look at the effects of passive and active HIV-prevention interventions since the beginning of the epidemic. Psychological Bulletin, 131, 856-897

Andersen, R. M. (1995) Revisiting the behavioural model and access to medical care: does it matter. Journal of Health and Social Behaviour, 36; 1-10.

Angeles, G., Guilkey, D.K., Mroz, T.A. (2005) The impact of community-level variables on individual-level outcomes: Theoretical results and applications. Sociological Method & Research 34:76-121.

Angotti, N., Bula, A., Gaydosh, L., Kimch. E.Z, Thornton RL, Yeatman, S.E. (2009) Increasing the acceptability of HIV counselling and testing with three C's: Convenience, Confidentiality and Credibility. Soc Sci Med, 68(12):2263–70.

Anyait, A., Mukanga. D., Oundo, G.B., Nuwaha, F. (2012) Predictors for health facility delivery in Busia district of Uganda: a cross sectional study. BMC Pregnancy and Childbirth 12: 132.

Aremu, O., Lawoko, S., Dalal, K. (2011) Neighbourhood socioeconomic disadvantage, individual wealth status and patterns of delivery care utilization in Nigeria: A multilevel discrete choice analysis. Women's Health, 3, 167-174.

Arias, M.G., Nove, A., Schuldt, M., Bernis, L. (2017) Current and future availability of and need for human resources for sexual, reproductive, maternal and newborn health in 41 countries in Sub-Saharan Africa. International Journal for Equity in Health, 16:69.

Arthur, E. (2012) Wealth and antenatal care use: implications for maternal health care utilization in Ghana. Health Economics Review, 2:4.

Asefa, A., Beyene, H. (2013) Awareness and knowledge on timing of mother-to-child transmission of HIV among antenatal care attending women in Southern Ethiopia: a cross sectional study. Reproductive Health, 10:66.

Askew, I., Berer, M. (2003) The Contribution of Sexual and Reproductive Health Services to the Fight against HIV/AIDS: A Review, Reproductive Health Matters, 11(22), 51– 73

Babalola. S., Fatusi, A. (2009) Determinants of use of maternal health services in Nigerialooking beyond individual and household factors. BMC Pregnancy & Childbirth, 9: 43-103

Bayer, R. (1991) Public health policy and the AIDS epidemic. An end to HIV exceptionalism?. New England Journal of Medicine, 234:1500–1504.

Bayer, R., Edington, C. (2009) HIV Testing, Human Rights, and Global AIDS Policy: Exceptionalism and Its Discontents. J Health Polit Policy Law, 34(3), 301–323

Bayer, R., Fairchild, A.L. (2006) 'Changing the paradigm for HIV testing- The end of exceptionalism'. INEJM 335(7), 647 -649.

Bbaale, E. (2011) Factors influencing timing and frequency of antenatal care in Uganda. AMJ, 4 (8), 431-438.

Beauclair R., Petro, G., Myer, L. (2014). The association between timing of initiation of antenatal care and stillbirths: a retrospective cohort study of pregnant women in Cape Town, South Africa. BMC Pregnancy and Childbirth, 14:204.

Berkman, A. (2001) Confronting Global AIDS: Prevention and Treatment. American Journal of Public Health, 91, 1348–1349.

Bertrand, J.T., O'Reilly,K., Denison, J., Anhang, R., Sweat, M. (2006) Systematic review of the effectiveness of mass communication programs to change HIV/AIDS-related behaviours Health Education Research, 21(4), 567-597.

Bertrand, J.T., Sullivan, T.M., Knowles, E.A., Zeeshan, A.F., Shelton, J.D. (2014) Contraceptive method skew and shifts in method mix in low- and middle-income countries, Int Perspect Sex Reprod Health, 40, 144–153

Birhane, T., Tessema. G., Alene. A., Dadi, A. (2015) Knowledge of Pregnant Women on Mother-to-Child Transmission of HIV in Meket District, Northeast Ethiopia. Journal of Pregnancy.http://dx.doi.org/10.1155/ 2015/960830. 14.

Bisika, T. (2008) Cultural factors that affect sexual and reproductive health in Malawi. J FAM Plann Reprod Health Care, 34(2)

Blackie, I.R., Balekang, G.B. (2016) Desires, Intentions and Decisions of HIV-Positive Women in Attaining Motherhood in Botswana. American Journal of Social Sciences, 1, 1 Bogale, G.W., Boer, H., Seydel, E.R. (2010) Condom use among low-literate, rural females in Ethiopia: the role of vulnerability to HIV infection, condom attitude, and self-efficacy. AIDS Care, 22, 851–857.

Bond, L., Lauby, J., Batson, H. (2005) HIV Testing and the role of individual and structural level barriers and facilitators. AIDS Care, 17(2), 125-140.

Breslow, N.E., Clayton, D.G. (1993) Approximate Inference in Generalized Linear Mixed Models. Journal of the American Statistical Association, 88 (421), 9-25.

Browne, W.J. (2017) MCMC estimation in MLwiN Version 3.01.Centre for Multilevel Modelling, University of Bristol.

Buzdugan. R., McCoy. S.I., Webb.K. Mushavi, A., Mahomva, A., Padian.N.S. Cowan. F.M (2015) Facility-based delivery in the context of Zimbabwe's HIV epidemic – missed opportunities for improving engagement with care: a community-based serosurvey .BMC Pregnancy and Childbirth, 15:338.

Byamugisha, R., Tumwine, J., Ndeezi, G, Karamagi C, Tylleska, T. (2010) Attitudes to routine HIV counselling and testing, and knowledge about prevention of mother to child transmission of HIV in eastern Uganda: a cross-sectional survey among antenatal attendees. Journal of the International AIDS Society, 13:52.

Byamugisha, R., Åstrøm. A.N., Ndeezi, G., Karamagi.C.A.S. Tylleskär.T. (2011) Tumwine.J.K. (2011) Male partner antenatal attendance and HIV testing in eastern Uganda: a randomized facility-based intervention trial. Journal of the International AIDS Society, 14:43

Car, L.T., Velthoven, M., Brusamento, S., Elmoniry H, Car J, Majeed A, et al. (2012). Integrating prevention of mother-to-child HIV transmission programs to improve uptake: a systematic review. PLoS One, 7(4):e35268.

Carneiro, I., Howard, N. (2011). Introduction to Epidemiology: Open University Press, UK

Catania, J., Kegeles, S., Coates, T. (1990) Towards an understanding of risk behavior: an AIDS Risk Reduction Model (ARRM). Health Education Quarterly, 17 (1), 53-72.

Chatiza, K., Mlalazi, A. (2009) Human Settlement Needs Assessment in Zimbabwe: Critical review and proposed methodology. Government of Zimbabwe, UN-Habitat.<u>https://www.informationvine.com/index?o=603201&l=sem&qo=pagination&q=urba nisation%20in%20zimbabwe&qsrc=998&page=3</u>. Accessed May, 2017.

Chevo, T., Bhatasara, S. (2012) HIV and AIDS Programmes in Zimbabwe: Implications for the Health System. Official journal of the California Perinatal Association 30(11), 763

Chirawu, P., Langhaug, L., Mavhu, W, et al. (2008). Acceptability and challenges of implementing voluntary counselling and testing (VCT) in rural Zimbabwe: evidence from the Regai Dzive Shiri Project. AIDS Care, 22:81–88

Chirwa, Y., Witter, M., Munjoma, M., Mashange, W., Munyati, S. (2013) "The human resource implications of improving financial risk protection for mothers and newborns in Zimbabwe," BMC Health Services Research, 13; 197.

Chitalu et al. (2013) Utilisation of Focused ANC in Zambia: Examining individual and community level factors using a multilevel analysis, Health policy and Planning, 1-10.

Chiu, J., Grobbelaar, J., et al. (2008) HIV-related stigma and social capital in South Africa. AIDS Education and Prevention, 20(6), 519–530.

Church, K., Wringe, A., Fakudze, P., Kikuvi, J., Nhlabatsi, Z., Masuku, R., Integra Initiative., Mayhew, S.H.(2014) Reliance on condoms for contraceptive protection among HIV care and treatment clients: a mixed methods study on contraceptive choice and motivation within a generalised epidemic. Sex Transm Infect, 0:1–7.

Church, K., Wringe, A., Fakudze, P., Kikuvi, J., Simelane, D., Kikuvi, J., Mayhew S, et al.(2013) Are integrated HIV services less stigmatising than standalone models of care? A comparative case study from Swaziland. Journal of the International AIDS Society, 16:17981

Claudia, H. (2013) The Epidemiology of Maternal Mortality in Southern Tanzania. Doctoral thesis, London School of Hygiene and Tropical Medicine.<u>http://researchonline.lshtm.ac.uk/1012993/</u>. Assessed June 2017.

Cleland, J., Ali, M.M. (2006) Sexual abstinence, contraception, and condom use by young African women: a secondary analysis of survey data. Lanct 368, 1788-1793.

Curtis, K.M., Nanda, K., Kapp, N. (2009) Safety of hormonal and intrauterine methods of contraception for women with HIV/AIDS: a systematic review. AIDS, 23:55–S67

Daftary, A., Padayatchi, N., Padilla, M. (2007) .HIV testing and disclosure. A qualitative analysis of TB patients in South Africa. AIDS Care, 19(4), 572–7.

Dahl, B. (2012) Beyond the Blame Paradigm: Rethinking Witchcraft Gossip and stigma around HIV-Positive Children in South Eastern Botswana: African Historical Journal Review, 44 (1), 53-79.

Dahlgren. G., Whitehead, M. (1991) Policies and Strategies to Promote Social Equity in Health. Stockholm, Sweden: Institute for Futures Studies

De Cock, K.M., Johnson, A.M (1998) From Exceptionalism to Normalization: A Reappraisal of Attitudes and Practice around HIV Testing. British Medical Journal, 316:290–293.

Delprato, M. (2016) Preschool attendance: a multilevel analysis of individual and community factors in 21 low and middle income countries.intl.J. Quantitative Research in Education; 3(1:2).

Demisse .A. Abera, A et al. (2009) Determinats of VCT among ANC clinic attendees at Ail ChoraHosp, Dire Dawa East Ethiopia, Ethio J Health Dev, 23 (2).

Denison, J.A., McCauley, A.P., Dunnett-Dagg, W.A., Lungu, N., Sweat, M.D. (2008) The HIV testing experiences of adolescents in Ndola, Zambia: Do families and friends matter? AIDS Care, 20(1),101–5.

Dias JG, de Oliveira IT (2015) Multilevel Effects of Wealth on Women's Contraceptive Use in Mozambique. PLoS ONE 10(3): e0121758. doi:10.1371/journal.pone.0121758.

Diez Roux, A.V. (2002) Invited commentary: Places, people, and health.American Journal of Epidemiology, 155(6), 516-519.

Diez-Roux, A.V. (1998) Bringing context back into epidemiology: Variables and fallacies in multilevel analysis. American Journal of Public Health, 88(2), 216-222.

Do, M., and N. Kurimoto (2012) "Women's Empowerment and Choice of Contraceptive Methods in Selected African Countries." International Perspectives on Sexual and Reproductive Health 38 (1), 23–33.

Druce .N. Dickinson, D., Attawell, K., Campbell, A., Standing, H. (2006) Strengthening linkages for sexual and reproductive health, HIV and AIDS: progress, barriers and opportunities for scaling up.PLOS ONE DOI: 10.1371/journal.pone.0035268.

Duflo, E., Dupas, P., Kremer, M. (2015) Education, HIV, and Early Fertility: Experimental Evidence from Kenya. American Economic Review, 105(9), 2757–2797.

Dugas, M., Bédard, E., Georges, B., Adolphe C, Fernand .A, Dubé, E., Alary, M(2015) Outreach Strategies for the Promotion of HIV Testing and Care: Closing the Gap Between Health Services and Female Sex Workers in Benin. AIDS Journal of Acquired Immune Deficiency Syndromes 68: 198-205.

Duri, K., Stray-Pedersen, B., Muller, F. (2013) HIV/AIDS The Zimbabwean Situation and Trends American Journal of Clinical Medicine Research, 1(1), 15-22.

Eijk, A., Hanneke, M., Odhiambo, B. (2006). Use of antenatal services and delivery care among women in rural western Kenya: a community based surveyReproductive Health 3:2

Elul, B., Delvaux, T., Munyana, E., Lahuerta, M., Horowitz, D., Ndagije, F, et al. (2009) Pregnancy desires, and contraceptive knowledge and use among prevention of mother-to-child transmission clients in Rwanda. AIDS. 23(1), 19–26.

Farahani, M., Vable, A., Lebelonyane, R., Seipone, K., Anderson, M., Avalos, A et al. (2014) Outcomes of the Botswana national HIV/AIDS treatment programme from 2002 to 2010: a longitudinal analysis. Lancet Glob Health 2(1), 44–50.

Farquhar, C., Kiarie, J.N., Richardson, B.A., Kabura, M.N., John, F.N., Nduati, R.W., Mbori-Ngacha, D., John-Stewart, G.C. (2004) Antenatal couple counselling increases uptake of interventions to prevent HIV-1 transmission. J Acquir Immune Defic Syndr. 37(5), 1620–6.

Faye, A., Niane, M. (2011) Home birth in women who have given birth at least once in a health facility: Contributory factors in a developing country. Acta ObstetGynecolScand 90 (11), 1239-1243

Femnet (2016) Sexual and reproductive health and rights (SRHR) and the 2030 agenda for sustainable development (sdgs). http://femnet.co/wp-content/uploads/2016/06/Briefing-on-SRHR-and-SDGs-FEMNET.pdf. Accessed June 2017.

Fetene, N.W., Feleke, A.D. (2010) Missed opportunities for earlier HIV testing and diagnosis at the health facilities of Dessie town, North East Ethiopia. BMC Public Health 10(1), 362.

Fotso, J.C., Ezeh, A., Madise, N., Ziraba, A., Ogollah R. (2009) What does access to maternal care mean among the urban poor? Factors associated with use of appropriate maternal health services in the slum settlements of Nairobi, Kenya. Matern Child Health J 13 (1), 130-137.

Fotso, J.C., Ezeh, A., Oronje, R. (2008) Provision and use of maternal health services among urban poor women in Kenya: what do we know and what can we do? J Urban Health 85: 428–442.

Gabrysch, S., Cousens, S., Cox, J., Campbell, O. (2011) Distance and quality of care strongly influence choice of delivery place in rural Zambia: A study linking national data in a

GARPR (2016) GLOBAL Aids response progress reporting 2016.Construction of core indicators for monitoring the 2011 United Nations Political Declaration on HIV and AIDS.<u>https://aidsreportingtool.unaids.org/static/docs/GARPR_Guidelines_2016_EN.pdf.</u> <u>Assessed June 2017</u>.

Gausset, Q., Mogensen, H. O., Yameogo, W. M. E., Berthe, A., & Konte, B. (2012). The ambivalence of stigma and the double-edged sword of HIV/AIDS interventions in Burkina Faso. Social Science & Medicine, 74, 1037-1044.

Godif, M., Assefa, H., Alemayehu, M., etal. (2015) Factors Associated with HIV Counselling and Testing among Males and Females in Ethiopia: Evidence from Ethiopian Demographic and Health Survey Data: AIDS& Clinical Research, 6:3

Golden, S.D., Anne L. Earp, J.A.L. (2012) Social Ecological Approaches to Individuals and Their Contexts: Twenty Years of Health Education & Behavior Health Promotion Interventions. Health Education & Behavior, 39(3), 364–372

Goldstein, H. and Rasbash, J. (1996).Improved approximations for multilevel models with binary responses.J. Roy. Statist.Soc. A 159, 505-513.

Goldstein.H (2003) Multilevel statistical models.3rd ed. London, Arnold

Goli, S., Doshi, R., Perianayagam, A. (2013) Pathways of economic inequalities in maternal and child health in urban India: a decomposition analysis. PLoS One; 8: e58573.

Gombachika, B.C., Fjeld, H. (2014). In search of a face: Childbearing decisions among couples living with HIV from rural southern Malawi. Afr J AIDS Res. 13(1), 21–9

Gourlay, A., Wringe, A., Todd, J., Cawley, C., Machemba, R., Clark, R., Zaba, B (2016) Uptake of services for prevention of mother-to-child transmission of HIV in a community cohort in rural Tanzania from 2005 to 2012. BMC Health Services Research, 16:4

Granich, R., Gupta S, Hersh B, et al. (2015). Trends in AIDS deaths, new infections and ART coverage in the top 30 countries with the highestAIDS mortality burden; 1990–2013.*PLoS* One, 10:e0131353.

Gregson, S., Mushati, P., Gruish, H., Nhamo, M., Skovdal, M. (2011). Social capital and reduced female vulnerability to HIV infection in rural Zimbabwe. Population and Development Review, 37(2), 333–359.

Gross, S., Alba, S., Glass, T.R., Obrist, B. (2012). Timing of antenatal care for adolescent and adult pregnant women in south-eastern Tanzania BMC Pregnancy and Childbirth 12:16

Gupta, S., Yamada, G., Mpembeni, R., Frumence, G., Callaghan-Koru, J.A, et al. (2014) Factors Associated with Four or More Antenatal Care Visits and Its Decline among Pregnant Women in Tanzania between 1999 and 2010. PLoS ONE 9(7)

Gustav, A.S.P., Odberg Pettersson, K., Sandberg, J., Kabakyenga, J., & Agardh, A. (2014). Associations between mass media exposure and birth preparedness among women in southwestern Uganda: a community-based survey. Global Health Action, 7(9), 1-9.

Gwauya, A.G. (2013) Urban poverty in Zimbabwe: Unmasking severe deprivations in cities, UNICEF Zimbabwe.<u>http://www.urb.im/blog/unicef/140224</u>. Accessed June, 2017.

Habte, D and Namasasu, J. (2015) Family planning use among women living with HIV: Knowing HIV positive status helps - results from a national survey, Reproductive Health 12(41

Haddad, L.B. (2015) Contraception and prevention of HIV transmission: a potential conflict of public health principles. J Fam Plann Reprod Health Care, 41:20–23.

Hardon, A., Vernooij, E., Bongololo-Mbera, G., Cherutich, P., Desclaux A, Kyaddondo D, Ky-Zerbo, O., Neuman M, Wanyenze R, Obermeyer, C .(2012) Women's views on consent, counseling and confidentiality in PMTCT: a mixed-methods study in four African countries. BMC Public Health, 12:26.

Harrison, A., Newe, M., Imrie, J., Hoddinott, G. (2010) HIV prevention for South African youth: which interventions work? A systematic review of current evidence. BMC Public Health, 10:102

Hartwig, K. A., Kissioki, S., & Hartwig, C. D. (2006). Church leaders confronting HIV/AIDS and stigma: A case study from Tanzania. Journal of Community & Applied Social Psychology, 16, 492-497.

Hazemba, A.N., Ncama, B.P., Sithole, S.L. (2016) Promotion of exclusive breastfeeding among HIV-positive mothers: an exploratory qualitative study. International Breastfeeding Journal.

Heemelaar, S., Habets, N., Makukuka, Z., Van Den Akker, T. (2014) Repeat HIV testing during pregnancy and delivery: Missed opportunities in a rural district hospital in Zambia. Tropical Medicine & International Health, 20(3)

Hendriksen, E., Hlubinka, D., Chariyalertsak, S., Chingono, A et al. (2009) Keep talking about it: HV/AIDS-related community and prior HIV testing in Tanzania, Zimbabwe, South Africa and Thailand. Aids and behaviour, 13:1213-1221

Heys, J., Kipp, W., Jhangri, G.S, et al. (2009) Fertility desires and infection with the HIV: results from a survey in rural Uganda. AIDS, 23:S37–45.

Heywood, M.J. (2005) The Routine Offer of HIV Counseling and Testing: A Human Right. Health and Human Rights, 8:13–19

Hoffman I, Martinson F, Powers K, Chilongozi D, Msiska E, Kachipapa E, et al. (2008) The year-long effect of HIV-positive test results on pregnancy intentions contraceptive use, and pregnancy incidence among Malawian women. J Acquir Immune Defic Syndr. 47(4), 477–83.

Homsy, J., Bunnell, R., Moore, D., King, R., Malamba, S., Nakityo, R. et al. (2009) Reproductive intentions and outcomes among women on antiretroviral therapy in rural Uganda: a prospective cohort study, PLoS One. 4:e4149.

Hope, R., Kendall, T., Langer, A., Barnighausen, T. (2014) Health systems integration of sexual and reproductive health and HIV services in Sub-Saharan Africa: a scoping study.J Acquir Immune Defic Syndr. 67(4), 259–70

Houweling, T.A., Ronsmans, C., Campbell, O. (2007) Huge poor rich inequalities in maternity care: an international comparative study of maternity and child care in developing countries. Bull World Health Organ, 85:745-754.

Ipas (2015) The youth-friendly corner: Where all young people find safe, respectful health care. <u>http://www.ipas.org/en/News/2015/August/The-youth-friendly-corner--Where-all-young-people-find-safe--respectful-health-care.aspx</u>. Assessed May 2017.

Iriyana, S., Nakahara, S., Jimba, M et al. (2007) AIDS Health beliefs and intention for sexual abstinence among male adolescent students in Kathmandu, Nepal: A test of perceived severity and susceptibility, Science Direct, 121(1), 64-72

Izugbara, C.O., Undie, C.C., Mudege, N.N., Ezeh, A.C. (2009) Male youth and voluntary counselling and HIV-testing: The case of Malawi and Uganda. Sex Educ 9(3), 243–59.

Jacobi, C. A., Atanga, P. N. J. I., Bin, L. K., Mbome, V. N., Akam, W., Bogner, J. R., Kropf, S., & Malfertheiner, P. (2013). HIV/AIDS-related stigma felt by people living with HIV from Buea, Cameroon. AIDS Care: Psychological and Social-medical Aspects of AIDS/HIV, 25, 173-180.

Jayleen, K. L.G., Asaolu, I.O., Center E.K., Gibson S.C., Wightman, P., Echezona, E., Ehiri, J.E. (2016) Antenatal care and uptake of HIV testing among pregnant women in sub-Saharan Africa: a cross-sectional study: Journal of the International AIDS Society, 19:20605

Jean, K., Anglaret, X., Moh, R et al. (2012) Barriers to HIV Testing in Cote d'Ivoire: The Role of Individual characteristics and Testing Modalities, PLoS medicine .7(1)

Jesmin,S.S., Chaudhuri,S., Abdullah. S. (2013). Educating women for HIV prevention: does exposure to mass media make them more knowledgeable? Health Care Women Int 34: 30331.

Johnson, F.A., Padmadas, S.S., Brown, J.J. (2009): On the spatial inequalities of institutional versus home births in Ghana: a multilevel analysis. J Community Health, 34: 64-72.

Johnson, K., Akwara, P., Rutstein, S., Bernstein, S. (2009) Fertility preferences and the need for contraception among women living with HIV: the basis for a joint action agenda. AIDS. 23 (1), S7–17.

Jones, K. (1991) Specifying and Estimating Multilevel Models for Geographical Research.Transactions of the Institute of British Geographers, 16(2), 148-159.

Jürgensen. M., Tuba .M., Fylkesnes. K., Blystad. A. (2012) The burden of knowing: balancing benefits and barriers in HIV testing decisions. A qualitative study from Zambia. BMC Health Serv Res, 12:2.

Kabagenyi, A., Reid, A., Ntozi, J., Atuyambe, L. (2016). Socio-cultural inhibitors to use of modern contraceptive techniques in rural Uganda: a qualitative study. The Pan African Medical Journal, 25:78.

Kaida.A. Andia.I., Marissa, M et al. (2006). The potential impact of antiretroviral therapy on fertility in Sub-Saharan Africa, Current HIV/AIDS Reports 3(4), 187-194

Kakaire, O., Osinde, M., Kaye, D. (2010) Factors that predict fertility desires for people living with HIV infection at a support and treatment centre in Kable, Uganda. Reproductive Health 7:27

Kalembo F.W., Zgambo. M., Mulaga, A.N., Yukai, D., Ahmed, N.I. (2013): Association between male partner involvement and the uptake of prevention of mother-to-child transmission of HIV (PMTCT) interventions in Mwanza District, Malawi: a retrospective cohort study. PLoS ONE 8 (6), 1–7.

Kalichman, S.C., Simbayi, L.C. (2003) HIV testing attitudes, AIDS stigma, and voluntary counselling and testing in a black township in Cape Town, South Africa. Sex Tranasm Infect, 79(6), 442–7.

Kasenga, F., Hurtig, A.K., Emmelin M.H. (2007) Home deliveries: implications for adherence to nevirapine in a PMTCT programme in rural Malawi. AIDS Care 19:646-652.

Kawale, P., Mindry, D., Phoya, A., Jansen, P., Hoffman, R.A. (2012) Provider attitudes about childbearing and knowledge of safer conception at two HIV clinics in Malawi . Reproductive Health 12:17.

Kayeyi, N, Sanday, I.F., Fykesnes, K. (2014) Effects of neighbourhood-level educational attainment on HIV prevalence among young women in Zambia. BMC Public Health; 9:310

Kayiki, S.P, Forste, R. (2011) HIV/AIDS related knowledge and perceived risk associated with condom use among adolescents in Uganda. African Journal of Reproductive Health, 15: 57–63.

Kennedy, C.E., Spaulding, A.B., Brickley.D.B., Almers, L., Mirjahangir, J. et al. (2010) Linking sexual and reproductive health and HIV interventions: a systematic review. J Int AIDS Soc 13: 26. Keogh, S.C., Urassa, M., Kumogola, Y., Mngara, J., Zaba,B. (2009) Reproductive behaviour and HIV status of antenatal clients in northern Tanzania: opportunities for family planning and preventing \zx mother-to-child transmission integration. AIDS.23:S27–35.

Kevany, S., Murimo, O., Singh, B et al. (2012) Socio- economic Status and Health care Utilisation in Rural Zimbabwe:Findings from Project Accept (HPTN 043). Journal of public Health in Africa, 3: 13.

Kevin, A., Mutugi, M., Wanzala. P. (2014) Knowledge and attitude of women on the available PMTCT services at the antenatal clinic of the Coast Province General Hospital.Pan African Medical Journal. 18:4

Killam, W.P, Tambatamba, B.C., Chintu, N., Rouse, D., Stringer, E., Bweupe, M, et al. (2010) antiretroviral therapy in antenatal care to increase treatment initiation in HIV-infected pregnant women: a stepped-wedge evaluation. AIDS. 24(1), 85–91.

Kurewa, N. E., Munjoma. M. W., Hussain. A., Chirenje .Z.M., Sebit .M.B., Stray-Perdesen.B.(2011) Contraceptive uptake, reproductive choices and sexual behavior of HIV positive compared to HIV negative women participating in the prevention of mother to child transmission of HIV program in Zimbabwe. Journal of AIDS and HIV Research, 3(3), 51-56

Laher, F., Todd, C., Stibich, M et al. (2009) A qualitative assessment of decisions affecting contraceptive utilization and fertility intentions among HIV-Positive women in Soweto, South Africa. AIDS Behav 13:47–54.

Laird, N. M. (1978) "Empirical Bayes Methods for Two-Way Contingency Tables," Biometrika, 65, 581-590.

Lepine, A., Prestholt, F.T., Vickerman, P. (2014) Determinants of HIV testing among Nigerian couples: a multilevel modelling approach: Health Policy and Planning, 1-14.

Leyland, A.H., Groenewegan, P.P. (2003) Multilevel modelling and public health policy. Scandinavian Journal of Public Health, 31(4), 267-274.

Liddel, C., Barret, L., Bydawell, M (2004) Indigenous representations of illness and AIDs in Sub-Saharan Africa. Social Science & Medicine, 60 (4), 691-700.

Loewenson, R., Kadungure, A., Shamu, S. (2012) Assessment of facilitators and barriers to maternal and child health services in four rural and urban districts of Zimbabwe. Report.Ccore.Website. <u>www.tarsc.org</u>. Assessed December, 2016

Lugalla, J., Yoder, S., Sigala, H., Madihi, C. (2012) Social context of disclosing HIV test results in Tanzania. Culture, Health & Sexuality: An International Journal for Research, Intervention and Care, 14(1), S53-S66.

Luginaah, I.N., Yiridoe, E.K., Taabazuing, M.M. (2005) From mandatory to voluntary testing: Balancing human rights, religious and cultural values, and HIV/ AIDS prevention in Ghana. Soc Sci Med, 61(8), 1689–700.

MacPhail, C., Pettifor, A., Moyo, W et al. (2009) Factors associated with HIV testing among sexually active South African youth aged 15-24 years. AIDS Care.21:456–467

Magadi, M.A, Agwanda, A.O., Obare, F.O. (2007) A comparative analysis of the use of maternal health services between teenagers and older mothers in sub-Saharan Africa: Evidence from Demographic and Health Surveys (DHS). SocSci Med, 64 (6), 1311-1325.

Magadi .M and Desta. M (2011) A multilevel analysis of determinants and cross-national variations of HIV sero positivity in sub-Saharan Africa: Evidence from the DHS; Health and Place 17:1067-1083

Magadi, A, M., Madise, N. J., Rodrigues R. N. (2001) "Frequency and timing of antenatal care in Kenya: explaining the variations between women of different communities," Social Science and Medicine, 51 (4), 551–561.
Magadi, M. A., Magadi, W. A. (2017). "HIV/AIDS and contraceptive use: factors associated with contraceptive use among sexually active HIV-positive women in Kenya". *Contraception* 95:312–321.

Magadi, M.A. (2016b) "HIV/AIDS and contraceptive method choice: Demographic and socioeconomic correlates of contraceptive method choice among HIV-positive women practising family planning in Kenya". *HIV/AIDS Res Treat Open J*. 3(1), 1-12. doi: 10.17140/HARTOJ-3-119.

Makate, M., Makate, C. (2017) The Evolution of Socioeconomic-Related Inequalities in Maternal Healthcare Utilization: Evidence from Zimbabwe, 1994-2011.Global Health Research and Policy 2:1

Maleche, A. (2015) SDG SERIES: Are SDGs the Vehicle to End AIDS by 2030? Only if Driven by Human Rights. Health and Human rights Journal.https://www.hhrjournal.org/2015/09/sdg-series-are-sdgs-the-vehicle-to-end-aidsby-2030-only-if-driven-by-human-rights/. Access June 2017

Mantell, J.E., Exner, T.M., Cooper, D., Bai, D., Leu, C-S., Hoffman, S et al. (2014) Pregnancy intent among a sample of recently diagnosed HIV-positive women and men practicing unprotected sex in Cape Town, South Africa. J Acquir Immune Defic Syndr, 67 (4)202.

Manzi, A., Munyaneza, F., Mujawase, F., Banamwana, L., Sayinzoga, F., Thomson, D.R, Ntaganira, J., Hedt-Gauthier, B et al. (2014). Assessing predictors of delayed antenatal care visits in Rwanda: a secondary analysis of Rwanda demographic and health survey 2010. BMC Pregnancy and Childbirth, 14:290

Martin-Herz, S.P., Shetty, A.K., Bassett,M.T., Ley C., Mhazo, M., Moyo, S., Herz,A.M., Katzenstein, D. (2006) Perceived risks and benefits of HIV test and predictors of acceptance of HIV counselling and testing among pregnant women in Zimbabwe International Journal of STDS and AIDS 17(12),835-41.

Maruva .M., Gwavuya .S., Marume, M., Musarandega.R., Madzingira .N .(2014) Knowledge of HIV status at ANC and utilisation of Maternal health services in the 2010-11 Zimbabwe demographic and Health services, ICF International, DHS working papers No 107

Mashuba, S., Hemalata, K. (2015) HIV status disclosure among HIV positive women in Zimbabwe. The International Asian Research Journal, 03 (01), 44.50.

Mason, L., Dellicour, S., Kuile, F.T., Ouma, P., Howard, P.P., Were, F., Laserson, K., Desai, M. (2015) Barriers and facilitators to antenatal and delivery care in western Kenya: a qualitative study. BMC Pregnancy and Childbirth, 15:26

Matthews, C., Guttmacher, S., Fisher, A., Mtshizana, Y., Nelson, T., McCarthy, J., Daries, V (2009) The quality of HIV testing services for adolescents in Cape Town, South Africa: do adolescent-friendly services make a difference?J Adolesc Health. 44, 188-190.

Matthews, Z., Brookes, M., Stones, R.W., Hossain, M.B. (2005) Village in the city: autonomy and maternal health-seeking among slum populations of Mumbai. In: Moore S, ed. A focus on gender: collected papers on gender. Calverton (Maryland): ORC Macro.

Matthews, Z., Channon, A., Neal, S., Osrin, D., Madise, N et al. (2010) Examining the "Urban Advantage" in Maternal Health Care in Developing Countries. PLoS Med 7:9.

Matthews, L. T., T. Crankshaw, J. Giddy, A. Kaida, J. A. Smit, N. C. Ware and D.
R. Bangsberg. (2013) "Reproductive Decision-making and Periconception Practices among HIV-positive Men and Women Attending HIV Services in Durban, South Africa. *AIDS Behavior* 17(2), 461-470.

Mayondi, G., Wirth, K., Morroni, C., Moyo, M., Makhema.J et al. (2016) Unintended pregnancy, contraceptive use, and childbearing desires among HIV infected and HIV-uninfected women in Botswana: across-sectional study. BMC Public Health, 16:44

McCoy, S.I., Buzdugan, R., Ralph, L.J., Mushavi, A., Mahomva, A., Hakobyan, A., Watadzaushe, C., Dirawo, F., Cowan, F.M., Padian, N.S. (2014) Unmet need for family

planning, contraceptive failure, and unintended pregnancy among HIV-infected and HIV-Uninfected women in Zimbabwe. Journal of Southen African studies 27, 225-243.

Meiberg, A.E., Bos, A.E.R., Onya, H.E., Schaalma, H.P. (2008) Fear of stigmatization as barrier to voluntary HIV counselling and testing in South Africa. East Afri J Public Health, 5(2), 49–54.

Mitchel, S., Cockcroft, A., Lamothe, G, et al. (2010) Equity in HIV testing: evidence from a cross-sectional study in ten Southern African Countries. BMC International Health and Human Rights. 10:23.

Mlay, R., Lugina, H., Becker, S. (2008) Couple counselling and testing for HIV at antenatal clinics: Views from men, women and counsellors. AIDS Care, 20(3), 356–60.

Mohlala, B.K., Boily, M.C., Gregson, S. (2011) The forgotten half of the equation: randomized controlled trial of a male invitation to attend couple voluntary counselling and testing. AIDS. 25(12), 1535–41.

Moindi, R.O., Mbakaya, C., Nyambati, V., Habtu, M. (2015) Qualitative study of factors associated with home deliveries and practices in Kilifi County-Kenya. International Journal of Health Sciences and Research. 5(12), 369–376.

Moodley, D., Esterhuizen, T.M., Pather, T., Chetty, V., Ngaleka, L. (2009) High HIV incidence during pregnancy: compelling reason for repeat HIV testing.AIDS 23: 1255–1259.

Mopembeni ,R.N., Killewo, J.Z., Leshabari, M.T., Massawe ,S.N., Jahn, A., Mushi, D., Mwakipa, H.(2007) Use pattern of maternal health services and determinants of skilled care during delivery in Southern Tanzania: Implications for achievement of MDG-5 targets. BMC Pregnancy Childbirth 7: 29-10.

Morrill, A.C., Noland, C. (2006). Interpersonal Issues Surrounding HIV Counseling and Testing, and the Phenomenon of "Testing by Proxy". Journal of Health Communication, International Perspectives, 11(6) 2.

Morrison, C.S., Chen, P.L., Nankya, I., Rinaldi, A., Van Der Pol, B et al. (2011) Hormonal contraceptive use and HIV disease progression among women in Uganda and Zimbabwe. J Acquir Immune Defic Syndr 57:157–164.

Moyer, C.A., Mustafa, A. (2013) Drivers and deterrents of facility delivery in sub-Saharan Africa: a systematic review. Reprod Health. 10: 40.

Mrisho, M., Schellenberg, J.A., Mushi, A.K., Obrist, B., Mshinda, H., Tanner, M., Schellenberg, D. (2007): Factors affecting home delivery in rural Tanzania. Trop Med Int Health 12(7), 862–872.

Mtumbuka, E., Maluwa, A., Pindani, A., Bultemeier, K. (2012)Factors influencing pregnant women to undergo HIV testing and counselling during antenatal clinic in Malawi.Journal of Research in Nursing and Midwifery (JRNM), 1(4), 56-61.

Muchabaiwa, L., Muzambani, D., Chigusiwa. L., Bindu, S., Mudavanhu V. (2012) Determinants of maternal healthcare utilisation in Zimbabwe; Intl journal of Economic science and applied Research, 5 (2), 145:162.

Mucheto, P., Chadambuka, A., Shambira, G et al. (2011) Determinants of nondisclosure of HIV status among women attending the prevention of mother to child transmission programme, Makonde district, Zimbabwe.PanAfr Med J. 8: 51.

Muchini, B., Benedikt, C., Gregson. S., Gomo, E., Mate, R., Mugurungi, O, et al. (2010) Local perceptions of the forms, timing and causes of behaviour change in response to the AIDS epidemic in Zimbabwe AIDS and Behaviour.AIDS Behav, 15:487.

Mugurungi, O., Mahomva, A., Chikhata, F., Gregson, S.(2008) Estimating vertically acquired HIV infections and the impact of the prevention of mother-to-child transmission program in Zimbabwe: insights from decision analysis models. J Acquir Immune Defic Syndr, 48:72-81.

Muheriwa, S., Chimwaza, A., Maluwa, A., Kamanga, M. (2012. Prevention of Mother to Child Transmission of HIV services in Malawi. Afri. J. Midwifery and Women's Health 6 (3),139-144.

Mumah, J.N., Ziraba, A.K., Sidze, E.M (2014) Effect of HIV status on fertility intention and contraceptive use among women in nine Sub-Saharan African Coutries: Evidence from Demographic health surveys; Glob Health action 7

Munjoma, M.W., Mhlanga, F.G., Mapingure, M.P., Kurewa, E, N., Mashavave, G et al. (2010) The incidence of HIV among women recruited during late pregnancy and followed up for six years after childbirth in Zimbabwe. BMC Public Health 10: 668.

Murire, M.N., Campbell, C., Gregson, C. (2014). Community group membership and stigmatising attitudes towards people living with HIV in Eastern Zimbabwe.Journal of Community Health. 39(1), 72-8

Musheke, M., Ntalasha, H., Gari , S et al. (2013) . A systematic review of qualitative findings on factors enabling and deterring uptake of HIV testing in Sub-Saharan Africa BMC Public Health, 13:220.

Mwangi, M., Kellogg, T.A., Sufia, S et al. (2014). Factors Associated with Uptake of HIV Testing Results in a Nationally Representative Population-based AIDS Indicator Survey. The Open AIDS journal, 8: 7-16.

Naik, R., Doherty, T., Jackson, D et al. (2015) Linkage to care following a home-based HIV counselling and testing intervention in rural South Africa. J Int AIDS Soc 18:19843.

Nakanjako, D., Kamya, M., Kyabayinze, D et al. (2006). Acceptance of routine testing for HIV among adult patients at the medical emergency unit at a national referral hospital in Kampala. AIDS Behav, 11:753–758

Nakayiwa, S., Abang, B., Packel, L., Lifshay, J., Purcell, D. et al. (2006).Desire for children and pregnancy risk behavior among HIV-infected men and women in Uganda.AIDS and Behavior 10: 95-104.

Nanda K, Morrison CS, Kwok C, Byamugisha J, Jones L, Sriplienchan S, et al. (2011) Discontinuation of oral contraceptives and depot medroxyprogesterone acetate among women with and without HIV in Uganda, Zimbabwe and Thailand. Contraception, 83(6), 542–8.

Nattabi, B., Li, J., Thompson, S.C. (2009) A systematic review of factors influencing fertility desires and intentions among people living with HIV/AIDS: implications for policy and service delivery. AIDS Behav, 13:949–968.

Ngome, E, Odimegwu, C. (2014) The social context of adolescent women's use of modern contraceptives in Zimbabwe: a multilevel analysis. Reproductive Health Journal, 11:64

Nkonki, L.L., Doherty, T.M., Hill Z, et al. (2007) Missed opportunities for participation in prevention of mother to child transmission programmes: simplicity of nevirapine does not necessarily lead to optimal uptake, a qualitative study. AIDS Res Ther, 4: 27.

Nyblade, L., Menken, J., Wawer, M.J et al. (2001) Population-based HIV testing and counseling in rural Uganda: participation and risk characteristics. J Acquir ImmuneDefic Syndr 28 (5):463–470.

O'Meara, W. P., Platt, A., Naanyu, V., Ndege, S. (2013) Spatial autocorrelation in uptake of antenatal care and relationship to individual, household and village-level factors: results from a community-based survey of pregnant women in six districts in western Kenya.International Journal of Health Geographics, 12:55

Obermeyer, A.M., Neuman, M., Hardon, A et al. (2013) .Socio-economic determinants of HIV testing and counselling: a comparative study in four African countries Tropical Medicine and International Health. 18 (9), 1110-1118.

Obure, A.D., Jacobs, R., Guinness, L., Mayhew, S., Vassall, A. (2016) Does integration of HIV and sexual and reproductive health services improve technical efficiency in Kenya and Swaziland? An application of a two-stage semi parametric approach incorporating quality measures. Social Science & Medicine 151:147e156

Obure, C.D., Sweeney, S., Darsamo, V., Michaels-Igbokwe, C., Guinness, L., Terris-Prestholt, F., Muketo, E., Nhlabatsi, Z., Warren, C.E., Mayhew, S. (2015) The Costs of Delivering Integrated HIV and Sexual Reproductive Health Services in Limited Resource Settings. PLoS ONE. 10(5)

Ochako, R., Gichuh, W. (2016) Pregnancy wantedness, frequency and timing of antenatal care visit among women of childbearing age in Kenya. Reproductive Health, 13:51

Olalekan, A., Uthman, A. (2010). Does It Really Matter Where You Live? A Multilevel Analysis of Social Disorganization and Risky Sexual Behaviours in Sub-Saharan Africa. Demographic and Health Research, 78

OPHI. (2013) Country Briefing: Zimbabwe, Multidimensional Poverty Index (MPI). http://www.ophi.org.uk/wp-content/uploads/Zimbabwe-2013.pdf. Accessed June 2017.

Orne-Gliemann. J., Mukotekwa. T., Perez.F. Miller.A. Sakarovitch., Dabis. F. (2006) Improved knowledge and practices among end-users of mother-to-child transmission of HIV prevention services in rural Zimbabwe. Trop Med Int Health.11 (3), 341-9.

Oseni, Z., Seedat, F., & Kandala, N. (2018) HIV Epidemic Heterogeneity in Zimbabwe: Evidence from Successive Demographic and Health Surveys. Journal of Biosocial Science, 1-13. <u>https://doi.org/10.1017/S0021932017000657</u>

Oshi, S.N., Ezugwu, F.O., Oshi, D.C., Dimkpa,U., Korie, F.C., Okperi ,B.O. (2007) Does self perception of risk of HIV infection make the youth to reduce risky behaviour and seek voluntary counselling and testing services? A case study of Nigerian Youth. J SocSci, 14 (2), 195–203.

Peltzer, K., Matseke, G., Mzolo, M et al. (2009). Determinants of knowledge of HIV status in South Africa: results from a population-based HIV survey, BMC Public Health, 9:174.

Perez, F., Aung, K.D., Ndoro, T., Engelsmann, B., Dabis, F. (2008). Participation of traditional birth attendants in prevention of mother-to-child transmission of HIV services in two rural districts in Zimbabwe: a feasibility study. BMC Public Health, 8.

Piot, P., Abdool Karim, S.S., Hecht, R, et al (2015) Defeating AIDS—advancing global health. Lancet, 386: 171–218.

Rademacher, K.H., Solomon, M., Brett, T., Bratt, J.H., Pascual, C., Njunguru, J (2016) Expanding access to a new, more affordable Levonorgestrel intrauterine system in Kenya: service delivery costs compared with other contraceptive methods and perspectives of key opinion leaders Glob Health Sci Pract, 4 (2), 83–S93.

Rasbash, J; Steele,F., Browne,W., Goldstein, H.(2016). A users Guide to MLwiN, Version 2.36. Centre for Multilevel Modelling, University of Bristol.

Renju, J., Andrew, B., Kishamawe, C., Kato, C., Changalucha, J., Obasi A. (2010): A process evaluation of the scale up of a youth friendly health services initiative in northern Tanzania. J Int AIDS Soc, 13: 32-10.

Reynolds, H., B. Janowitz, R. Wilcher and W. Cates. (2008) "Contraception to Prevent HIVpositive Births: Current Contribution and Potential Cost Savings in PEPFAR Countries." Sexually Transmitted Infections, 84: 49-53.

Rochat, T., Richter, L., Doll, H., Buthelezi, N., Tomkins, A., Stein, A. (2006). Depression among Pregnant Rural South African Women Undergoing HIV Testing. JAMA, 295:176–178.

Rockers, P.C., Wilson, M.L., Mbaruku, G., Kruk, M.E. (2009) Source of antenatal care influences facility delivery in rural Tanzania: A population-based study. Maternal Child Health J, 13 (6), 879-885.

Rosenberg, N.E., Pettifor, A.E., De Bruyn, G., Westreich, D., Delany-Moretlwe, S., Behets, F et al. (2013) HIV testing and counselling leads to immediate consistent condom use among South African stable HIV-discordant couples. J Acquir Immune Defic Syndr, 62: 226–233.

Rutaremwa, G., Kabagenyi, A. (2016) Utilization of integrated HIV and sexual and reproductive health services among women in Uganda.BMC Health Serv Res. 16:494.

Rutstein, S.O. (2008). The DHS wealth index: Approaches for rural and urban areas. DHS Working Paper No. 60. Calverton, Maryland, USA: Macro International.

Sahlu, I., Howe, C.J., Clark, M.A., Marshall, B.D.L. (2014) HIV Status, Knowledge of Motherto-Child Transmission of HIV and Antenatal Care Use among Ethiopian Women. Journal of Epidemiology and Global Health, 4, 144- 184

Sama, C-B., Feteh , V.F., Tindong, M., Tanyi, J.T., Bihle, N.M., Angwafo, F.F. (2017) Prevalence of maternal HIV infection and knowledge on mother– to–child transmission of HIV and its prevention among antenatal care attendees in a rural area in northwest Cameroon. PLoS ONE, 12: 2.

Sambisa W. (2008) AID's stigma and Uptake of HIV Testing in Zimbabwe. ZDHS Working Paper 49. Web site.<u>http://www.dhsprogram.com/publications/publication-WP49-Working-</u> <u>Papers.cfm.</u> Accessed January 10, 2017

Sanga, E.S., Lerebo, W., Mushi, A.K et al. (2017) Linkage into care among newly diagnosed HIV-positive individuals tested through outreach and facility-based HIV testing models in Mbeya, Tanzania: a prospective mixed-method cohort study. BMJ Open 7:e013733.

Schneider, M.T., Birger, M., Haakenstad, A et al. (2016). Tracking development assistance for HIV/AIDS: the international response to a global epidemic.AIDS, 30:1475–79.

Schnippel, K., Mongwenyana, C., Long, L.C., Larson, B.A. (2015) Delays, interruptions, and losses from prevention of mother-to-child transmission of HIV services during antenatal care in Johannesburg, South Africa: a cohort analysis. BMC Infectious Diseases, 15:46

Semali, I.A., Jeremia, D., Saronga, H., Malamsha, D. (2014) Factors associated with HIV testing and receiving results during antenatal care in Tanzania. African Population Studies, 28(2)

Sennott, C., Majola, S.A. (2017) 'Behaving well': the transition to respectable womanhood in rural South Africa: Culture, Health & Sexuality, 19 (7), 781-795

Sharma, M., Ying, R., Tarr, G., Barnabas, R. (2015) Systematic review and meta-analysis of community and facility-based HIV testing to address linkage to care gaps in sub-Saharan Africa. Nature, 528(7580), 77–85

Sharma, M., Barnabas, R.V., Celum, C. (2017) Community-based strategies to strengthen men's engagement in the HIV care cascade in subSaharan Africa. PLoS Med, 14(4)

Sherr, L., Lopman, B., Kakowa, M et al. (2007). Voluntary counselling and testing: uptake, impact on sexual behavior, and HIV incidence in a rural Zimbabwean cohort. AIDS. 21: 851-860.

Shetty. A.K., Chandisarewa .W. Mahomva. A., Mushava.A. (2010)Preventing mother-to-child transmission of HIV in urban Zimbabwe.J Perinatol, 30(11), 763-764.

Shiferaw, S., Spigt, M., Godefrooij, M., Melkamu, Y., Tekie, M. (2013) Why do women prefer home births in Ethiopia? BMC Pregnancy Childbirth, 13:5.

Sibanda, E.L., Weller, I.V.D., Hakim, J.G., Cowan, F.M. (2013) The magnitude of loss to follow-up of HIV-exposed infants along the prevention of mother-to-child HIV transmission continuum of care: a systematic review and meta-analysis. AIDS. 27:2787–97.

Simbayi, L., Shisana, J., Chauveau. J., Ramiagan, S. (2003) Determinants of the use of voluntary and testing services among the sexually active adults of South Africa. Paper presented at the 2nd International AIDS Society Conference on HIV Pathogenesis and Treatment. Paris, France, July 13-16.

Singh, D., K. Anastos, D. Hoover, R. Burk, Q. Shi, L. Ngendahayo, E. Mutimura, A. Cajigas, V. Bigirimani, X. Cai, J. Rwamwejo, M. Vuolo, M. Cohen and P. Castle. (2009) "Human Papillomavirus Infection and Cervical Cytology in HIV-Infected and HIV-Uninfected Rwandan Women." Journal of Infectious Disease, 199: 1851-1861.

Snijders, T., Bosker, R. (1999) Multilevel Analysis: An introduction to basic and advanced multilevel modelling. 1999: SAGE Publications.

Snow, R., Mutumba, M., Resincow, K., Mugyenyi, R.G (2012) The social legacy of AIDS: Fertility aspirations among HIV-Affected women in Uganda, American Journal of Public Health 103(2)

Staveteig, S., Wang, S., Head, S.K et al. (2013) Demographic Patterns of HIV Testing Uptake in Sub-Saharan Africa. DHS COMPARATIVE REPORTS 30. Demographic and Health Research Division, ICF International,

Stekler, J.D., Golden, M.R. (2009) Learning from the missed opportunities for HIV testing. Sex Transm Infect, 85: 2-3

Stephenson, R., Elfstrom, K.M., Winter, A. (2013). Community influence on Married Men's uptake of HIV Testing in Eight African Countries: AIDs Behav 17(7), 2352-2366.

Stokols, D. (1996) Translating social ecological theory into guidelines for community health promotion. Am. J. Health Promotion. 10: 282-298.

Stringer, E.M., Kaseba, C., Levy, J et al. (2007) A randomized trial of the intrauterine contraceptive device vs hormonal contraception in women who are infected with the human immunodeficiency virus. Am J Obstet Gynecol, 197: 1–8

Sweeney, S., Obure, C.D., Maier, C.B., Greener, R., Dehne, K., Vassall, A.(2012) Costs and efficiency of integrating HIV/AIDS services with other health services: a systematic review of evidence and experience. Sex Transm Infect, 88:85–99

Taegtmeyer, M., Kilonzo, N., Mung'ala, L et al.(2006). Using gender analysis to build voluntary counselling and testing responses in Kenya.Trans R Soc Trop Med Hyg. 100(4), 305–11.

Takarinda. K.C., Madyira, L.K., Mhangara. M., Makaza, V., Maphosa-Mutsaka, M., Rusakaniko, S et al. (2016) Factors Associated with Ever Being HIV Tested in Zimbabwe: An Extended Analysis of the Zimbabwe Demographic and Health Survey (2010–2011). PLoS ONE, 11 (1).

Tarekegn, .M.S., Lieberman, .L.S., Giedraitis, V. (2014). Determinants of maternal health service utilisation in Ethiopia: Analysis of the 2011 Ethiopian Demographic and Health Survey, BMC pregnancy and Childbirth, 14:161

Taulo, F., M. Berry, A. Tsui, B. Makanani, G. Kafulafula, Q. Li, C. Nkhoma, J. Kumwenda,N. Kumwenda and T. Taha. (2009). "Fertility Intentions of HIV-1 Infected and UninfectedWomen in Malawi: A Longitudinal Study." AIDS & Behavior 13 (1), 20-27.

Tedrow, V.A., Denison, C.E., Kennedy. K., O'Reilly.M. (2011). "Voluntary counseling and testing (VCT) for changing HIV-related risk behavior in developing countries."The Cochrane Library.

Tenthan, L., Haas, H.D., Egger, M. (2015) HIV Testing Among Pregnant Women Who Attend Antenatal Care in Malawi. J Acquir Immune Defic Syndr, 69:610–614 Theilgaard, Z.P., Katzenstein, L., Chiduo, M.G., Pahl, C., Bygbjerg, G., Gerstoft, L., Lemnge. M., Britt, P. T. (2011) Addressing the fear and consequences of stigmatization - a necessary step towards making HAART accessible to women in Tanzania: a qualitative study. AIDS Research and Therapy, 8:28

Theuring S., Jefferys, L.F., Nchimbi, P., Mbezi, P., Sewangi. J. (2016) Increasing Partner Attendance in Antenatal Care and HIV Testing Services: Comparable Outcomes Using Written versus Verbal Invitations in an Urban Facility-Based Controlled Intervention Trial in Mbeya, Tanzania. PLoS ONE, 11(4)

Titaley, C. R., Hunter, C. L., Heywood, P., & Dibley, M. J. (2010). Why don't some women attend antenatal and postnatal care services?: a qualitative study of community members' perspectives in Garut, Sukabumi and Ciamis districts of west Java province Indonesia. BMC Pregnancy and Child Birth. 10, 61.

Townsen, C.L., Cortina-Borja, M., Peckham, C.S, de Ruiter A, Lyall H, Tookey PA (2008) Low rates of mother-to-child transmission of HIV following effective pregnancy interventions in the United Kingdom and Ireland, 2000–2006. AIDS, 22(8), 973–981.

Turan J.M, Miller. S, Bukusi E.A, Sanded. J, Cohen. C.R, (2008) HIV/AIDS and maternity care in Kenya: how fears of stigma and discrimination affect uptake and provision of labour and delivery services. AIDS Care, 20(8):938–45.

Turan, J.M., Onono M, Steinfeld ,R.L, Shade, S.B, Owuor K, Bukusi, E.A, et al. (2015). Effects of antenatal care and HIV treatment integration on elements of the PMTCT cascade: results from the SHAIP cluster-randomized controlled trial inKenya. J Acquir Immune Defic Syndr. 69:e172_81.

Turan.J.M. Hatcher, A.B., Medema-Wijnveen, J., Onono, M., Miller, S., Elizabeth, A., Cohen, C.R. (2012) The Role of HIV-Related Stigma in Utilization of Skilled Childbirth Services in Rural Kenya: A Prospective Mixed Methods Study, PLOS Medicine, 9:8

UNAIDS 2016–2021 Strategy: On the Fast-Track to End AIDS. UNAIDS. (2015).http://aidsdatahub.org/unaids-2016%E2%80%932021-strategy-fast-track-end-aids-unaids-2015. Access June 2017

Underwood, C.R., Schwand, H.M. (2016) Assessing girls' HIV vulnerability: evidence from Botswana, Malawi and Mozambique. Health Policy and Planning, 31, 729–735

Usman, A.R. (2016) Understanding the spatial context of sustainable urban health in Africa for the SDGs: some lessons from the corridors of deprivation in Ilorin, Nigeria, From the Millennium Development Goals (MDGs) to Sustainable Development Goals (SDGs):Africa in the Post-2015 Development Agenda, A Geographical Perspective. African Geographical Review, 36 (2), 216-235

Uthman, O. A. (2017) Does it really matter where you live? A multilevel Analysis of Social Disorganisation and Risky Sexual Behaviours in Sub-Saharan Africa. 2010. Working papers No.78: Demographic and health research.Web site. https://dhsprogram.com/pubs/pdf/WP78/WP78.pdf. Accessed January 10, 2017

Uwakwe, K.A., Iwu, C.A., Duru, C.B., Nnebue, C.C., Diwe, K.C., Merenu, I.A., Nwapi, J.I (2016) Sero - Status Of HIV-Exposed Infants Delivered By HIV-Positive Mothers In Health Facilities Of Imo State, Nigeria: A 5year Survey. Journal of Dental and Medical Sciences, 15(3), 121-127

Valpine, P.J. (2012) Frequentist analysis of hierarchical models to population dynamics and demographic data. Journal of Ornithology, 152:393-408.

Venkatesh, K.K., Madiba, P., Guy De Bruyn et al. (2011) Who Gets Tested for HIV in a South African UrbanTownship? Implications for Test and Treat and Gender-Based Prevention Interventions, J Acquir Immune Defic Syndr, 56:151–165

Vu, L. T. H. (2005) Multilevel determinants of children's health outcomes. PhD, thesis. University of Saskatchewan, Canada.

Wagner, G.J., Wanyenze, R. (2013). Fertility desires and intentions and the relationship to consistent condom use and provider communication regarding childbearing among HIV clients in Uganda. ISRN Infect Dis. 2013:7

Wanga, H., Wolock, T.M., Carter, A. et al. (2016). Estimates of global, regional, and national incidence, prevalence, and mortality of HIV, 1980–2015: the Global Burden of Disease Study 2015

Wanjira, C., Mwangi,M, Mathenge E, Mbugua G, Ng'ang'a Z (2011) Delivery practices and associated factors among mothers seeking child welfare services in selected health facilities in Nyandarua South District, Kenya. BMC Public Health 11: 360-10.

Warren, C.E., Abuya, T., Askew, I. (2013). Family planning practices and pregnancy intentions among HIV-positive and HIV-negative postpartum women in Swaziland: a cross sectional survey. BMC Pregnancy Childbirth. 13(1), 150.

Washington, S., Owuor, K., Turan. J.M., Steinfeld, R.L., Onono, M., Shade, S.B et al. (2015) The effect of integration of HIV care and treatment into antenatal care clinic son mother-tochild HIV transmission and maternal outcomes in Nyanza, Kenya: results from the SHAIP cluster randomized controlled trial. J Acquir Immune Defic Syndr. 69:e164_71.

Weiser, S.D., Heisler, M., Leiter, K. (2006). Routine HIV testing in Botswana: A populationbased study on attitude, practices, and Human Rights Concerns. Medicine journal.3 :(7) 261

Wekesa, E., Coast, E. (2014) Fertility desires among Men and Women Living with HIV/AIDs in Nairobi Slums: A mixed Methods Study. Population council, 9 (8)

Wheeler, T., Wolf, R.C., Kapesa, L., Cheng Surdo, A., Dallabetta, G., (2015) Scaling-up HIV responses with key populations in West Africa. J Acquir Immune Defic Syndr, 68 (2), 69–73.

White, D., Dynes, M., Rubardt, M., Sissoko, K., Stephenson, R. (2013) The influence of intra familial power on maternal health care in Mali: perspectives of women, men and mothers-inlaw. Int Perspect Sex Reprod Health,, 39 (2), 58-68. ITS (2015) Data and Statistics, Geneva. http://www.who.int/hiv/data/en/. Access June 2017.

WHO (2015) New HIV testing guidance to help reach the 90-90-90 targets.<u>http://www.who.int/hiv/mediacentre/news/hiv-testing-90targets/en/</u>. Accessed June 2016

WHO, UNAIDS, UNICEF. Towards universal access: scaling up priority HIV/AIDS interventions in the health sector. Progress Report. 2010 Available from: http://www.who.int/hiv/pub/2010progressreport/en/.

Wiegert, K., Dinh, T., Mushavi ,A. (2014) Integration of Prevention of Mother-to-Child Transmission of HIV (PMTCT) Postpartum Services with Other HIV Care and Treatment Services within the Maternal and Child Health Setting in Zimbabwe.PLoS ONE 9(6), e98236.

Wilcher, R., Cates, W.J.R., Gregson, S. (2012) Family planning and HIV: strange bedfellows no longer. AIDS, 23 (1), 1–6.

Woldemicael, G., Tenkorang, E. Y. (2009) Women's autonomy and maternal health-seeking behaviour in Ethiopia.Maternal Child Health Journal, 14(6), 988-998.

World Health Organisation (2014) Global update on the health sector response to HIV, 2014.Geneva:WorldHealthOrganisation.Available:http://apps.who.int/iris/bitstream/10665/128494/1/9789241507585_eng.pdf?ua=1Assessed May 2017

World Health Organisation (2012) Guidelines on maternal, newborn, child and adolescent health, Recommendations on maternal and perinatal health, Geneva. <u>http://www.who.int/maternal_child_adolescent/documents/guidelines-recommendations-</u>maternal-health.pdf. Assessed July 2017.

World Health Organisation (WHO), United Nations Programme on HIV/AIDS (UNAIDS) and United Nations Children's Fund (UNICEF). Towards universal access: Scaling up Priority HIV/AIDS interventions in the Health Sector. Progress report 2010.Web site.<u>http://www.who.int/hiv/pub/2010progressreport/en/</u>. Accessed December 4, 2016.

Wringe, A., Isingo, R., Urassa, M et al. (2008). Uptake of HIV voluntary counselling and testing services in rural Tanzania: implications for effective HIV prevention and equitable access to treatment. Tropical Medicine and International Health, 13(3), 319-327

Yargawa, J., Leonardi-Bee, J. J. (2015) Male involvement and maternal health outcomes: systematic review and meta-analysis. Epidemiol Community Health, 69:604–612

Yeatman, S.E. (2009) The Impact of HIV Status and Perceived Status on Fertility Desires in Rural Malawi. AIDS and Behaviour, 13, 12-19. http://dx.doi.org/10.1007/s10461-009-9534-1

Zere, E., Oluwole, D., Kirigia, J.M., Mwikisa, C.N., Mbeeli, T. (2011) Inequities in skilled attendance at birth in Namibia: A decomposition analysis. BMC Pregnancy Childbirth, 11: 34-10.

Zimbabwe National Family Planning Council (ZNFPC) (2016) Half of pregnancies unplanned.<u>https://drive.google.com/file/d/0B9-C8CiaTO-cTnJrN1ZkMW5FV1U/view</u>. Assessed June 2017

Zimbabwe National HIV and AIDS Strategic Plan (ZNASP) (2015 – 2018). Commitment Towards Fast tracking 90.90.90 Targets by 2020 and Ending aids by 2030. <u>http://www.nac.org.zw/sites/default/files/ZNASP%20III%20Final%20(1).pdf</u>. Assessed May 2017.

ZIMBABWE NATIONAL HIV AND AIDS STRATEGIC PLAN [ZNASP II] (2011-2015) .Delivering our commitment to, Zero new HIV infections, Zero discrimination, Zero AIDSrelated deaths, Republic of Zimbabwe. Web site.<u>http://www.nac.org.zw/.</u> Accessed December 3, 2016

253

Zimbabwe National Network of PLHIV (ZNNP+) (2014), ZIMBABAWE STIGMA INDEXRESEARCHREPORTDECEMBER2014.http://www.hivsharespace.net/resource/zimbabwe-people-living-hiv-stigma-index.AssessedMay 2017.

Zimbabwe National Statistics Agency (ZIMSTAT) and ICF International (2012) Zimbabwe Demographic and Health Survey 2010–11. Calverton, Maryland: ZIMSTAT and ICF International Inc. Available: <u>http://dhsprogram.com/pubs/pdf/FR254/FR254.pdf</u>. Assessed <u>May 2017</u>.

Appendices

Appendix 4.1 Average odds of HIV testing from multilevel logistic regression models (95% confidence intervals are given in square brackets) males and females

Parameters	Model 1	Model 2	Model 3	
Survey (2005-6)				
2010-11	4.16[3.80-4.55]*	5.30 [4.67-6.02]*	5.60 [4.90-6.37]*	
Gender (males)				
Females	1.98[1.85-2.13]*	1.84 [1.63-2.07]*	1.83 [1.63-2.07]*	
Residence (urban)				
Rural	0.79[0.72-0.86]*	1.12 [1.00-1.26]*	1.18 [1.04-1.34]*	
Age groups (15-24)				
25-29	1.30[1.17-1.43]*	1.20[1.08-1.33]*	1.20 [1.08-1.33]*	
30-34	0.90 [0.80-1.01]	0.83 [0.74-0.94]*	0.83 [0.74-0.94]*	
35-39	0.75[0.66-0.86]*	0.69 [0.61-0.79]*	0.69 [0.61-0.79]*	
40+	0.60[0.53-0.69]*	0.55 [0.49-0.63]*	0.55 [0.49-0.63]*	
Marital status (never mar	ried)			
Married	2.01[1.78-2.26]*	2.30 [2.03-2.60]*	2.30 [2.03-2.59]*	
Widowed	2.39[1.98-2.88]*	2.50 [2.07-3.03]*	2.51 [2.07-3.04]*	
Divorced	1.83[1.56-2.16]*	1.80 [1.52-2.13]*	1.80 [1.52-2.13]*	
Occupation(unemployed))			
Professional	1.31[1.18-1.46]*	1.21 [1.08-1.35]*	1.21 [1.09-1.35]*	
Manual work	1.28[1.15-1.42]*	1.20 [1.08-1.34]*	1.21 [1.08-1.34]*	
Agriculture	1.03 [0.93-1.14]	1.00 [0.91-1.11]	1.01 [0.91-1.11]	
Religion (Catholic)				
Protestant	1.08 [0.98-1.18]	1.04 [0.95-1.15]	1.05[0.96-1.15]	
Pentecostal	1.18[1.08-1.29]*	1.17 [1.06-1.28]*	1.17 [1.06-1.29]*	
Apostolic	0.98 [0.90-1.05]	1.01 [0.93-1.09]	1.01 [0.93-1.09]	
Secondary	1.67[1.55-1.80]*	1.49 [1.38-1.61]*	1.50 [1.39-1.62]*	
Higher	3.62[3.04-4.31]*	2.82 [2.36-3.38]*	2.84 [2.37-3.40]*	
Enabling and perceived/need factors				
Wealth (poorest)				
Poorer		1.00 [0.90-1.11]	1.01 [0.90-1.12]	
Middle		1.12 [1.01-1.25]*	1.13 [1.01-1.26]*	
Richer		1.25 [1.10-1.42]*	1.24 [1.09-1.41]*	
Richest		1.40 [1.20-1.62]*	1.38 [1.18-1.60]*	

Continued

Appendix 4.1 continued			
Media exposure (low)			
Medium		1.22 [1.12-1.34]*	1.21 [1.10-1.32]*
High		1.62 [1.36-1.92]*	1.52 [1.28-1.82]*
HIV/AIDS Awareness(lo	w)		
Medium		1.75 [1.51-20.2]*	1.76 [1.52-2.03]*
High		1.85 [1.58-2.16]*	1.86 [1.59-2.17]*
Observed AIDS stigma(n	10)		
Yes		0.81 [0.73-0.91]*	0.81 [0.73-0.91]*
HIV disclosure concern(n	10)		
Yes		0.83 [0.76-0.91]*	0.83 [0.76-0.91]*
Knows someone died wit	h Aids(no)		
Yes		1.30 [1.14-1.48]*	1.29 [1.13-1.48]*
History of STI (no)			
Yes		1.66 [1.38-2.00]*	1.66 [1.38-1.99]*
Community factors			
HIV/AIDS stigma-gm			0.62 [0.42-0.92]*
Media exposure-gm			1.47 [1.14-1.89]*
Random Variance (SE)	0.129(0.016)*	0.112(0.015)*	0.11(0.015)*
(VPC)=ICC(%)	3.8	3.3	3.3

VPC=Variance Partition Co-efficient, ICC=intra-cluster correlation, *Statistical significance

at 5% level p<0.05

Appendix 6.1 Average odds of HIV testing from multilevel logistic regression models (95% confidence intervals are given in square brackets) institutional and professional delivery

Parameters	Institutional Delivery	Professional Delivery
Predisposing factors		
Survey 2010	0.41 [0.32-0.53]*	0.44 [0.34-0.56]*
Place of residence (rural)		
Urban	0.39 [0.30-0.50]*	0.38 [0.29-0.49]*
Age groups(15-24)		
25-29	1.68 [1.43-1.97]*	1.74 [1.49-2.04]*
35+	1.77 [1.44-2.19]*	1.99 [1.61-2.45]*
Marital status(never married		
Married	1.39 [1.03-1.88]*	1.27 [0.94-1.72]
Widowed	1.75 [1.19-2.57]*	1.42 [0.97-2.09]
Divorced	1.11 [0.81-1.52]	0.95[0.69-1.30]
Women's education (primary)		
Secondary	1.69 [1.50-1.91]*	1.67 [1.48-1.89]*
Religion(Catholic)		
Protestant	1.53 [1.26-1.84]*	1.59 [1.32-1.92]*
Pentecoastal	1.30 [1.07-1.57]*	1.33 [1.10-1.61]*
Apostolic	0.94 [0.82-1.07]	0.94 [0.82-1.08]
Wanted child (now)		
Later	0.78 [0.68-0.89]*	0.76 [0.66-0.87]*
Not wanted	0.77 [0.67-0.92]*	0.71 [0.59-0.85]*
Parity		
2	0.60 [0.50-0.70]*	0.59 [0.50-0.70]*
3+	0.38 [0.32-0.47]*	0.36 [0.30-0.44]*
Wealth (poor)		
Medium	1.64 [1.35-2.00]*	1.70 [1.40-2.07*
Rich	2.66 [1.94-3.65]*	2.76 [2.00-3.80]*
Employment status(No)		
Yes	0.94 [0.83-1.07]	0.97 [0.85-1.10]
Media exposure (Lowl)		
Medium	1.23 [1.07-1.40]*	1.21 [1.06-1.39]*
High	1.29 [1.10-1.51]*	1.28 [1.09-1.50]*
Decision making (husband)		
Women alone	0.98 [0.80-1.20]	0.96 [0.78-1.17]
Jointly	0.95 [0.80-1.13]	0.98 [0.82-1.17]

Continued

Access problem (no)		
Yes	0.79 [0.69-0.91]*	0.83 [0.72-0.95]*
No. of ANC visits (no)		
4 plus	4.14 [3.28-5.21]*	5.90 [4.82-7.24]*
ANC-HIV test (no)		
Yes	2.77 [2.43-3.16]*	2.75 [2.41-3.13]*
HIV sero-status (negative)		
Positive	1.01 [0.86-1.19]	0.10 [0.85-1.18]
HIV awareness (low)		
Medium	1.14 [0.99-1.31]	1.13 [0.98-1.30]
High	0.84 [0.72-0.98]	0.84 [0.72-0.97]*
HIV stigma(low)		
Medium	1.04 [0.86-1.27]	1.01 [0.3-1.23]
High	0.95 [0.77-1.17]*	0.89 [0.73-1.10]
Confidential concern (yes)		
No	1.18 [1.05-1.32]*	1.19 [1.06-1.33]*
Knows someone with AIDS (no)		
Yes	1.25 [1.05-1.49]*	1.21 [1.02-1.45]*
Contextual Factors		
HIV awareness-gm	1.22 [0.73-2.04]	0.73[0.33-1.51]
HIV stigma-gm	1.26 [0.71-2.24]	1.07 [0.60-1.92]
HIV prevalence-gm	0.68 [0.31-1.46]	1.2]6 [0.75-2.13
Random effects parameters		
Community-level		
Variance (SE)	0.447[0.061]*	0.526 [0.083]*
(VPC)=ICC (%)	12	14